UTAH STATE IMPLEMENTATION PLAN

SECTION X

VEHICLE INSPECTION AND MAINTENANCE PROGRAM

PART A

GENERAL REQUIREMENTS AND APPLICABILITY

Adopted by the Utah Air Quality Board December 5, 2012

Table of Contents

1.	General Requirements
2.	Applicability
3.	General Summary

UTAH STATE IMPLEMENTATION PLAN SECTION X VEHICLE INSPECTION AND MAINTENANCE PROGRAM PART A GENERAL REQUIREMENTS AND APPLICABILITY

1. General Requirements

Federal I/M Program requirements: Utah was previously required by Section 182 and Section 187 of the Clean Air Act to implement and maintain an Inspection and Maintenance (I/M) program in Davis, Salt Lake, Utah, and Weber counties that met the minimum requirements of 40 code of federal regulation (CFR) Part 51 Subpart S and that was at least as effective as the EPA's Basic Performance Standard as specified in 40 CFR 51.352. The Basic Performance Standard requirement is no longer applicable as the relevant nonattainment areas in Davis, Salt Lake, Utah, and Weber counties have been redesignated to attainment / maintenance for the carbon monoxide (CO) National Ambient Air Quality Standards (NAAQS) and the 1-hour ozone NAAQS. Parts A, B, C, D, and E of Section X, together with the referenced appendices, continue to demonstrate compliance with the 40 CFR Part 51 provisions for Inspection and Maintenance Program Requirements for Davis, Salt Lake, Utah, and Weber counties and produce mobile source emission reductions that are sufficient to demonstrate continued maintenance of the applicable CO and 1-hour ozone NAAQS. In addition, the Cache, Davis, Salt Lake, Utah, and Weber counties' I/M programs are also utilized as a control measure to attain and maintain EPA's particulate NAAQS (PM_{2.5} and PM₁₀).

On-Board Diagnostics (OBD) Checks: By January 1, 2002, OBD checks and OBDrelated repairs are required as a routine component of Utah I/M programs on model year 1996 and newer light-duty vehicles and light-duty trucks equipped with certified onboard diagnostic systems. The federal performance standard requires repair of malfunctions or system deterioration identified by or affecting OBD systems.

Utah I/M program history and general authority: The legal authority for Utah's I/M programs, Utah Code Annotated Section 41-6-163.6¹, was enacted during the First Special Session of the Utah legislature in 1983. I/M programs were initially implemented by Davis and Salt Lake counties in 1984, by Utah County in 1986, and by Weber County in 1990.

In 1990, the legislature enacted Section 41-6-163.7² that requires that counties with I/M programs use computerized I/M testing equipment, adopt standardized emission standards, and provide for reciprocity. Those requirements were fully implemented by Davis, Salt Lake, and Utah counties on September 1, 1991, and by Weber County on January 1, 1992.

¹ Renumbered and recodified in 2005 at Utah Code Annotated 41-6a-1642

² Renumbered and recodified in 2005 at Utah Code Annotated 41-6a-1643

Section 41-6-163.6 was again amended by the legislature in 1992 to include vehicles owned and operated by the federal government, federal employees, and students and employees of colleges and universities. The 1992 revision of 41-6-163.6 also established more stringent restrictions for vehicles that qualify for a farm truck exemption.

Section 41-6-163.6 requires that, if identified as necessary to attain or maintain any NAAQS, a county must create an I/M program that follows the criteria outlined in 41-6-163.6. Once a county enacts regulations or ordinances, amendments to Section 19-2-104 in 1992 authorized the Utah Air Quality Board to formally establish those requirements for county I/M programs after obtaining agreement from the affected counties. Section 41-6-163.6 was also amended to allow the counties to subject individual motor vehicles to inspection and maintenance at times other than the annual inspection.

Section 41-6-163.6 was amended in 1994 to authorize implementation of I/M programs stricter than minimum federal requirements in counties where it is necessary to attain or maintain ambient air quality standards. Section 41-6-163.6 requires preference be given to a decentralized program to the extent that a decentralized program will attain and maintain ambient air quality standards and meet federal requirements. It also requires affected counties and the Air Quality Board to give preference to the most cost effective means to achieve and maintain the maximum benefit with regard to air quality standards and to meet federal air quality requirements related to motor vehicles. The legislature indicated preference for a reasonable phase-out period for replacement of air pollution test equipment made obsolete by an I/M program in accordance with applicable federal requirements and if such a phase-out does not otherwise interfere with attainment of ambient air quality standards.

House Concurrent Resolution No. 9 of the 1994 General Session of the legislature (H.C.R. 9) was a concurrent resolution of the legislature and the governor expressing opposition to the EPA position regarding the implementation of enhanced vehicle inspection and urging the EPA to recognize the benefits of other vehicle inspection program options and to work with the state to develop workable plans for attaining ambient air quality standards and protecting public health.

In 1995, the legislature amended Section 41-6-163.7 to rescind the requirement for I/M program standardization and reciprocity between counties. While advantageous, standardization and reciprocity between I/M counties is no longer required, and each I/M county is free to develop an I/M program that best meets the respective county's needs.

In 2002, the Legislature amended Section 41-6-163.7 to allow for inspection every other year for cars that are six years old or newer on January 1 each year. This provision is applicable to the extent allowed under the current state implementation plan for each area.

In 2005, the Legislature renumbered Section 41-6-163.6 and re-codified it as Section 41-6a-1642. The Legislature also amended Section 41-6a-1642 to allow counties with an

I/M program to require college students and employees who park a motor vehicle on college or university campus that is not registered in a county subject to emission inspection to provide proof of compliance with an emission inspection.

Section 41 6a-1642 was amended in 2008 to provide an exemption for vintage vehicles, which are defined in Section 41-21-1. Section 41 6a-1642 was again amended in 2009 to provide an exemption for custom vehicles, which are defined in Section 41-6a-1507.

In 2010, the legislature enacted Section 41-1a-1223 that allows counties with an I/M program to impose a local emissions compliance fee of up to three dollars. This same bill amended Section 41-6a-1642 to require I/M counties that impose the fee to use revenues generated from the fee to establish and enforce an emission inspection and maintenance program.

Section 41-6a-1642 was amended in 2011 to require I/M counties' regulations and ordinances to be compliant with the analyzer design and certification requirements contained in the SIP.

In 2012, the Legislature amended Section 41-6a-1642 to allow a motor vehicle that is less than two years old as of January 1 of any given year to be exempt from being required to obtain an emission inspection. This provision is applicable to the extent allowed under the current SIP for each area. This bill went into effect on October 1, 2012. In addition, the legislature also amended Section 41-1a-205 to allow a safety and emissions inspection issued for a motor vehicle during the previous 11 months may be used to satisfy the safety and emissions inspection requirements³. The effective date of this bill is January 1, 2013. The legislature also amended Section 41-1a-1223 to allow the counties to collect a \$2.25 fee for those vehicles that are registered for a six-month period under Utah Code Annotated 41-1a-215.5. The effective date of this bill is July 1, 2013.

2. Applicability

General Applicability: Utah Code Annotated 41-6a-1642 gives authority to each county to implement and manage an I/M program to attain and maintain any national ambient air quality standard (NAAQS). Davis, Salt Lake, Utah, and Weber counties were required under Section 182 and 187 of the Clean Air Act to implement an I/M program to attain and maintain the ozone and carbon monoxide NAAQS. All of Utah's ozone and carbon monoxide maintenance areas are located in Davis, Salt Lake, Utah, and Weber counties. In addition, a motor vehicle I/M program is a control measure for attaining the particulate matter NAAQS in Cache, Davis, Salt Lake, Utah, and Weber counties. Utah's SIP for I/M is applicable county-wide in Cache, Davis, Salt Lake, Utah, and Weber counties.

³ Utah Code 41-6a-1642(7) states that "the emissions inspection shall be required within the same time limit applicable to a safety inspection under Section 41-1a-205."

3. General Summary

Below is a general summary of Utah's I/M programs. Part B, C, D, E and F of this section of the SIP provide a more specific summary of I/M programs for Cache, Davis, Salt Lake, Utah, and Weber counties. These parts also incorporate the individual county I/M ordinances/regulations and policies that provide for the enforceability of the respective I/M programs.

Network Type: All Utah I/M programs are comprised of a decentralized, test-and-repair network.

I/M program funding requirements: Counties with I/M programs allocate funding as needed to comply with the relevant requirements specified in Utah's SIP; the Utah statutes; county ordinances, regulations and policies; and the federal I/M program regulation. Program budgets include funding for resources necessary to adequately manage the programs conduct covert and overt audits, including repairs; assist and educate inspectors, station owners, and the public; manage, analyze, and report data; ensure compliance with the program by inspectors, stations, and vehicle owners; and evaluate and upgrade the programs.

Funding mechanisms: Utah's I/M programs are funded through several mechanisms including, but not limited to, a fee which is collected at the time of registration by the Utah Tax Commission Division of Motor Vehicles or the county Assessor's Office. Those monies are remitted to the county in which the vehicle is registered. The collection of fees for various permitting activities and the selling inspection certificates to inspection stations are the other mechanisms. A fee schedule can be found in an appendix to each county I/M ordinance or regulation.

Government fleet: Section 41-6a-1642(1)(b) of the Utah Code requires that all vehicles owned or operated in the I/M counties by federal, state, or local government entities comply with the I/M programs.

Vehicles owned by students and federal employees: Section 41-6a-1642(5) provides a provision that counties may require universities and colleges located in Utah's I/M areas to require proof of compliance with the I/M program for vehicles which are permitted to park on campus regardless of where the vehicle is registered. Vehicles operated by federal employees and operated on a federal installation located within an I/M program area are also subject to the I/M program regardless of where they are registered. Proof of compliance consists of a current vehicle registration in an I/M program area, an I/M certificate of compliance or waiver, or evidence of exempt vehicle status.

Rental vehicles: All vehicles available for rent or use in an I/M county are subject to the county I/M program. To the extent practicable, all vehicles principally operated in the county are subject to the I/M program.

Section X, Part A, page 4

Farm truck exemption: Eligibility for the farm truck exemption from the I/M programs is specified in Section 41-6a-1642(4) and must be verified in writing by county I/M program staff.

Out-of-state exemption: Vehicles registered in an I/M county but operated out-of-state are eligible for an exemption. The owner must complete Utah State Tax Commission form TC-810 in order to be registered without inspection documentation. The owner must explain why the vehicle is unavailable for inspection in Utah. Common situations include Utah citizens that are military personnel stationed outside of the state, students attending institutions of higher education elsewhere, and people serving religious assignments outside the area. If the temporary address of the owner is located within another I/M program area listed on the back of the form, the owner must submit proof of compliance with that I/M program at the time of, and as a condition precedent to, registration or renewal of registration. The vehicle owner must identify his or her anticipated date of return to the state and is required to have the vehicle inspected within ten days after the vehicle is back in Utah.

Motorist Compliance Enforcement Mechanism: The I/M programs are registrationenforced on a county-wide basis. A certificate of emissions inspection or a waiver or other evidence that the vehicle is exempt from the I/M program requirements must be presented at the time of, and as a condition precedent to, registration or renewal of registration of a motor vehicles as specified in Section 41-6a-1642(1)(a). Owners of vehicles operated without valid license plates or with expired license plates are subject to ticketing by peace officers at any time. Proof of compliance consists of a current vehicle registration in an I/M program area or an I/M certificate of compliance or waiver, or evidence of exempt vehicle status.

Valid registration required: A certificate of emissions inspection or a waiver or other evidence that the vehicle is exempt from the I/M program requirements must be presented at the time of, and as a condition precedent to, registration or renewal of registration of a motor vehicles as specified in Section 41-6a-1642 and 41-1a-203(1)(c). The I/M inspection is required within two months prior to the month the registration renewal is due as specified in Section 41-6a-1642(7) and 41-1a-205(2)(a). Owners of vehicles operated without valid license plates or with expired license plates are subject to ticketing by peace officers at any time. Registration status is also checked on a random basis at roadblocks and in parking lots at various locations around the state. Per Section 41-1a-402, Utah license plates indicate the expiration date of the registration. Per Section 41-1a-1303, it is a Class C misdemeanor for a person to drive or move, or for an owner knowingly to permit to be driven or moved, upon any highway any vehicle of a type that is required to be registered in the state that is not registered in the state. Section 41-1a-1315 specifies that it is a third degree felony to falsify evidence of title and registration.

Change of ownership: Vehicle owners are not able to avoid the I/M inspection program by changing ownership of the vehicle. Upon change of vehicle ownership the vehicle must be re-registered by the new owner. The new owner must present an emissions

Section X, Part A, page 5

certificate, waiver, or proof of exemption from the I/M program as a condition precedent to registration⁴. The new annual registration and I/M inspection dates for the vehicle will be the date of registration.

Utah Tax Commission, and County Assessors roles: The Utah Tax Commission Motor Vehicle Division and county assessor deny applications for vehicle registration or renewal of registration without submittal of a valid certificate of compliance, waiver, or verified evidence of exemption. Altered or hand-written documents are not accepted. All certificate data is collected by county I/M program auditors and subjected to scrutiny for evidence of any improprieties.

Database quality assurance: The vehicle registration database is maintained and quality assured by the Utah Division of Motor Vehicle (DMV). Each county I/M inspection database is maintained and quality assured by the county I/M program staff. The county I/M program has access to the DMV database and utilizes it for quality assurance purposes. All databases are subject to regular auditing, cross-referencing, and analysis. The databases are also evaluated using data obtained during roadblocks and parking lot surveys. Evidence of program effectiveness may trigger additional joint enforcement activities.

Oversight provisions: The oversight program includes verification of exempt vehicle status through inspection, data accuracy through automatic and redundant data entry for most data elements, an audit trail for program documentation to ensure control and tracking of enforcement documents, identification and verification of exemption-triggering changes in registration data, and regular audits of I/M inspection records, I/M program databases, and the DMV database.

Enforcement staff quality assurance: County I/M program auditors and DMV clerks involved in vehicle registration are subject to regular performance audits by their supervisors. All enforcement personnel (direct and indirect) involved in the motorist enforcement program are subject to disciplinary action, additional training, and termination for deviation from procedures. Specific provisions are outlined in the DMV procedures manual which is available upon request. The county I/M audit policy documents are provided in their respective part of this section.

Quality Control: The I/M counties maintain records regarding inspections, equipment maintenance, and the required quality assurance activities. The I/M counties analyze I/M program data and submit annual reports to the U.S. Environmental Protection Agency and UDAQ upon request.

Analyzer data collection: Each county's I/M analyzer data collection system meets the requirements specified under 40 CFR 51.365.

4 See Utah Code Section 41-6a-1642 (7) and 41-1a-205(2)(b) and (c)

Data analysis and reporting- Annual: The I/M counties analyze and submit to EPA and UDAQ an annual report for January through December of the previous year, which includes all the data elements listed in 40 CFR Subpart S 51.366 by July of each year. If a report is required earlier than annually, the counties will accommodate the request.

General enforcement provisions: The county I/M programs are responsible for enforcement action against incompetent or dishonest stations and inspectors. Each county I/M ordinance or regulation includes a penalty schedule.

General public information: The I/M counties have comprehensive public education and protection programs, including providing strategies to educate the public on Utah's air quality problems; ways that people can reduce emissions; the requirements of state and federal law; the role of motor vehicles in the air quality problem; the need for and benefits of a vehicle emissions inspection program; ways to operate and maintain a vehicle in a low-emission condition; how to find a qualified repair technician; and the requirements of the I/M program. Information is provided via county websites and direct response to inquiries for information, reports, classes, pamphlets, fairs, school presentations, workshops, news releases, posters, signs, and public meetings. Utah Department of Environmental Quality also provides information on its website about ways to operate and maintain a vehicle in a low-emission condition.

County I/M technical centers: Each I/M county operates an I/M technical center staffed with trained auditors and capable of performing emissions tests. A major function of the I/M technical centers is to serve as a referee station to resolve conflicts between permitted I/M inspectors, stations, and motorists. Auditors actively protect consumers against fraud and abuse by inspectors, mechanics, and others involved in the I/M program. Complaints are received and investigated fully. Auditors advise motorists regarding emissions warranty provisions and assist the owners in obtaining warranty covered repairs for eligible vehicles. The I/M technical centers also provide motorists with information regarding the I/M program, general air pollution issues, and emissions-related vehicle repairs.

Vehicle inspection report: A vehicle inspection report (VIR) is printed and provided to the motorist after each vehicle inspection. The VIR includes a public awareness statement about vehicle emissions and lists additional ways that the public can reduce air pollution. The test results are detailed on the VIR. Information about vehicle emissions warranties and the benefits of emissions-related repairs are printed for vehicles that failed the test. If the vehicle fails a retest, information about wavier requirements, application procedures and the address and telephone number of the applicable I/M technical center are printed on the VIR.

Reciprocity between County I/M programs: Utah I/M programs are conducting the same test procedures and thereby agreed to recognize the validity of a certificate granted by any Utah I/M program.

Section X, Part A, page 7

UTAH STATE IMPLEMENTATION PLAN SECTION X VEHICLE INSPECTION AND MAINTENANCE PROGRAM PART B DAVIS COUNTY

> Adopted by the Utah Air Quality Board February 5, 1997

Ta	h		of	2	atonte	
10	D	e.	UL.	COL	nems	

1. I/M Performance Standard	1	
Federal Requirements	1	
Basic I/M program performance standard	1	
Basic I/M Program MOBILE modeling	1	
Davis County Enhanced (DC98) I/M program Performance Standard	2	
DC98 MOBILE modeling	2	
· · · · · · · · · · · · · · · · · · ·		
2. Network type	3	
Basic I/M Program	3	
DC98 Enhanced I/M Program	3	
3. Tools and resources	4	
Funding mechanisms	4	
Funding requirements	4	
4. Test convenience	4	
Basic I/M program	4	
DC98 I/M program	5	
5. Vehicle Coverage	5	
Subject fleet	5	
Alternative fuels	5	
Government fleet	5	
Vehicles owned by students and federal employees	5	
Farm truck exemption	6	
Diesel vehicles	6	
New vehicle exemption	6	
Out-of-state exemption	6	
Exempt vehicle statistics	6	
Unregistered vehicles	7	
Roadside I/M program element	7	
6 Test procedures and standards	7	
Specifications	7	
Basic I/M Program test procedure and analyzer	7	
Enhanced I/M (DC98) test procedure and analyzers	8	
Pre-inspection emissions-related renairs	8	
Safety issues	8	
Exhaust leaks	8	
Emission standards	9	
Stringency	9	
Re-test standards	9	
Anti-tampering provisions	9	
Engine changes	9	
Fuel switching	9	
T del Switching	5	
7 Test Equipment	10	
Specifications	10	
UTAH01 analyzer access restrictions	10	
UTAH91 data security provisions	10	
UTAH91 automated test procedure	10	
ITAH91 security lockouts	10	
LITAH01 certified analyzer use restriction	10	
ט זא אוסד טפרנוווכע מוזמוץ בפו עספ דפסגווטגוטוו	10	
8. Quality Control	11	
Ceneral quality control specifications	11	
LITAH01 automatic electronic quality assurance features	11	
o manor automatic electronic quality assurance realures	11	

UTAH91 analyzer maintenance	11
UTAH91 document security	11
UTAH91 analyzer certification	12
General UTAH91 analyzer security provisions	12
9. Waivers	12
Waiver rate	12
Waiver procedures	12
10.Motorist compliance enforcement	13
Registration denial	13
Certificate of Compliance	13
Fuel changes to non-subject status	13
Title transfers	13
11.Motorist compliance enforcement program oversight	14
Utah Tax Commission, tax assessors and county roles	14
Database quality assurance	14
	14
Enforcement staff quality assurance	15
Co-operative enforcement oversight effort	15
12.I/M Program guality assurance	15
Station/inspector audits	15
Covert audits	15
Electronic audit capabilities	16
13.Enforcement against stations and inspectors	16
General enforcement provisions	16
Suspension and revocation	17
Enforcement records	17
14.Data collection	17
UTAH91 analyzer inspection data	17
IM240 inspection data	17
UTAH91 analyzer quality assurance data	18
IM240 quality assurance data	18
UTAH91 analyzer database specifications	18
DC98 data collection	18
15.Data analysis and reporting	18
Annual	18
Biennial	18
16.Inspector training and permitting	18
Inspector permitting and initial training	18
Basic inspector permit renewal	19
Inspector permit suspension and revocation	19
Inspector training authority and materials	19
17.Public information and consumer protection	19
General public information	19
Davis County I/M technical center	20
Vehicle inspection report	20
Davis County co-operative public education tools	20
	20
18.Improving repair effectiveness	20
High priority	20
······································	20

Continuing education	20
Pollution prevention	21
19 Basic and DC98 I/M SIP implementation	21

SECTION X, PART B Appendices

- 1 Motor Vehicle I/M Program Ordinance
- 1.a Motor Vehicle I/M Program Ordinance, revised 1994
- 1.b Enhanced Vehicle I/M Program Ordinance, 1996
- 1.c Amendment to I/M Program Ordinance, 1997
- 2 Attorney General's letter
- 3 County Attorney Letters
- 4 (reserved)
- 5 Waiver Policy
- 6 Audit Policies
- 7 Out-of-state Vehicle Notification Letter
- 8 Inspector Training
- 9 Auditor Training
- 10 Resources, Staff, Budget, Facilities and Equipment
- 11 MOBILE5. Modeling
- 11.a Basic I/M Mobile 5.a Modeling Methodology and Documentation
- 11.b DC98 Enhanced I/M Mobile 5.a.h Modeling Methodology and Documentation
- 12 IM240 Specification

UTAH STATE IMPLEMENTATION PLAN SECTION X AUTOMOTIVE INSPECTION AND MAINTENANCE (I/M) PROGRAM PART B DAVIS COUNTY

1. I/M Performance Standard

Federal requirements EPA's I/M regulation, 40 CFR Part 51, Inspection and Maintenance Program Requirements, Final Rule November 5, 1992, specifies a model Basic I/M program. Utah is required by Section 182 of the Clean Air Act to implement an I/M program in Davis County that is at least as effective as the EPA's Basic Performance Standard. The Basic I/M performance standard is specified in 40 CFR 51.352. Regulators are not required to implement the exact elements specified in EPA's model I/M programs. EPA's I/M regulations instead require a performance demonstration that local I/M programs result in automotive emissions equal to or less than predicted for the EPA model I/M program. State and local governments may choose options best suited for their area to meet the performance standard.

Basic I/M program performance standard Davis County's Basic I/M program exceeds the Basic I/M performance standard for all pollutants, although the EPA only requires the demonstration for each pollutant which caused an area to be subject to an I/M program. Davis County is in a moderate ozone non-attainment area. Achieving EPA's Basic I/M performance standard in Davis County will result in no increase in oxides of nitrogen (NOX) as a result of the Basic I/M program.

Basic I/M Program MOBILE modeling The performance standard demonstration made use of EPA's MOBILE5.a model. The MOBILE5.a model is able to calculate emission factors, grams of a particular pollutant per vehicle mile traveled across the fleet in an area (G/VMT), given information about the fleet, climate, fuel characteristics, and I/M programs in a local area. MOBILE5.a was used for the Basic I/M performance standard demonstration analysis. The MOBILE5.a input and output files for the modeling performed to evaluate the emission reduction benefits for Davis County's Basic I/M program are found in the Appendices, Section X, Part B.¹ Table X.B.1 summarizes the attainment milestones, the applicable performance standard and program. The modeling demonstrates compliance with the federal Basic I/M performance standard as required by the ozone maintenance plan until the DC98 enhanced I/M program is implemented.

	DAVIS COUNTY BASIC I/M PERFORMANCE STANDARD ANALYSIS SUMMARY (Basic I/M Program effective until replaced by DC98 I/M program					
				no later	than January 1, 1998)	
pollutant	program modeled	emission factors in grams/mile				
		January 1	1997	2000	2003	
VOC	Basic Performance Standard		2.70	2.47	2.30	
	Basic Program Target		2.64	2.40	2.22	
NOx	No I/M or ATP Program		2.60	2.38	2.23	
	Basic Program Target		2.53	2.32	2.17	
		January 1	1996	2000		
CO	Basic Performance Standard		22.85	18.69		
	Basic Program Target		21.02	17.92		

TABLE X.B.2

The PM10 contingency plan adopted on July 1, 1994, requires implementation of an enhanced I/M program in Davis and Salt Lake Counties if Salt Lake County violates the PM10 standard. Commitments from the Davis County Commissioners to implement an enhanced I/M program, as required by the SIP, are in Section IX, Part A, Appendices. The enhanced I/M program can be revised in the future by the Davis County Commissioners as long as the revised program meets all the applicable performance standards documented in the Ozone Maintenance Plan, Section IX.D.2.

2. Network type

Basic I/M Program The Davis County Basic I/M program is a decentralized, test-and-repair network consisting of approximately 93 stations using a two-speed idle test and anti-tampering inspection. Letters of opinion from the Utah Attorney General's Office and the Davis County Attorney's office verifying authority to implement the specified Basic I/M program network in Davis County are provided in Section X, Part A, Appendices.

DC98 Enhanced I/M Program Beginning no later than January 1, 1998, Davis County will implement the DC98 I/M program. The hybrid DC98 network will include a centralized, test-only component where the applicable fraction of the vehicle fleet will be subject to an IM240 test. The hybrid program will include an enhanced Basic I/M program component using UTAH91 analyzers at decentralized, test-and-repair facilities to test the remainder of the vehicle fleet. The enhanced UTAH91 program includes NOx emission control device functional tests, a gas gap pressure test, more efficient data recovery and analysis, and more audit emphasis on effectiveness of repairs. A training and certification program for technicians and a rating system to grade the capability of a facility to repair failed vehicles will be established across the entire DC98 network. Letters of opinion from the Utah Attorney General's Office and the Davis County Attorney's office verifying authority to implement the specified DC98 I/M program network in the subject area are provided in Section X, Part B, Appendices, along with the Davis County I/M ordinance.

Projected Number and types of I/M Stations in Davis County			
Effective Date	Program	Number	
or to Jan 1, 1998	Basic	93	
anuary 1, 1998	Decentralized DC98	75	
anuary 1, 1998	Centralized DC98	1 (5 lanes)	

3. Tools and resources

Funding mechanisms Davis County's I/M program is funded through two mechanisms. At the time of registration, a fee of \$1 per car is collected by the Davis County Tax Assessor's Office or Utah Tax Commission Motor Vehicle Customer Service Division. Davis County sells I/M inspection certificates to inspection stations for \$2.25 each. The fees are dedicated to I/M needs. Furthermore, Davis County charges fees for various permitting activities. A fee schedule can be found in the appendices to Davis County's I/M ordinance, which is found in the Appendices for Section X, Part B. The fee schedule is placed into an appendix so that it can be revised quickly, as needed, to support the program without taking the entire ordinance through rulemaking. Davis County began its I/M program in 1984. Past performance has demonstrated that adequate funding of Davis County's I/M program can be maintained.

Funding requirements Davis County will continue to allocate funding as needed to comply with the relevant requirements specified in Utah's SIP; Utah statutes; Davis County ordinances and policies; and the federal I/M program regulation. Program budgets will include funding for resources necessary to adequately: manage the program; conduct covert and overt audits, including repairs as specified in the I/M Program quality assurance section; assist and educate inspectors, station owners, and the public; manage, analyze, and report data; ensure compliance with the program by inspectors, stations, and vehicle owners; and evaluate and upgrade the programs. Budgets and descriptions of personnel resources, facilities, and equipment for Davis County's I/M program are provided in Section X, Part B, Appendices.

4. Test convenience

Basic I/M program There are approximately 100 permitted Basic I/M stations currently available within Davis County. Specific operating hours for the Basic I/M stations are not specified. Some stations that test and service only one type of vehicle are permitted. It may not be practical to have a sports car tested at a heavy duty truck repair facility. To date, no complaints have been received about test inconvenience. Also, there are government and private fleet permitted stations that are not open to the public.

DC98 I/M program It is anticipated that only about 80% of the existing decentralized, Basic I/M stations will convert over to Davis County's DC98 program starting on January 1, 1998. Davis County will have at least one centralized, IM240 test station ready by January 1, 1998. Davis County has determined that the IM240 facility can handle up to 50,000 vehicles per year. In its first year, Davis County projects about 45,000 vehicles being tested in its IM240 facility. The IM240 facility operating hours are planned to run from Monday through Friday 8 AM to 6 PM, and on Saturday from 9 AM to 4 PM. The operational hours may be extended, if necessary. The IM240 facility is located near the geographical center of the county and no further than 13 miles from any residence in the county. The facility is adjacent to the freeway. The location was chosen with consumer convenience as the primary consideration.

5. Vehicle Coverage

Subject fleet The Davis County I/M ordinance specifies that all model year 1968 and newer model year light duty vehicles, light duty trucks, and heavy duty trucks registered or principally-operated in Davis County are subject to the I/M programs except for exempt vehicles. Vehicle coverage is discussed in greater detail in the Davis County I/M ordinance provided in Section X, Part B, Appendices. Statistics for the subject vehicle fleet by vehicle type, model year, vehicle class, and weight class are included in Section X, Part B, Appendices. The data was compiled for the 1990 emissions inventory and has been subjected to a comprehensive quality assurance effort. Vehicles older than model year 1986 will be tested only at the decentralized Basic I/M stations. Beginning on January 1, 1998, vehicles of model year 1986 and newer will be periodically tested at the DC98 centralized, 5-lane, IM240 facility. Each year, vehicles for three model years will be selected for inspection at the test-only IM240 facility. If required to meet the ozone performance standard, additional model years may be added. OBD II testing will be performed as part of the DC98 program and as described in the Davis County ordinance, Section X, Part B, Appendices.

Alternative fuels Vehicles operated on alternative fuels such as propane, alcohol, and natural gas are also subject to the program. Dual-fueled vehicles are tested twice, once on each fuel.

Government fleet Section 41-6-163.6(1)(b) requires that all vehicles owned or operated in Davis County by federal, state, or local government entities comply with the Davis County I/M program. Under its Basic I/M program, Davis County permits government stations and inspectors to perform I/M inspections. The I/M station and inspector permit requirements are the same for government fleets as for private or commercial stations and inspectors. Some

government agencies may choose to have their vehicles inspected at a decentralized I/M station. Davis County requires submittal of a list of subject vehicles and a certificate of compliance or waiver for each vehicle every year. See Section X, Part B, Appendices, for the waiver policy developed by Davis County. However, beginning on January 1, 1998, if the government vehicle is of a model year subject to centralized, IM240 testing, it will have to be tested at a Davis County approved IM240 test facility.

Vehicles owned by students and federal employees Section 41-6-163.3(5) requires universities and colleges located in Utah's I/M areas to require proof of compliance with the I/M program for vehicles which are permitted to park on campus regardless of where the vehicle is registered. Vehicles operated by federal employees and operated on a federal installation located within an I/M program area are also subject to the I/M program regardless of where they are registered. Proof of compliance consists of a current vehicle registration in an I/M program area or an I/M certificate of compliance or waiver, or evidence of exempt vehicle status as specified in in this section.

Farm truck exemption Eligibility for the farm truck exemption from the I/M programs is specified in Section 41-6-163.6(4) and must be verified in writing by the Davis County I/M program staff. The owner must sign an affidavit on Utah State Tax Commission form TC-838 that vehicle use will be limited to agricultural activities. A copy of the form is provided in Appendices of Section X, Part A. Due to past abuses by vehicle owners, the counties strictly limit use of the farm truck exemption.

Diesel vehicles Diesel vehicles are no longer exempt from I/M. Davis County implemented its diesel I/M program in 1994.

New vehicle exemption Proof that a vehicle is new and being registered for the first time is established by presentation of a Manufacturer's Statement of Origin (MSO) at the time of registration.

Out-of-state exemption Vehicles registered in Davis County but operated out-of-state are eligible for an extension. The owner must complete Utah State Tax Commission form TC-810 in order to be registered in Davis County. The owner must explain why the vehicle is unavailable for inspection. Common situations include Davis County residents who are military personnel stationed outside of the state, students attending institutions of higher education elsewhere, and people serving missions. If the temporary address of the owner is located within another I/M program area listed on the back of the form, the owner must submit proof of compliance with that I/M program at the time of, and as a condition precedent to, registration or renewal of registration. The vehicle owner must identify their anticipated date of return to the state and is required to have the vehicle inspected within 10 days after the vehicle is back in Davis County. The Davis County maintains a record of such exemptions and requires submission of an I/M inspection certificate or waiver at the indicated time. A copy of the Tax Commission form is found in Section X, Part A, Appendices.

Exempt vehicle statistics Motorcycles, farm vehicles, and new vehicles being registered for the first time are exempt. Statistics for exempt vehicles are provided in Table X.B.4.

Vehicles Exem (provided by Utah	ot from I/M Program Requirements in Davis Cou Motor Vehicle Customer Service Division January	nty 1995)
motorcycles	2,687	
farm trucks (over 12,000 GVW)	157	
farm trucks (< 12,000 GVW)	112	
new vehicles	1,275	
total	4,231	

TABLE X.B.4

Unregistered vehicles Law enforcement agencies conducted random roadblock surveys in 1992. The number of unregistered vehicles or vehicles with expired registrations in Davis County is unknown. A data summary of the 1992 survey is provided below.

	1992 Registered and Unregistered vehicle data (Utah Highway Patrol and Motor Vehicle Customer Service Division data)						
County	Vehicles Registered	Registration Citations	Registration Warnings				
Davis	126,013	1,507	375				

TABLE X.B.5

Roadside I/M program element The Davis County I/M ordinance requires that vehicles available for rent or use in Davis County are subject to its I/M program. To the extent practicable, all vehicles principally-operated in the county are subject to the I/M program.

6. Test procedures and standards

Specifications Detailed specifications for the I/M test procedures and standards are described in the Davis County I/M ordinance provided in Section X, Part B, Appendices. The UTAH91 Analyzer specifications are provided in Section X, Part A, Appendices. The IM240 test will be compliant with EPA-AA-RSPD-IM-96-1 provided in Section X, Part A, Appendices. Specifications for the test procedure and equipment for the Basic I/M program were developed according to good engineering practices to ensure test accuracy. Specifications for the test procedure and equipment for the Enhanced (DC98) I/M program are being developed according to good engineering practices to ensure test accuracy 1, 1998 program implementation.

Basic I/M Program test procedure and analyzer Davis County's Basic I/M program uses EPA's PRECONDITIONED TWO SPEED IDLE TEST as specified in EPA-AA-TSS-I/M-90-3 March 1990, Technical Report, "Recommended I/M Short Test Procedures for the 1990's: Six Alternatives." All Basic emissions inspections are performed using the UTAH91 Analyzer, a BAR90-type emissions analyzer. The UTAH91 Analyzer calibration specifications and emissions test procedures meet the minimum standards established in Appendix A of the EPA's I/M Guidance Program Requirements, 40 CFR Part 51 Subpart S. Covered vehicles are defined in X.B.5. All covered vehicles in Davis County are subject to the Basic I/M test procedure and inspected using the UTAH91 analyzer as specified in Section X, Part A, Appendices.

Enhanced I/M (DC98) test procedure and analyzers A major component of the DC98 program will be to identify vehicle model years most likely to be high emitters of oxides of nitrogen (NOX) and hydrocarbons (HC). Davis County estimates that the selected model years will include between 40,000 and 45,000 vehicles. The selected model years will be tested using IM240 at Davis County's centralized facility in accordance with EPA-AA-RSPD-IM-96-1. The IM240 test will be for pass/fail for these vehicles. Davis County will also interrogate vehicles with On Board Diagnostics II (OBD II) capability at its centralized facility. Once a vehicle passes an IM240 test, it may be inspected for 2 -3 years at one of the decentralized, DC98 I/M facilities. The decentralized, DC98 I/M facilities will use a two-speed idle test using an enhanced UTAH91 Analyzer, a BAR90-type emissions analyzer. The Enhanced UTAH91 Analyzer calibration specifications and emissions test procedures meet the minimum standards established in Appendix A of 40 CFR Part 51 Subpart S. Gas cap and EGR valve function tests will be included in the DC98 program.

Covered vehicles are defined in Section X.B.5. Until January 1, 1998, all covered vehicles in Davis County are subject to the Basic test procedure and inspected using the UTAH91 analyzer as specified in this section. On January 1, 1998, all covered vehicles will be tested using IM240 or an Enhanced Basic test procedure under Davis County's DC98 program as designated by county ordinance. Vehicles subject to the Enhanced Basic program will be inspected using the Enhanced UTAH91 analyzer as specified in this section.

Pre-inspection emissions-related repairs Inspectors in the Davis County test-and-repair network are required to perform the emissions test prior to making any emissions-related repairs when a vehicle is presented for an emissions inspection.

Safety issues Vehicles presented in unsafe condition must be repaired before inspection. Vehicles are also subject to an annual safety inspection administered by the Highway Patrol. Submission of proof of compliance with the safety program is also required as a condition for registration or renewal of registration. Most owners in Davis County's test-and-repair network have the safety and emissions inspection performed at the same time as the emissions inspection. Data relative to the safety inspection can be recorded in the UTAH91 Analyzer. The Davis County I/M program is administered with close cooperation with the Utah Highway Patrol Safety Program. The IM240

program equipment, including dynamometers, shall be operated in accordance with manufacturer's specifications and EPA-AA-RSPD-IM-96-1 to prevent injury or damage to people or equipment.

Exhaust leaks The UTAH91 analyzer measures exhaust carbon monoxide (CO) and carbon dioxide (CO2). Exhaust CO + CO2 readings of less than 6% indicate a leaky exhaust system and cause the UTAH91 analyzer to abort the inspection. See section 3.3.30C of the UTAH91 analyzer specifications in Appendices of Section X, Part A.

Emission standards Davis County's I/M ordinance includes hydrocarbon (HC), oxide of nitrogen (NOX) and carbon monoxide (CO) emission standards in an appendix to allow for quick adjustment of the standards in case actual failure rates fall below the level specified in the State Implementation Plan. Only vehicles tested using IM240 will be tested for NOX. Vehicles must pass HC, NOX and CO emission standards, as applicable. The emission standards for the Basic I/M and DC98 programs were used in the MOBILE5.a and MOBILE5.a.h modeling that was conducted to demonstrate compliance with the respective I/M performance standards.

Stringency The Davis County I/M program will adjust its tailpipe emission standards as necessary to maintain a stringency rate of at least 22% for pre-81 model year vehicles, the stringency rate used in the Basic I/M performance standard modeling demonstration.

Re-test standards The same test procedure and emission standards are used for initial tests and retests, regardless of which part a vehicle may have failed during an initial test. The UTAH91 and DC98 test procedure requires an official test, once initiated, to be performed in its entirety regardless of intermediate outcomes, except in the case of invalid test conditions, unsafe conditions, or the fast pass/fail algorithms.

Anti-tampering provisions The Davis County I/M program requires a visual emissions control device inspection to determine whether the air system, catalyst, fuel inlet, exhaust gas recirculation (EGR) valve, evaporative system, positive pressure crankcase valve (PCV), and gas cap are present, appear to be properly connected, and appear to be the correct type for the certified vehicle configuration. Regardless of the vehicle model year, Davis County does not allow waivers for tampered vehicles or money spent to repair tampered or missing emission control devices to be applied towards a minimum waiver cost. Davis County requires repair of any tampering on model year 1984 and newer vehicles including repair of any tampering of the air system, catalyst, fuel inlet, exhaust gas recirculation (EGR) valve, evaporative system, positive pressure crankcase valve (PCV) and gas cap. The catalytic convertor must be replaced on vehicles that fail due to a tampered fuel inlet restrictor.

Engine changes After an engine change, vehicles are tested to the tailpipe emission standards and anti-tampering requirements applicable to vehicles of the chassis model year. Mixing vehicle classes (e.g., light-duty with heavy-duty) and certification types (e.g. California with federal) within a single vehicle is considered tampering.

Fuel switching Vehicles that are switched to a fuel type for which there is no certified configuration are tested according to the most stringent emission standards for that vehicle model year and vehicle type.

7. Test Equipment

Specifications Written technical specifications for the UTAH91 Analyzer, a BAR90-type computerized emissions analyzer, are provided in Appendix 10. The DC98 IM240 test equipment will be compliant with EPA-AA-RSPD-IM-96-1. Detailed specifications for the DC98 test equipment will be developed in time to support the January 1, 1998 program implementation date.

UTAH91 analyzer access restrictions An inspector access code is required to use the UTAH91 analyzer for official tests, a service access code to repair or service the analyzer, and an auditor access code to access the audit functions. DOS functions are not accessible to station owners, inspectors, or analyzer service personnel. Programming changes are made by county I/M auditors from disks supplied by the analyzer manufacturer.

UTAH91 data security provisions Manual data entry is minimized. For initial inspections, the inspector enters vehicle registration and vehicle information from the keyboard. Data elements are described in the UTAH91 analyzer specifications. For retests, the inspector calls up the initial test file, compares the vehicle and owner data, and confirms the VIN and license plate data. Data regarding inspections, analyzer calibration and service, lock-out activities, and audit information are stored to a secured disk drive and retrieved by county auditors at least once a month.

UTAH91 automated test procedure The UTAH91 analyzer automatically reads all test measurements, records test results in the computer database, determines whether the vehicle has passed or failed a test, and prints vehicle inspection reports and inspection certificates for all subject vehicles. The analyzers are capable of simultaneously

sampling dual exhaust vehicles. The analyzer bench includes two non-dispersive infrared (NDIR) analyzers for carbon monoxide, carbon dioxide, and hydrocarbon measurements (one low range and one high range), and one NDIR analyzer for carbon dioxide measurement. The test procedure is automated to the highest degree practical to minimize the potential for intentional fraud and/or human error.

UTAH91 security lockouts The analyzers are programmed to trigger lock-outs when abuse or tampering occur. Lock-outs occur after any security system is tampered, failure to conduct or pass periodic calibration tests, or the data recording medium is full. The analyzer can not be used until the lock-out has been cleared by a Davis County I/M auditor. The analyzer automatically keeps an electronic record of all lock-outs including the date of the lock-out, the reason for the lock-out, and the date and person that cleared the lock-out.

UTAH91 certified analyzer use restriction Since September 1, 1991, the Basic I/M program has required that official emissions tests be conducted only on registered and certified UTAH91 analyzers. A description of the certification procedure is provided in Appendices of Section X, Part A. There have been several updates of the UTAH91 Analyzer specifications to date and more will follow, as necessary, to accommodate new technology vehicles and changes to the program.

8. Quality Control

General quality control specifications The UTAH91 Analyzer specifications and Davis County I/M Program ordinance were carefully designed to insure that emission measurement equipment is calibrated and maintained properly, and that inspection, calibration records, and maintenance records are accurately created, recorded, and maintained. The specifications meet the test equipment quality assurance practices described in 40 CFR 51 Subpart S Sec. 51.359 and Appendix A.

UTAH91 automatic electronic quality assurance features Operational analyzer quality assurance measures such as analyzer calibration, zero and span check, hydrocarbon hang-up check, and leak check are mandatory automatic analyzer capabilities. Gas accuracy tolerances, dilution limits, analyzer warm up requirements, system response time requirements, optical correction factors, and interference effects are also addressed in the analyzer specifications. If the checks are not performed on schedule or identify measurements outside of acceptable limits established in the specifications, a lock-out occurs preventing use of the analyzer until such problems are corrected. See Sections 2.12, 2.13, and 2.18 of the UTAH91 Analyzer specifications. Records of all quality assurance activities with respect to the analyzer are automatically recorded in the analyzer's electronic database and evaluated by Davis County I/M auditors on a regular basis. Section 1.7 discusses requirements for assurance that unauthorized access to the I/M database in the analyzer is prevented. Attempts to deliberately avoid or defeat analyzer or inspection quality assurance provisions result in disciplinary action against the I/M mechanic and/or station.

UTAH91 analyzer maintenance Section 1.8 of the UTAH91 Analyzer specifications describes required services, warranty provisions, and documentation that analyzer manufacturers must provide to customers. It includes ensuring that the analyzer meets the quality assurance specifications at the time of delivery, that routine quarterly preventative maintenance is performed, training on how to use, maintain, and operate the analyzer is provided by the manufacturer, and that if repair of defects can not be made promptly a temporary analyzer replacement is provided. Service activities are recorded in the analyzer's electronic database. Davis County has conducted a survey of analyzer owners to determine compliance with these provisions. Failure of an analyzer manufacturer to meet quality assurance specifications could result in de-certification of that manufacturer's product for use in Davis County.

UTAH91 document security Document security was a high priority during the UTAH91 analyzer design phase. The analyzer tracks the unique certificate numbers and ensures that the certificate printed matches the test number. Missing certificate numbers are stored in the analyzer database for auditor review. The certificates are printed on a dedicated and locked printer. Only permitted inspectors have access to the certificate printer and storage area. Access to the certificates is only possible for the purpose of loading or aligning certificates in the printer. Attempts to access this area at other times or without an access code sets a lock-out that only Davis County auditors can clear. The certificate storage area is designed with redundant security systems including both hardware and software locks. See Section 2.16 of the UTAH91 analyzer specifications. The blank certificates are commercially printed with sequential and unique serial numbers on counterfeit-resistant security paper.

UTAH91 analyzer certification Sound engineering practices were followed during the design and certification of the UTAH91 analyzer to insure accurate and repeatable inspections under a range of environmental conditions. Manufacturer owner's manuals, operating instructions, and warranty provisions were also reviewed during the certification process. Comprehensive records of the certification process have been maintained.

General UTAH91 analyzer security provisions The Davis County I/M ordinance requires use of a certified and registered UTAH91 analyzer for official inspections at the decentralized facilities. Inspection records include the analyzer registration number. The ordinance makes it illegal to alter analyzer software or hardware without written approval. Analyzer calibration requirements, maintenance, and warranty provisions are also specified in the Davis County I/M ordinance. A copy is provided in the Section X, Part A, Appendices.

9. Waivers

Waiver rate Davis County will take corrective action as needed to maintain a maximum waiver rate of 1% of the initially failed vehicles or the Utah Air Quality Board will revise the SIP and emission reductions claimed based on the actual waiver rate. The conditions for issuing waivers are legally authorized and specified in the Davis County I/M ordinance to meet the minimum waiver issuance criteria specified in 40 CFR Subpart S 51.360.

Waiver procedures The Vehicle Inspection Report (VIR), printed after each inspection and provided to the vehicle owner/operator, includes warranty and waiver information, if the vehicle failed the emissions inspection. A waiver document may be issued only by Davis County I/M technical center staff and only after verification of required documentation. Any tampered, missing, or inoperable emission control devices must have been replaced or repaired. At least \$100 for 1968 through 1980 model year vehicles and \$200 for 1981 and newer model year vehicles must have been spent on acceptable emission repairs as verified by a Davis County I/M program auditor by physical examination of the vehicle and review of the repair documentation. Repair documentation, such as receipts, are copied and retained by auditor to prevent reuse. Repairs made by a permitted I/M inspector include the cost of labor. The retest must reflect a reduction of carbon monoxide, oxides of nitrogen (for IM240 only) and/or hydrocarbon emissions after repairs. Vehicles still under the federal emissions warranty are not eligible for a waiver until all warranties are exhausted. Under IM240, vehicles displaying an OBD II fault code should be repaired before a waiver can be issued for the vehicle. Warranted repair and tampering repair may not be applied to the repair cost waiver limits. Waivers are only valid for one test cycle. The vehicle owner surrenders the original waiver document at the time of registration; copies are not accepted for registration purposes. Specific provisions regarding waivers may be found in the Davis County I/M ordinance and the Utah Tax Commission Division of Motor Vehicle policy manual which is available upon request. The I/M program does not provide for time extensions to relieve economic hardships in obtaining emission-related repairs.

10. Motorist compliance enforcement

Registration denial Davis County's I/M program is enforced by means of registration denial. Vehicle owners must present proof of compliance with the I/M program, a waiver, or evidence of exemption from the I/M program as a condition precedent to vehicle registration or registration renewal. See Section X, Parts E and G for a more detailed discussion of inspection frequency, inspection scheduling, license plate requirements, and enforcement of the registration requirements. Citations are routinely issued to operators of vehicles with expired or missing license plates during routine traffic stops, parking lot inspections, and roadblocks. As specified in Section 41-1a-1303 (Section X, Part A, Appendices), driving without registration is a Class C misdemeanor. The penalty for a Class C misdemeanor is imprisonment of no more than 90 days and \$750 for persons or up to \$1000 for corporations, associations, partnerships, or government instrumentalities. In addition to paying a fine, the motorist must register the vehicle. It is currently a Class B misdemeanor to violate a county I/M ordinance. The penalty for a Class B misdemeanor is an imprisonment not exceeding six months and for persons a fine of up to \$1000 or for corporations, associations, partnerships, or government instrumentalities a fine of up to \$5000. Copies of the relevant statute are provided in Section X, Part A, Appendices. In Utah, the magnitude of such penalties is a judicial rather than an administrative decision. Per Section 41-1a-1315 falsification of evidences of title and registration is a second degree felony.

Certificate of Compliance The Certificate of Compliance is dated by the UTAH91 or DC98 analyzer immediately after a passing inspection is completed. The certificate is only valid for registration purposes for two months. At the same time the analyzer also prints the following information on the certificate to ensure unambiguous vehicle identification: the vehicle identification number (VIN), license number, model year, make, and model. A sample of the Certificate of Compliance is in Appendix C of the UTAH91 specifications. The certificates are only printed in the event that the vehicle passed the emissions inspection. Separate documentation, including the same vehicle information, is used for waivers.

Fuel changes to non-subject status Vehicle changes that would result in registration changes from a subject to exempt status require physical confirmation by Davis County I/M program personnel at the I/M technical center. Falsification of registration or title information is a felony offense.

Title transfers Proof of compliance with the I/M program is required for a title transfer. The system ensures that owners are not able to avoid the program by extending the inspection date through manipulation of the title and registration system.

Davis County I/M program staff, peace officers, and Utah Tax Commission's Motor Vehicle Customer Service Division routinely work together to ensure that motor vehicle owners that move into an I/M program area complete registration transfer including compliance with the I/M program. Except for higher education students and active duty military personnel, people are required to register their vehicles in the county in which they are domiciled. As discussed in the Vehicle Coverage section, although these two exempted classes of vehicle owners do not have to register their vehicles in Utah, they do have to comply with the I/M programs. Employment status, maintenance of a residence, enrollment of children in local schools, and voting districts are considered when identifying persons in violation of this requirement.

The I/M program staff work with citizens, the Motor Vehicle Customer Service Division and county attorneys to identify and prosecute people that illegally transfer registration to a non-subject area to avoid the I/M program. The process is very labor intensive. There are many legitimate reasons to be operating a vehicle in an I/M program area that is registered elsewhere. Violators must be dealt with on a case-by-case basis. Persons caught to date have been subject to fines of around \$700. Those prosecuted and convicted could end up with a criminal record and actual jail time. Fraudulent registration of a motor vehicle is a felony offense. Most people confronted with evidence of their guilt and the seriousness of their offense, to date, have complied promptly. The involved agencies are developing more efficient methods of dealing with illegal registrations that result in exemption from the I/M programs.

Davis County is committed to a cooperative aggressive effort to ensure that vehicles operated in the county comply with the I/M program to ensure a compliance rate of at least 96%.

11. Motorist compliance enforcement program oversight

Utah Tax Commission, tax assessors, and county roles The Utah Tax Commission Motor Vehicle Customer Service Division and Davis County tax assessor deny applications for vehicle registration or renewal of registration without submittal of a valid certificate of compliance, waiver, or verified evidence of exemption. Proof is retained by the tax clerk, micro-photo-copied, and then destroyed. Altered or hand-written documents are not accepted. All certificate data is collected by Davis County I/M program auditors and subjected to scrutiny for evidence of any improprieties.

Database quality assurance The vehicle registration database is maintained and quality assured by the Motor Vehicle Customer Service Division. The I/M inspection database is maintained and quality assured by the Davis County I/M program staff. The Davis County I/M program has access to the Motor Vehicle Customer Service Division database and utilizes it for quality assurance purposes. The databases are subject to regular auditing, cross-referencing, and analysis. The databases are also evaluated using data obtained during roadblocks and parking lot surveys. Evidence of program effectiveness problems trigger additional joint enforcement activities.

Oversight provisions The oversight program includes verification of exempt vehicle status through inspection, data accuracy through automatic and redundant data entry for most data elements, an audit trail for program documentation to ensure control and tracking of enforcement documents, identification and verification of exemption-triggering changes in registration data, and regular audits of I/M inspection records, I/M program databases, and the Motor Vehicle Customer Service Division database.

Enforcement staff quality assurance Davis County I/M program auditors and tax clerks involved in vehicle registration are subject to regular performance audits by their supervisors. All enforcement personnel (direct and indirect) involved in the motorist enforcement program are subject to disciplinary action, additional training, and termination for deviation from procedures. Specific provisions are outlined in the Motor Vehicle Customer Service Division procedures manual which is available upon request, the Davis County I/M audit policy documents provided in Davis County I/M ordinance provided in Section X, Part A, Appendices.

Co-operative enforcement oversight effort The Motor Vehicle Customer Service Division, Utah Division of Air Quality, Utah Highway Patrol and Davis County I/M program staff meet at least once a month to ensure on-going high quality oversight of joint motorist compliance programs. EPA audit of this process is authorized if measures to protect tax-payer confidentiality acceptable to Motor Vehicle Customer Service Division are exercised.

12. I/M Program quality assurance

Station/inspector audits Davis County regularly audits all permitted I/M inspectors and stations to ensure compliance with the county I/M ordinance and policies. Particular attention is given to identifying and correcting any fraud or

incompetence with respect to vehicle emissions inspections. Compliance with record keeping, document security, analyzer maintenance, and program security requirements are scrutinized. The inspector's skill level is also evaluated during audits. Another major purpose of the audits is to retrain inspectors, as necessary, as soon as problems are identified. Documentation sufficient to support a legal case to suspend or revoke a permit is also collected in the event of serious and/or repeated violations. Most stations and inspectors are audited every month and all at least quarterly.

Covert audits Davis County, to the extent possible, performs a covert audit of each inspector and station at least once a year. The number of covert audits at least equals the number of permitted inspectors. Covert audits are performed using a variety of vehicles that are representative of the subject fleet that are set to fail across a full range of malfunctions. Suspected problem stations and inspectors are targeted for earlier and more frequent audits. Complaints also trigger additional audits.

Covert performance audits shall include:

Remote visual observation of inspector performance, which may include the use of aids such as binoculars or video cameras, at least once per year per inspector in high-volume stations (i.e., those performing more than 4000 tests per year);

Site visits at least once per year per number of permitted inspectors (per inspector FTE) using covert vehicles set to fail (this requirement sets a minimum level of activity not a requirement that each inspector be involved in a covert audit); and

For stations that conduct both testing and repairs, at least one covert vehicle visit per station per year including purchase of repairs and subsequent retesting if the vehicle is initially failed for tailpipe emissions.

Electronic audit capabilities The Davis County I/M program equipment perform various analyses to identify statistically inconsistent data indicative of problem stations and inspectors. Overt audit records are maintained electronically in the analyzer. After overt audits, the auditor retrieves the data on the analyzer diskette containing the audit, vehicle inspection, and analyzer service, maintenance, and calibration records dating back to the previous audit. The data from each audit is added to the comprehensive central Davis County I/M database. Further analysis of the central database results in identification of stations and inspectors for which additional audits are performed.

Auditor quality assurance Auditors receive 24 hours of formal classroom instruction and are provided on-the-job training in: the use of the UTAH91 analyzer; the Davis County I/M program ordinance; basic air pollution control; basic principles of emissions-related motor vehicle engine repair; emission control systems; evidence gathering; administrative procedures and laws; quality assurance practices; and covert audit procedures. Davis County sends auditors to additional automotive emissions-related training and meetings on a regular basis. Auditor supervisors audit the I/M program auditors by reviewing their documentation and also auditing a number of their stations at least once every year.

Written audit procedures Copies of Davis County's I/M program overt and covert audit procedures are provided in Section X, Part B, Appendices. A detailed description of the audit capabilities of the UTAH91 analyzer is found in Section 3.9 of the UTAH91 analyzer specifications. It is anticipated that the DC98 Enhanced I/M program written audit procedures and analyzer's automatic audit capabilities will be substantially similar to the UTAH91 program.

13. Enforcement against stations and inspectors

General enforcement provisions The Davis County I/M program is responsible for enforcement action against incompetent or dishonest stations and inspectors. The Davis County I/M ordinance includes a penalty schedule. For serious or repeated offenses, auditors are authorized to immediately suspend the station or inspector by locking out their UTAH91 analyzer(s). The county does not have legal authority to impose direct fines on stations or inspectors, but suspension or revocation of a station permit results in a substantial loss of income that is far in excess of \$100 fine suggested by the EPA guidance. The county can negotiate a fee settlement rather than a station suspension. Fee settlements are at least as much the station's anticipated income for emissions testing for the time during which the station would be suspended. A station permit may be suspended or revoked even if the owner/operator had no direct knowledge of the violation. In the case of incompetence, re-training is required before the permit is restored.

Davis County revised its penalty schedules to comply with the more stringent specifications included in 40 CFR 51.364. The Utah Air Quality Board adopted the revised penalty schedule for Davis County into Appendices of Section X, Part B, on January 30, 1995. At a minimum, inspector and station permit suspension shall be imposed for

at least 6 months (or a fee retainage or settlement penalty equivalent to the inspector's salary for that period) whenever a vehicle is intentionally improperly passed for any portion of the required test.

Suspension and revocation Suspension or revocation effectively bars an individual from further inspections because the auditor removes the inspector's authorization code from the UTAH91 analyzer. Evidence of indirect participation in emissions inspections by an individual while suspended or revoked would result in legal action against the station. If the station is suspended or revoked the analyzer is totally locked-out. The analyzers are initialized by an auditor for use at a single permitted station and only by inspectors permitted for that station. A record of the serial numbers of all registered analyzers and their locations is maintained by Davis County.

Enforcement records Davis County keeps comprehensive records of all audit activities, warnings, suspensions, and revocations and report enforcement activity statistics to the EPA and the executive secretary on an annual basis.

14. Data collection

UTAH91 analyzer inspection data The UTAH91 analyzer creates a detailed record of each emissions inspection performed including, but not limited to the following data, for each vehicle tested: test record number; inspection station number; inspector number; test system number; date of the test; emission test start time; the time final emission scores are determined; vehicle identification number (VIN); license plate number; test certificate number; gross vehicle weight rating (GVWR); model year, make, and type of vehicle; number of cylinders or engine displacement; transmission type; odometer reading; category of test performed (i.e., initial, first retest, or subsequent retest); fuel type of the vehicle; emission scores for HC, CO, and CO2 at idle and 2500 RPM; and results (pass/fail/not applicable) for visual inspection of the catalytic convertor, air system, gas cap, evaporative system, positive crankcase (PCV) valve; and the fuel inlet restrictor. The tailpipe emission standards for each type of vehicle is included in a look-up table in the UTAH91 analyzer. The UTAH91 analyzer automatically uses the appropriate standards for the type of vehicle being tested and makes a pass/fail determination. The inspection data is recorded by the UTAH91 analyzer during the inspection procedure.

IM240 inspection data The IM240 inspection data will include the UTAH91 analyzer inspection data modified to be in compliance with EPA-AA-RSPD-IM-96-1. Detailed Davis County IM240 inspection data parameters will be developed in time to support the January 1, 1998, DC98 program implementation.

UTAH91 analyzer quality assurance data Quality assurance data including a detailed history of all calibration (including the concentration values of the calibration gases), service, lockout, and document security events are also recorded and maintained by the UTAH91 analyzer. Each UTAH91 record includes, as applicable, the station number, mechanic access number, auditor access number, service access number, analyzer serial number, date, and activity time.

IM240 quality assurance data The IM240 quality assurance data will include the UTAH91 analyzer inspection data modified to be in compliance with EPA-AA-RSPD-IM-96-1. Detailed Davis County IM240 quality assurance data parameters will be developed in time to support the January 1, 1998, DC98 program implementation.

UTAH91 analyzer database specifications The programming criteria for the analyzer database is described in Section 3 of the UTAH91 analyzer specifications. Appendix F of the UTAH91 analyzer specifications contains a complete description of the electronic data records. The data disk containing inspection and quality assurance information is removed from the UTAH91 analyzer by an auditor at least once a month during overt audits and maintained permanently in the county's central I/M database.

DC98 data collection The DC98 data collection system will meet the requirements specified under 40 CFR 51.365 and be substantially similar to the UTAH91 Basic I/M program data collection system. Detailed Davis County DC98 database specifications will be developed in time to support the January 1, 1998, DC98 program implementation.

15. Data analysis and reporting

Annual Davis County shall analyze I/M program data and submit annual reports to the U.S. Environmental Protection Agency and the executive secretary upon request. Beginning in July of 1995, Davis County will submit to EPA and the executive secretary an annual report, for January through December of the previous year, which provides statistics on the testing, quality assurance, and enforcement activities of its I/M program. At a minimum the annual reports will include all of the data elements listed 40 CFR Subpart S 51.366.

Biennial Beginning in July of 1996, and biennially thereafter, the Davis County shall submit a report to EPA and the executive secretary discussing all changes made in the program design, funding, personnel levels, procedures,

regulations, and legal authority. The report will also supply a detailed discussion of the impact of such changes upon the program, any weaknesses or problems discovered in the program over the previous two-year period, the steps that were taken to address those problems, the result of those corrective actions, and any future efforts planned.

16. Inspector training and permitting

Inspector permitting and initial training No person may conduct an official I/M inspection unless they are permitted. Davis County requires formal training prior to permitting inspectors. Each class includes at least the following information: the causes and effects of air pollution; the purpose, function, and goal of the I/M program; I/M inspection ordinances, policies, and procedures; technical details of the test procedures and the rational for their design; emission control device function, configuration, and maintenance; quality control procedures and their purposes; public relations; and safety and health issues related to the I/M inspection process. Weber State University teaches an inspector training class on a contract basis for Davis County. Davis County regularly monitors and evaluates Weber State University's inspector training program delivery. Inspector candidates will not be issued a permit unless they have passed a written test with at least 80% (or lower if an occupational analysis justifies it) correct responses and a hands-on test during which the trainee demonstrates the ability to properly conduct all test procedures, calibrate the analyzer, properly utilize equipment, and to follow other I/M program requirements. The county will take appropriate steps to insure the security of the testing process.

Basic inspector permit renewal Inspector permits are valid for a period of one year, at which point refresher training and testing, are required prior to permit renewal. An auditor enters the inspector's permit expiration date in the analyzer(s) that the inspector is authorized to use. Starting 60 days prior to the inspector's permit expiration date the analyzer displays the message "Your mechanic permit expires MM/DD/YY". The analyzer locks-out inspectors that attempt to use the analyzer after their permit expires and displays the following message. "Your mechanic permit expired (date). You are not authorized to perform any emissions inspections at this time. Please contact your local I/M office." Auditors will not clear the lock-out until the inspector has renewed the permit. Davis County may require evidence of more comprehensive emissions-related automotive training as a prerequisite to inspector permit renewal.

Inspector permit suspension and revocation A determination of inspector incompetence or failure to comply with I/M program requirements may result in suspension or revocation or an inspector's permit prior to the annual expiration date. A permit to conduct I/M inspections is not a legal right but rather a privilege bestowed by Davis County and is conditional upon adherence to its I/M program requirements.

Inspector training authority and materials Authority to require mandatory I/M inspector training is established and described in the Davis County I/M ordinance. A description of the I/M inspector training programs and the written and hands-on tests is provided in Section X, Part B, Appendices.

17. Public information and consumer protection

General public information Davis County, along with the Utah Department of Environmental Quality, provides a comprehensive public education and protection program including strategies to educate the public on: Utah's air quality problems; ways that people can reduce emissions; the requirements of state and federal law; the role of motor vehicles in the air quality problem; the need for and benefits of a vehicle emissions inspection program; ways to operate and maintain a vehicle in a low-emission condition; how to find a qualified repair technician; and the requirements of the I/M program. Information is provided via direct response to inquiries for information, reports, classes, pamphlets, fairs, school presentations, workshops, news releases, posters, signs, and public meetings.

Davis County I/M technical center Davis County operates an I/M technical center staffed with trained auditors and capable of performing emissions tests. A major function of the I/M technical center is to serve as a referee station to resolve conflicts between permitted I/M inspectors, stations, and motorists. Auditors actively protect consumers against fraud and abuse by inspectors, mechanics, and others involved in the I/M program. Complaints made on a confidential basis are investigated and resolved in a manner that conceals the person's identity to ensure protection of whistle blowers. Auditors advise motorists regarding emissions warranty provisions and assist the owners in obtaining warranty-covered repairs for eligible vehicles. Applications for waivers are evaluated by auditors at the I/M technical center and issued only after visual verification that all the requirements for a waiver have been met. The I/M technical center also provides motorists with information regarding the I/M program, general air pollution issues, and emissions-related automotive repairs.

Vehicle inspection report A vehicle inspection report (VIR) is printed and provided to the motorist after each vehicle inspection. The VIR includes a public awareness statement about automotive emissions and lists additional ways that the public can reduce air pollution. The test results are detailed on the VIR. Information about vehicle emissions warranties and the benefits of emissions-related repairs are printed for vehicles that failed the test. Information about

waiver requirements and application procedures are printed on the VIR, if the vehicle has failed a retest, including the address and telephone number of the applicable I/M technical center. A complete description of the VIR is included in Appendix E of the UTAH91 analyzer specifications.

Davis County co-operative public education tools A variety of pamphlets and radio, television, and newspaper advertisements about automotive air pollution issues are developed and distributed by the Davis County I/M program in cooperation with other I/M counties and the Utah Division of Air Quality. The legislature authorizes funding each year for pass-through money from the state to Davis County for public education to help reduce vehicle emissions.

18. Improving repair effectiveness

High priority Davis County, along with the other three Utah I/M counties, initially implemented a major Basic I/M program revision on September 1, 1991. Shortly thereafter, the county I/M managers and the Utah Division of Air Quality staff jointly identified improvement of repair effectiveness as a high priority action item. The Governor's Clean Air Commission also recommended making affordable additional emissions-related training available. Full emission reductions will only be realized if the repair industry is able to competently diagnose and repair emissions-related defects.

Continuing education To that end, I/M program managers have worked with Utah's higher education institutions to develop and provide emissions-related automotive technology classes to technicians. Inspectors are also encouraged to take classes offered by trade organizations, automobile manufacturers, and dealers. Davis County's I/M program waives the re-permitting test for inspectors that submit transcripts reflecting that they have taken and passed one of the classes. The permit renewal tests are difficult enough to make this provision a good incentive. The classes are advertised in the Davis County I/M technical bulletins. Appendices for Section X, Part B include descriptions of some of the classes available in the community.

Pollution prevention The Utah Department of Environmental Quality Pollution Prevention Program is funding various projects related to pollution control curriculum being developed by the Weber State Automotive Technology department and students.

I/M program repair support activities In initiating improved automotive educational opportunities, the counties work on a day-to-day basis to ensure that repair information is available. I/M stations are required to have available up-to-date relevant automotive diagnostic references and tools as a condition for obtaining a permit. Davis County maintains a hot line to its I/M technical center that any mechanic can call for technical assistance related to vehicle inspection, diagnosis, and repair. Technical bulletins are regularly mailed to each permitted inspector with information regarding training schedules, common problems found with particular engine families, and diagnostic tips.

19. Basic and DC98 I/M SIP implementation

As required by 40 CFR Part 51.373(a) the Basic I/M SIP requirements not included in the September 30, 1993, adoption of Section X by the Utah Air Quality Board have been funded and implemented, including but not limited to the covert audits requirements specified in Section X.B.12 and the penalty provisions specified in Section X.B.13. The Utah Air Quality Board adopted the changes into Appendix 6 of Section X on January 30, 1995.

The Basic I/M program health regulations, policies, procedures, and activities specified in this I/M SIP revision have been implemented. The Enhanced DC98 I/M program requirements will be implemented no later than January 1, 1998. Davis County shall continue to implement and operate the I/M program until a maintenance plan, without an I/M program, is approved by EPA in accordance with Section 175 of the Clean Air Act as amended.

UTAH STATE IMPLEMENTATION PLAN

SECTION X

VEHICLE INSPECTION AND MAINTENANCE PROGRAM

PART C

SALT LAKE COUNTY

Adopted by the Utah Air Quality Board October 6, 2004

Table of Contents

1.	I/M performance standard	1
2.	Network type	2
3.	Tools and resources	3
4.	Test convenience	3
5.	Vehicle Coverage	4
6.	Test procedures and standards	5
7.	Test Equipment	8
8.	Quality Control	9
9.	Waivers	10
10.	Motorist compliance enforcement	11
11.	Motorist compliance enforcement program oversight	12
12.	I/M Program quality assurance	13
13.	Enforcement against stations and inspectors	14
14.	Data collection	15
15.	Data analysis and reporting	16
16.	Inspector training and certification	16
17.	Public information and consumer protection	17
18.	Improving repair effectiveness	18
19.	Improved I/M SIP implementation	19
18. 19.	Improving repair effectiveness Improved I/M SIP implementation	

SECTION X, PART C SALT LAKE COUNTY Appendices

- 1 Motor Vehicle I/M Program Ordinances
 - Salt Lake City-County Health Department Regulation #22A Governing the Motor Vehicle Emissions Inspection Maintenance Program for the Control of Air Contaminant Emissions from Motor Vehicles, March 5, 1998
 - 1.b Salt Lake County Council Resolution #3480, August 5, 2003- Fees
 - 1.c Salt Lake County Commission Resolution #2367, September 18, 1996

UTAH STATE IMPLEMENTATION PLAN SECTION X AUTOMOTIVE INSPECTION AND MAINTENANCE (I/M) PROGRAM PART C SALT LAKE COUNTY

1. I/M performance standard

Federal requirements EPA's I/M regulation, 40 CFR Part 51, Inspection and Maintenance Program Requirements, last amended at 66 FR 18156, April 5, 2001, specifies a model Basic I/M program. Utah is required by Section 182 of the Clean Air Act to implement an I/M program in Salt Lake County that is at least as effective as the EPA's Basic Performance Standard. The Basic I/M performance standard is specified in 40 CFR 51.352. Regulators are not required to implement the exact elements specified in EPA's model I/M programs. EPA's I/M regulations instead require a performance demonstration that local I/M programs result in automotive emissions equal to or less than predicted for the EPA model I/M program. State and local governments may choose options best suited for their area to meet the performance standard.

UM Performance Standard Salt Lake County began its *I/M* program in 1984. The *I/M* program exceeds the Basic *I/M* performance standard for all pollutants. Salt Lake County was redesignated as attainment for ozone on July 17, 1997 and is not classified for carbon monoxide.

Salt Lake Salt Lake County I/M program requirements The Utah Air Quality Board adopted an ozone maintenance plan for Salt Lake and Davis Counties on November 5, 1993. The plan was reorganized and adopted on January 5, 1995. Revisions to the ozone maintenance plan were adopted by the Board on June 5, 1996, and January 7, 1997; EPA approved the plan on July 17, 1997. The ozone maintenance plan required implementation of an improved I/M program no later than January 1, 1998. The ozone maintenance plan established a performance standard for both counties that is more stringent than the federal Basic I/M performance standard. Parts A and C of Section X, together with referenced appendices, demonstrate compliance with the improved IM program for Salt Lake County as specified in Part IX.D.2.g of the Ozone SIP and herein after referenced as Salt Lake County's Program.

I/M Program MOBILE modeling The MOBILE model is able to calculate emission factors, grams of a particular pollutant per vehicle mile traveled across the fleet in an area (G/VMT), given information about the fleet, climate, fuel characteristics, and I/M programs in a local area. The latest approved release of the MOBILE model was used to demonstrate the I/M performance standard. This analysis and documentation is included in the respective criteria pollutant section of the SIP. The modeling demonstrates compliance with both the federal Basic I/M performance standard and the Ozone Maintenance Plan Basic I/M performance standard. Attainment of this

performance standard only required I/M emission benefits derived from a test-andrepair network. Subsequent to this demonstration, EPA promulgated Additional Flexibility Amendments to the Vehicle Inspection Maintenance Requirements effective August 23, 2000. This rule change removed the mandatory I/M rule provision establishing the decentralized, test-and-repair network credit discount and permits areas to demonstrate increased I/M program effectiveness of their networks.

2. Network type

Salt Lake County's I/M program comprises a decentralized, test-and-repair network with approximately 340 stations. The network provides ASM2 tailpipe inspections in a decentralized test-and-repair network for light duty gasoline vehicles older than 1996. On-Board Diagnostic (OBD) compliant 1996 and newer model year vehicles undergo an applicable OBD inspection. Non-OBD compliant vehicles exceeding 85900 lbs GVWR and/or with full-time four wheel or all wheel drive undergo a Two-Speed-Idle (TSI) inspection. The Salt Lake Valley Health Department regulation administering the program is provided in Appendix C.1.

3. Tools and resources

Funding mechanisms Salt Lake County's I/M program is funded through two mechanisms. At the time of registration, the Utah State Tax Commission Motor Vehicle Customer Service Division collects an Air Pollution Control (APC) fee of \$3 per vehicle.. Those monies are remitted to the county and the fees are dedicated to I/M needs. In addition, the County charges fees for various permitting activities. A fee schedule can be found in Appendix C of the Salt Lake County Regulation #22A for the program.

Funding requirements Salt Lake County will continue to allocate funding as needed to comply with the relevant requirements specified in Utah's SIP; Utah statutes; county regulations and policies; and the federal I/M program regulation. Program budgets will include funding for resources necessary to adequately: manage the program; conduct covert and overt audits, including repairs as specified in Section 13 below; assist and educate inspectors, repair technicians, station owners, and the public; manage, analyze, and report data; ensure compliance with the program by inspectors, stations, and vehicle owners; and evaluate and upgrade the programs.

4. Test convenience

There are approximately 340 I/M stations currently permitted within Salt Lake County. Operating hours are not specified by the county. However, an I/M technician must be available for at least 40 hours per week at facilities open to the public. Some stations that test and service only one type of vehicle are permitted. Also there are government and private fleet permitted stations that are not open to the public.

5. Vehicle Coverage

Subject fleet All model year 1968 and newer model year light duty vehicles, light duty trucks, and heavy duty trucks registered or principally-operated in Salt Lake County are subject to the I/M program except for exempt vehicles. All vehicles six years and older on January 1 are tested annually; newer vehicles are tested every other year as per Utah 41-6-163(6), last amended in 2002. Modeling demonstrates that annual testing for newer vehicles is not needed to maintain the National Ambient Air Quality Standards in Salt Lake County.

Alternative fuels Vehicles operated on alternative fuels such as propane, alcohol, and natural gas are also subject to the program. Dual-fueled vehicles are tested twice, once on each fuel.

Government fleet Section 41-6-163.6(1)(b) of the Utah Code requires that all vehicles owned or operated in the I/M counties by federal, state, or local government entities comply with the I/M programs. Salt Lake County permits government stations and certifies inspectors to perform I/M inspections. The I/M station permit and inspector certification requirements are the same for government fleets as for private or commercial stations and inspectors. Some government agencies choose to have their vehicles inspected at a commercial I/M station. Salt Lake Salt Lake County requires submittal of a list of subject vehicles and a certificate of compliance or waiver for each vehicle every year.

Vehicles owned by students and federal employees Section 41-6-163.3(5) requires universities and colleges located in Utah's I/M areas to require proof of compliance with the I/M program for vehicles which are permitted to park on campus regardless of where the vehicle is registered. Vehicles operated by federal employees and operated on a federal installation located within an I/M program area are also subject to the I/M program regardless of where they are registered. Proof of compliance consists of a current vehicle registration in an I/M program area or an I/M certificate of compliance or waiver, or evidence of exempt vehicle status as specified in this section.

Rental vehicles Allvehicles available for rent or use in Salt Lake County are subject to its I/M program. To the extent practicable, all vehicles principally operated in the county are subject to the I/M program.

Farm truck exemption Eligibility for the farm truck exemption from the I/M programs is specified in Section 41-6-163.6(4) and must be verified in writing by Salt Lake County I/M program staff. The owner must sign an affidavit on Utah State Tax Commission form TC-838 that vehicle use will be limited to agricultural activities. A copy of the form is provided in the Technical Support Document for Section X, Part A. Due to past abuses by vehicle owners, Salt Lake County strictly limits use of the farm truck exemption.

Diesel vehicle exemption Salt Lake County implemented its diesel I/M program on January 1, 1997 in accordance with Salt Lake City-County Health Department Regulation #28.

New vehicle exemption Proof that a vehicle is new and being registered for the first time is established by presentation of a Manufacturer's Statement of Origin (MSO) at the time of registration.

Out-of-state exemption Vehicles registered in Salt Lake County, but operated out-ofstate are eligible for an exemption. The owner must complete Utah State Tax Commission form TC-810 in order to be registered without inspection documentation. The owner must explain why the vehicle is unavailable for inspection in Utah. Common situations include Utah citizens that are military personnel stationed outside of the state, students attending institutions of higher education elsewhere, and people serving religious assignments outside the area. If the temporary address of the owner is located within another I/M program area listed on the back of the form, the owner must submit proof of compliance with that I/M program at the time of, and as a condition precedent to, registration or renewal of registration. The vehicle owner must identify their anticipated date of return to the state and is required to have the vehicle inspected within 10 days after the vehicle is back in Utah. Salt Lake County maintains a record of such exemptions and requires submission of an I/M inspection certificate or waiver at the indicated time.

Exempt vehicles The following vehicles are exempt from inspection: motorcycles, electric powered vehicles, farm vehicles and equipment, construction equipment, offroad vehicles and new vehicles being registered for the first time.

6. Test procedures and standards

Specifications Detailed specifications for the I/M test procedures and standards are described in the Salt Lake Valley Health Department regulations provided in Section X, Part C, Appendices.

I/M Program test procedure The inspection for vehicles older than 1996 consists of a loaded-mode emissions test for concentrations of hydrocarbons (HC), carbon monoxide (CO), and oxides of nitrogen (NO_X), a functional inspection of the gas cap and a visual/tampering inspection of the PCV, EGR, AIR and catalytic converter systems. OBDII testing is performed on 1996 and later model year vehicles in compliance with federal statute. All emissions inspections are performed using the BAR97-compliant UTAH98 Analyzer. The UTAH98 Analyzer calibration specifications and emissions test procedures meet the minimum standards established in above referenced

specification. ASM2 testing uses a BAR97-compliant dynamometer. Gas cap and EGR valve function tests will be included in the UTAH98 program. Full-time and All-wheel four wheel drive and vehicles with a GVWR exceeding 8500 lbs undergo a Two-Speed Idle test unless they are OBD-compliant.

Pre-inspection emissions-related repairs Inspectors in the county's test-and-repair networks are required to perform the emissions test prior to making any emissionsrelated repairs when a vehicle is presented for an emissions inspection. All inspectors who conduct test-only inspections, are required to ask the vehicle owner or operator whether a tune-up or other emissions-related repairs have been performed within 6 weeks prior to the emissions inspection and to document the owner's response in the computer vehicle information database (VID).

Safety issues Vehicles presented in unsafe condition must be repaired before inspection. Vehicles are also subject to an annual safety inspection administered by the Highway Patrol. Submission of proof of compliance with the safety program is also required as a condition for registration or renewal of registration. Most owners in Salt Lake County's test-and-repair network have the safety and emissions inspection performed at the same time. Data relative to the safety inspection can be recorded in the UTAH98 I/M Analyzers. The Salt Lake County I/M program is administered with close cooperation with the Utah Highway Patrol Safety Program. UTAH98 I/M program equipment, including dynamometers, are required to be operated in accordance with manufacturer's specifications to prevent injury or damage to people or equipment. Exhaust gases are to be safely ventilated in accordance with EPA-AA-RSPD-IM-96-2.

Exhaust leaks Vehicles with leaking exhaust systems will be rejected.

UM program emission standards The Salt Lake Valley Health Department Regulation #22A, Appendix D, includes hydrocarbon, oxides of nitrogen and carbon monoxide emission standards. These emission standards allow for quick adjustment of the standards in case actual failure rates fall below the level specified in the State Implementation Plan. The emission standards for the UTAH98 I/M program were used in the MOBILE modeling to demonstrate compliance with the federal Basic I/M performance standard.

Stringency Salt Lake Valley Health Department will adjust tailpipe emission standards as necessary to maintain a stringency rate of at least 22% for pre-81 model year vehicles, the stringency rate used in the UTAH98 I/M performance standard modeling demonstrations.

Re-test standards The same test procedure and emission standards are used for initial tests and retests, regardless of which part a vehicle may have failed during an initial test. The test procedure requires an official test, once initiated, to be performed in its

entirety regardless of intermediate outcomes, except in the case of invalid test conditions, unsafe conditions, or the fast pass/fail algorithms.

Anti-tampering provisions Salt Lake County requires a visual emissions control device inspection to determine whether the air system, catalyst, fuel inlet, exhaust gas recirculation (EGR) valve, evaporative system, positive pressure crankcase valve (PCV), and gas cap are present, appear to be properly connected, and appear to be the correct type for the certified vehicle configuration. Regardless of the vehicle model year, Salt Lake County does not allow waivers for tampered vehicles or money spent to repair tampered or missing emission control devices to be applied towards a minimum waiver cost. The County requires repair of catalyst, and air pump system for model year 1984 and newer vehicles. The county requires repair of any tampering of the air system, catalyst, exhaust gas recirculation (EGR) valve, evaporative system, positive pressure crankcase valve (PCV), and gas cap on model year 1990 and newer vehicles. 1996 and newer vehicles also are required to have emission-related malfunction indicator lights (MIL) extinguished.

Engine changes The Salt Lake County health regulations address engine changes. After an engine change, vehicles are tested to the tailpipe emission standards and antitampering requirements applicable to vehicles of the chassis model year. Mixing vehicle classes (e.g., light-duty with heavy-duty) and certification types (e.g. California with federal) within a single vehicle is considered tampering.

Fuel switching Vehicles that are switched to a fuel type for which there is no certified configuration are tested according to the most stringent emission standards for that vehicle model year and vehicle type.

7. Test Equipment

Analyzer access restrictions An inspector access code is required to use the analyzer for official tests, a service access code to repair or service the analyzer, and an auditor access code to access the audit functions. Operating system functions are not accessible to station owners, inspectors, or analyzer service personnel. Programming changes are made by Salt Lake County I/M program auditors from disks supplied by the analyzer manufacturer.

Data security provisions Manual data entry is minimized. For initial inspections, the inspector enters vehicle registration and vehicle information from the keyboard. For retests, the inspector calls up the initial test file, compares the vehicle and owner data, and confirms the VIN/license plate data. Data regarding inspections, analyzer calibration and service, lock-out activities, and audit information are transmitted via phone line to the county every night.

Automated test procedure The analyzer automatically reads all test measurements, records test results in the computer database, determines whether the vehicle has passed or failed a test, and prints vehicle inspection reports and inspection certificates for all subject vehicles. The analyzers are capable of simultaneously sampling dual exhaust vehicles. The analyzer will measure carbon monoxide, carbon dioxide, nitric oxide and hydrocarbon emissions. The test procedure is automated to the highest degree practical to minimize the potential for intentional fraud and/or human error in compliance with ASM2.

Security lockouts The analyzers are programmed to trigger lock-outs when abuse or tampering occur. Lock-outs occur after any security system is tampered, failure to conduct or pass periodic calibration tests, or the data recording medium is full. The analyzer can not be used until the lock-out has been cleared by a Salt Lake County I/M program auditor. The analyzer automatically keeps an electronic record of all lock-outs including the date of the lock-out, the reason for the lock-out, and the date and person that cleared the lock-out.

Analyzer use restriction Salt Lake County requires official emissions tests to be conducted only on registered analyzers. Updates to the analyzer specifications may occur, as necessary, to accommodate new technology vehicles and changes to the program.

Analyzer design and certification The analyzer is BAR97 designed and certified and OBD-compliant. The analyzer performs ASM2 testing in compliance with the Acceleration Simulation Mode Test Procedures, Emission Standards, Quality Control Requirements, and Equipment Specifications Technical Guidance, EPA-AA-RSPD-IM-96-2, July 1996 and 40 CFR 51.358.

8. Quality Control

General quality control specifications The UTAH98 analyzer specification was carefully designed to insure that emission measurement equipment is calibrated and maintained properly, and that inspection, calibration records, and maintenance records are accurately created, recorded, and maintained.

Automatic electronic quality assurance features Operational analyzer quality assurance measures such as analyzer calibration, zero and span check, hydrocarbon hang-up check, and leak check are mandatory automatic analyzer capabilities. Gas accuracy tolerances, dilution limits, analyzer warm up requirements, system response time requirements, optical correction factors, and interference effects are also addressed in the analyzer specifications. If the checks are not performed on schedule or identify measurements outside of acceptable limits established in the specifications, a lock-out occurs preventing use of the analyzer until such problems are corrected. Records of all quality assurance activities with respect to the analyzer are automatically recorded in the analyzer's electronic database and evaluated by Salt Lake County I/M auditors on a regular basis. The analyzer specifications discuss requirements for assurance that unauthorized access to the I/M database in the analyzer is prevented. Attempts to deliberately avoid or defeat analyzer or inspection quality assurance provisions result in disciplinary action against the I/M mechanic and/or station. The automatic electronic quality assurance features of the analyzer are in compliance with the referenced ASM2 specification, EPA-AA-RSPD-IM-96-2.

Analyzer maintenance The analyzer specifications describes required services, warranty provisions, and documentation that analyzer manufacturers must provide to customers. It includes ensuring that the analyzer meets the quality assurance specifications at the time of delivery, that routine quarterly preventative maintenance is performed, training on how to use, maintain, and operate the analyzer is provided by the manufacturer, and that if repair of defects can not be made promptly a temporary analyzer replacement is provided. Service activities are recorded in the analyzer's electronic database. Maintenance of the analyzer is in compliance with the Salt Lake City-County Health Department Regulation #22A.

Document security Document security for the analyzer is in compliance with the Salt Lake City-County Health Department Regulation #22A.

Analyzer certification Sound engineering practices were followed during the design and certification of the analyzer to insure accurate and repeatable inspections under a range of environmental conditions. Manufacturer owner's manuals, operating instructions, and warranty provisions were also reviewed during the certification process. Comprehensive records of the certification process have been maintained.

General analyzer security provisions Salt Lake City-County Health Department Regulation #22A requires use of a certified and registered UTAH98 I/M analyzer for official inspections. Inspection records include the analyzer registration number. The regulations make it illegal to alter analyzer software or hardware without written approval. Analyzer calibration requirements, maintenance, and warranty provisions are also specified in the above Salt Lake Valley Health Department regulations.

9. Waivers

Waiver rate Salt Lake County will take corrective action as needed to maintain a maximum waiver rate of 1% of the initially failed vehicles or the Utah Air Quality Board will revise the SIP and emission reductions claimed based on the actual waiver rate. The conditions for issuing waivers legally authorized and specified in the Salt Lake County regulations meet the minimum waiver issuance criteria specified in 40 CFR Subpart S 51.360.
Waiver procedures The Vehicle Inspection Report (VIR) printed by the I/M analyzer after each inspection and provided to the vehicle owner/operator includes warranty and waiver information, if the vehicle failed the emissions inspection. A waiver document may be issued only by Salt Lake County I/M Technical Center staff and only after verification of required documentation. Any tampered, missing, or inoperable emission control devices must have been replaced or repaired. At least \$100 for 1968 through 1980 model year vehicles and \$200 for 1981 and newer model year vehicles must have been spent on acceptable emission repairs as verified by a Salt Lake County I/M program auditor by physical examination of the vehicle and review of the repair documentation. Repair documentation, such as receipts, are copied and retained by auditor to prevent reuse. Salt Lake County requires signed documentation on official stationery of a business involved in the automotive repair industry to include labor costs. In Salt Lake County, the retest must reflect a reduction of carbon monoxide oxides of nitrogen and/or hydrocarbon emissions after repairs. Emissions defects indicated by OBD fault codes must be repaired for the vehicle to qualify for a waiver. Vehicles still under the federal emissions warranty are not eligible for a waiver until all warranties are exhausted. Warranted repair and tampering repair may not be applied to the repair cost waiver limits. Waivers are only valid for one test cycle. The vehicle owner surrenders the original waiver document at the time of registration; copies are not accepted for registration purposes. Specific provisions regarding waivers may be found in the Salt Lake County health regulations and the Utah Tax Commission Division of Motor Vehicle policy manual that is available upon request. Salt Lake County does not provide for time extensions to relieve economic hardships in obtaining emission-related repairs.

10. Motorist compliance enforcement

Registration denial Salt Lake County's I/M program is enforced by means of registration denial. Vehicle owners must present proof of compliance with the I/M program, a waiver, or evidence of exemption from the I/M program as a condition precedent to vehicle registration or registration renewal. Citations are routinely issued to operators of vehicles with expired or missing license plates during routine traffic stops, parking lot inspections, and roadblocks. As specified in Section 41-1a-1303 of the Utah Code, driving without registration is a Class C misdemeanor. The penalty for a Class C misdemeanor is imprisonment of no more than 90 days and \$750 for persons or up to \$1000 for corporations, associations, partnerships, or government instrumentalities. In addition to paying a fine, the motorist must register the vehicle. It is currently a Class B misdemeanor to violate a County health regulation. The penalty for a Class B misdemeanor is an imprisonment not exceeding six months and for persons a fine of up to \$1000 or for corporations, associations, partnerships, or government instrumentalities a fine of up to \$5000. In Utah, the magnitude of such penalties is a judicial rather than an administrative decision. Per Section 41-1a-1315 falsification of evidences of title and registration is a second degree felony.

Certificate of Compliance The Certificate of Compliance is dated by the I/M analyzer immediately after a passing inspection is completed. The certificate is only valid for registration purposes for two months. At the same time the analyzer also prints the following information on the certificate to ensure unambiguous vehicle identification: the vehicle identification number (VIN), license number, model year, make, and model. A sample of the Certificate of Compliance is in the UTAH98 specifications. The certificates are only printed in the event that the vehicle passed the emissions inspection. Separate documentation, including the same vehicle information, is used for waivers.

Fuel changes to non-subject status Vehicle changes that would result in registration changes from a subject to exempt status require physical confirmation by Salt Lake County I/M program personnel at the I/M technical center. Falsification of registration or title information is a felony offense.

Title transfers Proof of compliance with the I/M program is required for a title transfer. The system ensures that owners are not able to avoid the program by extending the inspection date through manipulation of the title and registration system.

Salt Lake County I/M program staff, peace officers, and Utah Tax Commission Motor Vehicle Customer Service Division routinely work together to ensure that motor vehicle owners that move into an I/M program area complete registration transfer including compliance with the I/M program. Except for higher education students and active duty military personnel, people are required to register their vehicles in the county in which they are domiciled. As discussed in the Vehicle Coverage section, although these two exempted classes of vehicle owners do not have to register their vehicles in Utah, they do have to comply with the I/M programs. Employment status, maintenance of a residence, enrollment of children in local schools, and voting districts are considered when identifying persons in violation of this requirement.

Salt Lake County I/M program staff work with citizens, the Utah Motor Vehicle Customer Service Division and county attorneys to identify and prosecute people that illegally transfer registration to a non-subject area to avoid the I/M program. The process is very labor intensive. There are many legitimate reasons to be operating a vehicle in an I/M program area that is registered elsewhere. Violators must be dealt with on a case-by-case basis. Persons caught to date have been subject to fines of around \$700. Those prosecuted and convicted could end up with a criminal record and actual jail time. Fraudulent registration of a motor vehicle is a felony offense. Most people confronted with evidence of their guilt and the seriousness of their offense, to date, have complied promptly. The involved agencies are developing more efficient methods of dealing with illegal registrations that result in exemption from the I/M program. Salt Lake County is committed to a cooperative aggressive effort to ensure that vehicles operated in the county comply with the I/M program to ensure a compliance rate of at least 96%.

11. Motorist compliance enforcement program oversight

Utah Tax Commission, tax assessors, and county roles The Utah Tax Commission Motor Vehicle Customer Service Division and Salt Lake County tax assessor deny application for vehicle registration or renewal of registration without submittal of a valid certificate of compliance, waiver, or verified evidence of exemption. Proof is retained by the tax clerk, micro-photo-copied, and then destroyed. Altered or handwritten documents are not accepted. All certificate data is collected by Salt Lake County I/M program auditors and subjected to scrutiny for evidence of any improprieties.

Database quality assurance The vehicle registration database is maintained and quality assured by the Motor Vehicle Customer Service Division. The I/M inspection database is maintained and quality assured by the Salt Lake County I/M program staff. The Salt Lake County I/M program has access to the Motor Vehicle Customer Service Division database and utilize it on a regular basis for quality assurance purposes. The database is subject to regular auditing, cross-referencing, and analysis. The database is also evaluated using data obtained during roadblocks and parking lot surveys. Evidence of program effectiveness problems trigger additional joint enforcement activities.

Oversight provisions The oversight program includes verification of exempt vehicle status through inspection, data accuracy through automatic and redundant data entry for most data elements, an audit trail for program documentation to ensure control and tracking of enforcement documents, identification and verification of exemptiontriggering changes in registration data, and regular audits of I/M inspection records, I/M program databases, and the Motor Vehicle Customer Service Division database.

Enforcement staff quality assurance I/M program auditors and tax clerks involved in vehicle registration are subject to regular performance audits by their supervisors. All enforcement personnel (direct and indirect) involved in the motorist enforcement program are subject to disciplinary action, additional training, and termination for deviation from procedures. Specific provisions are outlined in the Motor Vehicle Customer Service Division procedures manual that is available upon request.

Co-operative enforcement oversight effort The Motor Vehicle Customer Service Division, Utah Division of Air Quality, Utah Highway Patrol and Salt Lake County I/M program staff meet at least once per month to ensure on-going high quality oversight of joint motorist compliance program. EPA audit of this process is authorized if measures to protect tax-payer confidentiality acceptable to Motor Vehicle Customer Service Division are exercised.

12. I/M Program quality assurance

Station/inspector audits Salt Lake County's I/M program regularly audits all certified I/M inspectors and stations to ensure compliance with Salt Lake County health regulations and policies. Particular attention is given to identifying and correcting any fraud or incompetence with respect to vehicle emissions inspections. Compliance with recordkeeping, document security, analyzer maintenance, and program security requirements are scrutinized. The inspector's skill level is also evaluated during audits. Another major purpose of the audits is to retrain inspectors, as necessary, as soon as problems are identified. Documentation sufficient to support a legal case to suspend or revoke a certification is also collected in the event of serious and/or repeated violations. Most stations and inspectors are audited every month and all at least quarterly.

Covert audits Salt Lake County, to the extent possible, performs a covert audit of each inspector and station at least once a year. The number of covert audits at least equals the number of certified inspectors. Covert audits are performed using a variety of vehicles that are representative of the subject fleet that are set to fail across a full range of malfunctions. Suspected problem stations and inspectors are targeted for earlier and more frequent audits. Complaints also trigger additional audits.

Covert performance audits shall include:

Remote visual observation of inspector performance, which may include the use of aids such as binoculars or video cameras, at least once per year per inspector in high-volume stations (i.e., those performing more than 4000 tests per year);

Site visits at least once per year per number of certified inspectors (per inspector FTE) using covert vehicles set to fail (this requirement sets a minimum level of activity not a requirement that each inspector be involved in a covert audit); and For stations that conduct both testing and repairs, at least one covert vehicle visit per station per year including purchase of repairs and subsequent retesting if the vehicle is initially failed for tailpipe emissions.

Electronic audit capabilities The Salt Lake County I/M program equipment performs various analyses to identify statistically inconsistent data indicative of problem stations and inspectors. Overt audit records are maintained electronically in the analyzer. After overt audits, the auditor retrieves the data on the analyzer diskette containing the audit, vehicle inspection, and analyzer service, maintenance, and calibration records dating back to the previous audit. The data from each audit is added to the comprehensive central Salt Lake Valley Health Department I/M database. Further analysis of the central database results in identification of stations and inspectors for which additional audits are performed.

Auditor quality assurance Auditors receive 24 hours of formal classroom instruction and are provided on-the-job training in: the use of the analyzer; the Salt Lake County I/M health regulations, basic air pollution control; basic principles of emissions-related motor vehicle engine repair; emission control systems; evidence gathering; administrative procedures and laws; quality assurance practices; and covert audit procedures. Salt Lake County sends auditors to additional automotive emissionsrelated training and meetings on a regular basis. Auditor supervisors audit the I/M program auditors by reviewing their documentation and also auditing a number of their stations at least once every year.

Written audit procedures Copies of the Salt Lake County I/M program overt and covert audit procedures are provided in Section X, Part C, Appendices. A detailed description of the audit capabilities is found in Section 3.9 of the UTAH98 analyzer specifications.

13. Enforcement against stations and inspectors

General enforcement provisions The Salt Lake County I/M program is responsible for enforcement action against incompetent or dishonest stations and inspectors. The Salt Lake County health regulations include a penalty schedule. For serious or repeated offenses, auditors are authorized to immediately suspend the station or inspector by locking out their UTAH98 analyzer. The County does not have legal authority to impose direct fines on stations or inspectors, but suspension or revocation of a station permit results in a substantial loss of income that is far in excess of \$100 fine suggested by the EPA guidance. Fee settlements are at least as much the station's anticipated income for emissions testing for the time during which the station would be suspended. A station permit may be suspended or revoked even if the owner/operator had no direct knowledge of the violation. In the case of incompetence, re-training is required before the permit is restored.

Salt Lake County revised its penalty schedules to comply with the more stringent specifications included in 40 CFR 51.364; it is found in Appendix E of Salt Lake City-County Health Department Ordinance 22A. At a minimum, inspector certification and station permit suspension shall be imposed for at least 6 months (or a fee retainage or settlement penalty equivalent to the inspector's salary for that period) whenever a vehicle is intentionally improperly passed for any portion of the required test.

Suspension and revocation Suspension or revocation effectively bars an individual from further inspections because the auditor removes the inspector's authorization code from the UTAH98 analyzer. Evidence of indirect participation in emissions inspections by an individual while suspended or revoked would result in legal action against the station. If the station is suspended or revoked the analyzer is totally locked-out. The analyzers are initialized by an auditor for use at a single permitted station and only by

inspectors certified for that station. A record of the serial numbers of all registered analyzers and their locations is maintained by Salt Lake County.

Enforcement records Salt Lake County keeps comprehensive records of all audit activities, warnings, suspensions and revocations, and enforcement activity statistics to the EPA and the executive secretary annually.

14. Data collection

Analyzer inspection data A detailed record of each emissions inspection is performed including, but not limited to the following data, for each vehicle tested: test record number; inspection station number; inspector number; test system number; date of the test; emission test start time; the time final emission scores are determined; vehicle identification number (VIN); license plate number; test certificate number; gross vehicle weight rating (GVWR); model year, make, and type of vehicle; number of cylinders or engine displacement; transmission type; odometer reading; category of test performed (i.e., initial, first retest, or subsequent retest); fuel type of the vehicle; emission scores for HC, CO, NO and CO₂ at 25 mph and 15 mph; and results (pass/fail/not applicable) for visual inspection of the catalytic converter, air system, gas cap, evaporative system, and positive crankcase (PCV) valve. The tailpipe emission standards for each type of vehicle is included in a look-up table in the analyzer. The analyzer automatically uses the appropriate standards for the type of vehicle being tested and makes a pass/fail determination. The inspection data is recorded by the analyzer during the inspection procedure.

Analyzer quality assurance data Quality assurance data including a detailed history of all calibration (including the concentration values of the calibration gases), service, lockout, and document security events are also recorded and maintained by the analyzer. Each record includes, as applicable, the station number, mechanic access number, auditor access number, service access number, analyzer serial number, date, and activity time.

Analyzer data collection The I/M analyzer data collection system meets the requirements specified under 40 CFR 51.365.

15. Data analysis and reporting

Annual reports Salt Lake County shall analyze I/M program data and submits annual reports to the U.S. Environmental Protection Agency and the executive secretary upon request. Beginning in July of 1995, Salt Lake County will submit to EPA and the executive secretary an annual report, for January through December of the previous year, which provides statistics on the testing, quality assurance, and enforcement activities of each I/M program. At a minimum the annual reports will include all of the data elements listed 40 CFR Subpart S 51.366.

Biennial reports Beginning in July of 1996, and biennially thereafter, Salt Lake County shall submit a report to EPA and the executive secretary discussing all changes made in the program design, funding, personnel levels, procedures, regulations, and legal authority. The report will also supply a detailed discussion of the impact of such changes upon the program, any weaknesses or problems discovered in the program over the previous two-year period, the steps that were taken to address those problems, the result of those corrective actions, and any future efforts planned.

16. Inspector training and certification

Inspector certification and initial training No person may conduct an official I/M inspection unless they are certified. Salt Lake County requires all persons desiring to become UM technicians to pass a pretest to insure they have a basic understanding of automotive engine operation and repair. Only about one half of those attempting to become certified pass the pretest and are allowed to take the formal training class. Salt Lake County requires formal training prior to certifying inspectors. Each class includes at least the following information: the causes and effects of air pollution; the purpose, function, and goal of the I/M program; I/M health regulations, policies, and procedures: technical details of the test procedures and the rational for their design; emission control device function, configuration, and maintenance; quality control procedures and their purposes; public relations; and safety and health issues related to the I/M inspection process. Salt Lake County provides the training directly. Inspector candidates will not be issued a certificate unless they have passed a written test with at least 80% (or lower if an occupational analysis justifies it) correct responses and a hands-on test during which the trainee demonstrates the ability to properly conduct all test procedures, calibrate the analyzer, properly utilize equipment, and to follow other I/M program requirements. Salt Lake County will take appropriate steps to insure the security of the testing process.

Inspector certification renewal Inspector certification is valid for a period of one year, at which point refresher training and testing, are required prior to certification renewal. An auditor enters the inspector's certification expiration date in the analyzer(s) that the inspector is authorized to use. The analyzer locks out the inspector upon expiration of the certification. Auditors will not clear the lock-out until the inspector has renewed the certification. Salt Lake County may require evidence of more comprehensive emissions-related automotive training as a prerequisite to inspector certification renewal.

Inspector certification suspension and revocation A determination of inspector incompetence or failure to comply with I/M program requirements may result in suspension or revocation or an inspector's certification prior to the annual expiration date. A certification to conduct I/M inspections is not a legal right but rather a privilege bestowed by Salt Lake County conditional upon adherence to its I/M program requirements.

Inspector training authority and materials Authority to require mandatory I/M inspector training is established and described in the Salt Lake County health regulations.

17. Public information and consumer protection

General public information Salt Lake County, along with the Utah Department of Environmental Quality, provides a comprehensive public education and protection program including strategies to educate the public on: Utah's air quality problems; ways that people can reduce emissions; the requirements of state and federal law; the role of motor vehicles in the air quality problem; the need for and benefits of a vehicle emissions inspection program; ways to operate and maintain a vehicle in a lowemission condition; how to find a qualified repair technician; and the requirements of the I/M program. Information is provided via direct response to inquiries for information, reports, classes, pamphlets, fairs, school presentations, workshops, news releases, posters, signs, and public meetings.

Salt Lake County I/M Technical Center Salt Lake County operates an I/M Technical Center staffed with trained auditors and capable of performing emissions tests. A major function of the I/M technical center is to serve as a referee station to resolve conflicts between certified I/M inspectors, permitted stations, and motorists. Auditors actively protect consumers against fraud and abuse by inspectors, mechanics, and others involved in the I/M program. Complaints made on a confidential basis are investigated and resolved in a manner that conceals the person's identity to ensure protection of whistle blowers. Auditors advise motorists regarding emissions warranty provisions and assist the owners in obtaining warranty-covered repairs for eligible vehicles. Applications for waivers are evaluated by auditors at the I/M technical center and issued only after visual verification that all the requirements for a waiver have been met. The I/M technical center also provides motorists with information regarding the I/M program, general air pollution issues, and emissions-related automotive repairs.

Vehicle inspection report A vehicle inspection report (VIR) is printed and provided to the motorist after each vehicle inspection. The VIR includes a public awareness statement about automotive emissions and lists additional ways that the public can reduce air pollution. The test results are detailed on the VIR. Information about vehicle emissions warranties and the benefits of emissions-related repairs are printed for vehicles that failed the test. Information about waiver requirements and application procedures are printed on the VIR, if the vehicle has failed a retest, including the address and telephone number of the applicable I/M technical center. A complete description of the VIR is included in the UTAH98 analyzer specifications. Co-operative public education tools A variety of public outreach materials about automotive air pollution issues are developed and distributed by the Salt Lake County I/M program in cooperation with other I/M counties and the Utah Division of Air Quality.

18. Improving repair effectiveness

High priority Salt Lake County implemented its first major Basic I/M program revision on September 1, 1991. Shortly thereafter, the Salt Lake County and the Utah Division of Air Quality staff jointly identified improvement of repair effectiveness as a high priority action item. The Governor's Clean Air Commission also recommended making affordable additional emissions-related training available. In 1998, Salt Lake County implemented further improvements when it established its UTAH98 program. Full emission reductions will only be realized if the repair industry is able to competently diagnose and repair emissions-related defects.

Continuing education To that end, Salt Lake County's I/M staff has worked with Utah's higher education institutions to develop and provide emissions-related automotive technology classes to technicians. Inspectors are also encouraged to take classes offered by trade organizations, automobile manufacturers, and dealers. Salt Lake County subsidizes the tuition for certified I/M inspectors. The certification renewal tests are difficult enough to make this provision a good incentive. The classes are advertised in the county I/M technical bulletins.

L/M program repair support activities In initiating improved automotive educational opportunities, Salt Lake County works on a day-to-day basis to ensure that repair information is available. *L/M* stations are required to have available up-to-date relevant automotive diagnostic references and tools as a condition for obtaining a permit. Salt Lake County maintains a hot line to its *L/M* technical center that any mechanic can call for technical assistance related to vehicle inspection, diagnosis, and repair. Technical bulletins are regularly mailed to each certified inspector with information regarding training schedules, common problems found with particular engine families, and diagnostic tips.

19. I/M SIP implementation

The I/M program regulations, policies, procedures, and activities specified in this I/M SIP revision have been implemented. Salt Lake County shall continue to implement and operate the I/M program until a maintenance plan without an I/M program is approved by EPA in accordance with Section 175 of the Clean Air Act as amended.

UTAH STATE IMPLEMENTATION PLAN

SECTION X

VEHICLE INSPECTION AND MAINTENANCE PROGRAM

PART D

UTAH COUNTY

Adopted by the Utah Air Quality Board March 31, 2004

و

_____.

Table of Contents

1	I/M performance standard
	Federal requirements
	I/M Program MOBILE modeling
	I/M Program Performance Standard
	I/M Program Improvements
	Enhanced I/M Program requirement
	I/M Program Improvements MOBILE modeling
	Improved I/M Program Performance Standard
2.	Network type
3.	Tools and resources
	Funding mechanisms
	I/M program funding requirements
4.	Test convenience
5.	Vehicle Coverage
	Subject fleet
	Alternative fuels
	Government fleet
	Vehicles owned by students and federal employees
	Farm truck exemption
	Diesel vehicles4
	New vehicle exemption4
	Out-of-state exemption4
	Exempt vehicle4
	Unregistered vehicles4
6.	Test procedures and standards
	Specifications5
	Test procedure and analyzer5
	Pre-inspection emissions-related repairs
	Safety issues
	Exhaust leaks
	Emission standards
	Stringency
	Re-test standards
	Anti-tampering provisions
d	Engine changes
_''	Fuel switching
7.	Test Equipment
	Specifications
	Analyzer access resultions

Automated test procedu	re	7
Security lockouts		7
Certified analyzer use re	estriction	7
8. Quality Control		8
General quality control	specifications	8
Automatic electronic qu	ality assurance features	8
Analyzer maintenance.	· · · · · · · · · · · · · · · · · · ·	8
Document security		8
Analyzer certification		8
Analyzer security provis	sions	9
9 Waivers		9
Waiver rate		
Waiver procedures		
10 Motorist compliance enfor	cement	
Registration denial		
Certificate of Complian	се	
Fuel changes to non-sub	piect status	
Title transfers	J	
11 Motorist compliance enfor	cement program oversight	
Utah Tax Commission.	tax assessors, and county roles	
Database quality assurat	псе	
Oversight provisions		
Enforcement staff qualit		
Co-operative enforceme	nt oversight effort	
12 I/M Program quality assura	nce	
Station/inspector audits.		
Covert audits		
Electronic audit capabili	ities	
Auditor quality assurance	۲ <u>۰</u> ۰۰	
Written audit procedures	s	
13 Enforcement against station	is and inspectors	
General enforcement pro	avisions	14
Suspension and revocati	on	14
Enforcement records	~	14
14 Data collection		14
I/M data collection		
Analyzer inspection data	1	
Analyzer quality assuran	ice data	15
Analyzer database specif	fications	
15. Data analysis and reporting		15
Annual Reports		
Biennial Reports		
Data link		
16. Inspector training and perm	itting	
Utah County I/M SIP	FINAL	March 31, 2004

ii

• •

Inspector permitting and initial training	16
Inspector Training	16
Inspector permit renewal	16
Inspector permit suspension and revocation	
Inspector training authority and materials	17
17. Public information and consumer protection	
General public information	
County I/M Technical Center	
Vehicle inspection report	
I/M county co-operative public education tools	
18. Improving repair effectiveness	
High priority	18
Continuing education	
I/M program repair support activities	
19 I/M SIP implementation.	
20. On-road Testing	19

SECTION X, PART D UTAH COUNTY Appendices

- 1 Vehicle Emissions Inspection/Maintenance Program, Ordinance 2003-28, revised June 10, 2003.
- 2 Provo I/M Ordinance 1994-106, December 12, 1994
- 3 Audit Policies

•

• .

4 Utah County Remote Sensing Ordinance, April 30, 1997

UTAH STATE IMPLEMENTATION PLAN SECTION X AUTOMOTIVE INSPECTION AND MAINTENANCE (I/M) PROGRAM PART D UTAH COUNTY

1. I/M performance standard

Federal requirements EPA's I/M regulation, 40 CFR Part 51, Inspection and Maintenance Program Requirements last amended at 66 FR 18156, April 5, 2001, specifies a model Basic I/M program. Utah is required by Section 182 of the Clean Air Act to implement an I/M program in Utah County that is at least as effective as the EPA's Basic Performance Standard. The Basic I/M performance standard is specified in 40 CFR 51.352. While local governments have flexibility to implement programs best suited for their area, EPA's regulations require a performance demonstration that local I/M programs result in automotive emissions equal to or less than predicted for the EPA model I/M program. State and local governments may choose options best suited for their area to meet the performance standard.

I/M Program MOBILE Modeling Mobile source emission factors are determined by use of the most recent release of EPA's MOBILE model. The MOBILE model is able to calculate emission factors, grams of a particular pollutant per vehicle mile traveled across the fleet in an area (G/VMT), given information about the fleet, climate, fuel characteristics and I/M programs in a local area. The latest approved release of the MOBILE model was used for the I/M performance standard demonstration analysis and the documentation is included in the respective criteria pollutant section of the SIP.

I/M Program Performance Standard Utah County's I/M program exceeds the Basic I/M performance standard for all pollutants, although the EPA only requires the demonstration for each pollutant which caused an area to be subject to an I/M program. Utah County is a moderate carbon monoxide (CO) National Ambient Air Quality Standard (NAAQS) non-attainment area.

I/M Program Improvements On December 18, 1995, the Utah County Commission adopted Ordinance No. 1995-29, which adopted the Diesel Vehicle Emissions Inspection/Maintenance Program Rules and Regulations and the Vehicle Emissions Inspection/Maintenance Program Rules and Regulations in book form. Ordinance 2003-28, adopted June 10, 2003, accommodates the current analyzer and is On-Board Diagnostics (OBD) compliant. It requires individuals whose primary residence in Utah County to register their motor vehicles in Utah County, removes the exemption for diesel vehicles older than model year 1968 (except for vintage vehicles), establishes waiver cut points, allows the county to recall specific vehicles for quality assurance testing, and allows the county to require repair of vehicles following the additional testing. Provo

d

City ordinance requires that the vehicles operated by people staying in Provo for more than sixty days be inspected and repaired as specified in the Utah County I/M ordinance regardless of where the vehicle is registered. These ordinances are provided in Section X, Part D, Appendix D.1.

Enhanced I/M Program requirement At the time the Provo CO SIP revision was developed in 1994, EPA assumed only 50% credit for a decentralized test-and-repair I/M program. To qualify for 100% credit, an enhanced vehicle emission I/M program was identified as a control strategy, with Utah County demonstrating its decentralized I/M program with enhancements would provide emission reductions equal to or greater than a test-only program. Following the provisions of Section 348 of the National Highway System Designation Act of 1993 (NHSDA), Utah County performed additional testing and analysis using methodology developed by the Environmental Council of the States (ECOS), State and Territorial Air Pollution Program Administrators (STAPPA), and EPA I/M Workgroup in response to the NHSDA requirements.

Utah County's NHSDA analysis was submitted to EPA on May 27, 1999. On September 12, 2002 (67 FR 57744), EPA published approval of the Utah County I/M program, including approval of the demonstration of full emission reduction credit for the program. This allowed Utah County to claim 100% emissions test-only credit for its I/M program.

LIM Program Improvements MOBILE modeling The performance standard demonstration is made by use EPA's MOBILE model. The MOBILE model is able to calculate emission factors, grams of a particular pollutant per vehicle mile traveled across the fleet in an area (G/VMT), given information about the fleet, climate, fuel characteristics, and I/M programs in a local area. The latest approved version of the MOBILE model was used for the performance standard demonstration analysis.

Improved I/M Program Performance Standard Utah County's Improved I/M program exceeds the federal Basic I/M performance standard for all pollutants, although the EPA only requires the demonstration for each pollutant which caused an area to be subject to an I/M program. Utah County has incorporated Basic I/M improvements to demonstrate compliance with the I/M performance standard for carbon monoxide.

2. Network type

Utah County's I/M program is a basic, decentralized, test-and-repair network, as approved by EPA on September 12, 2002 (67 FR 57744).

3. Tools and resources

Funding mechanisms Utah County's I/M program is funded through several mechanisms including a \$1 air pollution control fee for each non-diesel-fueled vehicle registered in the county. I/M Certificates are sold to I/M test stations for \$2.25 each. The county also charges fees for various permitting activities. The fees are dedicated to the I/M program. A fee schedule can be found in an Appendix to Utah County's I/M Program ordinance.

I/M program funding requirements Utah County will allocate funding as needed to comply with the relevant requirements specified in Utah's SIP; Utah statutes; county ordinances, regulations and policies; and the federal I/M program regulation. Budgets and descriptions of personnel resources, facilities, and equipment for Utah County's I/M program are included in the TSD.

4. Test convenience

There are approximately 140 permitted Basic I/M stations within Utah County. Specific operating hours are not specified by the county. Some stations that test and/or service only one type of vehicle are permitted. There are also government and private fleet permitted stations that are not open to the public.

5. Vehicle coverage

Subject fleet All model year 1968 and newer model year light duty vehicles, light duty trucks, and heavy duty trucks registered or principally-operated in Utah County are subject to the I/M programs except for exempt vehicles as specified in Section 6.6 of the Utah County I/M Ordinance. All vehicles six years old and older on January 1 are tested annually; newer vehicles are tested every other year as per Utah Code 41-6-163.6(6) amended in 2002. Modeling conducted for the Provo Carbon Monoxide Maintenance Plan adopted by the Air Quality Board on April 7, 2003 demonstrates that annual testing for newer vehicles is not needed to maintain the National Ambient Air Quality Standard for carbon monoxide in Provo.

Alternative fuels Vehicles operated on alternative fuels such as propane, alcohol, and natural gas are also subject to the program. Dual-fueled vehicles are tested twice, once on each fuel.

Government fleet Section 41-6-163.6(1)(b) of the Utah Code requires that all vehicles owned or operated in the county by federal, state, or local government entities comply with the I/M programs.

Vehicles owned by students and federal employees Section 41-6-163.3(5) of the Utah Code requires universities and colleges located in Utah's I/M areas to require proof of compliance with the I/M program for vehicles which are permitted to park on campus regardless of where the vehicle is registered. Vehicles operated by federal employees and operated on a federal installation located within an I/M program area are also subject to the I/M program regardless of where they are registered. Proof of compliance consists of a current vehicle registration in an I/M program area or an I/M certificate of compliance or waiver, or evidence of exempt vehicle status as specified in this section.

Farm truck exemption Eligibility for the farm truck exemption from the I/M programs is specified in Section 41-6-163.6(4) of the Utah Code and must be verified in writing. The owner must sign an affidavit on Utah State Tax Commission form TC-838 that vehicle use will be limited to agricultural activities.

Diesel vehicles A light and heavy duty diesel I/M program was implemented in 1994 and is defined in SIP Section XXI.

New vehicle exemption Proof that a vehicle is new and being registered for the first time is established by presentation of a Manufacturer's Statement of Origin (MSO) at the time of registration.

Out-of-state exemption Vehicles registered in an I/M county but operated out-of-state are eligible for an extension. The owner must complete Utah State Tax Commission form TC-810 in order to be registered without inspection documentation from Utah County. The owner must explain why the vehicle is unavailable for inspection in Utah. Common situations include Utah citizens that are military personnel stationed outside of the state, students attending institutions of higher education elsewhere, and people serving missions. If the temporary address of the owner is located within another I/M program area listed on the back of the form, the owner must submit proof of compliance with that I/M program at the time of, and as a condition precedent to, registration or renewal of registration. The vehicle owner must identify their anticipated date of return to the state and is required to have the vehicle inspected within 10 days after the vehicle is back in Utah, unless they can demonstrate that the vehicle had passed an I/M inspection in another area. Utah County maintains a record of such exemptions and requires submission of an I/M inspection certificate or waiver at the indicated time.

Exempt vehicle The following vehicles are exempt from inspection: motorcycles, electric powered vehicles, new vehicles registered for the first time, model year 1968 and older vehicles, farm vehicles and equipment, construction equipment, and other off-road vehicles.

įi.

Unregistered vehicles I/M ordinances and regulations require that vehicles available for rent or use in Utah County are subject to its I/M program. To the extent practicable, all vehicles principally-operated within the county are subject to the I/M program.

6. Test procedures and standards

Specifications Detailed specifications for the I/M test procedures and standards are described in the Utah County I/M ordinance provided in Section X, Part D, Appendix D.1. Specifications for the test procedure and equipment were developed according to good engineering practices to ensure test accuracy.

Test procedure and analyzer The Basic I/M program is compatible with EPA's PRECONDITIONED TWO SPEED IDLE TEST as specified in EPA-AA-TSS-I/M-90-3 March 1990, Technical Report, "Recommended I/M Short Test Procedures for the 1990's: Six Alternatives. 1996 and newer vehicles are tested using OBD II test procedures. All Basic emissions inspections are performed using the UTAH2000 Analyzer, a BAR97-type emissions analyzer. The UTAH2000 Analyzer calibration specifications and emissions test procedures meet the minimum standards established in Appendix A of the EPA's I/M Guidance Program Requirements, 40 CFR Part 51 Subpart S.

Covered vehicles are defined in Section 5 above. All covered vehicles in Utah County are subject to the Basic test procedure and inspected using the UTAH2000 analyzer as specified in this section.

Pre-inspection emissions-related repairs Inspectors in the county's test-and-repair networks are required to perform the emissions test prior to making any emissions-related repairs when a vehicle is presented for an emissions inspection. All inspectors who conduct test-only inspections, are required to ask the vehicle owner or operator whether a tune-up or other emissions-related repairs have been performed within 6 weeks prior to the emissions inspection and to document the owner's response in the UTAH2000 computer database.

Safety issues Vehicles presented in unsafe condition must be repaired before inspection. Vehicles are also subject to an annual safety inspection administered by the Highway Patrol. Submission of proof of compliance with the safety program is also required as a condition for registration or renewal of registration. Most owners in Utah's test-and-repair networks have the safety and emissions inspection performed at the same time as the emissions inspection. Data relative to the safety inspection can be recorded in the UTAH2000 Analyzer. Utah County's I/M program is administered with close cooperation with the Utah Highway Patrol Safety Program.

Exhaust leaks The UTAH2000 analyzer measures exhaust carbon monoxide (CO) and carbon dioxide (CO₂). Exhaust CO + CO₂ readings of less than 6% indicate a leaky exhaust system and cause the UTAH2000 analyzer to abort the inspection.

Emission standards The Utah County proposed I/M ordinance includes hydrocarbon and carbon monoxide emission standards in an appendix to allow for quick adjustment of the standards in case actual failure rates fall below the level specified in the State Implementation Plan. Vehicles must pass both the hydrocarbon and carbon monoxide emission standard regardless of the NAAQS attainment status of the county of registration. The emission standard for the Basic I/M program was used in the MOBILE modeling that was conducted to demonstrate compliance with the Basic I/M performance standard. Utah County also established waiver emission standard for carbon monoxide that can be found in Appendix F of Utah County's Vehicle Emission Inspection Maintenance Program ordinance.

Stringency The Utah County I/M program will adjust tailpipe emission standards as necessary to maintain a stringency rate of at least 22% for pre-81 model year vehicles, the stringency rate used in the Basic I/M performance standard modeling demonstration.

Re-test standards The same test procedure and emission standards are used for initial tests and retests, regardless of which part a vehicle may have failed during an initial test. Utah County's I/M test procedure requires an official test, once initiated, to be performed in its entirety regardless of intermediate outcomes, except in the case of invalid test conditions, unsafe conditions, or the fast pass/fail algorithms.

Anti-tampering provisions Regardless of the vehicle model year, Utah County does not allow waivers for tampered vehicles or money spent to repair tampered or missing emission control devices to be applied towards a minimum waiver cost. Utah County requires repair of any catalyst and air pump system tampering on vehicles of model year 1977 through 1989. The county also requires repair of any tampering of the air system, catalyst, exhaust gas recirculation (EGR) valve, evaporative system, positive pressure crankcase valve (PCV), and gas cap on model year 1990 and newer vehicles.

Engine changes Utah County's proposed I/M ordinance has a section that addresses engine changes performed prior to 1991. After an engine change, vehicles are tested to the tailpipe emission standards and anti-tampering requirements applicable to vehicles of the engine model year. Mixing vehicle classes (e.g., light-duty with heavy-duty) and certification types (e.g. California with federal) within a single vehicle is considered tampering.

Fuel switching Vehicles that are switched to a fuel type for which there is no certified configuration are tested according to the most stringent emission standards for that vehicle model year and vehicle type.

Utah County I/M SIP

FINAL Section X, Part D, page 6

7. Test Equipment

Specifications The UTAH2000 Analyzer is a BAR97-type computerized emissions analyzer. Additional written technical specifications for Utah County's I/M test equipment are specified in Utah County's I/M ordinance.

Analyzer access restrictions An inspector access code is required to use the UTAH2000 analyzer for official tests, a service access code to repair or service the analyzer, and an auditor access code to access the audit functions. DOS functions are not accessible to station owners or inspectors. Programming changes are made by county I/M auditors from disks supplied by the analyzer manufacturer.

Data security provisions Manual data entry is minimized. For initial inspections, the inspector enters vehicle registration and vehicle information from the keyboard. Data elements are described in the UTAH2000 analyzer specifications. For retests, the inspector calls up the initial test file, compares the vehicle and owner data, and confirms the VIN/license plate data. Data regarding inspections, analyzer calibration and service, lock-out activities, and audit information are downloaded to the county vehicle identification database daily; data from each analyzer is downloaded once or twice weekly.

Automated test procedure The UTAH2000 analyzer automatically reads all test measurements, records test results in the computer database, determines whether the vehicle has passed or failed a test, and prints vehicle inspection reports and inspection certificates for all subject vehicles. The analyzers are capable of simultaneously sampling dual exhaust vehicles. The analyzer bench includes two non-dispersive infrared (NDIR) analyzers for carbon monoxide, carbon dioxide, and hydrocarbon measurements (one low range and one high range), and one NDIR analyzer for carbon dioxide measurement. The test procedure is automated to the highest degree practical to minimize the potential for intentional fraud and/or human error.

Security lockouts The analyzers are programmed to trigger lock-outs when abuse or tampering occur. Lock-outs occur after any security system is tampered, failure to conduct or pass periodic calibration tests, or the data recording medium is full. The analyzer cannot be used until the lock-out has been cleared by a Utah County I/M auditor. The analyzer automatically keeps an electronic record of all lock-outs including the date of the lock-out, the reason for the lock-out, and the date and person that cleared the lock-out.

Certified analyzer use restriction Since March 1, 2000, the Utah County Basic I/M program requires that official emissions tests be conducted only on registered UTAH2000 analyzers jointly certified by Utah, Davis and Weber Counties. A description of the

Utah County I/M SIP

li.

certification procedure is provided in the TSD. There have been several updates of the UTAH2000 Analyzer specifications to date and more will follow, as necessary, to accommodate new technology vehicles and changes to the program.

8. Quality Control

General quality control specifications Utah County's I/M Program, the UTAH2000 Analyzer specifications, and current I/M program ordinances and regulations were carefully designed to insure that emission measurement equipment is calibrated and maintained properly, and that inspection, calibration records, and maintenance records are accurately created, recorded, and maintained. The specifications meet the test equipment quality assurance practices described in 40 CFR 51 Subpart S Sec. 51.359 and Section X, Appendix A.

Automatic electronic quality assurance features Operational analyzer quality assurance measures such as analyzer calibration, zero and span check, hydrocarbon hang-up check, and leak check are mandatory automatic analyzer capabilities. Gas accuracy tolerances, dilution limits, analyzer warm up requirements, system response time requirements, optical correction factors, and interference effects are also addressed in the analyzer specifications. If the checks are not performed on schedule or identify measurements outside of acceptable limits established in the specifications, a lock-out occurs preventing use of the analyzer until such problems are corrected. See Sections 2.12, 2.13, and 2.18 of the UTAH2000 Analyzer specifications. Records of all quality assurance activities with respect to the analyzer are automatically recorded in the analyzer's electronic database and evaluated by Utah County I/M auditors on a regular basis. Section 1.7 discusses requirements for assurance that unauthorized access to the I/M database in the analyzer is secure. Attempts to deliberately avoid or defeat analyzer or inspection quality assurance provisions result in disciplinary action against the I/M mechanic and/or station.

Analyzer maintenance Section 1.8 of the UTAH2000 Analyzer specifications describes required services, warranty provisions, and documentation that analyzer manufacturers must provide to customers. It includes ensuring that the analyzer meets the quality assurance specifications at the time of delivery, that routine quarterly preventative maintenance is performed, training on how to use, maintain, and operate the analyzer is provided by the manufacturer, and that if repair of defects can not be made promptly a temporary analyzer replacement is provided. Service activities are recorded in the analyzer's electronic database. Utah County has conducted a survey of analyzer owners to determine compliance with these provisions. Failure of an analyzer manufacturer to meet quality assurance specifications could result in de-certification of that manufacturer's product for use in Utah.

Document security Document security was a high priority during the UTAH2000 analyzer design phase. The analyzer tracks the unique certificate numbers and ensures

Utah County I/M SIP

ſł.

FINAL Section X, Part D, page 8

that the certificate printed matches the test number. Missing certificate numbers are stored in the analyzer database for auditor review. The blank certificates are commercially printed on counterfeit-resistant security paper.

Analyzer certification Sound engineering practices were followed during the design and certification of the UTAH2000 analyzer to insure accurate and repeatable inspections under a range of environmental conditions. Manufacturer owner's manuals, operating instructions, and warranty provisions were also reviewed during the certification process. Comprehensive records of the certification process have been maintained.

Analyzer security provisions Utah County's I/M ordinance requires use of a certified and registered UTAH2000 analyzer for official inspections. Inspection records include the analyzer registration number. The ordinances and regulations make it illegal to alter analyzer software or hardware without written approval. Analyzer calibration requirements, maintenance, and warranty provisions are also specified in the Utah County I/M ordinance.

9. Waivers

Waiver rate Utah County will take corrective action as needed to maintain a maximum waiver rate of 5% of the initially failed vehicles or the Utah Air Quality Board will revise the SIP and emission reductions claimed based on the actual waiver rate. The conditions for issuing waivers legally authorized and specified in the Utah County I/M ordinance meets the minimum waiver issuance criteria specified in 40 CFR Subpart S 51.360.

Waiver procedures A waiver document may be issued only by Utah County I/M technical center staff and only after verification of required documentation. Any tampered, missing, or inoperable emission control devices must have been replaced or repaired. At least \$100 for 1968 through 1980 model year vehicles, \$200 for 1981 through 1995 and \$400.00 for 1996 and newer model year vehicles must have been spent on acceptable emission repairs as verified by a Utah County I/M program auditor by physical examination of the vehicle and review of the repair documentation. Repair documentation, such as receipts, are copied and retained by auditor to prevent reuse. Utah County requires that emissions-related repairs be performed by a licensed auto repair business in order to count the labor costs. Any vehicle that experiences an increase in all emissions levels is not eligible for an emissions repair waiver regardless of the amount spent to repair the vehicle. Also, before a waiver can be issued, the vehicle meet a specific waiver cutpoint. Utah County's waiver policy on emission standards for carbon monoxide can be found in Appendix E of Utah County's Vehicles Emission Inspection/Maintenance Program Ordinance. In the state of Utah, vehicles still under the federal emissions warranty are not eligible for a waiver until all warranties are exhausted. Warranted repair and tampering repair may not be applied to the repair cost waiver limits. Waivers are only valid for one test cycle. The vehicle owner surrenders the

Utah County I/M SIP

original waiver document at the time of registration; copies are not accepted for registration purposes. Specific provisions regarding waivers may be found in Utah County's I/M ordinance and the Utah Tax Commission Division of Motor Vehicle policy manual which is available upon request. The I/M program in Utah County does not provide for time extensions to relieve economic hardships in obtaining emission-related repairs.

10. Motorist compliance enforcement

Registration denial Utah County's I/M program is enforced by means of registration denial. Vehicle owners must present proof of compliance with the I/M program, a waiver, or evidence of exemption from the I/M program as a condition precedent to vehicle registration or registration renewal. See sections 4 and 6 above for a more detailed discussion of inspection frequency, inspection scheduling, license plate requirements, and enforcement of the registration requirements. Citations are routinely issued to operators of vehicles with expired or missing license plates during routine traffic stops, parking lot inspections, and roadblocks. As specified in Section 41-1a-1303 of the Utah Code, driving without registration is a Class C misdemeanor. The penalty for a Class C misdemeanor is imprisonment of no more than 90 days and \$750 for persons or up to \$1000 for corporations, associations, partnerships, or government instrumentalities. In addition to paying a fine the motorist must register the vehicle. It is currently a Class B misdemeanor to violate a county I/M regulation or ordinance. The penalty for a Class B misdemeanor is a imprisonment of not exceeding six months and for persons a fine of up to \$1000 or for corporations, associations, partnerships, or government instrumentalities a fine of up to \$5000. In Utah, the magnitude of such penalties is a judicial rather than an administrative decision. Per Section 41-1a-1315 falsification of evidences of title and registration is a second degree felony.

Certificate of Compliance The Certificate of Compliance is dated by the UTAH2000 analyzer in Utah County immediately after a passing inspection is completed. The certificate is only valid for registration purposes for two months. At the same time the analyzer also prints the following information on the certificate to ensure unambiguous vehicle identification: the vehicle identification number (VIN), license number, model year, make, and model. A sample of the Certificate of Compliance is in Appendix C of the UTAH2000 specifications. The certificates are only printed in the event that the vehicle passed the emissions inspection. Separate documentation, including the same vehicle information, is used for waivers.

Fuel changes to non-subject status Vehicle changes that would result in registration changes from a subject to exempt status require physical confirmation by Utah County I/M program personnel at the I/M technical center. Falsification of registration or title information is a felony offense.

-fi

Title transfers Proof of compliance with the I/M program is required for a title transfer. The system ensures that owners are not able to avoid the program by extending the inspection date through manipulation of the title and registration system.

Utah County I/M program staff, peace officers, and the Utah Tax Commission Motor Vehicle Customer Service Division routinely work together to ensure that motor vehicle owners that move into an I/M program area complete registration transfer including compliance with the I/M program. Except for higher education students and active duty military personnel, people are required to register their vehicles in the county in which they are domiciled. As discussed in the Vehicle Coverage section, although these two exempted classes of vehicle owners do not have to register their vehicles in Utah, they do have to comply with the I/M programs. Employment status, maintenance of a residence, enrollment of children in local schools, and voting districts are considered when identifying persons in violation of this requirement.

The Utah County I/M program staff work with citizens, the Motor Vehicle Customer Service Division and county attorneys to identify and prosecute people that illegally transfer registration to a non-subject area to avoid the I/M program. The process is very labor intensive. There are many legitimate reasons to be operating a vehicle in an I/M program area that is registered elsewhere. Violators must be dealt with on a case-by-case basis. Persons caught are subject to fines. Those prosecuted and convicted could end up with a criminal record and actual jail time. Fraudulent registration of a motor vehicle is a felony offense. Most people confronted with evidence of their guilt and the seriousness of their offense, to date, have complied promptly. The involved agencies are developing more efficient methods of dealing with illegal registrations that result in exemption from the I/M programs.

Utah County is committed to a cooperative aggressive effort to ensure that vehicles operated in the county comply with the I/M program to ensure a compliance rate of at least 95%.

11. Motorist compliance enforcement program oversight

Utah Tax Commission, tax assessors, and county roles The Utah Tax Commission Motor Vehicle Customer Service Division and county tax assessors deny application for vehicle registration or renewal of registration without submission of a valid certificate of compliance, waiver, or verified evidence of exemption. Proof is retained by the tax clerk, micro-photo-copied, and then destroyed. Altered or hand-written documents are not accepted. All certificate data is collected by Utah County I/M program auditors and subjected to scrutiny for evidence of any improprieties.

Database quality assurance The vehicle registration database is maintained and quality assured by the Motor Vehicle Customer Service Division. The I/M inspection database is

Utah County I/M SIP

maintained and quality assured by Utah County I/M program staff. See Appendix F of the UTAH2000 analyzer specifications for a file layout description. The Utah County I/M program has access to the Motor Vehicle Customer Service Division database and utilizes it on a regular basis for quality assurance purposes. The databases are subject to regular auditing, cross-referencing, and analysis. The databases are also evaluated using data obtained during roadblocks and parking lot surveys. Evidence of program effectiveness problems triggers additional joint enforcement activities.

Oversight provisions The oversight program includes verification of exempt vehicle status through inspection, data accuracy through automatic and redundant data entry for most data elements, an audit trail for program documentation to ensure control and tracking of enforcement documents, identification and verification of exemptiontriggering changes in registration data, and regular audits of I/M inspection records, I/M program databases, and the Motor Vehicle Customer Service Division database.

Enforcement staff quality assurance I/M program auditors and tax clerks involved in vehicle registration are subject to regular performance audits by their supervisors. All enforcement personnel (direct and indirect) involved in the motorist enforcement program are subject to disciplinary action, additional training, and termination for deviation from procedures. Specific provisions are outlined in the Motor Vehicle Customer Service Division procedures manual, the county I/M audit policy documents contained in the Utah County I/M ordinances, and Section 3.9 of the UTAH2000 analyzer specifications.

Co-operative enforcement oversight effort Motor Vehicle Customer Service Division, Utah Division of Air Quality, Utah highway patrol, and Utah County I/M program staff meet as needed to ensure on-going high quality oversight of joint motorist compliance program. EPA audit of this process is authorized if measures to protect taxpayer confidentiality acceptable to Motor Vehicle Customer Service Division are exercised.

12. I/M Program quality assurance

Station/inspector audits Utah County regularly audits all permitted I/M inspectors and stations to ensure compliance with the Utah County I/M ordinance. Particular attention is given to identifying and correcting any fraud or incompetence with respect to vehicle emissions inspections. Compliance with record keeping, document security, analyzer maintenance, and program security requirements are scrutinized. The inspector's skill level is also evaluated during audits. Another major purpose of the audits is to retrain inspectors, as necessary, as soon as problems are identified. Documentation sufficient to support a legal case to suspend or revoke a permit is also collected in the event of serious and/or repeated violations. Most stations and inspectors are audited every month and all at least quarterly.

Covert audits Utah County, to the extent possible, performs a covert audit of each inspector and station at least once a year. The number of covert audits at least equals the number of permitted inspectors. Covert audits are performed using a variety of vehicles that are representative of the subject fleet that are set to fail across a full range of malfunctions. Suspected problem stations and inspectors are targeted for earlier and more frequent audits. Complaints also trigger additional audits.

Covert performance audits shall include:

Remote visual observation of inspector performance, which may include the use of aids such as binoculars or video cameras, at least once per year per inspector in high-volume stations (i.e., those performing more than 4000 tests per year);

Site visits at least once per year per number of permitted inspectors (per inspector FTE) using covert vehicles set to fail (this requirement sets a minimum level of activity not a requirement that each inspector be involved in a covert audit); and

For stations that conduct both testing and repairs, at least one covert vehicle visit per station per year including purchase of repairs and subsequent retesting if the vehicle is initially failed for tailpipe emissions.

Electronic audit capabilities The UTAH2000 performs various analyses to identify statistically inconsistent data indicative of problem stations and inspectors. Overt audit records are maintained electronically in the UTAH2000. After overt audits the auditor retrieves the data containing the audit, vehicle inspection, and analyzer service, maintenance, and calibration records dating back to the previous audit. The data from each audit is added to the comprehensive central county I/M database. Further analysis of the central database results in identification of stations and inspectors for which additional audits are performed.

Auditor quality assurance Auditors receive on-the-job training in: the use of the UTAH2000 analyzer; the I/M program regulations; basic air pollution control; basic principles of emissions-related motor vehicle engine repair; emission control systems; evidence gathering; administrative procedures and laws; quality assurance practices; and covert audit procedure. Utah County sends auditors to additional automotive emissions-related training and meetings on a regular basis. Auditor supervisors audit the I/M program auditors by reviewing their documentation and also auditing a number of their stations at least once every year.

Written audit procedures The Utah County I/M program overt and covert audit procedures are contained in the Utah County I/M ordinances. A detailed description of

Utah County I/M SIP

p.

FINAL Section X, Part D, page 13

the audit capabilities of the UTAH2000 analyzer are found in Section 3.9 of the UTAH2000 analyzer specifications.

13. Enforcement against stations and inspectors

General enforcement provisions The Utah County I/M program is responsible for enforcement action against incompetent or dishonest stations and inspectors. The Utah County I/M ordinance includes a penalty schedule. For serious or repeated offenses, auditors are authorized to immediately suspend the station or inspector by locking out their UTAH2000 analyzer(s). The County does not have legal authority to impose direct fines on stations or inspectors, but suspension or revocation of a station permit results in a substantial loss of income that is far in excess of \$100 fine suggested by the EPA guidance. Station fee settlements are based on 50% of the expected revenue from I/M testing during the suspension, up to a maximum of \$3,000. Fee settlements for the inspectors are \$100 for any portion of a 15-day period, up to a maximum of \$500. A station permit may be suspended or revoked even if the owner/operator had no direct knowledge of the violation. In the case of incompetence, re-training is required before the permit is restored.

The County revised its penalty schedule to comply with the more stringent specifications included in 40 CFR 51.364; it is found in Appendix D of Utah County Ordinance 1999-28. Inspector suspensions may not be reduced by more than 75 days through a negotiated fee settlement.

Suspension and revocation Suspension or revocation effectively bars an individual from further inspections because the auditor removes the inspector's authorization code from the UTAH2000 analyzer. Evidence of indirect participation in emissions inspections by an individual while suspended or revoked could result in legal action against the station. If the station is suspended or revoked the analyzer is totally locked-out. The analyzers are initialized by an auditor for use at a single permitted station and only by inspectors permitted for that station. A record of the serial numbers of all registered analyzers and their locations is maintained by Utah County.

Enforcement records Utah County keeps comprehensive records on all audit activities, warnings, suspensions, and revocations and report enforcement activity statistics to the EPA and the executive secretary on an annual basis.

14: Data collection

li.

I/M data collection Utah County maintains records regarding inspections, equipment maintenance, and the required quality assurance activities.

Utah County 1/M S1P

Analyzer inspection data The UTAH2000 analyzer creates a detailed record of each emissions inspection performed including, but not limited to the following data, for each vehicle tested: test record number; inspection station number; inspector number; test system number; date of the test; emission test start time; the time final emission scores are determined; vehicle identification number (VIN); license plate number; test certificate number; gross vehicle weight rating (GVWR); model year, make, and type of vehicle; number of cylinders or engine displacement; transmission type; odometer reading; category of test performed (i.e., initial, first retest, or subsequent retest); fuel type of the vehicle; emission scores for HC, CO, and CO₂ at idle and 2500 RPM; and results (pass/fail/not applicable) for visual inspection of the catalytic convertor, air system, gas cap, evaporative system, and positive crankcase (PCV) valve. The tailpipe emission standards for each type of vehicle is included in a look-up table in the UTAH2000 analyzer. The UTAH2000 analyzer automatically uses appropriate standards for the type of vehicle being tested and makes a pass/fail determination. The inspection data is recorded by the UTAH2000 analyzer during the inspection procedure.

Analyzer quality assurance data Quality assurance data including a detailed history of all calibration (including the concentration values of the calibration gases), service, lockout, and document security events are also recorded and maintained by the UTAH2000 analyzer. Each UTAH2000 record includes, as applicable, station number, mechanic access number, auditor access number, service access number, analyzer serial number, date, and activity time.

Analyzer database specifications The programming criteria for the analyzer database is described in Section 3 of the UTAH2000 analyzer specifications. Appendix A of the UTAH2000 analyzer specifications contains a complete description of the electronic data records. The data containing inspection and quality assurance information is transferred electronically nightly and maintained permanently in the county's central I/M database.

15. Data analysis and reporting

Annual Reports Utah County shall analyze I/M program data and submit annual reports to the U.S. Environmental Protection Agency and the executive secretary upon request. Beginning in July of 1995, Utah County will submit to EPA and the executive secretary an annual report, for January through December of the previous year, which provides statistics on the testing, quality assurance, and enforcement activities of each I/M program. At a minimum the annual reports will include all of the data elements listed 40 CFR Subpart S 51.366.

Biennial Reports Beginning in July of 1996, and biennially thereafter, Utah County shall submit a report to EPA and the executive secretary discussing all changes made in the program design, funding, personnel levels, procedures, regulations, and legal authority.

Utah County I/M SIP

FINAL Section X, Part D, page 15

The report will also supply a detailed discussion of the impact of such changes upon the program, any weaknesses or problems discovered in the program over the previous twoyear period, the steps that were taken to address those problems, the result of those corrective actions, and any future efforts planned.

Data link Utah County requires all certified station owners to provide a computer data link between their station(s) and the Utah County health department in a manner approved by the health department and consistent with the requirements of 40 CFR 51 Subpart S.

16. Inspector training and permitting

÷ *

Inspector permitting and initial training No person may conduct an official I/M inspection unless they are certified and subsequently permitted. Utah County requires formal training prior to certifying inspectors. Each class includes at least the following information: the causes and effects of air pollution; the purpose, function, and goal of the I/M program; I/M inspection ordinances, policies, and procedures; technical details of the test procedures and the rational for their design; emission control device function, configuration, and maintenance; quality control procedures and their purposes; public relations; and safety and health issues related to the I/M inspection process. Inspector candidates will not be issued a permit unless they have passed a written test with at least 70% correct responses and a hands-on test during which the trainee demonstrates the ability to properly conduct all test procedures, calibrate the UTAH2000 analyzer, properly utilize equipment, and to follow other I/M program requirements. Utah County will take appropriate steps to insure the security of the testing process.

Inspector Training The Utah County I/M ordinance requires an inspector training program, to include both classroom and hands-on training, with provisions for initial and periodic in-service training. Utah County requires in house training for each inspector before the inspector may perform inspections periodic in-service training, over a period established by the health department.

Inspector permit renewal Inspector permits are valid for a period of one year, at which point refresher testing is required prior to permit renewal. An auditor enters the inspector's permit expiration date in the UTAH2000 analyzer(s) that the inspector is authorized to use. Starting 60 days prior to the inspector's permit expiration date the analyzer displays the message "Your mechanic permit expires MM/DD/YY". The analyzer locks-out inspectors that attempt to use the UTAH2000 analyzer after their permit expires and displays the following message. "Your mechanic permit expired (date). You are not authorized to perform any emissions inspections at this time. Please contact your local I/M office." Auditors will not clear the lock-out until the inspector has

Utah County I/M SIP

renewed the permit. Utah County may require evidence of more comprehensive emissions-related automotive training as a prerequisite to inspector permit renewal.

Inspector permit suspension and revocation A determination of inspector incompetence or failure to comply with I/M program requirements may result in suspension or revocation of an inspector's permit prior to the annual expiration date. A permit to conduct I/M inspections is not a legal right but rather a privilege bestowed by Utah County conditional upon adherence to its I/M program requirements.

Inspector training authority and materials Authority to require mandatory I/M inspector training is established and described in the Utah County I/M ordinances.

17. Public information and consumer protection

General public information The Utah County, along with the Utah Department of Environmental Quality, provides a comprehensive public education and protection program including strategies to educate the public on: Utah's air quality problems; ways that people can reduce emissions; the requirements of state and federal law; the role of motor vehicles in the air quality problem; the need for and benefits of a vehicle emissions inspection program; ways to operate and maintain a vehicle in a low-emission condition; how to find a qualified repair technician; and the requirements of the I/M program. Information is provided via direct response to inquiries for information, reports, classes, pamphlets, fairs, school presentations, workshops, news releases, posters, signs, and public meetings.

County I/M Technical Center Utah County operates an I/M Technical Center staffed with trained auditors and capable of performing emissions tests. A major function of the I/M technical center is to serve as a referee station to resolve conflicts between permitted I/M inspectors, stations, and motorists. Auditors actively protect consumers against fraud and abuse by inspectors, mechanics, and others involved in the I/M program. Complaints made on a confidential basis are investigated and resolved in a manner that conceals the person's identity to ensure protection of whistle blowers. Auditors advise motorists regarding emissions warranty provisions and assist the owners in obtaining warranty-covered repairs for eligible vehicles. Applications for waivers are evaluated by auditors at the I/M technical center and issued only after visual verification that all the requirements for a waiver have been met, including retest of the vehicle. The I/M technical centers also provide motorists with information regarding the I/M program, general air pollution issues, and emissions-related automotive repairs.

¹ Vehicle inspection report A vehicle inspection report (VIR) is printed and provided to the motorist after each vehicle inspection. A description of the VIR is included in the UTAH2000 analyzer specifications.

Utah County I/M SIP

I/M county co-operative public education tools A variety of pamphlets and radio, television, and newspaper advertisements about automotive air pollution issues are developed and distributed by the Utah County I/M program in cooperation with other I/M counties and the Utah Division of Air Quality.

18. Improving repair effectiveness

...

High priority Utah County (along with other I/M counties) and the Utah Division of Air Quality staff jointly identified improvement of repair effectiveness as a high priority action item. The Governor's Clean Air Commission also recommended making affordable additional emissions-related training available. Full emission reductions will only be realized if the repair industry is able to competently diagnose and repair emissions-related defects.

Continuing education I/M program managers have worked with Utah's higher education institutions to develop and provide emissions-related automotive technology classes to mechanics. Inspectors are also encouraged to take classes offered by trade organizations, automobile manufacturers, and dealers. The permit renewal tests are difficult enough to make this provision a good incentive. The classes are advertised in the Utah County I/M technical bulletins.

I/M program repair support activities In initiating improved automotive educational opportunities, Utah County works on a day-to-day basis to ensure that repair information is available. I/M stations are required to have available up-to-date relevant automotive diagnostic references and tools as a condition for obtaining a permit. Utah County maintains a hot line to its I/M technical center so that any mechanic can call for technical assistance related to vehicle inspection, diagnosis, and repair. Technical bulletins are regularly mailed to each permitted station with information regarding training schedules, common problems found with particular engine families, and diagnostic tips.

19. I/M SIP implementation

The I/M program ordinances or regulations, policies, procedures, and activities specified this I/M SIP revision have been implemented and shall continue until a maintenance plan without an I/M program is approved by EPA in accordance with Section 175 of the Clean Air Act as amended.

20. On-road Testing

Utah County operates Remote Sensing Device (RSD) units to help quantify I/M program effectiveness and provide additional program flexibility in the event additional emission credits and/or contingency measures are required to meet program objectives.

*i*l

UTAH COUNTY

VEHICLE EMISSION INSPECTION/MAINTENANCE PROGRAM

Adopted by the Board of County Commissioners Utah County, Utah

Date June 10, 2003

Ordinance No.2003-28

Under Authority of Sections 41-6-163.6 through 41-6-163.8. Utah Code Annotated, 1953, as amended VEHICLE EMISSIONS INSPECTION/MAINTENANCE PROGRAM

INDEX

SECTIONS PAGE					
1.0	Definit	ions 1			
2.0	Purpose	e 5			
3.0	Jurisdic	tion of the Department			
4.0	Powers	and Duties			
5. <u>0</u>	Scope				
6.0	General	Provisions			
7.0	Standar	ds and Specifications for Exhaust Gas Analyzers and Gases			
8.0	Permitt	ing Requirements of the Vehicle Emissions Inspection/Maintenance Program Stations 13			
9.0	Inspect	ion Procedures			
10.0	Adjustr	nent Procedures			
11.0	Testing	and Certification of Applicant for Emissions Mechanics			
12.0	Emissic	ons Standards for Motor Vehicle Exhaust Gases			
13.0	Certific	ates of Compliance and Waivers 23			
14.0	Discipli	inary Penalties and Right to Appeal			
15.0	Penalty				
16.0	Severat	pility			
17.0	Effectiv	ve Date			
Append	lix A	UTAH2000 Analyzer Specifications A-1			
Append	lix B	Fee Schedule			
Append	lix C	Utah County Emission Standards Cutpoints			
Append	lix D	Penalty Schedule D-1			
Append	lix E	Utah County Emission Standards Waiver Cutpoints E-1			
Appendix F		OBD Test Procedures F-1			

1.0 DEFINITIONS.

For the purpose of these Regulations, the following terms, phrases, and words shall have the following meanings, unless otherwise defined:

- 1.1 Accreditation: Certification that the instrument and instrument manufacturer meet the operating criteria, specifications and requirements of the Utah County Health Department;
- 1.2 Accuracy: The degree by which an instrument is able to determine the true concentration of pollutants of interest;
- 1.3 Air Intake Systems: Systems that allow for the induction of ambient air, including preheated air into the engine combustion chamber for the purpose of mixing with a fuel for combustion;
- 1.4 AIR System: (Air Injection Reaction) A system for providing supplementary air into a vehicle's exhaust system to promote further oxidation of HC and CO gases and to assist catalytic reaction;
- 1.5 Analyzer: An engine exhaust gas analyzer;
- 1.6 Bar 97: Refers to California Bureau of Automotive Repair Exhaust Gas Analyzer Specifications, which became effective in 1997;
- 1.7 Basic Engine Systems: Parts or assemblies that provide efficient conversion of a compressed air/fuel charge into useful power, including but not limited to, valve train mechanisms, cylinder head to block integrity, piston-ring-cylinder sealing integrity and post-combustion emissions control device integrity;
- 1.8 Bench: The main sample processing assembly of an engine exhaust gas analyzer including detectors, sampling tubes, processor boards, infrared sources and power supply;
- 1.9 Calibration: The process of establishing or verifying the accuracy of an exhaust gas analyzer to perform a consistent evaluation of engine exhaust by using different calibration gases having precisely known concentrations;
- 1.10 Calibration Gases: Gases of accurately known concentration that are used as references for establishing or verifying the calibration curve and accuracy of an exhaust gas analyzer and are approved by the Department for use. May also be referred to as a span gas;
- 1.11 Catalytic Converter: A post-combustion device that oxidizes HC and CO gases and/or reduces oxides of nitrogen gases;
- 1.12 Certificate: A Certificate of Compliance or Certificate of Waiver;
- 1.13 Certificate of Compliance: A document used in the Vehicle Emissions Inspection/Maintenance Program to certify that a vehicle meets all applicable requirements of the program;
- 1.14 Certification: Assurance by an authorized source, whether it be a laboratory, the manufacturer, the State, or the Department, that a specific product or statement is in fact true and meets all required requirements;
- 1.15 Certified Emissions Mechanic: A person who has successfully completed all certification requirements and has been issued a current, valid Mechanic Certificate of Qualification by the Department;
- 1.16 CO: Carbon monoxide;
- 1.17 Compliance: Verification that certain submission data and hardware submitted by a manufacturer for accreditation consideration, meets all required accreditation requirements;
- 1.18 County: Utah County, Utah;
- 1.19 Cutpoints: The maximum allowable concentration of carbon monoxide (CO) and hydrocarbons (HC) for a given weight class and model year of a motor vehicle, as determined by the Department, using an approved infrared exhaust gas analyzer;
- 1.20 DLC: Data Link Connector;

4

- 1.21 DTC: Diagnostic Trouble Codes;
 - 1.22 Dedicated printer: The printer on the approved analyzer which is used solely to print certificates and vehicle inspection reports;
 - 1.23 Department: The Utah County Health Department;
 - 1.24 Director: The Director of the Utah County Health Department or his authorized representative;
 - 1.25 Drift: The amount the analyzer reading changes expressed as a percentage of full scale over a period of time. Zero Drift refers to no change of the zero reading in the zero mode. Span Drift refers to the amount of change in reading of hydrocarbons or carbon monoxide when the analyzer is in the span mode;
 - 1.26 EGR System: The Exhaust Gas Recirculation System An emissions control system that recycles or recirculates a portion of the exhaust gases back to the engine combustion chambers;
 - 1.27 Emissions Control Systems: Parts, assemblies or systems originally installed by the manufacturer in or on a vehicle for the sole or primary purpose of reducing emissions;
 - 1.28 Engine Exhaust Gas Analyzer: An instrument that is capable of measuring the concentrations of certain air contaminants in the exhaust gas emanating from a motor vehicle which is approved by the Department for this use in accordance with these Regulations as an official test instrument;
 - 1.29 Evaporative Control System: An emissions control system that prevents the escape of fuel vapors from the fuel tank or air cleaner and stores them in a charcoal canister to be burned in the combustion chamber;
 - 1.30 Exemption Form: A document used to verify that a vehicle is exempt from the testing and repair/adjustment requirements of these Regulations;
 - 1.31 Fuel Control Systems: Mechanical, electro-mechanical, galvanic or electronic parts or assemblies that regulate the air/fuel ratio in an engine to provide a combustible charge;

- 1.32 Gas Span Check: The checking and adjustment, as necessary, of an exhaust gas analyzer to correspond with known concentrations of HC and CO span gases;
- 1.33 Gas Calibration Check: A procedure using known concentrations of HC and CO span gases to verify the accuracy of an analyzer in measuring HC and CO;
- 1.34 Gaseous Fuel: Means, but is not limited to, liquefied petroleum gases and natural gases in liquefied or gaseous forms;
- 1.35 Hang-up: A situation in which hydrocarbons cling to the surface of the sampling and analyzer systems in contact with the exhaust gas sample stream resulting in errors in HC readings;
- 1.36 Hexane Equivalency Value: The value derived from multiplying the propane equivalency factor (P.E.F.), as labeled on the analyzer, by the concentration of propane recorded on the calibration gas cylinder that is used to determine the proper HC reading when calibration/span gas is introduced into the analyzer bench;
- 1.37 High Altitude Specifications: Tune-up specifications that have been provided by the manufacturer to the Environmental Protection Agency for cars operating above 4,000 feet sea level;
- 1.38 HC: Hydrocarbons;
- 1.39 Idle Mode: A condition where the vehicle engine is warm and running at the rate specified by the manufacturer's curb idle, where the engine is not propelling the vehicle, and where the throttle is in the closed or idle stop position. This condition must be achieved without placing a load on the vehicle to decrease the RPM to the specified rate;
- 1.40 Ignition Systems: Parts or assemblies that are designed to cause and time the ignition of a compressed air/fuel charge;
- 1.41 I/M Program Station: A stationary Vehicle Emissions Inspection and Maintenance Station that qualifies and has a valid permit, issued by the Department, to operate as an emissions inspection and maintenance station in the Vehicle Emissions Inspection/Maintenance Program;
- 1.42 Inspection: An official vehicle emissions test performed for the purpose of issuing a Certificate of Compliance or Waiver;
- 1.43 Inspection Area: The area that is occupied by the analyzer, sample hose, and the vehicle being inspected;
- 1.44 Inspector: A Certified Emissions Mechanic;
- 1.45 Instrument: The complete analyzer system that samples and displays the concentration of pollutant hydrocarbon and carbon monoxide gas. The instrument includes the sample handling system, the exhaust gas analyzer and the enclosure cabinet;
- 1.46 Mechanic: A Certified Emissions Mechanic possessing a current permit to perform emissions testing at a location permitted by the Department;
- 1.47 MIL: Malfunction Indicator Light;

- 1.48 Motor Vehicle: A self-propelled motorized vehicle with an internal combustion powered engine which is licensed for operation on public roads and/or streets. Motor Vehicles exempted from the inspection requirements of these Regulations are listed in Section 6.6 of these Regulations;
- 1.49 Motorcycle: Every motor vehicle having a saddle for the use of the rider and designed to travel with not more than three wheels in contact with the ground, but excluding a tractor;
- 1.50 Non-certified Mechanic: Any person who has not been certified by the Department to perform official emissions tests;
- 1.51 OBD: On Board Diagnostic System;

4

- 1.52 OBD Deficient: A vehicle in which the OBD system does not function as it was intended for the purpose of diagnosing emission failures or vehicle readiness accuracy;
- 1.53 Off-highway Vehicles: A vehicle licensed to operate exclusively off highways;
- 1.54 Original Condition: The condition of the emission control system(s) as installed by the manufacturer, but not necessarily to the original level of effectiveness;
 - 1.55 PCV System: Positive Crankcase Ventilation System an emissions control system which returns crankcase vapors and blowby gases to the combustion chamber to be burned;
 - 1.56 Primary Residence: Is the place where an individual intends to permanently reside, maintains a permanent residence more than six (6) months during a calendar year, or where an individual lives more than six (6) months during a calendar year;
 - 1.57 Publicly-owned vehicles: A motor vehicle owned by a government entity, including but not limited to the federal government or any agency thereof, the State of Utah or any agency or political subdivision thereof;
 - 1.58 Readiness: Codes set in the OBD system that indicate a vehicles readiness to be OBD tested;
 - 1.59 Repeatability: The instrument's capability to provide the same value, within specified tolerances, for successive measurements of the same sample;
 - 1.60 Response Time: The period of time in seconds for an instrument to measure and display a pollutant concentration after a concentration of gases is introduced or removed from the sample probe;
 - 1.61 Span Gases: Same as calibration gases (see Section 1.10);
 - 1.62 Stabilization: The process of bringing an instrument into equilibrium with the ambient environment and operating conditions;
 - 1.63 Station: An I/M Program Station including all station personnel, employees, and owner(s);
 - 1.64 Technical Bulletin: A document, issued to Certified Emissions Mechanics and/or I/M Program Stations by the Department to update, clarify or establish policies and/or procedures for their implementation in the Vehicle Emissions Inspection/Maintenance Program;
 - 1.65 Temporary Waiver: A waiver that may be issued by the Director which will allow the temporary registration of a vehicle based upon a vehicle owner's compliance with the conditions of the waiver;

- 1.66 Training Program: A formal program administered, conducted, or approved by the Department for the education of emission inspectors/mechanics in basic emission control technology, inspection procedures, diagnosis and repair of emissions related problems, I/M Program policies, procedures, and these Regulations;
- 1.67 Utah County Board of Health: The Utah County Health Council of Utah County as authorized by Title 26A, Chapter 1 Part 1, Utah Code Annotated, 1953, as amended;
- 1.68 Utah County Board of Commissioners: The elected Utah County Commissioners;
- 1.69 UTAH2000 Analyzer: A computerized exhaust analyzer approved by the Department for use in the areas of Utah requiring inspections as specified in Section 41-6-163.6 and 41-6-163.7, Utah Code Annotated, 1953, as amended;
- 1.70 Vehicle Emissions Inspection/Maintenance Program: The program established by the Department pursuant to Section 41-6-163.6, Utah Code Annotated 1953, as amended;
- 1.71 Waiver or Certificate of Waiver: A document used to verify that a vehicle has met the repair or adjustment requirements of the I/M Program Rules and Regulations even though specific emission standards have not been met.

2.0 PURPOSE.

It is the purpose of these Regulations to reduce air pollution levels by requiring inspections in accordance with the schedule adopted by the Board of County Commissioners. and by requiring emission related repairs/adjustments for those vehicles that fail to meet prescribed standards so as to:

- 2.1 Protect and promote the public health, safety and welfare;
- 2.2 Improve air quality;
- 2.3 Comply with federal regulations contained in the Clean Air Act of 1970, 42 USC 7401-7671; and the Amendments to the Act, Amendments of 1977, PL 95-95, PL 95-190; and Amendments of 1990, PL 101-549; and
- 2.4 Comply with the law enacted by the Legislature of the State of Utah, Sections 41-6-163.6 and 41-6-163.7 Utah Code Annotated, 1953, as amended.

3.0 JURISDICTION OF THE DEPARTMENT.

All aspects of the Vehicle Emissions Inspection/Maintenance Program within Utah County enumerated in Section 2.0 shall be subject to the direction and control of the Department.

4.0 **POWERS AND DUTIES.**

- 4.1 The Department, by the Director, shall be responsible for the enforcement and administration of these Rules and Regulations and any other powers vested in it by law and shall:
 - 4.1.1 Require the submission of information, reports, plans, and specifications from I/M Program Stations as necessary to implement the provisions, requirements, and standards of these Regulations;
 - 4.1.2 Issue permits, certifications, and charge fees as necessary to implement the provisions, requirements and standards of these Regulations;

- 4.1.3 Perform inspections (audits) of any I/M Program Station, issue orders and/or notices, hold hearings, and levy administrative penalties, as necessary to effect the purposes of these Regulations,
- 4.1.4 Take samples and make analyses required to ensure that the provisions of these Regulations are met; and
- 4.1.5 Make policies and procedures necessary to ensure that the provisions of these Regulations are met and that the purposes of these Regulations are accomplished. All policies and procedures must be approved by the Utah County Commission.
- 4.2 The Department may suspend, revoke, or deny a permit, subject to the Penalty Schedule in Appendix D, of an I/M Program Station and/or require the surrender of the permit and unused Certificates of Compliance of such I/M Program Station upon showing that:

.,

- 4.2.1 A vehicle was inspected and issued a Certificate by the station personnel that did not, at the time of inspection, comply with all applicable policies, procedures, Technical Bulletins, and these Regulations;
- 4.2.2 A vehicle was inspected and rejected by the station when, in fact, the vehicle was determined, by the Director, to be in such condition that it did comply with the requirements of these Regulations;
- 4.2.3 A vehicle was inspected and was passed for the tampering inspection as detailed in Section 9.10 of these Regulations that did not at the time of inspection comply with the requirements of the Section 9.10 tampering inspection.
- 4.2.4 The station is not open and available to perform inspections during a major portion of the normal business hours of 8:00 AM to 5:00 PM Mondays through Fridays (except stations which only test their own vehicles);
- 4.2.5 The station has violated any provisions of these Regulations, or any Rule, Regulation, or Department policy properly promulgated for the operation of an I/M Program Station;
- 4.2.6 The station was not equipped as required by Section 8.0 of these Regulations;
- 4.2.7 The station is not operating from a location specified on the permit;
- 4.2.8 An official inspection was done by a non-certified mechanic or a non-certified mechanic has gained access to the official testing portion of the analyzer or a non-certified mechanic signed a Certificate;
- 4.2.9 The computerized analyzer has been tampered with or altered in any way contrary to the certification and maintenance requirements of the analyzer;
- 4.2.10 The station denies access to a representative of the Department to conduct an audit/inspection or other necessary business during regular business hours; or
- 4.2.11 In accordance with Sections 41-6-163.6 and 41-6-163.7 Utah Code Annotated, 1953, as amended, an emissions inspection for a Weber, Salt Lake, or Davis County resident was performed but not as required by the Regulations/Ordinances adopted by the applicable county,

- 4.2.12 An engine change verification form was completed and signed when in fact the engine block number was not verified by a certified mechanic or other authorized personnel approved by the Department,
- 4.2.13 The emissions analyzer was not set in the standby mode as required by the Department for data collection or other administrative purposes.
- 4.2.14 The I/M fee signage procedures are not followed as specified in Section 6.8.3.
- 4.2.15 The I/M fee has been determined by the Department to be discriminatory. I/M fees cannot be discriminatory in that different fees are assessed dependent upon vehicle ownership, vehicle make or model, owner residence, etc.
- 4.3 The Department may suspend, revoke, or deny the certificate of an official emissions mechanic, subject to the Penalty Schedule in Appendix D, and require the surrender of this certificate upon showing that:
 - 4.3.1 The Certified Emissions Mechanic caused a Certificate of Compliance to be issued without an approved inspection being made;
 - 4.3.2 The Certified Emissions Mechanic denied the issuance of a Certificate of Compliance to a vehicle that, at the time of the inspection, complied with the law for issuance of said certificate;
 - 4.3.3 The Certified Emissions Mechanic issued a Certificate of Compliance to a vehicle that, at the time of issuance, was in such condition that it did not comply with these Regulations;
 - 4.3.4 The Certified Emissions Mechanic inspected and recorded passed on the tampering inspection for a vehicle that did not at the time of inspection comply with the tampering requirements of the tampering inspection detailed in Section 9.10, regardless of whether a Certificate of Compliance was issued or not;
 - 4.3.5 Inspections were performed by the Certified Emissions Mechanic, but not in accordance with applicable policies, procedures, Technical Bulletins, and these Regulations;
 - 4.3.6 The Certified Emissions Mechanic allowed a non-certified person to perform an official I/M test or gain access to the official testing portion of the analyzer;
 - 4.3.7 The Certified Emissions Mechanic signed an inspection form or certificate stating that he had performed the emissions test when, in fact, he did not;
 - 4.3.8 The Certified Emissions Mechanic performed a Weber, Salt Lake, or Davis County test as required by Sections 41-6-163.6 and 41-6-163.7, Utah Code Annotated, 1953, as amended, but did not perform it as required by the Regulations or Ordinances governing such testing in these counties;
 - 4.3.9 The Certified Emissions Mechanic signed a certificate prior to a test being performed and prior to the certificate being printed by the dedicated printer;
 - 4.3.10 The Certified Emissions Mechanic completed and signed an engine change verification form when in fact the engine block number was not verified.
- 4.4 The Department shall respond, according to the policies and procedures of the Department, to public complaints regarding the fairness and integrity of inspections they receive and shall provide a method that inspection results may be challenged if there is a reason to believe them to be inaccurate. If Department procedures are not followed the I/M station may not charge more than \$10.00 per I/M test.
- 5.0 SCOPE.

It shall be unlawful for any person to fail to comply with any policy, procedure, Technical Bulletin, or regulation promulgated by the Department, unless expressly waived by these Regulations.

6.0 GENERAL PROVISIONS.

<u>`</u> ^

Subject to the exceptions in Section 6.6 and pursuant to the schedule in Section 6.1, individuals with their primary residence in Utah County must register their motor vehicles in Utah County and motor vehicles (of model years 1968 and newer) that are or will be registered in Utah County, or operated from a facility within Utah County shall be subject to an annual exhaust gas emission inspection performed by an I/M Program station or other entity approved by the Director.

- 6.1 Beginning July 1, 1986, a Certificate of Compliance or Waiver, or evidence that the motor vehicle is exempt from the Inspection/Maintenance Program requirements (as defined in Section 6.6) shall be presented to the County Assessor or the Utah State Tax Commission and the Air Pollution Control Fee paid (See Section 6.10.1) as conditions precedent to annual registration or annual renewal of registration of a motor vehicle.
- 6.2 If Section 41-6-158 of the Utah Code Annotated, 1953, as amended, concerning safety inspections is in effect, the official vehicle emissions inspection shall occur and a Certificate of Compliance or Waiver shall be issued within the same time period as applicable in Section 41-6-158.
- 6.3 If Section 41-6-158 of the Utah Code Annotated, 1953, as amended, concerning safety inspections is <u>not</u> in effect, the official vehicle emissions inspection shall occur and a Certificate of Compliance or Waiver shall be issued within two months prior to the date of the motor vehicle registration.
- 6.4 A Certificate issued to a dealer licensed with the State of Utah and issued in the dealer's name, shall be valid for registration purposes for a period of six months as specified in Section 41-3-28.5, Utah Code Annotated, 1953, as amended. The purchaser's name, address, and phone number shall be recorded by the Dealer on the back of the Certificate.
- 6.5 <u>Publicly-Owned Vehicles.</u> Owners of publicly-owned vehicles shall comply with the inspection program requirements pursuant to a schedule determined by the Department. Federally-owned vehicles and vehicles of employees operated on a federal installation that do not require registration in the State of Utah shall comply with the emissions testing requirements on a basis pursuant to a schedule determined by the Department.
- 6.6 <u>Vehicle Exemption</u>. The following vehicles are exempt from these Vehicle Emissions Inspection/Maintenance requirements:
 - 6.6.1 Any vehicle of model year 1967 or older;
 - 6.6.2 All agricultural implements of husbandry and any motor vehicle that qualifies for an exemption as provided by Subsection 41-6-163.6(3) or (4) Utah Code Annotated, 1953, as amended;
 - 6.6.3 Any vehicle used for maintenance or construction and not designed or licensed to operate on the highway;
 - 6.6.4 Any motorcycle or motor driven cycle (including vehicles which operate with an engine normally used in a motorcycle);

- 6.6.5 Any vehicle that operates exclusively on electricity;
- 6.6.6 Any new vehicle being sold for the first time that has a valid MSO (Manufacturer's Statement of Origin) form.
- 6.6.7 Any motor vehicle which qualifies for legislative exemptions.
- 6.7 It shall be the responsibility of the emissions inspector/mechanic if a vehicle exempted from these Regulations by Section 6.6 of these Regulations is brought to the inspector/mechanic for an official emission test to inform the owner/operator of the vehicle that the vehicle is not required to have an official emission inspection for vehicle registration purposes.
- 6.8 Official Signs.
 - 6.8.1 All I/M Program Stations, except those stations authorized to inspect only their own motor vehicles as a fleet inspection station, shall display in a conspicuous location on the premises an official sign provided or approved by the Department;
 - 6.8.2 The exhaust emissions standards, as promulgated under authority of Section 12.0 of these Regulations shall be posted in a conspicuous place on the station's premises, if required by the Department;
 - 6.8.3 The station shall post on a clear and legible sign and in a conspicuous place at the station, the fees charged by that station for the performance of the emissions inspection and the J/M Program adjustments. Block lettering of the sign shall be a minimum of one inch in height on a sign that is easily visible;
 - 6.8.4 The signs required by Sections 6.8.1, 6.8.2, and 6.8.3 shall be located so as to be easily in the public view.
- 6.9 Equipment Available for Inspection.

Required tools and equipment as noted in Section 8.1.4, supplies, records, unused Certificates of Compliance, and other required forms, shall be kept at the official I/M Program Station at all times and shall be available for inspection and collection by the Department at any time the inspection station is open for business.

- 6.9.1 A periodic inspection and audit shall be made by a Department representative to verify compliance with these Regulations for each I/M Program Station. As part of the periodic inspection and audit of the I/M Program Station the Department representative shall, as applicable, observe the performance of a gas calibration and leak check performed by the Emissions Mechanic, examine leak check and gas calibration records, and examine inspection records and Certificates of Compliance as well as other required reports, forms, or records to see that the use of these items is in compliance with these Regulations and the policies and procedures of the Department.
- 6.9.2 During the time of the inspection and audit by the Department, the Department representative shall have exclusive access to the approved emissions testing analyzer(s).
- 6.9.3 The Department representative may check the accuracy of the analyzer using Department span gas to verify that the analyzer is reading within the tolerances established by the Department. Analyzers not reading within the tolerances shall be recalibrated to acceptable tolerances or placed "out of service".

9

6.10 Fees.

The fees assessed upon I/M Program Stations and Certified Emissions Mechanics shall be determined according to a fee schedule adopted by the Board of County Commissioners. The fee schedule is referenced in Appendix B to these Regulations and may be amended by the Board of County Commissioners as the Board deems necessary to accomplish the purposes of these Regulations.

- 6.10.1 The following fee is hereby assessed upon every motor vehicle registered in Utah County annually at the time of registration of the vehicle:
 Air Pollution Control Fee -- \$1.00
 This fee assessment is included upon all motorized vehicles including those that are exempted from the inspection requirements of these Regulations by Section 6.6 unless a separate fee is assessed on other motor vehicles by other Utah County Ordinances.
- 6.10.2 Those stations participating in the program hereunder may charge fees for the required service. Those fees may not exceed, for each vehicle inspected, the following amounts:
 - 6.10.2.1 Emissions inspection not to exceed the amount specified in Appendix B. Different fees may be assessed for the two-speed idle test and the OBD test. I/M fees must be uniformly applied and cannot be discriminatory in that different fees are assessed dependent upon vehicle ownership, vehicle type, owner residence, etc. If a vehicle fails the inspection, the owner is entitled to one free reinspection if he returns to the station that performed the original inspection within fifteen (15) calendar days from the date of the initial inspection. The station shall extend the fifteen day free reinspection time to accommodate the vehicle owner if the station is unable to schedule the retest of the vehicle within the fifteen day time period. The emissions inspection fee shall be the same whether the vehicle passes or fails the emission test. At the request of the Department, an official emissions station shall extend the free retest time for vehicle owners who were unable to complete emissions repairs because of the unavailability of parts to make the necessary repairs. In no case shall this extended time exceed the storage capacity time of the emissions analyzer.
 - 6.10.2.2 Emissions adjustments for vehicles 1980 and older, as specified in Section 10.0 of these Regulations, not to exceed \$5.00 per adjustment performed. If the air/fuel mixture is sealed, then the station may charge its customary rate to perform the adjustment and reseal it. This rate is to be posted at the station in a manner approved by the Department.
 - 6.10.2.3 Duplicate Certificates of Compliance and/or duplicate VIRs issued to a vehicle's owner(s)/operator(s), not to exceed \$1.00.
- 6.10.3 These fees are subject to change and may be amended as deemed necessary by the Board of County Commissioners.
- 6.10.4 If a vehicle fails the emissions test, and is within the time and mileage requirements of the federal emissions warranty contained in the Federal Clean Air Act, the mechanic shall inform the owner/operator that he may qualify for warranty coverage of emission related repairs as provided by the vehicle manufacturer and mandated by the Federal Environmental Protection Agency. The mechanic shall provide the owner with a copy or copies of the applicable emissions warranty pamphlets provided by the Department. The station shall display in an area readily accessible to the public any informational pamphlets required by the Department.

7.0 STANDARDS AND SPECIFICATIONS FOR EXHAUST GAS ANALYZERS AND SPAN GASES.

7.1 Approval of Exhaust Gas Analyzers.

A

- 7.1.1 No emissions inspection required by these Regulations shall be performed after February 29, 2000 unless the type of instrument used for measuring exhaust gases from motor vehicles is the UTAH2000 analyzer as specified by the Department. The analyzer shall meet the requirements of the analyzer specifications referenced in Appendix A to these Regulations. The analyzer shall also be certified by the manufacturer as meeting the criteria of all Federal warranty provisions of the Clean Air Act. The instrument shall be in good working condition, capable of meeting calibration requirements of the Department, and operated according to manufacturer's specifications and operating procedures.
- 7.1.2 Any analyzer used by an I/M Program Station shall be registered with and approved by the Department and, if required, shall bear a registration sticker issued by the Department. Registration stickers are not transferable or assignable. Any new or used exhaust gas analyzer put in use after station approval must be approved by the Department before use. Analyzers used temporarily during times of breakdown or repair of the registered analyzer do not require a registration sticker but shall meet all other requirements of this section including the approval of the Department before use.
 - 7.1.2.1 The analyzer printers shall be maintained in such a manner that the printing of the Certificate and inspection report shall be clearly visible on all copies. If any printer fails to properly function, then the station shall discontinue testing until the required repairs have been performed.
- 7.1.3 Propane Equivalency Factor (P.E.F.): Each instrument shall be labeled with a valid propane equivalency factor, shown with an accuracy of at least two decimal places, (i.e., 0.52). P.E.F. confirmation shall be made on each assembled analyzer by measuring both N-hexane and propane values on assembly line quality checks. If the analyzer bench is replaced, then a new P.E.F. label applicable to the replacement bench shall be appropriately attached to the analyzer.
- 7.1.4 Running Changes: Any changes to the design characteristics or component specifications that may affect the performance of an exhaust gas analyzer to be used as an official test instrument in the Utah County I/M Program shall be approved by the Department. It shall be the instrument manufacturer's responsibility to confirm that the changes have no detrimental effect on the performance of the exhaust gas analyzer.
 - 7.1.4.1 It shall be illegal for any person to modify the hardware or software of an approved emissions analyzer without written application and formal written approval by the Department.
- 7.1.5 Calibration/Span Gases
 - 7.1.5.1 General: The instrument manufacturer and/or his designated marketing vendors shall, on request, supply at a reasonable cost, span gases approved by the Department to any ultimate purchaser of his unit. Each new or used instrument sold by the instrument manufacturer or marketing vendor shall have approved full span gas containers installed and operational at the time of delivery. The Department shall establish necessary procedures for approving span/calibration gases.

- 7.1.5.2 Span Gas Blends: The span gases supplied to any I/M Program Station shall conform to the specifications of the Department. All span/calibration gases shall meet all Federal requirements for the emissions warranty coverage. Only gas blends supplied by Department approved blenders shall be used to calibrate official I/M test analyzers.
- 7.1.6 Documentation, Logistics, and Warranty Requirements.
 - 7.1.6.1 Instruction Manual: An instruction manual shall accompany each exhaust gas analyzer and shall contain at least the following information for the analyzer:
 - (a) A complete technical description;
 - (b) The functional mechanical and electrical schematics;
 - (c) The accessories and options that are included and/or available;
 - (d) The model number, identification marking and location;

(e) Operating maintenance including daily, weekly, and monthly, accommodations and procedures for maintaining sample system integrity including, but not limited to, leaks, hang-up, calibration and filters. The services to be performed only by the manufacturer shall be clearly identified;

(f) Field calibration procedures (i.e., Department Inspection procedure with separate gas supply);

(g) Cal-port gas inlet calibration, zero, and span instructions;

(h) Information concerning the nearest service facility where equipment can be serviced; and

(i) The warranty provisions for the analyzer, including a list of warranty repair stations by name, address and telephone number.

(j) The analyzer shall be maintained in accordance with the manufacturer's recommended maintenance schedule and records of this maintenance service shall be maintained for examination by the Department.

7.2 Calibration of Exhaust Gas Analyzers.

The Department shall use and require for use in the calibration of exhaust gas analyzers, calibration/span gases and containers meeting the guidelines contained in Section 7.1.5.

- 7.3 Gas Span and Leak Check:
 - 7.3.1 A Certified Emissions Mechanic shall perform a gas calibration of the exhaust gas analyzer, with an approved calibration/span gas, within three (3) days prior to performing any emissions test, and a leak check within twenty-four (24) hours prior to performing any emissions test. The gas calibration and leak check must be performed in accordance with the analyzer specifications as referenced in Appendix A.
 - 7.3.2 The analyzer instruction manual and other Department approved information shall be reviewed by the mechanic to ensure that proper procedures are being used for performing the gas calibration check.

7.4 At any time where the analyzer requires repairs in order to be used in accordance with these Regulations, the certified station shall notify the Department that the analyzer is "Out-of-Service" and shall again notify the Department after repairs are made and before testing is resumed.

8.0 PERMIT REQUIREMENTS OF THE VEHICLE EMISSIONS INSPECTION/MAINTENANCE PROGRAM STATION.

8.1 Permit Required.

. *^*

- 8.1.1 No person shall in any way represent any place as an official I/M Program Station unless the station is operated under a valid permit issued by the Department.
- 8.1.2 The Director is authorized to issue or deny permits for the emissions inspection of vehicles and the issuance of Certificate of Compliance and Waivers.
- 8.1.3 No permit for any official I/M Program Station may be assigned, transferred, or used by any person other than the original owner identified on the permit application for that specific I/M Program Station. The permit shall be posted in a conspicuous place within public view on the premises.
- 8.1.4 Application for an I/M Program Station permit shall be made to the Department upon a form provided by the Director. No permit shall be issued unless the Director finds that the facilities, tools and equipment of the applicant comply with the requirements of these Regulations and that competent personnel, certified under the provisions of Section 11.0, are employed and will be available to make inspections and adjustments, and the operation thereof will be properly conducted in accordance with these Regulations.
 - 8.1.4.1 An I/M Station shall notify the Department and place their analyzer in standby mode while the Department places the emissions analyzer(s) in lockout if the station does not have a Certified Emissions Mechanic employed.
 - 8.1.4.2 An I/M station shall comply with all the terms stated in the permit application and all the requirements of these Regulations.
 - 8.1.4.3 As a condition for permitting all I/M Program Stations, the following tools shall be available for performance of the inspection and maintenance of motor vehicles:

(a) A Department approved exhaust gas analyzer, as referenced in Appendix A;

(b) An accurate tachometer with a suitable pickup that allows for proper testing and adjustment of the vehicle to be tested;

(c) An accurate dwell meter;

- (d) An ignition timing light;
- (e) A propane enrichment kit for idle mixture adjustment;

(f)Department approved reference materials that contain idle speed, idle mixture, timing, dwell, fast idle speed specifications, and ready reference information covering the application of emissions control systems for the model years and makes of vehicles involved in the Vehicle Emissions Inspection/Maintenance Program; (g) Sufficient hand tools for proper performance of the inspection and maintenance of the vehicle;

(h) A Department approved calibration/span gas and equipment for performing the gas calibration/span check;

(i) A suitable non-reactive tail pipe extender or probe adapter for inspecting vehicles with screened or baffled exhaust systems;

(j) The analyzer manufacturer's maintenance and calibration manual which must be retained in the inspection area;

(k) All forms, Technical Bulletins, and other information materials provided by the Department;

(1) An approved adaptor for testing dual-exhaust vehicles;

- 8.1.5 All facets of the official I/M Program shall be performed by the Certified Emissions Mechanic including:
 - 8.1.5.1 Analyzer preparation, calibration checks, and leak checks;
 - 8.1.5.2 Exhaust gas sampling and analysis for purposes of an official emissions test for issuance of a Certificate of Compliance;
 - 8.1.5.3 The five parameter adjustments (if appropriate);
 - 8.1.5.4 Preparation of reports, forms, and certificates;
 - 8.1.5.5 Accessing the official emissions testing section of the analyzer; and
 - 8.1.5.6 All other aspects of the official emissions test including but not limited to: the tampering inspection, inserting the exhaust probe, hooking up the tachometer, entering data into the analyzer, preconditioning the vehicle, and signing certificates and inspection forms.
- 8.2 An I/M Program Station shall be kept in good repair and in a safe condition for inspection purposes free of obstructions and hazards.
- 8.3 Permit Duration and Renewal.
 - 8.3.1 The permit for I/M Program Stations shall be issued annually and shall expire on the last day of the year in which it was issued. The permit shall be renewable sixty days prior to the date of expiration.
 - 8.3.2 It is the responsibility of the owner/operator of the inspection station to pursue the permit renewal through appropriate channels.
 - 8.3.3 Inspection Station to hold County Harmless: In making application for a permit or for its renewal, such action shall constitute a declaration by the applicant that the County shall be held harmless from liability incurred due to action or inaction of I/M Program Station's owners or their employees.
- 8.4 Temporary Permits may be issued on a case by case basis for fleet testing only.

- 8.4.1 A fleet is defined as ten (10) or more vehicles with the same owner(s).
- 8.4.2 Only I/M Program stations permitted to operate in Utah County may apply for a temporary permit.
- 8.4.3 Any station requesting a temporary permit must notify the Vehicle Emissions Office 24 hours in advance by completing an application provided by the Department. The 24 hour notice may be waived at the Department's discretion.
- 8.4.4 Temporary permits may be purchased in accordance with the fee schedule for any 5-day calendar period or portion thereof. The station may charge the fleet owner the cost of the temporary permit plus testing fees in accordance with the fee schedule in Appendix B.
- 8.4.5 A temporary permit will expire at the end of the day specified on the application or when the analyzer is removed from the temporary permit location whichever occurs first.
- 8.4.6 Before any testing can be performed under a temporary permit, the analyzer must be certified on location by the Department. There must be a certified mechanic present to do the testing. The certified mechanic must have a current Emissions Control Systems Application handbook or manual approved by the Department, and other equipment as required by the Department.
- 8.4.7 The I/M Program station is subject to penalties in Section 4.2 for any violations of the Vehicle Emissions Regulations that may occur during operation with a temporary permit.

9.0 INSPECTION PROCEDURE.

4

- 9.1 The official emissions inspection shall be solely performed by a Certified Emissions Mechanic who has been certified at the station where the inspection is being performed and Department approved inspection procedures are to be followed.
- 9.2 If the mechanic is unable, unqualified, or unwilling to make the required repairs or adjustments, should the vehicle fail the emissions test, he shall notify the owner/operator of the vehicle before the emissions test is administered.
- 9.3 The entire inspection shall take place within the reach of the analyzer hose.
- 9.4 The temperature of the inspection area shall be between 41° Fahrenheit and 110° Fahrenheit during the inspection.
- 9.5 The analyzer shall be kept in an area that provides adequate protection from the weather, wind, and extreme temperatures.
- 9.6 The electrical supply to the analyzer shall be able to meet the analyzer manufacturer's requirements for voltage and frequency stability.
- 9.7 The emissions mechanic shall not inspect or test any motor vehicle with a mechanical condition which may cause injury to inspection personnel or damage to the inspection station or test equipment or which may affect the validity of the test, until such condition is corrected. Such conditions include, but are not limited to: coolant, oil, or fuel leaks, low oil or low fluid levels, and high visible emissions.
- 9.8 Any time an engine stalls during an emissions test, the test shall be restarted. If an inspector cannot complete a test because of continuous stalling, fluctuating RPM measurements, or RPM measurements that are not within the Department specified parameters, then these problems shall be corrected before the test is continued.
- 9.9 The mechanic shall verify the vehicle license plate and vehicle identification numbers by comparing the information on the vehicle's registration with those on the vehicle and shall accurately record them on the inspection analyzer.

- 9.9.1 The mechanic shall verify the owner's name and address and enter this information into the analyzer. The mechanic shall determine and enter the county in which the vehicle is registered.
- 9.9.2 The mechanic shall enter completely and accurately all the information required as part of the data entry procedure for the official vehicle emissions test on the approved UTAH2000 analyzer.
- 9.10 The mechanic shall:
 - 9.10.1 Examine the emissions/tune-up specification decal (sticker) under the hood and check an approved reference manual to determine if the vehicle was manufactured with a catalytic converter, air injection reaction (AIR) system, PCV System, EGR System, and Fuel Evaporative System.
 - 9.10.2 On 1996 and newer model year vehicles follow the OBD test procedures in accordance with Appendix F.
 - 9.10.3 On 1990 through 1995 vehicles, visually inspect for the presence and apparent operability of the AIR System, catalytic converter, EGR System, Evaporative Control System, PCV System, and gas tank cap in accordance with Department procedures and record the information in the emissions analyzer. If these parts or systems have been removed or are inoperable, the owner shall repair or replace the parts or systems before the emissions test may be continued.
 - 9.10.4 On 1989 and older vehicles, visually inspect for the presence and apparent operability of the AIR system and catalytic converter in accordance with Department procedures, and record the information in the emissions analyzer. For 1977 through 1989 model year vehicles, if these parts or systems have been removed or are inoperable, the owner shall repair or replace the parts or systems before an emissions test may be continued.
 - 9.10.5 For all vehicles, if a part or parts are necessary to bring a vehicle into compliance is/are not available by the time the vehicle's registration is due, the owner/operator may obtain a signed form to that effect from a manufacturer, dealer, or mechanic who has verified the non-availability of the part(s). The owner/operator shall then take such proof to the Department. The Director may issue a Temporary Waiver, for a period of time and under such conditions as he has determined, so that the vehicle may be registered. The owner shall have until the expiration of the time period specified by the Director to complete the necessary repairs or replacement, and submit a Certificate of Compliance or Waiver to the Director to verify that the part(s) have been installed and that the vehicle is in compliance with all provisions of these Regulations. The Director is under no obligation to issue these Waivers.
 - 9.10.6 Under certain conditions as determined by the Director, the Director may issue a Temporary Waiver enabling a vehicle to be registered. The person to whom the Temporary Waiver is issued shall comply with the conditions of the Temporary Waiver as agreed upon by a signed affidavit. Failure to comply with the affidavit shall be considered a violation of these Regulations. The Director is under no requirement or obligation to issue these waivers.
- 9.11 The analyzer shall be warmed up and stabilized prior to performing any inspection.
- 9.12 Each vehicle shall be checked to determine that it is at normal operating temperature by feeling the top radiator hose or by checking the temperature gauge. Each vehicle shall be at normal operating temperature before performing the emissions inspection.
- 9.13 The inspection shall be performed with the transmission in "park" or "neutral" and with all accessories off and the emergency brake applied.
- 9.14 The analyzer probe shall be inserted into the exhaust pipe at least twelve inches or as recommended by the analyzer manufacturer, whichever is greater.

X. Part D, App 1.18

- 9.15 If a baffle or screen prevents probe insertion of at least twelve inches, a suitable probe adapter or snug fitting, non-reactive hose which effectively lengthens the exhaust pipe shall be used.
- 9.16 For all vehicles equipped with a multiple exhaust system that does not originate from a common point, both sides shall be tested simultaneously with an approved adaptor.
- 9.17 When inspecting a vehicle under windy conditions, the tailpipe shall be shielded from the wind with a suitable cover.
- 9.18 For 1995 Model Year Vehicles and older:
 - 9.18.1 With the tachometer properly attached;
 - 9.18.2 The vehicle shall be tested according to the testing sequence as detailed in the Analyzer specifications referenced in Appendix A as programmed into the analyzer testing sequence. Vehicles failing because of excessive exhaust dilution shall repair the dilution problem prior to continuing the emission test. The dilution standard shall be contained in the analyzer specifications as referenced in Appendix A and adjusted when the Department determines by analysis that an adjustment is necessary to yield a more accurate level of emissions readings.
- 9.19 A Certificate of Compliance shall be issued if:

٨

- 9.19.1 The vehicle emissions levels are the same as or less than the applicable emissions standards; and
- 9.19.2 For 1977 through 1995 model year vehicles, the vehicle passes the visual inspection described in Section 9.10.
- 9.20 All testing procedures for restart testing, second chance testing, etc, shall be followed as contained in the analyzer specifications referenced in Appendix A.
- 9.21 If the vehicle fails the initial emissions inspection, the owner shall have fifteen calendar days in which to have repairs or adjustments made and return the vehicle to the I/M Program Station that performed the initial inspection for one (1) free reinspection. In order to be in compliance, the vehicle that failed the initial test shall meet the following conditions:
 - 9.21.1 A Certificate of Compliance shall be issued if all of the following are met:
 - 9.21.1.1 The vehicle is re-tested;
 - 9.21.1.2 The vehicle emissions levels are the same as or less than the applicable emissions standards; and
 - 9.21.1.3 For 1977 through 1995 model year vehicles, the vehicle passes the visual emissions inspection as provided for in Section 9.10.
- 9.22 A Certificate of Waiver shall be issued only by meeting cutpoints as established in Appendix E, and subject to the following conditions:
 - 9.22.1 A Certificate of Waiver shall be issued for 1968 to 1980 model year vehicles if all of the following requirements are met:
 - 9.22.1.1 Air pollution control devices (catalytic converter, and AIR system) are in place and apparently operable on the vehicle as specified in Section 9.10. If the devices have been removed or rendered inoperable, they shall be replaced or repaired before a Waiver is granted; and

X, Part D, App 1.19

- 9.22.1.2 The vehicle continues to exceed applicable emission standards after \$100 (one hundred dollars) of acceptable emissions related repairs have been performed and the adjustments required by Section 10.0 have been performed by a Certified Emissions Mechanic as part of the \$100 (one hundred dollars) in emissions related repairs. Proof of repair costs shall be provided for the vehicle to the Department in the form of an itemized bill, invoice, work order, manifest or statement in which emissions related parts are specifically identified. If repairs are made by the vehicle owner or by someone who does not possess a valid business license for automotive work, then the cost of labor shall not be included in the \$100.00.
- 9.22.2 A Certificate of Waiver shall be issued for 1981 through 1995 model year vehicles if all of the following requirements are met:
 - 9.22.2.1 Air pollution control devices applicable and specified in Section 9.10 of these Regulations are in place and operable on the vehicle. If the devices have been removed or rendered inoperable, they shall be replaced or repaired before a Waiver is granted; and
 - 9.22.2.2 At least two hundred dollars (\$200) has been spent on acceptable emissions related repair costs for that specific vehicle, and if proof of repair costs for that specific vehicle have been provided to the Department in the form of an itemized bill, invoice, work order, manifest or statement in which emissions related parts are specifically identified. If repairs are made by the vehicle owner or by someone who does not possess a valid business license for automotive work, then the cost of labor shall not be included in the \$200.00.
 - 9.22.2.3 The vehicle is not within the time and mileage requirements of the federal emissions warranties. Any vehicle that is within the time and mileage requirements of the federal emissions warranties shall not be eligible for an emissions repair waiver, but shall be repaired to pass the emissions standards.
- 9.22.3 A Certificate of Waiver shall be issued for 1996 and newer model year vehicles if all of the following requirements are met:
 - 9.22.3.1 The appropriate air pollution control devices installed by the manufacturer are in place and operable on the vehicle. If the devices have been removed or rendered inoperable, they shall be replaced or repaired before a Waiver is granted; and
 - 9.22.3.2 At least four hundred dollars (\$400) has been spent on acceptable emissions related repair costs for that specific vehicle, and if proof of repair costs for that specific vehicle have been provided to the Department in the form of an itemized bill, invoice, work order, manifest or statement in which emissions related parts are specifically identified. If repairs are made by the vehicle owner or by someone who does not possess a valid business license for automotive work, then the cost of labor shall not be included in the \$400.00.
 - 9.22.3.3 The vehicle is not within the time and mileage requirements of the federal emissions warranties. Any vehicle that is within the time and mileage requirements of the federal emissions warranties shall not be eligible for an emissions repair waiver, but shall be repaired to pass the emissions standards.

X, Part D, App 1.20

- 9.22.4 Any vehicle that experiences any increase in all emissions levels shall not be eligible for an emissions repair waiver regardless of the amount spent in attempting to repair the vehicle. Vehicles failing for HC must show a decrease in HC emission levels to be eligible for a waiver.
- 9.22.5 As used in Sections 9.22.1 and 9.22.2, acceptable emissions related repairs:

- 9.22.5.1 Refers to those expenditures and costs associated with the adjustment, maintenance, and repair of the motor vehicle which are directly related to reduction of exhaust emissions necessary to comply with the applicable emissions standards, cutpoints, and procedures;
- 9.22.5.2 Does <u>not</u> include adjustments, maintenance, or repairs performed prior to the official emissions test.
- 9.22.5.3 Does not include the fee paid for the test;
- 9.22.5.4 Does <u>not</u> include costs associated with the repairs or replacements required by Section 9.10 or the replacement, and/or repair of air pollution control equipment on the vehicle if the need for such adjustment, maintenance, replacement, or repair is due to disconnection of, tampering with, or abuse of the emissions control systems.
- 9.22.5.5 Does <u>not</u> include repairs performed to the vehicle's exhaust system to correct problems with excessive exhaust dilution.
- 9.22.5.6 Refers to repairs, maintenance, and diagnostic evaluations of the following systems, if done according to manufacturer's specifications, to the extent that the purpose is to reduce exhaust emissions:
 - (a) Air Intake Systems;

<u>,</u> ^

- (b) Ignition Systems:
- (c) Fuel Control Systems;
- (d) Emissions Control Systems except as noted in Section 9.22.4.4.
- (e) Basic Engine Systems.
- 9.23 Information regarding all performed repairs shall be entered into the appropriated data base of the analyzer prior to the vehicle being retested.
- 9.24 Certificates of Waiver shall only be issued by the Department unless the Department determines other acceptable methods of issuing the Waivers. A waiver shall only be issued after determining that the vehicle complies with the requirements of this Section for waiver issuance.
- 9.25 Prior to referring the owner to the Department for determining waiver eligibility, the I/M Station/Mechanic shall verify that the repair and eligibility requirements of this Section have been met.
- 9.26 The Certificate and Inspection records shall be completed accurately, signed immediately and filed and distributed, as required by the Department. The customer shall be given the appropriate copy.
- 9.27 Vehicles capable of being operated on both gaseous and liquid petroleum fuels shall be tested for both fuels in accordance with the UTAH2000 analyzer specifications as referenced in Appendix A to these Regulations.
- 9.28 When a vehicle owner requests an emissions test, the mechanic shall perform the inspection in the testing mode of the approved UTAH2000 analyzer. Performing a screening test in the manual mode of the approved UTAH2000 analyzer or on a non-approved analyzer shall be a violation of these Regulations if the vehicle owner requested an emissions test. No adjustments or repairs shall be made prior to a requested *I/M* inspection.

9.29 At the end of each business day the UTAH2000 analyzer shall be placed in a standby mode and be connected to the appropriate telecommunications line in order for the Department to collect data, load certificates, update station/mechanic information or any other administrative procedures.

10.0 ADJUSTMENT PROCEDURES.

- 10.1 The following adjustments should be performed on all 1980 and older vehicles (where applicable) that fail the I/M test and must be performed by an I/M Certified Mechanic before a vehicle will be eligible for an emissions waiver.
- 10.2 The high altitude specifications for idle speed, idle air/fuel mixture, ignition timing, and dwell, shall be determined for the purpose of adjustment. If no high altitude specifications are available, the mechanic shall refer to the emissions tune-up specification decal in the engine compartment or sea level specifications taken from a suitable reference manual for proper specifications. Fuel control systems designed with sealed, tamper-resistant adjustment screws for air/fuel mixture shall be adjusted according to manufacturer's specifications and resealed. On vehicles that have limiter caps on the fuel control systems, the limiter caps shall be removed and the air/fuel ratio adjusted to meet manufacturer's specifications and the proper limiter caps shall be reinstalled. The adjustment procedures shall be as follows:
 - 10.2.1 Failed vehicle readings of CO and HC shall be recorded BEFORE any adjustments are made;
 - 10.2.2 The dwell, if applicable, shall be checked with a dwell meter to determine if it is within the recommended tolerance of _+ 2 degrees of specifications. The dwell shall be reset if it exceeds this tolerance.
 - 10.2.3 The idle speed shall be checked with a tachometer to determine if it is within _+ 50 rpm of the manufacturer's specifications. If it is not, it shall be set to within _+ 50 rpm of the manufacturer's specifications;
 - 10.2.4 The ignition timing shall be checked, using a timing light or engine analyzer, to determine if it is within +4 degrees to -2 degrees of the recommended setting while the engine is idling at the specified speed. If the timing exceeds this tolerance, it shall be adjusted until it falls within +4 degrees to -2 degrees of the recommended setting;
 - 10.2.5 The idle air/fuel ratio shall be adjusted according to manufacturer's suggested procedures and/or specifications using an infrared analyzer, propane enrichment kit, or tachometer;
 - 10.2.6 The choke shall be checked and adjusted according to manufacturer's suggested procedures and/or specifications, if the choke is not sealed;
 - 10.2.7 After completing the preceding steps, the idle speed shall be readjusted to manufacturer's specifications;
 - 10.2.8 The performed adjustments shall be entered in the required data base of the analyzer.
- 10.3 Engine Changes.
 - 10.3.1 All vehicles which qualify for testing under this section shall be tested by the Department only.
 - 10.3.2 Vehicles qualifying for testing under this section shall not be eligible for a repair waiver.

- 10.3.3 After July 1, 1991 no new verifications of engine changes in accordance with Sections 10.3.4 and 10.3.5 of these Regulations shall be accepted unless it is demonstrated to the satisfaction of the Director that the changed system is more effective in controlling emissions than those systems originally manufactured on the vehicle. Installing an older engine will not qualify under any circumstances. Testing according to the guidelines of Sections 10.3.4 and 10.3.5 of these Regulations shall only pertain to those vehicles that received verification by the Department prior to July 1, 1991.
- 10.3.4 For 1968 to 1984 vehicles, having an engine older than the original, the owner may apply to the Director for a verification of an engine change. Verification will be made, provided that the owner can prove the actual model year of the engine. The model year of the engine shall be substantiated by a reference manual or an affidavit from a certified emissions mechanic. The cutpoints for the year of the engine may then be used rather than the year of the vehicle. Vehicles of model years 1968 or newer with engines older than 1968 shall be tested according to 1968 standards.
- 10.3.5 If a 1985 or newer vehicle has an engine other than the original, the vehicle owner must demonstrate to the satisfaction of the Director that the emission control systems on that engine are as effective in controlling emissions as those systems originally manufactured on the vehicle before a Certificate of Compliance is issued.

11.0 TESTING AND CERTIFICATION OF APPLICANT FOR EMISSIONS MECHANICS.

- 11.1 Certification Required.
 - 11.1.1 No person shall perform any part of the official emissions inspection for the issuance of a Certification of Compliance unless the person possesses a valid emissions mechanic certificate issued by the Department.
 - 11.1.2 Applications for an Official Emissions Mechanic Certificate shall be made upon a form to be prescribed by the Department. No certificate shall be issued unless the applicant has shown adequate competence by successfully completing the written and practical portions of the emissions mechanic certification requirements as specified in these Regulations.
 - 11.1.3 An applicant shall comply with all of the terms stated in the permit application and with all the requirements of these Regulations.
 - 11.1.4 An applicant shall complete a Department approved training course and shall demonstrate knowledge and skill concerning the performance of emissions inspections and adjustment of vehicles to manufacturer's specifications. Such knowledge and skill shall be shown by passing:
 - 11.1.4.1 A written qualification test including knowledge of the following:
 - (a) Operation and purposes of emission control systems;
 - (b) Relationship of HC and CO to timing and carburetion;
 - (c) Adjustment to manufacturer's and high altitude specifications;
 - (d) Inspection procedures as outlined in these Rules and Regulations;

(e) Operation of an exhaust gas analyzer including the performance of a gas calibration and leak check;

(f) The provisions of Section 207(b) warranty provisions of the Federal Clean Air Act; and

(g) The provisions of these Regulations and other Department policies and procedures.

11.1.4.2 A performance qualification test including the following:

(a) Visual inspection and knowledge of the function of the required emission control equipment;

(b) Demonstration of skill in the proper use, care, maintenance, and calibration and leak checking of approved analyzer;

(c) Demonstration of ability to conduct the emissions inspection;

(d) Demonstration of ability to adjust the engine system to manufacturer's and high altitude specifications; and

(e) Demonstration of ability to accurately and legibly complete the inspection forms.

- 11.1.5 A signed Hands-on Performance check sheet shall be necessary for successful completion of the Performance Qualification Test. The Hands-on Performance check sheet shall be signed by Department personnel or other person approved by the Department.
- 11.1.6 The Department shall issue an Emission Mechanic Certificate to an applicant upon successful completion of the requirements of Section 11.0.
- 11.1.7 The Emissions Mechanic Certificate shall be valid only at the station where the mechanic is presently employed. If the mechanic is later employed at another station, he shall notify the Department of the employment change. He shall also be required to be certified there prior to performing any emissions tests. A separate certificate number may be issued for use at the additional station as determined by the Department. Also, an additional mechanic certification fee shall be charged as specified in Appendix B. That certification will expire on the same date as the original.
- 11.2 Regualification Requirements for All Emissions Mechanics.
 - 11.2.1 Upon determination, by the Director, of the necessity of updating the qualifications for emissions mechanics, they shall be required to requalify.
 - 11.2.2 Emissions mechanics shall be required to requalify annually and within a specified time period, determined by the Director. Failure to requalify within the required period of time shall result in suspension or revocation of the emissions mechanic's certification as described in these Regulations.
- 11.3 Certification Expiration.
 - 11.3.1 The Mechanic Certificate shall be issued annually and shall expire one year from the date of issuance. The certificate shall be renewable sixty days prior to the date of expiration.
 - 11.3.2 It is the responsibility of the mechanic to pursue the renewal of the Mechanic Certificate.

12.0 EMISSIONS STANDARDS FOR MOTOR VEHICLES EXHAUST GASES.

- 12.1 In order to obtain a valid emissions Certificate of Compliance, exhaust emissions from a motor vehicle subject to an annual exhaust gas emission inspection shall not exceed the maximum concentrations for carbon monoxide (CO) and Hydrocarbons (HC) as established by the Director.
- 12.2 Maximum concentration cutpoints shall be determined by the County Commission as needed, to meet the National Ambient Air Quality Standards established by the Environmental Protection Agency. The established cutpoints contained in Appendix C, shall remain in effect until changed by order of the County Commission. Any change in cutpoints shall be effective upon the first day of any calendar month designated by the County Commission. Cutpoints shall be established by considering the following factors:
 - 12.2.1 To provide for the required stringency;
 - 12.2.2 The existing ambient air quality;

. *^*

- 12.2.3 The requirements for air quality currently in effect as promulgated by the Environmental Protection Agency, the Utah State Department of Health, and the Department. The cutpoints established shall be part of an overall progress in accordance with EPA guidelines to achieve the required tailpipe reduction of CO and HC from motor vehicles measured from the date this program is implemented;
- 12.2.4 The general level of emission control technology on vehicles registered in the county.
- 12.2.5 Population growth and other factors which may reasonably be expected to impact CO and HC concentrations in the atmosphere.
- 12.2.6 The likelihood of a particular cutpoint to achieve desired air quality goals.
- 12.2.7 To ensure compliance with the requirements of Section 41-6-163.6 and Section 41-6-163.7, Utah Code Annotated, 1953, as amended.
- 12.3 Upon determining the appropriate cutpoints (excluding waiver cutpoints), the Director shall cause notice thereof to be issued to each I/M Program Station and to the public by publication at least once in a newspaper of general circulation in the county at least thirty days prior to approval. Such notice shall indicate that written comment on the proposed cutpoint levels will be received by the County Commission until fifteen days prior to the approval of the cutpoints. The County Commission shall consider any written comment timely submitted and, should good cause appear, may alter or suspend the proposed cutpoints as appropriate. Otherwise, the proposed cutpoints shall take effect on the date determined by the County Commission.
 - 12.3.1 The cutpoints are referenced in Appendix C to these Regulations and may be amended pursuant to the provisions of Section 12.0 of these Regulations.

13.0 CERTIFICATES OF COMPLIANCE AND WAIVERS.

- 13.1 No person shall make, issue or knowingly use any imitation or counterfeit of an official Certificate of Compliance or Waiver/number.
- 13.2 No person shall knowingly use stolen or counterfeit certificates /numbers. Certificates of Compliance /numbers shall be purchased only from the Department.
- 13.3 No refund or credit shall be allowed for unused certificates/numbers, except as provided in Section 13.10.
- 13.4 Purchase of Certificates of Compliance/numbers.
 - 13.4.1 Certificates of Compliance/numbers shall be purchased in person or by mail with an approved check or money order. Sales shall only be made to a representative of the I/M

V Davi D Ann 1 95

Program Station possessing an acceptable form of identification. Certificates of Compliance numbers purchased by mail will be loaded into the I/M Program station by a representative of the Department.

- 13.4.2 Certificates of Compliance/numbers shall be sold at the cost adopted by the Board of County Commissioners and referenced in Appendix B, in lots to be determined by the Department. The Department may limit the number of certificate/numbers purchased to the number that the Department feels can be secured and stored safely.
- 13.4.3 Certificates of Compliance/numbers shall not be sold, loaned, transferred, or given to any other I/M Program Station, or any unauthorized individual. The I/M Program Station shall at all times account for all Certificate/numbers that have been purchased by the station.
- 13.5 Certificates of Compliance shall only be issued after being printed by the approved analyzer printer dedicated to the printing of certificates. Completion of certificates by handwritten information or the use of a non-dedicated printer by any person or station other than the Department or Director is strictly prohibited. The Certificates shall be signed only after being printed and shall be signed immediately after printing.
- 13.6 Certificates of Compliance and Waivers shall not be issued until an inspection has been performed as required by these Regulations.
- 13.7 Certificates of Compliance/numbers shall be used in sequential order by I/M Stations.
- 13.8 Certificates of Compliance/numbers found to be missing, stolen, or unaccounted for, shall be reported to the Department within twenty-four hours and the station shall cease performing emissions tests until an investigation by the Department has been completed and the Department reauthorizes the station to again begin testing.
- 13.9 I/M Program Stations shall have Certificates of Compliance on hand at all times.
- 13.10 Upon final cancellation, suspension or revocation of the I/M Program Permit of any station, the station owner, manager or other responsible person shall immediately surrender all unused Certificates of Compliance/numbers to the Department. The Department shall receipt and refund the fee paid for unused Certificates of Compliance/numbers to the station owner according to the Utah County Auditor's procedures. Upon transfer or termination of business ownership, the station permit and all Certificates of Compliance/numbers shall be immediately forwarded to the Department. Any person acquiring a business that has been permitted as an official I/M Program Station, is prohibited from using any Permit or emissions Certificate/numbers issued to the former business.
- 13.11 Any analyzer manufacturer or their authorized representative who repossesses or otherwise removes an approved I/M analyzer from an official I/M Station shall immediately notify the Department.

14.0 DISCIPLINARY PENALTIES AND RIGHT TO APPEAL.

- 14.1 When the Department, or its representative(s), receives information of a violation of any regulation contained herein which may result in a permit denial, revocation or suspension, the department shall, notify the affected entity, in writing, informing the entity of the violation and penalties to be enforced, and further informing the entity of the right of appeal and of the date, time, and location of an appeals hearing, if one has been scheduled, together with a copy of Appendix D. No appeal may be made on a formal warning.
 - 14.1.1 In considering the appropriate administrative action to be taken as indicated in Appendix D, the Director shall consider the following:

- 14.1.1.1 whether the violation was unintentional or careless;
- 14.1.1.2 the frequency of the violation or violations;
- 14.1.1.3 the audit and covert audit history of the station and the mechanic;
- 14.1.1.4 whether the fault lies with the mechanic or with the station.
- 14.1.2 After consideration of the factors in Section 14.1.1 the Director may take appropriate administrative action as indicated in Appendix D against either the I/M station, the I/M mechanic or both.
- 14.2 Appeals Hearing Procedure:
 - 14.2.1 A recorded appeals hearing shall be held at the request of the affected entity in order to determine the accuracy of information obtained by the department and whether there are mitigating factors which would justify a reduction of the imposed penalties.
 - 14.2.2 The requesting party may bring to the hearing any witnesses and any evidence believed to be pertinent to disciplinary action.
 - 14.2.3 The appeal shall be heard by the Vehicle Inspection and Maintenance Appeal Board ("I/M Board"), a Board consisting of at least three persons, who are not employees of the Utah County Health Department, appointed by the Utah County Commission. The I/M Board shall have the discretion to determine which witnesses shall be heard and what evidence is relevant.
 - 14.2.4 Violations determined to be intentional or flagrant shall result in the maximum enforcement of the penalty schedule pursuant to Appendix D herein.
 - 14.2.5 In considering whether to reduce a penalty indicated by Appendix D, the I/M Board and the department shall consider the following:
 - 14.2.5.1 whether the violation was unintentional or careless;
 - 14.2.5.2 the frequency of the violation or violations;
 - 14.2.5.3 the audit and covert audit history of the station and the mechanic
 - 14.2.5.4 whether the fault lies with the mechanic or with the station.
- 14.3 Written notice of the final determination of the I/M Board, including the Board's finding under Section 14.2.5 hereof, shall be made within ten (10) working days after the conclusion of the appeals hearing.
- 14.4 After receiving a suspension, a station may request a reduction in length of the suspension pursuant to a consent agreement. A substitute consent agreement allows the department to substitute a monetary penalty in lieu of the suspension time.
- 14.5 For stations, monetary penalties assessed pursuant to a consent agreement shall be as follows: (1/2)(average I/M tests per day [calculated from tests conducted over the last 12 months or length of time the station has been open, which ever is less)(number of suspension days)(dollar amount per test). The maximum monetary fee settlement cannot exceed \$3,000.00.
- 14.6 For mechanics, monetary penalties assessed pursuant to a consent agreement shall be as follows: \$100 minimum to \$500 maximum in \$100 increments (equal to 15 days or any portion thereof, of suspension time, up to 75 days) in lieu of all or a portion of the suspension.

X, Part D, App 1.27

15.0 PENALTY.

- 15.1 Any person who is found guilty of violating any of the provisions of these Rules and Regulations, either by failing to do those acts required herein or by doing a prohibited act, shall be guilty of a class B misdemeanor pursuant to Section 26A-1-123, Utah Code Annotated, 1953, as amended. If a person is found guilty of a subsequent similar violation within two years, he shall be guilty of a class A misdemeanor pursuant to Section 26A-1-123, Utah Code Annotated, 1953, as amended.
- 15.2 Each day such violation is committed or permitted to continue shall constitute a separate violation.
- 15.3 The city attorney, or, if appropriate, the County Attorney, may initiate legal action, civil or criminal, requested by the Department to abate any condition that exists in violation of these Rules and Regulations.
- 15.4 In addition to other penalties imposed by a court of competent jurisdictions, any person(s) found guilty of violating any of these Rules and Regulations shall be liable for all expenses incurred by the Department.
- 15.5 A Penalty Schedule for permit warning, suspension, or revocation is adopted as Appendix D and may be amended by the County Commission as the Commission deems necessary to accomplish the purposes of these Regulations.

16.0 SEVERABILITY.

If any provision, clause, sentence, or paragraph of these Regulations or the application thereof to any person or circumstances shall be held to be invalid, such invalidity shall not affect the other provisions or applications of these Regulations. The valid part of any clause, sentence, or paragraph of these Regulations shall be given independence from the invalid provisions or application and to this end the provisions of these Regulations are hereby declared to be severable.

17.0 EFFECTIVE DATE.

These Regulations shall become effective the day of their enactment by the Board of County Commissioners of Utah County. The Appendices become effective on the specific date found on each of the appendices.

APPENDIX A

REVISED UTAH 2000 EMISSION INSPECTION SYSTEM SPECIFICATIONS

January 2000

.

.. .

.

.

.

.

,

, A .

.

.

UTAH 2000 TWO SPEED IDLE & OBD II SPECIFICATION

SECTION 1. INTRODUCTION

1.1 BACKGROUND INFORMATION

The County has adopted Regulation __XXX_____ that changes the emission test used in the inspection/maintenance (I/M) program from that specified in UTAH-91. The previously required two-speed idle (TSI) test will be modified to conform to BAR-97 TSI standards, a series of enhanced two speed idle tests. Vehicle model years 1995 and older will now be subject to the enhanced TSI testing.

In general, vehicles 1996 and newer will use an OBD II test procedure. This document details the specifications for emission test systems to perform the two mode TSI, OBD II, safety, and gas cap tests.

1.2 COMMON TERMS AND DEFINITIONS

This document hence forth shall use the term "the County" to refer to the Counties of Davis, Utah and Weber as a single entity for the purposes of this specification.

The following words may have been used interchangeably within this document: For the physical cabinet and hardware;

Analyzer	Software
UTAH 2000	TAS-Test Analyzer System
EIS	Unit
Instrument	
For the Contractor;	
Contractor	Manufacturer

1.3 ELECTRONIC TRANSMISSION

1.3.1 Electronic Data Transmission (EDT) Overview

A required component of the enhanced program is the electronic transmission of data about the vehicle under test and its test results. Electronic Data Transmission (EDT) is the name that the County has given to the electronic network that enables the UTAH 2000 to automatically connect to the County's centralized Vehicle Information Database (HOST) via the modem and dial-up connection.

a) Mandatory EDT Service:

In order to comply with the EDT mandate, each inspection station shall obtain and maintain EDT services through the County's designated EDT contractor. The following criteria shall be met before an EIS is used for I/M test certification: (1) the EIS shall be connected to, and shall be fully functional with the EDT service and (2) the EIS shall maintain, and be operational with the most currently approved software and hardware updates. However, the UTAH 2000 units shall be capable of properly operating independently of the EDT service in the event the EDT services are not functioning for program start-up.

NOTE: The electronic certificate feature is required for authorization of certificate numbers. The UTAH 2000 will also have a similar function for bar code entry of certificates. The specification describes a manual method of achieving the same function. At least the manual method must exist at program start-up.

b) **Charges for EDT Services:**

Inspection stations must maintain EDT service in accordance with the terms specified by the contract between the County and the EDT contractor.

c) Communication Protocol

The Contractor will be required to finalize development of a communications protocol for the transmission of data to and from the HOST. The Contractor shall provide the algorithm/procedures or software that facilitates communication between the UTAH 2000 analyzers and the EDT host computers.

d) EDT Host Computer System (HCS)

The EIS will be required to interface to the EDT network host system. The HCS will run on a Microsoft Windows NT network. The transmitted data files will be stored in a Microsoft SQL Server database.

Each analyzer must be able to Communicate to the Windows NT 4.0 Server via a secured/encrypted method. The County prior to implementation must approve the proposed method. All transactions with the HCS will be accomplished using the approved method only.

1.3.2 Form, Manner and Frequency of Data Transmittals for ET

- a) Form: For each inspection, the data transmittal shall consist of the vehicle's test information and, when required, repair information, safety information, audit data, tamper data, calibration data, and lockout information.
- b) Manner: The manner of the data transmittal shall be using the method approved by the County and subsequently adopted into the UTAH 2000 specifications. The UTAH 2000 must be maintained to ensure proper operation and shall be connected to a fully operational connection during all times of operation.
- c) Frequency: The data shall be transmitted for inspection and shall include one transmission per scheduled time period. If the initial contact results in no match being found, an additional transmission may be required.

1.4 TAMPER RESISTANCE

Since the proposed EIS must be California BAR-97 Approved, the controlled access design shall be that method used by the manufacturer during California BAR-97 Approval. Measures for securing any additional hardware not included in the California BAR-97 Approval shall be submitted for approval by the County. Analyzer operators,

the County field representatives, Contractor's representatives, and manufacturer's representatives shall be prevented from creating or changing any test results, the County programs or the County data files contained in the EIS as called for in this specification. The EIS' shall utilize special BIOS, partitions (or equivalent approved by the County), as well as other appropriate software and hardware provisions deemed necessary by the County to protect the I/M files and programs. File and program protection may consist of mechanical systems in combination with electronic/software systems. The protection features shall prevent access to the secured disk drives and portions of the hard disk containing I/M programs and test data. The "control" key, or its functional equivalent giving access to the operating system (OS), shall not be activated except through the use of a special password algorithm securing the EIS' service-related page(s) and menu(s). The County shall approve the password algorithm at the time of certification testing. The Contractor may propose other security or protection alternatives, such as more sophisticated BIOS limitations and LPT port key, for approval by the County.

In addition, the emission analyzer and the sampling system shall be made tamperresistant. As a minimum, the manufacturer shall develop tamper-resistant features to prevent unauthorized access through the cabinet. Micro-switches, keyed locks, softwarecontrolled locks, and software algorithms requiring the use of an access code shall all be utilized where appropriate. Access codes for the County functions shall be changed daily based on an algorithm proposed by the Contractor and provided to the County. Service access codes shall be changed daily based on a unique algorithm provided by the Contractor. Algorithms must be changeable at the request of the County as part of a software update. Manufacturers must utilize electronic monitoring on the doors securing the floppy disk drives. The following examples illustrate ineffective, and therefore unacceptable, security measures: A mercury switch would not be effective if the analyzer can be tipped over to one side to trigger the switch.

A software-controlled solenoid sensor shall be used on the secured floppy drive door of all UTAH 2000 units. This sensor must be used in conjunction with any key or combination lock. The sensor shall be controlled by the EIS software, allowing the unlatching of the doors in response to authorized requests from the County Menu, always maintaining the appropriate levels of security. All UTAH 2000 EIS units shall have sensors, such as micro-switches, to detect the "open/closed" state of the doors, as well as other secured areas of the EIS. The EIS shall monitor these sensors and shall define an inappropriate state as a tamper.

Manufacturers may offer analyzers with additional disk drives that can run optional software application programs. However, the optional disk drives shall be secured from the BIOS, operating system and all other *I*/M related programs and test data (or equivalent acceptable to the County). If tampering occurs, a software lockout algorithm shall be activated which aborts any existing test sequence and prevents further I/M testing until the lockout is cleared by a the County field representative (or other representatives authorized by the County). In addition, manufacturers must describe, to the County's satisfaction, what security measures will be taken to prevent the unauthorized use of access codes, keys and combinations to the secured areas of the analyzer under each of the following circumstances:

- 1. Tampering has occurred.
- 2. A manufacturer's service technician quits or is fired.
- 3. 3. A combination, key or critical access code is obtained by an

unauthorized person(s) such as an inspection technician.

The County field representatives or other representatives authorized by the County may have access to the analyzer's OS or be able to modify files on the hard disk. At no point shall technicians have access to either the OS or the BIOS.

The use of micro-switches to detect unauthorized entry is acceptable. However, unauthorized access to the secured areas of the analyzer shall be detected even when the power is off. The analyzer shall record the type and location of each tamper. The tamper attempts shall be recorded in a tamper file which includes the date of the tamper-caused lockout, the type and location of the lockout, the date the lockout was cleared and who it was cleared by (the County or manufacturer's service representative). The specific tamper type and location shall only be accessible through the County Menu - LOCKOUT EIS function.

The lockout system shall be designed so that it can be activated by a the County field representative from the County Menu. Only the County field representatives (or other representatives authorized by the County) may remove lockouts put in place from the County Menu. Manufacturers shall develop a system by which their service technicians shall be prevented, by some method approved by the County, from clearing the County installed lockouts. In particular, the following policies shall apply to the manufacturers' field representatives:

- 1. They shall <u>not</u> be capable of clearing a County -installed lockout;
- 2. They shall <u>not</u> be capable of clearing a lockout due to a requirement for a three-day gas calibration/leak check;
- 3. They shall <u>not</u> be capable to add, delete or modify Station or Technician license number;
- 4. They shall <u>not</u> be capable of altering the calibration gas values;
- 5. They shall <u>not</u> clear a lockout when there is evidence of physical tampering. Furthermore, they shall report this or any other type of lockout to the County field office by the end of the next working day after discovering the lockout.

The access codes used by the manufacturer's service representatives shall be changed automatically by the EIS on a daily basis. The algorithm must not be available to manufacturer's field service personnel. The daily service access codes may only be given to authorized field service representatives and may not be provided more than one week in advance.

The tamper resistance features shall be designed so that software programs, especially those that deal with repair and diagnosis of vehicles, can be added at a later date.

Optional software packages supplied by the manufacturer shall: 1) not interfere with the normal operation of the I/M inspection and testing software; 2) not compromise the tamper-resistance of the analyzer or software (such as giving the technicians access to the OS); and 3) be approved by the County before they are delivered or installed in any UTAH 2000 analyzers.

Access to and from all required and mandatory-option programs shall be "seamless." These programs shall be accessed from the Main Menu or a submenu, and, when exited, shall return directly to the menu or submenu from which they were accessed, without requiring the EIS to reboot.

1.5 ORGANIZATION OF SPECIFICATION

This document provides the specifications for the UTAH 2000 equipment and procedures to be used for performing inspections required by the County Regulation XXX

Section 1 This section is an introduction, providing background about emission testing equipment, summarizing enhancements recently added to the UTAH 2000. System security and integrity are also included in this section.

Section 2 This section gives the specifications, including performance standards, for all test-related hardware such as the computer, the analyzer, the OBD II tester, the fuel cap tester, the analyzer cabinet, and the bar code scanner.

Section 3 This section describes in detail the software specification, including data storage; the form, manner and frequency of electronic transmission including transmission of test, calibration and vehicle records, sequences and procedures for performing required tests.

Section 4 This section describes in detail the requirements for the Vehicle Information Database (HOST) hardware and software functions.

Section 5 This section describes in detail the requirements for Warranty, Service and Documentation that must be adhered to in order to provide and guarantee a satisfactory level of protection for the UTAH 2000 customers.

The Appendices contain items referred to in the Specification such as the emissions standards table and the information that must be transmitted to and from the HOST, as well as highly technical and strictly confidential items.

SECTION 2 HARDWARE SPECIFICATIONS

2.1. OVERVIEW

Section 2 discusses the hardware performance requirements (and design requirements where necessary) for the UTAH 2000 Emission Inspection System (EIS) needed to perform emissions testing on the vehicles registered in the participating counties.

This section covers the computer and its peripherals, the emissions analytical instrument and its sample conditioning system, OBD II hardware, the cabinet and the hardware aspects of its security, bar code scanning, engine speed measurement, and other equipment.

The UTAH 2000 units must comprise of an IBM-compatible personal computer (PC) with a printer, a modem and software and hardware to perform both two-speed idle and OBD II testing; a five-gas analyzer and sample conditioning system; zero and calibration gases; a bar code reader; a fuel cap tester; a tachometer, an opacity measurement system (optional) and a cabinet.

2.1.1 Computer/Peripheral Compatibility

Computers shall be IBM-PC-compatible. They shall be able to reliably read and write IBM-compatible 1.44Mb 3.5" diskettes.

Systems must be capable of producing graphic output on CRT displays and printers. The computer and printer shall be capable of printing graphics and text displayed on CRT.

Systems must be capable of communicating with computers, specifically the HOST, using modems and a dial-up connection. The power supply must have the potential to handle at least 100 watts of additional upgrade devices.

2.2 GENERAL REQUIREMENTS

2.2.1 Availability of Circuitry

All components including circuit board and integrated circuits used in the EIS shall be types and brands that are presently in common usage. Custom ROM programs developed by the manufacturer for building the analyzer are allowed. Deviations may be allowed upon approval by the County.

2.2.2 Clock/Calendar

The EIS shall have a real time clock/calendar which shall make available the current date and time. Both time and date shall be in standard IBM PC format and used to set the computer's date and time on power up.

The EIS shall store the date and time in the *DateofTest*, *TestStartTime* and *TestEndTime* fields (or similarly named fields) of the EIS, Repair, and Safety records.

2-1

The communication software shall reset the current EIS date/time settings each time contact is made with the HOST except during communication diagnostics. The EIS clock shall be reset to the HOST clock at the beginning of each test. If the HOST determines that the EIS clock is not keeping correct time, the HOST shall set a lockout and a message shall be displayed indicating that service is required.

Resetting the clock after a lockout shall require controlled access available only to the County Representatives and the manufacturer's service technician. The access mechanism or procedures shall be approved by the County.

The analyzer clock/calendar shall be equipped with a battery backup feature that has a battery with at least a five-year expected life. The calendar shall be capable of handling the year roll-over from 1999 to 2000.

2.2.3 Data and File Transfer

All vehicle test, calibration, tamper and other required information shall be capable of being transferred from the EIS to the HOST:

- a) Via an IBM PC compatible modem (located inside the cabinet) and connection to a telephone line, electronically receiving and/or transmitting data from the HOST whenever the EIS connects to the HOST, and
- b) By use of the standard 3.5" IBM 1.44Mb compatible floppy disk.

2.2.4 **OBD II**

The EIS shall have a port to connect to any OBD II SAE Standardized Link. Hardware must be completely enclosed in EIS cabinet, less the cable to vehicle. The link shall enable the EIS to access vehicle readiness status, fault codes, MIL Request Status and engine RPM. The EIS must also have the ability to clear fault codes for all OBD II equipped vehicles.

For certification purposes, the County requires a description of the OBD II hardware and requires the hardware to undergo a series of tests to determine accuracy.

2.2.5 Analyzer Compatibility

The EIS shall be compatible with all types of automotive service operating environments. The analyzer shall operate under the conditions and performance requirements of this specification.

2.2.6 Testing Throughput Capability

The emissions analyzer shall be designed so that it is capable of performing at least 10 tests per hour for eight consecutive hours without experiencing excessive hang-up or other deleterious effects.

2.3 COMPUTERS & PERIPHERAL REQUIREMENTS

The EIS functions shall be controlled by an IBM PC-compatible computer. Each EIS must include the hardware and software needed to perform all functions required by this specification. The computer shall be capable of at least the following tasks:

- 1. Collect and record HC, CO, CO₂, and O₂, (NO upgrade) readings at a rate of once per second, at a minimum,
- 2. Monitor and clear OBD II functions,
- 3. Transmit test and other records to the HOST,
- 4. Read and interpret bar code labels from DMV registration documents, technician identification cards, calibration gas bottles, testing facility and technician licenses, referee labels, VIN and VEC labels,
- 5. Read data from compact discs (CD),
- 6. Provide storage for archived test and graphic files,
- 7. Test vehicle gas caps,
- 8. Recall as well as provide VIR reprint capability for emission test records and repair records,
- 9. Access engine RPM,
- 10. Interface with an optional partial-flow opacity-measuring device, display and record to the test record,
- 11. Demonstrate Year 2000 compliance.

The County reserves the right to add additional programs and functional performance requirements, up to the technical limits of the hardware, to improve the inspection and maintenance program.

Manufacturers may offer analyzers with additional disk drives that can run optional software/hardware application programs; however, the computer shall not be bootable from any additional drive, nor shall any program run from one of these drives have access to the computer's operating system. Programs run from an additional drive shall not be capable of interfering with, modifying, corrupting or interrupting any inspection-related program, procedure, or file.

2.3.1 Minimum Required Microcomputer Configuration

a) **Operating System**

Each unit must be supplied with an IBM PC-compatible, multi-tasking, operating system, which provides TCP/IP capability such as OS/2 connect or a MS WindowsTM variant. The County will not approve other systems which do not have full TCP/IP and multi-tasking capabilities.

Note: The multitasking capability of the UTAH 2000 computer system may be interpreted to include systems based on Windows 95/98 ®, Windows NT ®, OS2 ® or UNIX ® (the X Windows System).

b) **Processor**

The processor shall be IBM PC-compatible. Processing speed shall be equivalent to, or faster than, a computer equipped with an Intel Pentium 300MHz. The motherboard shall have Socket 7 or equivalent to allow processor upgrades.

c) RAM Memory

The system must contain at least 32Mb of user-available RAM and must be expandable to 128Mb. If configured with less than 64Mb, the system must be capable of adding at least 32Mb more memory to existing memory slots without discarding any existing memory.

d) **BIOS**

The system must include a ROM BIOS (basic input/output system) that provides a self diagnostic routine to check the performance of critical PC components (including, as a minimum, the processor, firmware, ROM, hard disk, keyboard, clock, set-up RAM and memory) upon power-up and which enables full use of the operating system. This BIOS must fully support all supplied components (an alternative may be approved by the County upon request).

e) Cache Memory

The system must contain at least 256K of external cache memory. If more than one processor is used for the central processing, then for each additional processor, 256K more cache memory must be added.

f) Bus

When equipped with all the County specified options, each unit must provide at least 100Mhz bus clock speed, two slots for future expansion, include at least 1 free PCI slot for future expansion. The PCI expansion slot or slots must be fully PCI-compliant ("plug-and-play") and be capable of mapping IRQ 14 & 15. If the video or hard drive interface is provided by the motherboard, it shall be capable of being disabled.

g) Monitors: Display Screen & Drive Trace

2-4

The active screen area must be in color, of .28 dot pitch or less, and measure at least 17" diagonally. The monitor must be capable of non-interlaced resolution up to 1024 X 768. The display must interface with a color graphics adapter fully compatible with the IBM SVGA color graphics adapter. This interface must be capable of operating in non-interlaced modes up to a resolution of 1024 X 768 while emulating 64K colors or more. The video adapter must be equipped with a 64-bit accelerator chip (or better) to increase its video processing speed and must be PCI bus-compliant. The video adapter must be easily upgradable to at least 4Mb and must be already equipped with 2Mb.

The above specifications do not apply to a second portable monitor that may be provided for the driver. However, this monitor must display all warnings and information required to perform the driving portion of the test (RPM, drive trace, etc.). This second monitor is subject to the County approval.

h) Floppy Disk

One 1.44Mb floppy drive is required. The floppy drives must have an external door protecting them from contamination (dust). The analyzer's cooling fan (if equipped) shall not create a negative pressure in the case unless the floppy drive(s) are sealed to prevent this negative pressure from drawing dust into the drive. The secured floppy disk shall be designated the "A" drive.

i) Compact Disc (CD)

Each analyzer must be equipped with one CD ROM drive. The disk drive must be protected from contamination in the shop environment. The CD ROM drive shall be capable of reading CD ROMs that are formatted per ISO 9660. The minimum acceptable sustained transfer rate is 600 kilobytes per second with a minimum acceptable average random access time of 225ms and must be multimedia and photo CD compatible as a minimum. A means for providing security to prevent unauthorized access to lower level system functions shall be submitted by the manufacturer for County approval.

j) Hard Disk

Each unit must come with at least 4 gigabytes of usable formatted uncompressed hard disk storage. The vendor must leave at least 2 gigabytes of usable storage for the County. Second-by-second data, emission inspection data (including graphics) and vehicle data will be stored in the County storage area. The system shall warn the technician with a screen prompt when the hard disk is within 10% of being full in any of the allotted storage areas. The hard disk is to be selfparking, shock mounted, and able to operate reliably in the expected hostile garage environment. The hard disk must also include a County-approved method of limiting logical access to the County data and programs. The hard drive's minimum acceptable burst transfer (external transfer) shall be 7,000 kilobytes per second. The hard drive's minimum acceptable sustained transfer (internal
transfer) shall be 2,000 kilobytes per second. The minimum acceptable average random access time shall be 14ms. No software cache can be used when measuring transfer rate or access times.

k) Hard Disk Interface

The hard disk interface must be PCI bus-compliant and use enhanced IDE Mode 4 (or better) or Fast SCSI-2 (or better) or alternative approved by the County. The hard disk interface must be capable of maintaining a minimum transfer rate of 8,000 kilobytes per second with all peripherals installed (including options).

1) I/O Ports

А

The unit must include at least one DOS/IBM compatible parallel printer port to which the printer may be connected.

Two baud-rate-programmable (300 to 115.2K or more) I/O serial ports using the County CPC female connectors with the following pin layouts must be provided. Both of these ports are to be free (the County-reserved) for future expansion or for use by the manufacturer upon approval by the County.

All the County-reserved serial ports (CPC and DB25) shall use 16550 UART chips or better. All I/O ports shall be clearly labeled and easily accessible and may be shared. All CPC pin layouts shall be as follows:

ANALYZER CPC REVERSE CONNECTOR

This connector must be compatible with an AMP 211398-1.

<u>PINS</u>	<u>SIGNAL</u>
1	GND
2	+12v
3	RTSRESET (request to send)
4	RESERVED (open)
5	SHIELD - GND
6	TXD TRANSMIT DATA
7	RCVRECEIVE DATA

The CPC ports will supply software switchable 12V DC to equipment attached. The +12V pin must provide circuit protection from shorts, or overload. The circuit protection can be in the form of a fuse, circuit breaker, etc. The circuit protection must be easily accessible to the operating technician for fuse replacement and or circuit beaker reset (unless automatic reset). The circuit must be capable of handling at least 6 watts.

2.3.2 Keyboard and Pointing Device

)

The EIS keyboard must be fully interfaced with the microcomputer and have all of the necessary normal, numeric, cursor, control, shift, alternate, and function keys needed to operate a standard IBM PC-compatible computer. A full-sized keyboard with at least 101 keys should be provided. The keyboard shall be readily available through retail outlets. The keyboard shall be removable and replaceable without requiring access to a secured area within the EIS cabinet. The keyboard must accept a standard keyboard connector. Provisions for a pointing device must be provided. If not built in, then a common connector (PS2, DB 9-pin, etc.) must be provided. The device driver must be active and compatible with an MS Mouse. Other pointing devices (such as light pens) may be approved by the County.

2.3.3 Modem

- a) The modem hardware must support the following protocols:
 - Modulation: ITU (International Telecommunications Union, formerly the CCITT) V.22, V.22bis, V.32, V.32bis, V.34.
 - Error control: ITU V.42, MNP (Microcom Network Protocol) 2, 3 and 4.
 - Compression: ITU V.42bis, MNP 5.
 - Connect Time: The modem must be capable of achieving a link with the HOST in less than 10 seconds at 56K baud or higher. The link time will be measured from the point the line is picked up to the point of connect.

The moderns must support at least the following baud rates: 1200, 2400, 4800, 9600, 12k, 14.4k, 19.2k, 21.6k, 24k, 26.4k, 28.8k, 33.6k and 56k asynchronous operation.

The modem must support the industry-standard AT command set.

If the modem is not using a common expansion bus slot or a common I/O jack (such as a modem that is an integral part of the motherboard), then a means of disabling the modem and an expansion slot or another high speed I/O port must be provided with the intent of supporting an upgraded modem if needed for future expansion. The modem lights, if equipped, shall not be visible from the outside of the EIS cabinet.

The analyzer shall have a standard female modular telephone connector located on the back of the analyzer. The telephone cord shall not be attached to the power cord. The telephone line shall be enclosed in a protective cable meeting County and UL approval. Alternative methods may be submitted to the County for approval.

2-7

2.3.4 Optional Diagnostic Assistance

This system shall be offered as an option. When analyzers are submitted to the County for certification, this option shall be installed.

Compatibility with H.324 (from International Telecommunications Union's Telecommunication Standardization Sector - ITU-T) and T120 (white boarding) is required. This may be provided by one multifunction device or multiple devices (video capture board, audio board, modem, etc.). The EIS must demonstrate ability to perform all functions.

a) Video

All video components listed in this section shall be capable of meeting the following requirements.

- 1. Capturing images in 256 shades of gray or, at a resolution of 320 x 240 pixels, at a minimum rate of one frame per second and saving the frames to the hard drive in TIFF-LZW format.
- 2. Receiving full motion video files and play them when triggered manually via CD for the optional electronic repair manual feature. These files shall be in a format that will run under Microsoft video for MS Windows Runtime Version.

b) Audio

A speaker is required on this optional system to provide the ability to play AVI files. This speaker shall also have the capability of providing audio for video teleconferencing for diagnostic assistance.

An external speaker connector is required to provide the ability to connect an external speaker or speakers to this audio system. An industry-standard speaker connector shall be used for the external connector and shall be easily accessible.

If equipped with a handset or headset and internal and/or external speakers, they shall be switchable on and off and shall have volume controls easily accessible to the technician.

An internal microphone may be provided at the manufacturer's discretion. The external microphone connector shall be a common type used for microphones. The audio system shall be capable of H.324 telecommunication. The microphone and handset/headset are not required at this time; however, the connectors and the functionality of the audio system with these components are required and must be demonstrated.

2.3.5 **Printer**

The Utah 2000 EIS shall have a single laser printer on board for printing both VIRs and certificates.

a) Certificates and VIR Printing Requirements

The EIS unit shall use a laser printer capable of printing at least 4 pages of text per minute on $8.5" \times 11"$ paper at 96 characters per line and 6 lines per inch. This laser printer will be used to print inspection reports and diagnostic information. The laser printer must print high-quality text and graphics at 600dpi or better. A Laser printer is required, no Ink-Jets or Bubble-Jets will be accepted. Printers must have as a minimum 2 Mbytes of memory and enough memory to print twelve 176 x 144 resolution (1.5" x 1.25") graphic images (pixels) in 64 shades of gray with the remainder of the 8.5×14 page filled with text. Page printers (printers that process total pages in memory before printing them) must be expandable to 4Mb of memory. Vehicle inspection reports (VIR) shall be printed for passing and failing vehicle inspections and as duplicates for a passing/failing inspection.

The printer shall print a VIR duplicating the font and clarity provided in the example VIRs (see Appendix IV). This is intended to ensure uniformity between manufacturers for style and size.

The printers shall be easily accessible to allow the clearing of paper jams, replacement of paper, toner, etc.

2.3.6 Running Changes and Other Hardware Modifications

Any changes to design characteristics, component specifications and any modifications to the hardware must be approved by the County. It will be the instrument manufacturer's responsibility to confirm that such changes have no detrimental effect on analyzer performance.

- a) Only County-approved hardware configurations and options may be used in UTAH 2000 analyzers.
- b) All proposed hardware modifications and options must be thoroughly tested before being submitted to the County.
- c) <u>ALL</u> proposed hardware modifications, including manufacturer-initiated modifications, must be submitted to the County for testing and approval as follows:
 - 1. Submit a modified UTAH 2000 analyzer to the County or arrange to update the Engineering test unit.
 - 2. All proposed hardware modifications shall be accompanied by a cover letter containing the following information:

1 1

- i. A description of <u>all</u> of the proposed modifications to be performed (including manufacturer-initiated modifications), a parts list and the installation instructions for the field service representative. Any modifications to the bench or sample system shall also be accompanied with test data and an engineering evaluation regarding the effects of the proposed modifications on the performance and reliability of the analyzer.
- ii. A timeline showing when the modifications are expected to be performed (start to finish), and how many existing units will be updated.
- iii. If any special procedures are needed to perform the hardware modifications, describe the procedures for performing the update.
- iv. If the proposed hardware modifications require changes or additions to the software, documentation for the software update shall be submitted as indicated above.
- 3. Depending on the type and number of modifications proposed, the County may require testing at the County-approved beta test sites prior to release of the software. The county will perform a preliminary review of the modifications prior to releasing it for beta-site testing.

2.4 EXHAUST GAS ANALYSIS EQUIPMENT FOR THE UTAH 2000

This section defines the requirements for the equipment needed to determine the concentrations of the exhaust gases of interest during the UTAH 2000 two-speed idle tests. It covers the analyzers/sensors and sampling systems, including sampling probes, hoses and filters.

2.4.1 General

,تر

The analyzer shall be compatible with all types of automotive service operating environments. The analyzer shall operate under the conditions and performance requirements listed below.

2.4.2 Measured Gases

Gases to be measured are hydrocarbons (HC), parts per million as hexane (ppmh); carbon monoxide (CO), percent; carbon dioxide (CO₂), percent; oxygen (O₂), upgradeble to percent; nitric oxide (NO), ppm. Opacity of diesel exhaust shall be offered as an option.

2-10

X, Part D, App 1 44

r ...

2.4.3 Types of Analyzers

HC, CO, and CO₂ shall be measured by means of non-dispersive infrared (NDIR) analysis.

2.4.4 Sampling Systems (except Opacity)

Sampling systems shall draw exhaust gas from the vehicle under test, shall remove particulate matter and aerosols from the sampled gas, shall drain the condensed water from the sample if necessary, and shall deliver the resultant gas sample to the analyzers/sensors for analysis. The sampling system shall, as a minimum, consist of a tailpipe probe, flexible sample line, a sample chiller, a continuously-draining water removal system, particulate trap, sample pump and flow control components. The sample system and its components shall be designed to conduct high through-put testing. Provisions shall be made for the introduction of zero air and calibration gases, as discussed below.

2.4.5 Analyzer Requirements

- a) Automatic Zero: The analyzer shall conduct an automatic zero adjustment (or equivalent, with County approval), prior to each test. The zero adjustment shall include the HC, CO, CO_2 and NO channels. The O_2 channel shall have its span adjusted while the other channels are being zeroed. The analyzer shall perform two steps while zeroing:
 - 1. <u>Zero Air</u>: The analyzer shall be zeroed, and the O₂ sensor spanned, using generated zero air. The Zero Air generator must be an integrated part of the sample system. See Section c.3.i for zero air requirements. System must be capable of detecting the presence of shop air.
 - 2. <u>Ambient Air:</u> Ambient air, filtered for particulates, shall be introduced to the analyzer before the sample pump, but after the sample probe, hose and filter/water trap. The analyzer shall record the concentrations of the four measured gases, but shall make <u>no</u> adjustments.
- b) **Zero Drift Lockout Threshold**: If zero and/or span drift cause the infrared signal levels to move beyond the adjustment range of the analyzer, the operator shall be locked out from testing and instructed to call for service. (The analyzer manufacturer shall indicate, in writing, at what point the drift lockout will occur.)
- c) Calibration and Leak Check: The analyzer shall, to the maximum extent possible, maintain accuracy between gas calibrations taking into account all errors including noise, repeatability, drift, linearity, temperature and barometric pressure.
 - 1. <u>General:</u> The analyzer shall automatically require and successfully pass a floppy drive check and a gas calibration for HC, CO, CO₂, and O₂ by a method that is approved by the County at least every three days or the

analyzer shall lock itself out from further I/M tests. The gas calibration shall ensure that accuracy specifications are satisfied or the analyzer shall be automatically prohibited from performing any portion of the I/M test. The gas calibration procedure shall correct the readings to the center of the allowable tolerance range. When a gas calibration is initiated, the analyzer channels shall actually be adjusted. It is not sufficient to merely check the calibration and do nothing if the analyzer is within allowable tolerances. The leak check shall be performed every 24 hours.

- 2. <u>Gas Calibration Procedure</u>: Gas calibration shall be accomplished by introducing gases traceable to the National Institute of Standards and Technology (NIST) into the analyzer either through the calibration port or through the probe. A High range calibration gas shall be introduced first, and the analyzer output shall be adjusted to the center of the tolerance range.
- 3. <u>Calibration Gases</u>: Calibration span gases and zero air utilized for calibration shall have a $\pm 2\%$ blend tolerance and a $\pm 1\%$ certified accuracy, and shall be provided by a County-certified gas blender. No more than 2 liters of each gas shall be required to successfully perform a gas calibration; exceptions shall be subject to County approval.

The analyzer shall be designed, in a manner approved by the county, to accommodate the gas cylinders, zero air generator and other hardware necessary to perform the three-day gas calibration. Other configurations may be submitted for the County's consideration. The analyzer shall be equipped with a gas calibration port. Gas cylinder mounting shall provide adequate room for routine access, servicing and replacement of cylinders, regulators, etc. Brackets and other hardware shall be located so that analyzer stability and impact protection is considered in the design. The gas cylinder storage area shall be actively ventilated to prevent gas buildup in case of leakage.

The analyzer manufacturers shall design the connectors used with the gas cylinders so that cylinders containing different concentrations or compositions of gas cannot be switched. As an alternative, manufacturers may use the same connectors on all required cylinders if they display a message instructing the operator to properly connect the hoses to the gas calibration cylinders when they are not connected correctly. In addition for this alternative, some type of reasonably permanent, prominent label or tag shall be used to readily identify which hose should be attached to which cylinder. Other alternatives may be presented to the county for consideration. In any event, disposable cylinders shall be equipped with CGA 165 connectors

· · _ |

2-12

Separate regulators shall be used for each cylinder necessary to perform a gas calibration. Regulator materials shall be compatible with the gases of interest.

The following calibration gases shall be used:

i. <u>Zero Air:</u> Concentrations: 20.9% O₂, balance N₂. Impurities: <1 ppm THC, CO, NO; <400 ppm CO₂.

ii.	<u>High Range</u> .		
		3200 ppm	propane
		8.00% carbon	monoxide
		12.0%	carbon dioxide
		Balance:	oxygen-free nitrogen

- 4. <u>Zero Air Supply Generators</u>: Zero air must be supplied to the analyzer from a California BAR-97 approved <u>zero air generator</u> meeting the following minimum requirements:
 - i. <u>Output Air Purity</u>: Generator output air shall meet the purity requirements of c) 3. i., above, when provided with inlet air containing no more than 100 ppm total hydrocarbons as methane, 100 ppm CO, 500 ppm CO₂, and 50 ppm NO_x.
 - ii. <u>Output Dewpoint:</u> $\leq -40^{\circ}$ C)
 - iii. <u>Output Particulates:</u> Filtration shall be 99.99% effective at 0.5 micron.
 - iv. <u>Operating Temperature Range:</u> +35°F to +110°F (2°C to 43°C)
 - v. <u>Warm-up Time:</u> The zero air generator shall be capable of providing a stabilized supply of air meeting the output purity and dewpoint requirements listed above in less than 30 minutes.
 - vi. <u>Mounting</u>: The air generator (1) shall comply with all applicable electrical and safety codes, (2) shall meet applicable Underwriters' Laboratories requirements (or BAR-approved equivalent), and (3) shall not cause the response time requirements of §2.4.5. r) and 2.4.6 g) to be exceeded.

2-13

X. Part D. App 1.47

- vii. <u>Connecting Hose</u>: At a minimum, a shop air connection is required from the EIS system and the EIS cabinet shall meet BAR-97 specifications.
- viii. <u>Certification</u>: At a minimum, the zero air generator must be BAR-97 Approved and the EIS must be BAR-97 Certified.
- 5. <u>Leak Check:</u> The analyzer shall require that a leak check be successfully passed every 24 hours. Refer to Section 2.4.6 Sampling System Components, subsection f) System Leak Check for leak check requirements.
- 6. <u>Other Requirements:</u> The gas calibration and leak check procedures shall require no more than five minutes. The analyzer shall provide adequate prompts on the display to guide the Inspection technician through the calibration procedure in a manner that minimizes the amount of gas used. The analyzer shall be designed to keep the loss of calibration gas to an absolute minimum (less than 0.1 liter in 24 hours) if the operator forgets to shut the valve off.
- 7. <u>Alternate Calibration Frequencies:</u> Proposals for less frequent gas calibrations will be subjected to lengthy accuracy and drift tests. Proposals of this type shall be thoroughly evaluated (e.g., lab as well as field testing in the range of the required span points for accuracy and drift for extended periods of time) and characterized prior to submission to the County.
- d) **Propane Equivalency Factor:** The nominal PEF range shall be between 0.470 and 0.560. For each audit, the nominal PEF shall be conveniently displayed for the County field representatives, in a manner acceptable to the County. Corrections shall be made automatically. The corrected PEF value may cover the range of 0.470 to 0.560.
- e) NDIR Beam Strength: The beam strength from the source to the detector for all channels shall be monitored such that when the beam degrades beyond the adjustment range of the analyzer, the analyzer shall be locked out from operation. The manufacturer shall specify at what point degradation occurs whereby the signal cannot be corrected.
- f) Date of Last Gas Calibration: The date of the last gas calibration shall be kept in non-volatile memory (or on the hard disk) and shall be displayed on the status page. When the system check is adjusted, if the date/time change, positive or negative, is greater than 48 hours, three-day gas calibration/leak check shall be required.

2-14

X. Part D, App 1.48

- g) Lockout Criteria: If the EIS has not successfully passed a 3-day calibration for a period of three days or more or a leak check for a period of 24 hours or more, it shall lock itself out from performing an official I/M test and shall display a message to the operator upon startup and attempting to initiate an official emissions inspection.
- h) Audit Gas Pressure: During a gas audit, analyzer readings shall not change by more than 1% of the reading if the audit gas pressure is modified by ±1.5 PSI from the atmospheric absolute pressure at the probe.

i) Audit Gas Blends and Gas Audit Procedure:

i.

ii.

There shall be four audit gas blends: Low Range, Mid Range #1, Mid Range #2, and High Range. Their concentrations, with $\pm 2\%$ blend tolerance and $\pm 1\%$ certified accuracy, shall be as follows:

<u>Zero Air</u> Same as zero air calibration gas, except that CO_2 impurity level shall be <1 ppm

<u>Low Range</u>	
200 ppm	propane
0.50%	carbon monoxide
6.0%	carbon dioxide
Balance:	oxygen-free nitrogen

iii.	<u>Mid Range</u>	<u>#1</u>
	960 ppm	propane
	2.40%	carbon monoxide
	3.6%	carbon dioxide
	Balance:	oxygen-free nitrogen

iv.	<u>Mid Range</u>	<u>+2</u> .
	1920 ppm	propane
	4.80%	carbon monoxide
	7.2%	carbon dioxide
	Balance:	oxygen-free nitrogen

v. <u>High Range</u> Same as High Range calibration gas

The audit procedure shall be as follows:

- 1. Enter the Audit mode or the corresponding field service mode.
- 2. Zero the analyzer.
- 3. Perform a leak check.

- 4. Flow the audit gas through the sample probe, ensuring that the pressure at the probe tip is equal to ambient barometric pressure ±0.1 in. Hg. (A balloon teed into the gas flow line is an acceptable pressure indicator. It should stand upright, but not inflated.)
- 5. When the HC, CO, and CO_2 readings have stabilized (no less than 60 seconds of gas flow), record them (HC in ppm propane).
- 6. Repeat Steps 4 and 5 for any other audit gases.
- 7. Compare the readings with the audit gas values. (Note: the HC reading should be taken in ppm propane before comparing.)
- j) Range and Accuracy:

.4

Gas	Range	Accuracy, % of point	Accuracy, absolute	Range	Accuracy, % of point	Accurac y, absolute
HC	0-4,000 ppm propane	±4%	12 ppm propane	4,001- 10,000 ppm propane 10,001 - 20,000 ppm propane	±6% ±11%	N/A N/A
CO	0 - 10.00%	±4%	0.03% CO	10.01-14.00%	<u>±</u> 6%	N/A
CO ₂	0 - 16%	±4%	0.4% CO ₂	16.1 - 18%	<u>±</u> 6%	N/A
O ₂	0 - 25%	±6%	0.2% O ₂	-	_	-

Emissions Analyzer Range and Accuracy

Rounding beyond the decimal places shown in the table shall follow the standard mathematical practice of going to next higher number for any numerical value of five or more. NOTE: This shall also hold true for pass/fail decisions during an I/M inspection. For example, if 2.00% CO passes but 2.01% CO fails and the reading is 2.0049%, the value shall be rounded down and the decision shall be "Pass;" however, if the reading is 2.0050, the value shall be rounded up and the decision shall be "Fail." Thus, the value displayed and printed on the VIR shall be consistent with the value used for the pass/fail decision.

Gas	Range	Repeatabilit y, % of point	Repeatabilit y, absolute	Range	Repeatabilit y, % of point	Repeatabilit y, absolute
HC	0-1400 ppmh	±2%	3 ppmh	1400-2000 ppmh	±3%	N/A
СО	0 - 7.00%	±2%	0.02% CO	7.01-10.00%	±3%	N/A
CO ₂	0 - 10%	±2%	0.1% CO ₂	10 - 16%	±3%	N/A
O ₂	0 - 25%	±3%	0.1% O ₂	•.	-	-

Emissions Analyzer Repeatability

Accuracy and repeatability shall be defined by the test procedures in Section 5.

1) Noise.

Emissions Analyzer Noise

Gas	Range	Noise, % of point	Noise, absolute	Range	Noise, % of point	Noise, absolute
HC	0-1400 ppmh	±0.8%	2 ppmh	1400-2000 ppmh	±1%	N/A
СО	0 - 7.00%	±0.8%	0.01% CO	7.01-10.00%	±1%	N/A
CO ₂	0 - 10%	±0.8%	0.1% CO ₂	10 - 16%	±1%	N/A
O ₂	0 - 25%	±1.5%	0.1% O ₂	-	-	-

Noise shall be defined operationally as follows: Sample Mid Range #1 Audit Gas for 20 seconds. Collect all the analyzer output readings for each channel over the 20 seconds. (For example, if the analyzer outputs are read by the EIS at the rate of twice per second, the total number of readings would be 40.) m) *Minimum Analyzer Display Resolution*: The analyzer electronics shall have sufficient resolution and accuracy to achieve the following:

HC	l ppm HC	2
CO	0.01%	CO
CO ₂	0.1%	CO ₂
O_2 (optional)	0.1%	O ₂
RPM	1	RPM

o)

- n) **Display Refresh Rate**: Dynamic information being displayed shall be refreshed at a minimum of twice per second. Alternatives may be submitted to the County for its approval.
 - **Interference Effects:** The interference effects from non-interest gases shall not exceed ± 4 ppm for HC, $\pm 0.02\%$ for CO, $\pm 0.20\%$ for CO₂. Corrections for collision-broadening effects of combined high CO and CO₂ concentrations shall be taken into account in developing the factory calibration curves, and is included in the accuracy specifications. Interference gases shall be as follows:

<u>Interfere</u>	nce Gases
16%	Carbon Dioxide in Nitrogen
1600 ppm	Hexane in Nitrogen
10%	Carbon Monoxide in Nitrogen
75 ppm	Hydrogen Sulfide in Nitrogen
75 ppm	Sulfur Dioxide in Nitrogen
18%	Carbon Dioxide and 9% Carbon Monoxide in
	Nitrogen
	Water-Saturated Hot Air

NOTE: Interference gases shall have a $\pm 2\%$ blend tolerance and $\pm 2\%$ certified accuracy.

- p) *Warm-up Time*: The analyzer shall reach stability within 30 minutes at 35°F from startup. If an analyzer does not achieve stability within the allotted time frame, it shall be locked out from I/M testing and a message shall be displayed instructing the operator to call for service.
- q) System Lockout During Warm-up: Functional operation of the gas sampling unit shall remain disabled through a system lockout until the instrument meets stability and warm-up requirements. The instrument shall be considered "warmed-up" when internal analyzer verifications are complete and the zero readings for HC, CO, CO₂, and O₂ have stabilized, within the allowable accuracy values, for five minutes without adjustment.

r) Analyzer/Sensor Response Times

Analyzer/sensor response times are defined as follows:

- 1. <u>Rise time:</u> When a gas is introduced to a sensor's sample cell inlet or inlet port (t₀), the time required by the sensor's output to rise from first indication of response to the input gas to a given percentage of the final stable reading of a gas's concentration. Two rise times are specified:
 - i. T_{90} : The time required to reach 90% of the final gas concentration reading from first indication of response to the input gas.
 - ii. T_{95} : The time required to reach 95% of the final gas concentration reading from first indication of response to the input gas.
- 2. <u>Fall Time:</u> When a gas is removed from a sensor's sample cell inlet or inlet port (t_s), the time required by the sensor's output to fall from first indication of withdrawal of the gas to a given percentage of the final stable reading of a gas's concentration. Two fall times are specified:
 - i. T_{10} : The time required to fall to 10% of the stable gas concentration reading from first indication of withdrawal of the gas.
 - ii. T_5 : The time required to fall to 5% of the stable gas concentration reading from first indication of withdrawal of the gas.

Requirements

	Maximum Response Time in Seconds For Each Channel
	HC, CO, CO ₂
T ₉₀	3.5
T ₉₅	4.5
T ₁₀	3.7
T5	4.7

The differences between T_{90} and T_{10} , and between T_{95} and T_5 , shall be no greater than 0.3 seconds.

- 1

2-19

X, Part D, App 1.53

Note that the oxygen (O_2) sensor's response time is specified as an overall system response time (see Section 2.4.6,g) in harmony with the generally-accepted European specifications.

s) HC Hang-up

When the analyzer performs a HC hang-up check before the start of an emissions inspection, the recorded ambient air readings shall be subtracted from the sampled readings to determine the amount of HC hang-up (residual HC) in the sampling system.

The analyzer shall be locked out from official emissions testing until (a) the ambient air has less than 15 ppm HC and 0.02% CO, and (b) until the residual HC obtained through the sample probe is less than 20 ppm propane.

t) Emissions Accounting/Accuracy

The manufacturer shall ensure that its analytical system provides an accurate accounting of the actual exhaust emissions produced during the test, taking into consideration the individual channel accuracy's, repeatability's, interference effects, sample transport times and analyzer response times.

2.4.6 Sampling System Components

a) General: The system shall be designed to ensure durable, leak-free operation and be easily maintained.

The sampling system shall be designed to withstand typical vehicle exhaust temperatures and high throughput, as when the vehicle is driven through a TSI test cycle for 120 seconds.

Materials that are in contact with the gases sampled shall not contaminate or change the character of the gases to be analyzed. The sampling system shall be designed to be corrosion-resistant for at least five years.

b) Sample Hose: The sample hose shall be 25 ft ± 0.5 ft in length, when measured from the front of the EIS cabinet. Other configurations may be submitted to the County for its consideration

The hose material in contact with the exhaust sample shall be nonporous, and shall not absorb, adsorb, react with, or affect the sample in any manner. The outer coating of the hose shall be abrasion-resistant and unaffected by the substances found in a typical service facility's environment.

The sample hose shall be flexible, yet shall resist kinking and crushing, as defined in Section 5.

The sample hose shall be connected to the probe and to the analyzer sample system with screw-type fittings.

- c) Sample Hose and Probe: The sample hose and probe shall withstand exhaust gas temperatures at the probe tip of up to 1100°F for five (5) minutes.
- d) Sample Probe: The analyzer manufacturer shall equip the analyzer with a sampling probe which meets the following criteria:
 - 1. <u>Retention</u> The probe shall incorporate a positive means of retention to prevent it from slipping out of the tailpipe when in use.
 - 2. <u>Hand Grip</u> A thermally-insulated, securely-attached hand grip shall be provided on the probe in such a manner that easy probe insertion using one hand is insured.
 - 3. <u>Flexibility</u> Manufacturers shall supply two types of removable probe tips with each analyzer sold. The probe and both probe tips shall meet the following criteria:
 - i. the probe shall be designed so that the tip extends 16 inches into the tailpipe;
 - ii. the probe and probe tip should be designed so the average garage operator can easily remove and reinstall them without special tools;
 - iii. a handle, made of thermally insulating materials, shall be attached to a rigid, reasonably non-crushable portion of tubing made of stainless steel or something equivalent, which can be easily removed from the sample line and reinstalled by the operator; and
 - iv. the probe tip shall be shielded so that debris is not scooped up by the probe when it is inserted into the tailpipe.
 - v. In addition, one of the probe tips supplied with the analyzer shall be of the traditional style meeting the following specifications:
 - a. flexible enough to extend into a 1¹/₂-inch diameter exhaust pipe having a three-inch radius, 45-degree bend; and
 - b. the flexible portion shall be constructed so that it is sealed to prevent any sample dilution.

. .

vi. Manufacturers shall also supply the analyzer with an essentially straight probe tip (no more than a 15° bend) meeting the following specifications:

- a. made of either stainless steel, 3/16 inch outside diameter (O.D.) solid-wall tubing, which is readily available; and
- b. designed so that the connector between the removable probe tip and the rigid portion of tubing is up inside the tailpipe at least three inches to reduce the effects of any leak that might occur.
- 4. <u>Serviceability</u> For the purposes of economical replacement, the flexible portion of the probe assembly shall be designed so it can be replaced. The probes supplied shall be readily available.
- 5. <u>Materials</u> The probe shall be made of materials that will withstand exhaust temperatures up to 1100°F. Use of dissimilar metals with thermal expansion factors of more than five percent shall not be used in either the construction of probes or connectors.
- 6. <u>Audit Gas Introduction</u> Probes shall be designed to allow, or shall be supplied with an adapter allowing, the introduction of audit gas from a one-half inch inside diameter flexible hose. The probe tip or the adapter shall be sized to provide a tight fit so that dilution cannot occur at the probe/hose connection.
- 7. <u>Probe Cap</u> A probe tip cap suitable for performing a system leak check shall be provided if the vacuum decay method of leak check is utilized. Otherwise, whatever hoses and connectors are necessary shall be provided to allow the operator to perform the leak check.

e) Particulate Filter, Water Trap and Sample Chiller

- 1. The particulate filter shall be capable of trapping 97% of all particulates and aerosols 5 microns or larger.
- 2. The filter element shall not absorb or adsorb hydrocarbons.
- 3. The water trap shall be sized to remove exhaust sample water from vehicles fueled with gasoline, gasohol, propane, compressed natural gas (CNG), as well as with alternative and oxygenated fuels, such as methanol (M85), ethanol (E85), and reformulated gasolines with MTBE as the oxygenate. The filter element, bowl and housing shall be inert to these fuels as well as to the exhaust gases from vehicles burning these fuels. The condensed water shall be continuously drained from the water trap's bowl. Sufficient water shall be trapped, regardless of fuel, to prevent condensation in the sample system or in the optical bench's sample cell.

- 4. The sample system shall incorporate a chiller to enhance water separation and system performance and extend analyzer life.
- f) System Leak Check: The analyzer shall require that a leak check be successfully passed every 24 hours.

During a leak check the analyzer shall not allow an error of more than $\pm 1\%$ of the High Range calibration gas reading (UTAH 2000 span gas). A leak equivalent to a reading error greater than $\pm 1\%$ shall be cause for failing the leak check.

g) System Response Time Requirements For Analyzer Channels:

The overall system response time of the analytical train comprises the Transport Time and the Analyzer/Sensor Response Time (see Section 2.4.5 r).

1. <u>Transport Time</u>: The time from the exhaust sample's entry into the tip of the sample probe until the analyzer/sensor first begins to respond to the sample. The Transport Time shall be no more than 5 seconds for HC, CO and CO₂ and no more than 7.5 seconds for O₂.

2. <u>System Response Time:</u>

- i. HC, CO, & CO₂ Channels: The response rise time (see §2.4.5.r)1) from the probe to the display shall be no more than eight (8) seconds to T_{90} . In addition, the response fall time shall be no greater than 8.3 seconds to T_{10} .
- ii. O_2 Channel: The response rise time shall be no greater than 15 seconds to T₉₀. The response fall time for a step change in concentration from 20.9% O₂ to 0.1% O₂ shall be not greater than 40 seconds.

h) Hang-up Check [Ref. Section 2.4.5 s)]

Activation of the emission measurements mode of the EIS shall be prevented unless a successful hang-up check has been performed immediately prior to the test sequence. The sample system's Hang-up shall not exceed 20 ppm propane prior to testing. A unit with a clean sample system shall have an HC hang-up time of no more than 120 seconds. If the HC hang-up does not drop below 20 ppm propane within 150 seconds, the following message shall be displayed: "POSSIBLE DIRTY FILTERS OR SAMPLE LINE."

i) **Dilution**

The analyzer supplier shall demonstrate to the satisfaction of the County that the flow rate on the EIS unit shall not cause more than 2% dilution during sampling of the exhaust of a 1.6L engine at normal idle. Two percent dilution is defined as a sample of 98% exhaust and 2% ambient air.

2-23

j) Back-Purge

The sample system must automatically back purge the sample system after and in between each test. A minimum pressure should be used for this purpose. The high pressure pack-purge serves to extend the life of the analyzer by flushing out particulates, moisture and HC's remaining in the sample line after each test

2.4.7 Temperature Operating Range

The analyzer, including all of the software/hardware enclosed in the cabinet, shall operate within the performance specifications described herein in ambient air temperatures ranging from 35° to 110°F. Analyzers shall be designed so that adequate air flow is provided around critical components to prevent overheating (and automatic shutdown) and to prevent the condensation of water vapor which could reduce the reliability and durability of the analyzer.

2.4.8 Humidity Operating Range

The analyzer, including all of the software/hardware enclosed in the cabinet, shall operate within the performance specifications described herein at up to 85% relative humidity throughout the required temperature range.

2.4.9 Opacity

An opacity option shall be offered for use in testing light- and medium-duty dieselpowered vehicles. It shall be a partial-flow device, meeting the performance requirements of ISO 11614, and shall interface seamlessly with the analyzer software via an RS232C port. A DB25 pin serial port or other County- approved connector is required. Adjustments such as electronic signal filtering shall be incorporated so as to correlate with other opacity-measuring devices and standards. Other methods of measuring opacity may be submitted for County consideration. The devices shall be calibrated by a method and at a frequency approved by the County.

2.4.10 Ambient Temperature Measurement

Ambient temperature shall be measured prior to the start of every inspection, and shall be recorded in the *Ambient Temperature* field of the test record. The temperature measuring device shall have the following minimum characteristics:

Range:	0 - 140°F
Accuracy:	±3⁰F

2.5 CABINET & PERIPHERAL REQUIREMENTS All cabinets, including modifications are subject to County approval and shall be tamper resistant as specified in §1.4.

2.5.1 **Power/Telephone Cord**

The modem shall be designed to connect to the EIS by means of a modular telephone connector with a standard wiring configuration. The connector shall be located on the back of the analyzer cabinet. Alternatives to this requirement to improve the durability of the connection interface and the telephone line are encouraged and may be proposed by the manufacturer for evaluation by the County. The telephone cord shall not be attached to the power cord. The telephone line shall be enclosed in a protective cable meeting County and UL approval. Alternative methods to protect the telephone line may be submitted to the County for approval.

The manufacturer shall include provisions to ensure that the power necessary to activate the modem at the appropriate time is available.

The analyzer shall be supplied with a 25-foot UL-approved power cord. The manufacturer shall design the cabinet so that convenient storage is provided for the excess cord not needed to reach the nearest power outlet.

2.5.2 Power Requirements

The EIS shall operate only on alternating current (AC). No direct current (DC) models will be acceptable. The EIS shall not be powered by a portable AC generating unit. An exception to this rule may be sought by the manufacturer if it can be shown, to the satisfaction of the County, that the analyzer is immune to the line frequency variations of the portable AC generating unit. Immunity to line frequency variations is defined here as line frequency variations which will not cause more than one percent of full scale (FS) disturbances on any of the analyzers. Additionally, any AC portable generating unit used with the EIS shall not have frequency excursions exceeding one hertz from 60 hertz.

Input power shall be 115 VAC, 60 hertz. All instruments shall meet the specified requirements over an input voltage variation of at least ± 12 volts. Maximum allowable performance change due to line voltage variations shall not exceed one-third of the accuracy requirements.

2.5.3 Instrument Construction

The instrument shall be designed and constructed to provide reliable and accurate service in the automotive repair environment. The analyzer shall be supplied with a cabinet which is equipped with a storage area large enough to secure all accessories and operating manuals.

a) Materials

The materials used in instrument construction shall be resistant to corrosive type substances found in the automotive repair environment and be designed to last for at least the period of the warranty.

b) Finish

The exterior and interior finish of the entire cabinet and console shall be sufficiently durable to withstand the chemicals and environmental conditions normally encountered in the automotive repair environment for the period of the warranty.

c) Mobility

The analyzer may be permanently mounted or mobile with wheels. Wheels shall be at least five inches in diameter and have a locking mechanism capable of preventing movement on a 15° incline.

If mobile, the analyzer shall be designed so that movement over rough surfaces (three-inch deep holes) and on 15° inclines, will not cause it to tip over. Analyzers shall not tip over when placed at the center of an inclined plane that makes an angle of 10 degrees with the horizontal and rotated 360° stopping in the position where it is most likely to tip over. In addition, the analyzer shall not become unstable or tip over when rolled straight off the edge of a two-inch high platform or when one wheel is rolled over a drain, two inches below the surface, inside an 18-inch diameter depression.

d) Identification

The analyzer serial number and the date of production shall be conveniently displayed to the quality assurance inspectors and the County field representatives, in a manner meeting the County's approval. The first two characters of the EIS number shall be alphas denoting the manufacturer's initials, and shall not be changeable from the keyboard even in the manufacturer's service mode. The initials chosen are subject to approval by the County to prevent duplication between manufacturers. The remaining six characters shall be numeric. The numbers shall be right justified. Zeroes shall be used to fill any blank spaces between the initials and the numerics. For example, the EIS number for analyzer #23 from Hobo Electronics would be "**HE000023**."

e) Electrical Design

Provisions shall be made for storing the power cord in a manner satisfactory to the County. Fuses or circuit breakers shall be used to protect individual electrical circuits and emission analyzers. Main circuit breakers and fuses shall be readily accessible from the exterior of the cabinet. Analyzer operation shall be unaffected by electrical line noise and voltage surges. The analyzer shall be sufficiently protected from voltage surges to prevent damage to the analyzer from the simultaneous start up of a 220-volt compressor, an arc welder, hydraulic controls and other equipment commonly found in the typical automotive test and/or repair environment.

f) Electromagnetic Isolation and Interference

Electromagnetic signals found in an automotive environment shall not cause malfunctions or changes in accuracy in the electronics of the EIS. The instrument design shall insure that readings do not vary as a result of electromagnetic radiation and induction devices normally found in the automotive garage environment (including high energy vehicle ignition systems, RF transmission radiation sources and building electrical systems).

In addition, the manufacturer shall ensure that the analyzer processor and memory components are sufficiently protected to prevent the loss of programs and test records.

g) Vibration and Shock Protection

System operation shall be unaffected by the vibration and shock encountered under the normal operating conditions encountered in an automotive environment. Instruments, motors, pumps, and disk drives shall be shock-mounted to absorb any vibration which might affect the system operation.

h) Instruction Manual & Accessories Storage

A storage area shall be provided to store the analyzer operating instruction manual and UTAH 2000 accessories.

2.6 BAR CODE SCANNER**

27

A non-contact two dimensional bar code scanner capable of reading both 1-D and 2-D bar codes, including code 39, PDF-417, UPC labels and 128 symbologies and all necessary interface software and hardware designed to read labels meeting SAE specifications J1877 and J1892 is required on all analyzers^{**}. The bar code scanner shall be able to auto-discriminate between the symbologies. The bar code scanner shall be capable of reading a VIN through a windshield. The bar code scanner shall be capable of reading a DMV bar code having a maximum length of 7¹/₄" (seven and one quarter inches).

In addition to collecting information from the VIN label, scanners may also be required to enter emission application information from the County recognized abbreviated lookup manuals.

The County recommends that the manufacturers contact the vehicle manufacturers to inquire about obtaining bar-coded labels for testing purposes.

2.6.1 Minimum Required Configuration for Bar Code Scanner

The analyzer shall be equipped with a standard port configuration and standard connector (such as DB9 or DB25 RS232C external connector) for the bar code scanner. Scanner and communication must be County approved (proprietary scanner systems will not be permitted). The bar code scanner will be used to load emission control system information from application manuals and from the permanent bar code labels placed on the vehicle by the manufacturer. The supplied bar code scanner shall come with at least a twenty (20) foot long self coiling cord and be able to read bar codes placed on the door frames and under the hoods. Manufacturers will be expected to include any software necessary to utilize the data gathered from labels.

[&]quot;The bar code scanner shall be standard, "off-the-shelf" technology approved by the County.

2.7 FUEL CAP TESTER

The UTAH 2000 EIS shall include a fuel cap testing system meeting the following specifications. The fuel cap tester must be provided as an integrated component located internally within a secured area of the EIS cabinet. All electronic components, flow sensors, controlling switches, and solenoids, with the exception of the hose and gas cap adapters, must be mounted within the secured area of the EIS cabinet. Fuel Cap Testers must be certified by the California Bureau of Automotive Repair or equivalent laboratories approved by the County.

- a) The fuel cap tester shall test the leak rate of fuel caps to prevent evaporative emissions.
- b) The tester shall be designed so that tethered caps can be accommodated without moving the EIS and shall be capable of pressurizing the fuel cap for this test. The pressurizing system shall apply a controlled pressure of 30 in. H_2O to the fuel cap. The system shall indicate a fail if the leak rate is greater than 60cc per minute. The system shall indicate a pass if the leak down rate is 60cc or less per minute. The leak test shall last no longer than 120 seconds.
- c) The tester shall have the capability to change the leak rate pass/fail setpoint if needed at a later date.
- d) The system shall be tamper resistant.
- e) Fuel cap test equipment shall indicate a pass/fail condition.
- f) The tester shall have an indicator and/or screen prompt informing the technician when the system is ready to test (pressurized).
- g) The tester shall have a means of controlling the maximum reservoir pressure and relieving overpressure.
 - 1. The fuel cap tester shall communicate with the UTAH 2000 EIS to record information such as pass/fail, calibration, etc. Communication and power to the unit shall be internal to the EIS cabinet.
- j) Calibration and Accuracy
 - 1. Each system will have a calibrated screened orifice for calibration PASS/FAIL determination. The set shall be individually calibrated; the calibration shall be traceable to the NIST. The master calibration set shall consist of a PASS MASTER flowing 52 to 56cc per minute and a FAIL MASTER flowing 64 to 68cc per minute (both measured at 30 in. H₂O pressure). The tester shall be calibrated every three days with the master calibration set. The calibration set shall be calibrated before initial usage.

The calibration method shall be NIST traceable. Equipment out of calibration may not be used.

- 2. The tester accuracy shall be $\pm 3cc$ per minute and shall be capable of maintaining its accuracy from 35° to $110^{\circ}F$ and at elevations from -60 to 7,000 feet.
- k) Adapters
 - 1. The system shall be capable of testing at least 95% of the County motor vehicle fleet (excluding pressurized fuel systems such as CNG, LPG, etc.) that are equipped with evaporative control systems.
 - 2. As soon as adapters are available for new designs they should be made available for sale to the EIS owners within a period of one year of their introduction.
 - 3. Adapter sets shall have a means of indicating or cross-referencing which vehicles they fit.

2.8 ENGINE RPM DETECTION

The analyzer shall utilize a tachometer capable of detecting engine RPM with a 0.5 second response time and an accuracy of $\pm 3\%$ of the true RPM. Prompts may be provided to assist the technician in locating an RPM signal on vehicles equipped with DIS. Based on the vehicle identification information entered by the technician, the analyzer shall advise the technician regarding which vehicles require a primary pick up, which require that an alternate counting algorithm be used and which require the use of an auxiliary piece of equipment. Analyzers shall be provided with all the software and hardware that is necessary to make them capable of reading engine speed on all vehicles manufactured prior to analyzer certification that are included in the Inspection Program (except those powered by diesel engines). As a minimum, analyzers must be equipped with a spark plug wire direct pickup, a non-contact pickup, and an OBD II interface connection. For analyzer certification, analyzers shall be capable of reading engine RPM on all spark ignition vehicles. Beginning with 1996 model year vehicles, the system shall be capable of detecting engine RPM via OBD II.

2.9 ACCESSING OBD II FAULT CODES

Analyzer manufacturers must incorporate provisions for reading fault codes from vehicles with on-board diagnostics II (OBD II) using the Standardized Link as defined by SAE J1979 and J1978. The SAE Standardized Link shall connect to the vehicle's on-board diagnostics port to automatically interrogate and retrieve fault codes. The OBD II hardware must adhere to the communications protocols and hardware requirements as defined by SAE J1979 and J1978 specifications. The OBD II hardware must be maintained in the secured area of the EIS cabinet with the exception of the cable to the vehicle. Hardware and software shall be seamless to the user.

2.10 TESTING HEAVY-DUTY GASOLINE-POWERED VEHICLES

Manufacturers shall supply the analyzer with the hardware and software necessary to test heavy-duty gasoline-powered vehicles manufactured prior to certification. At a minimum, accessories shall allow for 40-foot motorhomes to be tested without degrading the emission analyzer response time and provide the technician with an accurate indication of the engine speed.

2.11 DUAL EXHAUST

For vehicles with dual exhaust, the analyzer supplier shall provide a dual probe-and-hose arrangement, designed so that the flows from each tailpipe reach the main sample hose at the same time and shall have the same flow $\pm 10\%$. A quick-connect coupling may be used to connect an auxiliary probe and hose, but no quick-connect coupling shall be used in the primary single-exhaust path. The quick-connect fitting, if used, shall have a leak-proof shutoff when not in use.

- . -

ı,

SECTION 3. UTAH 2000 SOFTWARE SPECIFICATION

3.1 OVERVIEW

Section 3 specifies the software requirements for UTAH 2000 emission inspection systems (EIS). It includes inspection procedures, sequences, decisions, responses and prompts, as well as necessary information, security issues, lockouts, file structures, etc. It also contains requirements for communication with the County's Vehicle Information Database (HOST).

Important note: In general, when the wording of a prompt is not specified (or is paraphrased) it is up to the manufacturer to coordinate with the County to reach an agreement as to the specific wording to be used. This specification is designed to set the groundwork or foundation for the operation of the software for its functions, and at times is specific as to tasks, wording, processes and functions. It has been designed with flexibility in mind to allow for reaching the solution to each task using a method or methods deemed easier or more efficient by the manufacturer as long as the functionality is in place and the method(s) used do not undermine the intent of this specification.

3.2 UTAH 2000 SOFTWARE COMPONENTS

3.2.1 General

The program software used in the EIS shall consist of a process control system as well as data look-up files necessary to conduct official vehicle inspections that meet the UTAH 2000 specifications. The software shall consist of inspection test procedures and criteria; necessary station, technician, and vehicle information; security measures, utilities and ancillary software modules. The software's features shall include vehicular emission measurements of HC, CO, CO₂, and O₂, engine RPM measurements, exhaust dilution determinations, bar code scanning, interface with OBD II, communication with the HOST, safety testing, gas cap testing, etc.

The Manufacturer may offer additional features, which are not required by this specification, so long as these items do not defeat or interfere with the integrity or purpose of this specification.

3.2.2 Boot-up Configuration

On each power-up, the EIS shall automatically self-diagnose all computer systems, including memory, hard disk and loading of all necessary operating software without technician intervention. If any corruption is found on the hard disk during the boot sequence and if check files are created (usually saved as *.chk files), then the check files must be deleted so that the hard disk will not contain an excessive number of these files. Upon satisfactory computer component checkout (including hard disk and data structures), the application software shall present the Main Menu of the available EIS operations. All offered features shall be menu-driven. For inspection related features,

3-1

context-sensitive, on-line help shall be provided which can be accessed preferably with a single keystroke.

3.2.3 Software Modifications and Software Update Certification

Periodic software updates will be necessary. The County or the manufacturer may require software updates. In either case, the manufacturer is responsible for installing the software in all EIS units. The first four updates must be included as part of the EIS initial cost. The cost of subsequent software updates is the responsibility of the EIS owner if the software update is required by the County but is the responsibility of the manufacturer if they require the update to fix a problem that went undiscovered until after its release.

These revisions are likely to occur within the first two years of the enhanced vehicle inspection program and their costs shall be included in the quoted initial purchase price for the base system.

Updates to the software specifications will be provided to the manufacturer by the County. The manufacturer shall provide the software program to the County upon each update. The software version number is to be indicated on the EIS status screen, Main Menu, on each vehicle test record and the VIR. The version number shall consist of a four digit numeric code to be made up of the following format X.XX. Where the first version submitted will be 1.00 and each update will require the number to be incremented by a value of "0.01". For example, the 10th submittal will read 1.10.

All software updates shall cause the software version number to change. There will be a separate field in the test record indicating the software version currently in use

Areas in the software where changes or additions might be required include (but are not limited to):

- Preconditioning procedures and emission test sequences (as applicable for two-speed idle tests);
- Various lookup/reference tables;
- Functional tests;
- Diagnostic and repair procedures;
- Data communication procedures;
- Criteria affecting emission standards selection, vehicle exemptions and vehicle pass/fail criteria.

Other areas not specifically mentioned might also be impacted at some point, but we do not expect to request changes in all these areas at once.

To maintain the integrity of the County's I/M program, County field personnel will be instructed to lock out The UTAH 2000 units that have unauthorized modifications or are running unapproved software versions. The following criteria apply to software and software updates:

- a) Only the County-approved software shall be used in the UTAH 2000 units conducting official inspections. The County intends to accommodate software developed by third parties as long as system security and integrity are not compromised. In addition, the County may initiate the development of software updates by third parties for use in all UTAH 2000 units. If the County initiates development of a software update, manufacturers shall cooperate with the County and/or the County-approved third party.(This section does not prohibit manufacturers from charging reasonable fees for software updates are developed by third-parties.)
- b) All proposed software updates must be thoroughly tested by the manufacturer before being submitted to the County. Update disks as well as electronically transmitted updates shall be encrypted in a manner approved by the County. The EIS shall be capable of accepting software updates via CD.
- c) All proposed software updates generated by the manufacturer shall be submitted to the County with a written description list of the reason(s) for the update, such as the problem(s) that the update corrects.
- d) All proposed software updates, including manufacturer-generated updates, must be submitted to the County for testing and approval as follows:
 - 1. Software updates must be submitted on a mutually agreed upon medium.
 - 2. Each new software version submitted to the County, including minor revisions, must have a new and unique software version number.
 - 3. All proposed software updates must be accompanied by a cover letter with the following information:
 - i. A description of all of the changes contained in the proposed software update, including manufacturer-initiated modifications.
 - ii. A timeline of when the update is expected to be installed (start to finish) and how many units will be updated.
 - iii. If any hardware modifications or special procedures are needed to perform the software update, describe the procedures for performing the update.
 - 4. Depending on the type and number of changes contained in the proposed software update, the County may require testing at the County-approved beta sites prior to release of the software. The County will perform a preliminary review of the proposed software prior to releasing it for either full use or beta site testing.
- e) The manufacturer is allowed six months, from the date the County issues its finalized specifications, for periodic software updates to submit for approval and testing by the County. Prior to the six-month period, the manufacturer shall be permitted to review and to comment upon the proposed specifications. If,

. . .

3-3

however, a previously installed update does not meet the specifications, the County may require a shorter time period. Failure of a manufacturer to furnish or install software updates as specified, is cause for the County to decertify the manufacturer's EIS Certification, or to issue a citation and civil penalty up to \$1,000 per week that the manufacturer fails to furnish or install the software updates. (The County may allow additional time to review and comment and/or submit software updates if the modifications are more complex.)

3.2.4 Running Changes and Other Software Modifications

Any changes to design characteristics, component specifications, and any modifications to the software must be approved by the County. It will be the manufacturer's responsibility to confirm that such changes have no detrimental effect on the performance of the UTAH 2000.

3:2.5 Virus Detection Software

Each EIS unit shall contain a virus detection program, subject to the County approval, which shall verify the integrity (i.e. check for infection/corruption) of each update disk or decompressed file before it is applied to the EIS or allowed in memory. Infected/corrupted software shall be blocked from installation.

In lieu of this requirement, the EIS manufacturer may submit for the County's consideration written procedures clearly illustrating how the EIS manufacturer intends to meet the intent of the VIRUS PROTECTION PROGRAM requirement. These procedures shall demonstrate how the integrity of the EIS software and update software or decompressed file shall be protected under all circumstances.

3.2.6 Directory, File Structure and File Storage

When data is being stored or accessed, the computer shall display a message indicating that the disk is in operation and the EIS shall not be moved or disturbed. All information relating to vehicle emissions and OBD II inspections, calibrations, repairs, and safety inspections shall be stored onto the hard drive. Any information that must be sent to the HOST must also be stored onto the 3.5 inch floppy drive. The frequency of storage during testing shall be, at the very least, once per test immediately at the end of the inspection but before the printing of the inspection report. More frequent saves to the drives are permitted and encouraged. The EIS shall keep in storage on the hard drive, at the very least, the last one thousand (1,000) vehicle inspections, the last one thousand (1,000) safety inspections, the last one thousand (1,000) repair records and the last one hundred (100) calibration records.

All station related information and all information that remains unsent to the HOST shall be stored onto the 3.5 inch floppy drive (usually the A:Drive) as well as onto the hard drive. This shall allow the analyzer to recover any information that has not been sent to the HOST and station information in the event of a hard drive failure.

3.4

3.2.7 Display

a) Readability

The display, when in the test mode, shall be readable at a minimum distance of eight feet in a building that meets OSHA lighting standards for a garage environment. Display contrast and brightness shall be adjustable.

b) The County Messages (Technical Service Bulletins (TSB))

The County messages shall be transmitted by the HOST to the EIS during all communication sessions. All new messages shall be displayed immediately after the HOST contact. The EIS shall display these messages one-time-only and shall provide an option to print and an option to save. All displayed messages shall default to print and the technician must press a function key to continue. All messages from the HOST shall be deleted after the first viewing, unless specified otherwise by the technician.

c) Testing Messages

During the emissions test, the EIS shall display the word TESTING on the screen. The EIS shall also display messages such as test mode, vehicle engine speed, excessive exhaust dilution, low flow and engine RPM violations.

d) Information Not Permitted During Testing

The EIS shall not display the emission readings during the inspection. (However, during manual mode, the readings shall be displayed.)

e) Print Screen Capability

The EIS shall have a PRINT feature, which prints any current text or graph displayed on the screen, by depressing no more than three keys. The print feature shall <u>always</u> be active; however, there shall be no print capability during emissions testing.

3.2.8 Technician and Station License Numbers and Other Numbers

a) General:

The technician's license number and access code shall reside in both the EIS and the HOST. The EIS shall determine the validity of the technician's access code with the Technician Information Records.

The EIS shall have the capacity to store at least 99 technician access codes and 99 corresponding Technician License numbers. Only through the County Menu and the HOST can the technician's access code and corresponding license number be added, changed or deleted. Station and technician license numbers begin with three (3) alpha characters, followed by six numeric characters.

b) Technician Access Codes:

The EIS shall require the technician to enter a special access code before an inspection can begin. The access code shall neither be displayed nor printed on

· - · · - - -

the VIR. This special access code number shall be linked to the technician's license number.

c) Technician License Numbers:

A technician's license number reflects the type of license the technician possesses. The EIS shall automatically abort the inspection and display a message indicating that the technician has not obtained the proper license number and/or endorsement from the County.

d) Station License Number:

The station license number shall be entered into the EIS during initialization. Only valid station license numbers may be entered into the EIS. The station license numbers are unique to each station. The station license number shall be placed in the *Station Number I/M* field of the test record and on the VIR. This is field #2 and this field must be populated in the test record for every valid test record sent to the HOST.

e) Test Record Number

The EIS shall give each valid test a consecutive number. A valid test consists of a completed test with an overall pass or fail test result that shall be transmitted to the HOST. The record number shall be written to *Test Number* field, field #1 of the test record. This field is numeric and has a length of 6 digits. When the number reaches 999999, the number shall be reset to 000001. This field must be populated in the test record for every valid test record sent to the HOST.

f) EIS Number

The EIS number shall be unique for the EIS unit in The County. The first two characters of the EIS number are alpha. These two characters shall be assigned to the manufacturer upon certification of that manufacturer's EIS unit. The following six (6) digits shall be unique to each EIS made by the manufacturer. The EIS number shall be written to the *EIS ID* field, field #4, of the test record. This field must be populated in the test record for every valid emission test record sent to the HOST. Print the EIS number on the VIR.

g) Loaded Software Version Number

This field shall contain the version number of the software that is currently being used by the EIS. The loaded software version number shall be written to the *Software Version* field, field #31, of the test record and printed on the VIR. This field must be populated in the test record for every valid test record sent to the HOST.

h) HOST Identification

The HOST-ID is a record identifier generated by the HOST. The HOST shall assign an ID number to a test record which shall be transmitted to the EIS. The ID will be written to the HOST-ID field of the test record. The HOST-ID shall not be

3-6

modified by the EIS and shall be transmitted back to the HOST during transmission of data.

3.2.9 EIS Lockout Reasons

The EIS shall be prohibited from performing an inspection for any of the following reasons:

Warm-up in progress (see Note 1) Warm-up failure (see Note 1) Gas calibration required Gas calibration failure Gas analyzer failure Fuel cap tester failure Fuel cap tester out of calibration Oxygen sensor out of calibration Leak check required Leak check failure EIS tampering Out of certificates Hard disk is full Floppy disk or disk mechanism failure Hard disk or disk mechanism failure County EIS lockout No communication with HOST in XXX days and XXX tests (see Note 2) Failure to pay for test authorization numbers purchased Failure to pay monthly payment County disk drive tampering Technician License Expired Station License Expired

Notes:

- 1. An analyzer warm-up requirement or failure shall not restrict the UTAH 2000 units from allowing OBD II testing.
- 2. Example: A lockout shall be set whenever fifty (050) inspections (running total) have been performed by the EIS within five (005) consecutive days without communicating to the HOST. The HOST sets the no contact limit and number of inspections allowed. The lockout can be cleared by County personnel or by the HOST in accordance with pre-established procedures.

3.3 SOFTWARE MODULES

3.3.1 Utah Main Menu

Upon boot-up of an EIS unit and after the completion of the boot-up procedures, defined in Section 3.2.1b Boot-up Configuration, the software shall immediately query the analyzer for a warm-up requirement and check the system for any breaches in security that may have occurred while the power was shut off. The software shall display the Main Menu and the results of the warm-up and security checks. Somewhere on the Main Menu the software shall display the software version currently in operation and a clock with the current system date and time. The date and time shall be in the format of MM/DD/YYYY HH:MM AM/PM. In addition, the Main Menu shall display the County logo in which the EIS unit is operating under. The logos will be provided in electronic format at a later date.

a) Warm-up:

If the analyzer requires a warm-up, or a warm-up flag is set on the analyzer, the software shall go into a warm-up period whereby analyzer calibrations and official TSI inspections are not permitted. During the warm-up period, the Main Menu shall present a countdown timer displaying the time left in the warm-up period.

b) Security:

While in the Main Menu, the software shall periodically, no less than once every second, check for any security breaches. Once a security breach has occurred the software shall display a message on the screen notifying the operator of the violation and shall restrict any official vehicle inspections (safety, TSI and OBD II) and system calibrations. However, other functions shall be allowed after a security breach.

c) Menu Items:

The Main Menu shall display the title "Utah-2000 Main Menu" in larger font relative to that on the buttons and shall be located at the top of the Main Menu screen and centered horizontally on two lines. In addition, the logo of the County in which the individual EIS unit is operating in shall be displayed at the top right corner of the Main Menu screen.

Following are a list of options, along with a brief description, that shall be made available from this menu:

Inspection Menu... Gas Analyzer Menu... Shutdown Analyzer Communications Standby

1. <u>Inspection Menu</u>: With the selection of this feature the software shall display the Inspection Menu.

3-8

2. <u>Gas Analyzer Menu</u>: With the selection of this feature the software shall display the Gas Analyzer Menu.

3. <u>Shutdown Analyzer:</u> With the selection of this feature, the software allows the technician to select the Shutdown Analyzer procedure

4. <u>Communications Standby:</u> Selecting this feature allows the technician to place the analyzer in Communications Standby mode, which is required for data transfer.

3.3.2 Inspection Menu

Upon selection of this feature the software shall display the Vehicle Emissions Inspection menu. Somewhere on the Utah 2000 Inspection Menu the software shall display the software version currently in operation and a clock with the current system date and time. The date and time shall be in the format of MM/DD/YYYY HH:MM AM/PM. In addition, the Inspection Menu shall display the County logo in which the EIS unit is operating under. The logos will be provided in electronic format at a later date.

a) Warm-up:

If the analyzer requires a warm-up, or a warm-up flag is set on the analyzer, the software shall go into a warm-up period whereby analyzer calibrations and official TSI inspections are not permitted. During the warm-up period, the Vehicle Emissions Inspection menu shall present a countdown timer displaying the time left in the warm-up period.

b) Menu Items:

The Utah 2000 Inspection Menu shall display the title "Utah 2000 Inspection" in larger font relative to that on the buttons and shall be located at the top of the Utah 2000 Inspection Menu screen and centered horizontally. In addition, the logo of the County in which the individual EIS unit is operating in shall be displayed at the top right corner of the Utah 2000 Inspection Menu screen. Following are a list of options, along with a brief description, that shall be made available from this menu:

Emission & Safety Inspection Emission-Only Inspection Safety-Only Inspection Previous Vehicle Information Training Emission & Safety Inspection Analyzer Information Menu... Diagnostic Tools Menu... Communications Diagnostics... Station Manager Menu... Q/A County Menu...

1. <u>Emission & Safety Inspection</u>: Selecting this menu item allows the technician to perform official TSI Emission, OBDII, Gas Cap and Safety Inspections. (See Vehicle Inspection Procedures for details on inspection processes.)

. . /

- 2. <u>Emission_Only Inspection:</u> Selecting this menu item allows the technician to perform official TSI Emission, OBD II and Gas Cap Inspections.
- 3. <u>Safety-Only Inspection</u>: Selecting this menu item allows the technician to perform official Safety Inspections.
- 4. <u>Previous Vehicle Information:</u> Selecting this menu item allows the technician to query Previous Vehicle Information.
- 5. <u>Training Emission & Safety Inspection:</u> Selecting this menu item allows the technician to access training mode and perform unofficial TSI Emission, OBD II, Gas Cap and Safety Inspections. This feature shall allow technicians to conduct vehicle inspections as described in Section "Vehicle Inspection Procedures" with the following exceptions: (1) Inspections under Training Mode will not be stored and test results will not be transferred to the HOST; (2) The technician shall not be prompted for a password; (3) VIR's shall not display the certificate shading; (4) VIR's shall prominently display the words "NOT AN OFFICIAL INSPECTION" across the VIR; and (5) Certificate numbers shall not be issued.
- 6. <u>Analyzer Information Menu:</u> Selecting this menu item allows the technician to access the Analyzer Information Menu.
- 7. <u>Diagnostic Tools Menu</u>: Selecting this menu item allows the technician to access the Diagnostic Tools Menu.
- 8. <u>Communications Diagnostics:</u> Selecting this menu item allows the technician to access the Communications Diagnostics Menu.
- 9. <u>Station Manager Menu</u>: Selecting this menu item allows the technician to access the Station Manager Menu.
- 10. <u>QA/County Menu:</u> Selecting this menu item allows the technician to access the QA/County Menu.

3.3.3 Gas Analyzer Menu

, ^

Upon selection of this feature the software shall display the Gas Analyzer Menu. Somewhere on the Gas Analyzer Menu the software shall display the software version currently in operation and a clock with the current system date and time. The date and time shall be in the format of MM/DD/YYYY HH:MM AM/PM.

3-10

a) Warm-up:

If the analyzer requires a warm-up, or a warm-up flag is set on the analyzer, the software shall go into a warm-up period whereby analyzer calibrations and official TSI inspections are not permitted. During the warm-up period, the Gas Analyzer Menu shall present a countdown timer displaying the time left in the warm-up period. The manufacturer recommended warm-up period prior to calibration must be completed. This warm-up period must be submitted to the County for approval.

b) Menu Items:

1.

The Gas Analyzer Menu shall display the title "UTAH-2000 Gas Analyzer Menu" in larger font relative to that on the buttons and shall be located at the top of the Gas Analyzer Menu screen and centered horizontally on two lines. Following are a list of options, along with a brief description, that shall be made available from this menu:

Complete 3-Day Calibration Manual Gas Readings Mode Analyzer Gas Calibration Analyzer Leak Check Gas Cap Tester Calibration Floppy Disk/Drive Check

- 1. <u>Complete 3-Day Calibration:</u> Selection of this menu item shall allow technicians to conduct a 3-Day system calibration. Analyzer gas calibration and leak check procedures are described in Section 2.4.5 Analyzer Requirements.
- 2. <u>Manual Gas Readings Mode:</u> Selection of this menu item shall allow the technician to obtain Manual Gas Readings.
- 3. <u>Analyzer Gas Calibration</u>: Selection of this menu item shall allow the technician to perform an analyzer gas calibration as described in Section 2.4.5 Analyzer Requirements.

4. <u>Analyzer Leak Check:</u> Selection of this menu item shall allow the technician to test the system for leaks as described in Section 2.4.5 Analyzer Requirements.

- 5. <u>Gas Cap Tester Calibration:</u> Selection of this menu item shall allow the technician to conduct a calibration of the Gas Cap System. The calibration procedures shall adhere to Section 2.8 Fuel Cap Tester. The screen messaging and prompts shall be sufficient enough to lead the user easily through the steps of the calibration procedure.
- 6. <u>Floppy Disk/Drive Check:</u> Selection of this menu item shall initiate an integrity check of the 3.5 inch floppy drive. At a minimum, this feature shall check for bad sectors, lost clusters, and corrupt files. If the system can not recover the lost clusters and corrupt files or if the disk contains bad sectors, a calibration lockout shall be set and a message displayed to the technician that the floppy disk is damaged and to call for service.

3.3.4 Station Manager Menu

Upon selection of this menu item the software shall initiate the display of the Station Manager Menu. The Station Manager Menu shall be password protected with a 5 digit password that can be modified by the appropriate station staff once in this menu. Selecting this feature shall prompt the user to enter the 5 digit station password. A correct entry shall allow the display of the Station Manager Menu. An incorrect entry shall

3-11
inform the user that the password was incorrect and prompt the user to retry. Three incorrect password entries shall return the user to the Inspection Menu.

a) Menu Items:

The Station Manager Menu shall display the title "UTAH-2000 Station Manager Menu" in larger font relative to that on the buttons and shall be located at the top of the Station Manager Menu screen and centered horizontally on two lines. Following are a list of options, along with a brief description, that shall be made available from this menu:

Review Inventory Load Certificates Menu... Void Certificates Menu... Inventory Warning Settings Station Identification Update Safety Inspector Information Monthly Safety Inspection Report

1. <u>Review Inventory:</u> Selection of this menu item shall display Certificate Inventory.

- 2. Load Certificates Menu: Selection of this menu item displays the Load Certificates Menu and allows the Station Manager to load emission and safety inspection certificate numbers into inventory. Certificates will have the ability to be entered (loaded) manually, or may be entered by using a Bar Code Scanner. The "Bar Code Scan Certificate" feature shall allow entry of new emissions and safety certificate numbers using the bar code scanner in the event the HOST is not operational. The manufacturer shall develop a method whereby certificates can be embedded into a bar code and then scanned into the analyzer. This method must be secured and restrict unauthorized updates into the analyzer.
- 3. <u>Void Certificates Menu:</u> Selection of this menu item allows the Station Manager to void emission and safety inspection certificate numbers.
- 4. <u>Inventory Warning Settings:</u> Selection of this menu item allows the Station Manager to specify when the analyzer should display a warning of low certificate inventory (i.e. Five remaining certificates or 25 remaining certificates).
- 5. <u>Station Indentification:</u> Selection of this menu item displays station information and allows it to be changed. The only information that shall be allowed to be modified is limited to business name, street number and name, and regular business phone number. The HOST contact number, state, zip code, and station license number shall <u>not</u> be changed through this menu.

3-12

11

- 6. <u>Update Safety Inspector Information</u>: Selection of this menu item allows the Station Manager to change the access password to the Station Manager Menu.
- 7. <u>Monthly Safety Inspection Report</u>: Selection of this menu item shall allow the technician to print monthly safety inspection reports for submittal to the Highway Patrol. Upon entry to this feature, the software shall provide appropriate prompts to allow the user to select and print a monthly safety report for any month in which data exists on the UTAH 2000 unit. The monthly report shall be in the standard format adopted by the State's Highway Patrol, a sample copy of a monthly safety report is provided in Appendix I.

3.3.5 Diagnostic Tools Menu

Upon selection of this feature the software shall display the Diagnostic Tools Menu. This feature is designed to allow the technicians to test, diagnose and repair vehicles.

a) Menu Items:

The Diagnostic Tools Menu shall display the title "UTAH-2000 Diagnostic Tools Menu" in larger font relative to that on the buttons and shall be located at the top of the Diagnostic Tools Menu screen and centered horizontally on two lines. Following are a list of options, along with a brief description, that shall be made available from this menu:

Gas Cap Pressure Test Digital Multimeter Read OBD Fault Codes

- 1. <u>Gas Cap Pressure Test:</u> Selection of this menu item allow the technician to perform a diagnostic Gas Cap Pressure test.
- 2. <u>Digital Multimeter:</u> Selection of this menu item allows the technician to use the Digital Multimeter for diagnostic purposes.
- 3. <u>Read OBD II Fault Codes:</u> Selection of this menu item shall allow the technician to access the vehicle's readiness status code, any fault code, MIL request Status code, "freeze frame" and engine RPM via the OBD II port. This feature shall give the technician the ability to diagnose, repair and reset any OBD II related failure.

3.3.6 Q/A County Menu

Upon selection of this feature the software shall display the Q/A County Menu. This menu shall be password protected as previously described. NOTE: While in the Q/A County Menu, the software shall monitor the system. When the software detects inactivity for more than 10 minutes the software shall terminate its idle state and shall

3-13

exit the Q/A County Menu. This security feature shall prevent the County Representative from leaving the facility and mistakenly leaving the unit in the Q/A County Menu.

a) Menu Items:

The Q/A County Menu shall display the title "UTAH-2000 Q/A County Menu" in larger font relative to that on the buttons and shall be located at the top of the Q/A County Menu screen and centered horizontally on two lines. Following are a list of options, along with a brief description, that shall be made available from this menu:

Leak Check Gas Audit Update Station Information Update Inspector Information Install New Data Disk Reset Date & Time Hands-On Test Lockout Analyzer Perform Emergency Software Update Search and Retrieve Test Record Station Evaluation Report

- 1. <u>Leak Check:</u> Selection of this menu item shall allow the County Representative to perform Leak Check procedures for the analyzer hardware.
- 2. Gas Audit: Selection of this menu item shall allow the County 2. Representative to conduct a gas audit following procedures described in Section 2 of this specification. This screen shall allow the County Representative to zero the bench upon request, shall display HC, CO, CO2 and O2 emissions readings (uncompensated for the dilution correction factor), display a clock minute and second timer, allow the timer to be reset, display the status of the analyzer (analyzer error flags, zero-needed flag, out-of-calibration flag, warm-up ...) and allow the emissions readings to be printed upon request. This feature shall be "user friendly" by displaying the appropriate instructions for proper usage of the unit in this mode. Within the gas audit screen there shall be a function to enter (with capabilities of using the bar code scanner and key board) the gas audit bottle values. In addition, the gas audit screen shall have a function, manually selectable by the auditor, to capture and record the gas readings. Based on this capture, the analyzer must automatically make a pass/fail determination of the gas audit using the gas audit limits in Section 2 of this specification. Within this audit screen there shall also be a listing of the following items where the auditor may manually make a pass/fail or yes/no, etc, determination for each:

Station Sign Station Permit Technician Permit Emission Fee Chart Emissions Manual Certificates Public Relations Pamphlets Tune-Up Tools Propane Kit Technical Bulletins Dynamometer Audit Inspection Reason Inspection Result Flex Probes Aux. Flex Probes Current Repair Manuals Span Gas Change Access Code New Data Disk Reset Tamper Software Update Station Lockout Technician Lockout Remove Lockout Letter Delivered Follow-Up Action

There shall be adequate room for comments by the County Representative. Upon completion of the gas audit and station performance, the analyzer shall print out a copy of the results. On the print out, there shall be provisions for the signatures of the Station Manager and County auditor.

- 3. <u>Update Station Information</u>: Selection of this menu item shall allow the County Representative to modify the HOST phone number, the station license number, zip code and the UTAH 2000 serial number.
- 4. <u>Update Inspector Information</u>: Selection of this menu item shall allow the Inspector information assigned to that particular unit to be edited, deleted and added by the County Representative.
- 5. <u>Install New Data Disk:</u> Selection of this menu item shall give the County Representative the ability to replace the existing 3.5 inch diskette with a new one. Selection of this feature shall automatically open/unlock the secured area of the cabinet where the floppy drive is housed, prompt to open the secured door, replace the diskette and to close the secured area. The software shall not allow the County Representative to exit this function until the secured area has been closed.
- 6. <u>Reset Date & Time:</u> Selection of this menu item shall allow the County Representative to reset the analyzer's current date and time, if needed.
- 7. <u>Hands-On Test:</u> Selection of this menu item shall allow the County Representative to perform a manual TSI, OBD II, Gas Cap or Safety Inspection, without issuing or in any way using any certificate numbers.
- 8. <u>Lockout Analyzer:</u> Selection of this menu item shall allow the County Representative to set and reset the County and Tamper lockouts. Other lockouts shall not be able to be set from this menu.
- 9. <u>Perform Emergency Software Update:</u> Selection of this menu item shall allow the County Representative to perform an emergency software update.

10. <u>Search and Retrieve Test Record</u>: Selection of this menu item shall allow the County Representative to re-print previous emissions TSI, OBD II, Gas Cap, Repair and Safety Inspections.

3-15

X, Part D, App 1.79

11. <u>Station Evaluation Report</u>: Selection of this menu item shall display, with the ability to print, a report summarizing the system's use from the last gas audit to the current date. A summary of the following items shall be listed;

Total Initial Inspections Initial OBD II "Not Ready" Initial Pass OBD II Initial Pass Emissions Initial Pass Safety Initial Fail OBD II Initial Fail Emissions Initial Fail Safety Percent Fail OBD II Percent Fail Emissions Percent Fail Safety Initial Aborts **Total Re-Inspections** OBD II Re-Inspections **Emissions Re-Inspections** Safety Re-Inspections Failed OBD II Re-Inspections Failed Emissions Re-Inspections Failed Safety Re-Inspections Percent of Total Re-Inspections Pre-tuned **Re-Inspection** Aborts

For each of the listed headers the results shall be summarized in the following four (4) categories; 1968 to 1980, 1981 to 1995, 1996 & newer, and Total.

3.4 VEHICLE INSPECTION PROCEDURES

3.4.1 General Inspection Procedure

a) Upon selection of the item "Vehicle Emission Inspection" the software shall initiate the vehicle inspection process. First of all, the software shall check for any lockouts and security violations and the analyzer warm-up status and conduct system checks prior to proceeding with the inspection. If any lockouts are set or uncleared security violations remain or the bench requires a warm-up or any other condition exists that may negatively affect the integrity of the inspection the software shall prohibit the inspection from proceeding until the condition is fixed and shall notify the technician which condition(s) is the cause for termination of the inspection. If the corrective measure can not be carried-out by the technician then the software shall notify the technician to "call for service". Otherwise, if the corrective action can be performed by the technician (i.e., calibration....) the software shall allow an inspection once the corrective action(s) have been taken.

3-16

- b) If there exists no reason for the inspection to be terminated, the software shall prompt the technician to enter his/her unique access code. Upon entry of the access code, the software shall verify whether the access code exists in the Technician Information table. The technician shall be given three chances to enter a valid access code. After an incorrect third entry, the software shall terminate the inspection process. After each incorrect entry the software shall inform the technician of the incorrect entry and to retry. A correct entry shall allow the inspection process to proceed.
- c) If allowed to proceed, the software shall begin the HC Hang-up process and prompt the technician to enter the vehicle identification number (VIN) either manually or scanned using the bar code scanner. If scanned using the bar code scanner, only a single entry is required. The software shall allow three attempts to enter the VIN correctly, afterwards it should terminate the inspection process. The software shall record in the EIS test record which method was used to enter the VIN.
- d) If allowed to proceed, the software shall prompt the technician to enter the vehicle license plate number either manually or scanned using the bar code scanner. If scanned using the bar code scanner, only a single entry is required. The software shall allow three attempts to enter the license plate correctly, afterwards it should terminate the inspection process. The software shall record in the EIS test record which method was used to enter the license plate.
 - e) 1. The software shall conduct an internal search on the hard drive for any previous inspection containing the same VIN and/or license plate number. The software shall display on the screen a summary containing information of any matches found and allow the technician to select any one or to reject them all and proceed with a manual data entry. If the technician selects a matching record the software shall display the pertinent vehicle owner and vehicle specific information for review, possible modification, and acceptance by the technician as outlined in the following sub-section. If the technician to manually enter the vehicle owner and vehicle specific information as outlined in the following sub-section.
 - 2. If a match is found, the software shall display the pertinent vehicle owner and vehicle specific information for review, possible modification, and acceptance by the technician as outlined in the following sub-section.
- f) If allowed to proceed, the software shall display any matching inspection records found. Matching records available for review may be modified and accepted by the technician. The software shall require manual entry if a matching record was not found on the hard drive or if the technician rejected all matching records. The following information shall be displayed on two separate screens:

~ 10

X, Part D, App 1.81

- i. Vehicle Owner Information Screen. First Name, Last Name, Address, County, City, Zip Code.
 - Vehicle Specific Information Screen. Year, Make, Model, Type, GVWR, Body Type, Cylinders, Transmission, Engine Displacement (liters),

Odometer, Fuel Type.

- If allowed to proceed, the software shall make the determination as to whether to conduct a TSI emissions or an OBD II inspection. All 1996 and newer gasoline powered vehicles with a GVWR less than 8,500 pounds, which are flagged in the VRT to undergo OBD II testing shall be subject to an OBD II inspection. Other vehicles shall undergo a TSI inspection.
 - 1. If an OBD II inspection is required, the software shall follow the OBD II inspection procedures outlined in the section OBD II Inspection Procedures. Afterwards the software shall continue with the inspection process outlined in the next sub-section. An HC Hang-up failure shall not restrict OBD II inspections, however, the analyzer shall inform the technician of the problem.
 - 2. If a TSI inspection is required, the software shall not allow the inspection to continue if the system failed the HC Hang-up process and shall notify the technician of the failure and whether or not he/she would like to wait for the HC Hang-up process to repeat.

If the TSI inspection is allowed to continue, the software shall follow the inspection procedures outlined in the section Visual Inspection Procedures, followed by those in the section Gas Cap Pressure Test Inspection Procedures and the Two-Speed Idle (TSI) Inspection Procedures. Afterwards the software shall continue with the inspection process outlined in the next sub-section.

3-18

X, Part D, App 1.82

- - - -

g)

ii.

- h) If allowed to proceed, the software shall prompt the technician if he/she would like to conduct a safety inspection.
 - 1. If the technician chooses to conduct a safety inspection, the software shall follow the procedures outlined in the section Safety Inspection Procedures. Afterwards the software shall continue with the inspection process outlined in the next sub-section.
 - 2. If the technician chooses not to conduct a safety inspection, the software shall continue with the inspection process outlined in the next sub-section.
- i) If allowed to proceed, the software shall evaluate the test results for the emissions test and Safety inspection, if applicable, based on the procedures in this specification and display a summary screen indicating the results of the vehicle inspection. The software shall require the technician to press a button to continue with the inspection process and print the VIR.
- j) The software shall print the appropriate VIR as outlined in the section Vehicle Inspection Report (VIR).

3.4.2 Two-Speed Idle (TSI) Inspection Procedures

Two-speed idle testing in the enhanced program areas shall be used when OBD II testing is not required. A passing TSI test shall be if the final HC and CO emissions readings, as specified in the appropriate test sequence, do not exceed the respective emissions limits.

All emissions readings for HC and CO shall be corrected for dilution using the Dilution Correction Factor (DCF) prior to evaluation of emissions test results. The CO2 and O2 emissions readings shall <u>not</u> be compensated for the DCF. The following testing/sampling sequences shall be available in the software at the time of certification/approval:

a) <u>SEQUENCE #1</u>:

Testing period:	30 seconds for each stage
First stage:	2500 RPM (± 10 %)
Second stage:	Idle RPM
Basis for test results:	Average of last 5 seconds of each sampling period.
Units of test results:	Concentration measurements: PPM HC, $\%$ CO, $\%$ O ₂ and $\%$ CO ₂ .

Test Sequence # 1 shall be used to test all vehicles except those mentioned under the test sequences below.

b) <u>SEQUENCE #2</u>:

Testing period: 30 seconds for each stage

Note: Prior to initiating the test, the technician shall be informed that the vehicle they will be testing will require special test procedures and that it is important to follow directions carefully. The technician shall then be prompted to turn the key off for 10 seconds. At the end of 10 seconds, the analyzer shall prompt the technician to restart the engine and begin the 2500 RPM test. The software shall ensure that there is no RPM signal for 10 seconds prior to starting the 2500 RPM test.

First stage: 2500 RPM (±10 %)

Note: Between the test stages, the technician shall be prompted to turn the ignition off for 10 seconds. The analyzer shall ensure that there is no engine RPM signal for at least 10 seconds. At the end of 10 seconds, the software shall prompt the technician to restart the engine and begin the idle test.

Second stage:	Idle RPM (see standards for maximum)
Basis for test results:	After the first 15 seconds of each stage, any passing reading
	(averaged over 5 consecutive seconds) collected during
	each sampling period or if none, over the last 5 seconds.
Units of test results:	Concentration measurements: PPM HC, % CO, % O_2 and % CO_2

Test sequence #2 could take as little as 20 seconds if test conditions are satisfied and the vehicle meets the standards. If the emissions are not within the standards for any 5-second period (following the initial 15-second period), the test shall run the full 30 seconds.

All 1981-84 Ford passenger cars with 5.8L (351 CID) engines shall be tested using Sequence # 2.

c) <u>SEQUENCE #3</u>:

Testing period: 30 seconds for each stage

Note: Before the 2500 RPM test starts, the software shall display a message to the technician indicating that the engine RPM cannot exceed 2650 for this vehicle.

First stage:	2500 RPM (+ 6 % , - 10 %)
Second stage:	Idle RPM (see standards for maximum)
Basis for test results:	Average of the last 5 seconds of each sampling period.
Units of test results:	Concentration measurements: PPM HC, % CO, % O2 and
	% CO ₂

All 1984 Jeeps with a 2.5L (150 CID) light-duty trucks shall be tested using test Sequence # 3.

d) <u>SEQUENCE #4</u>:

Testing period:30 seconds for each stageFirst stage:2500 RPM (±10 %)

Note: A message shall be displayed to the technician indicating that the vehicle being tested will require special test procedures and that it is important that they follow directions carefully. The EIS shall display the following prompt only once, prior to the start of the emissions sampling:

DISPLAY PROMPT:

IS THE VEHICLE FUEL INJECTED? (YES/NO)

Programming Criteria:

- 1. If Yes, perform test sequence # 4.
- 2. If No, follow inspection sequence # 1.

The technician shall be prompted to set the parking brake, press the brake pedal and run the IDLE test with the transmission in DRIVE. When the idle test is complete, the technician shall be prompted to return the transmission to PARK.

Second stage:Idle RPM (see standards for max.)Basis for test results:Average of last 5 seconds of each sampling period.Units of test results:Concentration measurements: PPM HC, % CO, % O2 and
% CO2

All 1984 Chrysler/Dodge/Plymouth passenger cars with a 2.2L, fuel-injected engines with automatic transmissions shall be tested using Sequence # 4.

e) <u>SEQUENCE #5</u>:

Given the problems with the ZF automatic transmission, the County prefers that the affected vehicles be tested at their dealerships. Accordingly, if the technician enters an A (for automatic) for the transmission type, and if the vehicle make, model and model year match BMW/Peugeot/Volvo criteria, the EIS shall display the following message:

BECAUSE OF THE POSSIBILITY OF TRANSMISSION DAMAGE TO THIS VEHICLE, THE COUNTY PREFERS THAT IT BE INSPECTED AT

ITS DEALERSHIP. IF YOU STILL WISH TO PERFORM THE INSPECTION, YOU MAY DO SO AT YOUR OWN RISK OR YOU MAY ABORT THE TEST.

Note: If the technician chooses to continue testing this vehicle, display the following message before beginning the test sequence.

BEFORE BEGINNING THE EMISSIONS TEST, MAKE SURE THE ENGINE IS AT NORMAL OPERATING TEMPERATURE. IF NOT, THE VEHICLE SHOULD BE DRIVEN UNTIL IT IS. DO <u>NOT</u> WARM THE ENGINE BY RAISING THE RPM ABOVE IDLE WHILE THE TRANSMISSION IS IN PARK OR NEUTRAL.

Perform idle test only (delete first stage).

Testing period:	30 seconds for idle stage
Engine Speed:	Idle RPM [Note: One stage only.]
Basis for test results:	Average of the last 5 seconds of the sampling period.
Units of test results:	Concentration measurements: PPM HC, % CO, % O ₂ and
	% CO ₂

Note: All 1984-87 BMWs with automatic transmission, 1985-88 Volvo 740s with automatic transmission, and 1986-87 Peugeot 505s with automatic transmission shall be tested using test Sequence #5. If the engine has been changed to a different year, the special test sequence shall follow the year of the vehicle.

Example:

- 1985 BMW with a ZF transmission and original engine uses test sequence #5 and the emission standards for 1985.
- ¹ 1985 BMW with a ZF transmission and a 1990 engine uses test sequence #5 and emission standards for 1990.

f) <u>SEQUENCE #6</u>:

Testing period:	30 seconds for each stage	
First stage:	2500 RPM (±10%)	
Second stage:	Idle RPM	
Basis for test results:		
Stage 1:	Average of last 5 seconds of sampling period.	
Stage 2:	Same as stage 1; however, if the emissions are not within the standards and the idle RPM was below 900, then the technician shall be prompted to rev the engine so that the idle speed is a minimum of 900 RPM (but not to exceed the manufacturer's specifications), and to continue the test for	

another 30-second Second-Stage Idle Test. After the first 15 seconds of the repeated second stage, any passing reading (averaged over 5 consecutive seconds) collected during the sampling period, or, if none, the average reading over the last 5 seconds of the stage. Units of test results: Concentration measurements: PPM HC, % CO, % O₂ and

% CO₂

All 1985 Ford Ranger 2.3L (140 CID) light duty trucks and 1986 Ford Ranger and Aerostar 2.3L (140 CID) light duty trucks shall be tested using test sequence #6.

g) <u>SEQUENCE #7</u>:

Testing period: 25 seconds for each stage

.

Note: Prior to beginning the first stage, the technician shall be informed that the vehicle he/she will be testing will require special test procedures and that it is important to follow directions carefully (this information shall not be displayed prior to the "second-chance" test if preconditioning is required). The technician shall then be prompted to ensure the tach lead is connected, start the vehicle and allow it to idle. At the end of 156 seconds, the analyzer shall prompt the technician to insert the probe and begin the 2500 RPM test. The software shall ensure that there is an RPM signal for 156 seconds prior to starting the 2500 RPM test. This 156-second warm-up shall not be required prior to the "second-chance" test if preconditioning is required.

First stage:	2500 RPM (±10 %)
Second stage:	Idle RPM (see standards for maximum)
Basis for test results:	After the first 10 seconds of each stage, averaging shall
	begin. Any passing reading (averaged over 5 consecutive
	seconds) collected during each sampling period or if none,
	over the last 5 seconds.
Units of test results:	Concentration measurements: PPM HC, % CO, % O ₂ and
	% CO ₂

Test Sequence #7 could take as little as 15 seconds if test conditions are satisfied and the vehicle meets the standards. If the emissions are not within the standards for any 5-second period (following the initial 15-second period), the test shall run the 25 seconds.

All 1985-1986 GM passenger cars with VIN-Y ("Y" in eighth position of the <u>VIN</u>) engines shall be tested using Sequence # 7.

3.4.3 Vehicle Preconditioning Sequence For Two-Speed Idle Test

If a vehicle fails the TSI first chance and if all its emission readings are below 150% of their respective emissions limits, the analyzer shall instruct the technician to precondition the vehicle and run a second chance test. The EIS shall also use special test sequences for the second chance test if they were used for the first test. For example: if the EIS uses special test sequence #2 and the vehicle requires preconditioning, the EIS shall use special test sequence #2 for the second chance test. The EIS shall also follow any RPM restrictions that the special test sequence may require, i.e., a 1985 BMW with a ZF transmission shall NOT be preconditioned at high RPM. Based on the surveys conducted for the County, and on studies conducted on suspected pattern failures by the EPA, all model vehicles failing an initial test shall be preconditioned in the following manner, and re-tested:

DISPLAY PROMPT:

REMOVE THE EXHAUST PROBE FROM THE TAILPIPE.

a) Procedure #1: <u>For All Vehicles Except Those Covered by Procedures</u> <u>2 and 3</u> OPERATE THE VEHICLE AT 2500 ±300 RPM FOR THREE MINUTES WITH THE TRANSMISSION IN "PARK" OR "NEUTRAL."

> AT THE END OF THE THREE-MINUTE PERIOD, ALLOW THE VEHICLE TO RETURN TO IDLE AND STABILIZE FOR 10 SECONDS, BUT DO NOT TURN THE IGNITION SWITCH OFF.

INSERT THE PROBE INTO THE TAILPIPE.

AT THE END OF THE 10-SECOND PERIOD, IMMEDIATELY BEGIN THE EMISSIONS TEST.

Programming Criteria For Procedure # 1

Within 30 seconds of having completed the three-minute portion of the preconditioning sequence, the technician shall release the throttle, insert the probe and return the engine to $2500 (\pm 250)$ RPM. The 30-second time period shall begin when the engine RPM drops below 2200. The EIS shall provide prompts indicating when the technician should release the throttle, insert the probe and increase the engine RPM to the appropriate range as specified as soon as the probe has been inserted. The emissions test sequence shall begin as soon as the engine RPM reaches the appropriate range. The EIS shall display the time remaining before the preconditioning period will have to be restarted or the test aborted.

b) Procedure #2: For 1981-86 Fords and 1984-85 Honda Preludes

3-24 X, Part D, App 1.88

OPERATE THE VEHICLE AT 2500 ±300 RPM FOR 3 MINUTES WITH THE TRANSMISSION IN "PARK" OR "NEUTRAL."

AT THE END OF THE 3-MINUTE PERIOD, ALLOW THE VEHICLE TO RETURN TO IDLE AND IMMEDIATELY TURN THE IGNITION KEY OFF.

INSERT THE PROBE INTO THE TAILPIPE.

LEAVE THE IGNITION OFF FOR 10 SECONDS THEN RESTART THE ENGINE AND PROCEED IMMEDIATELY WITH THE EMISSIONS TEST.

Programming Criteria For Procedure # 2

Within 30 seconds of having completed the three-minute portion of the preconditioning sequence, the technician shall release the throttle, turn off the ignition for at least 10 seconds, insert the probe and return the engine to 2500 (± 250) RPM[•]. The 30-second time period shall begin when the engine RPM drops below 2200. The EIS shall provide prompts indicating when the technician should release the throttle, turn the ignition key off, insert the probe, and to restart the engine and immediately increase the engine RPM to the appropriate range as specified. The emissions test shall begin as soon as the engine RPM reaches the appropriate range. The EIS shall display the time remaining before the preconditioning period will have to be restarted or the test aborted.

c) Procedure #3: For "ZF" Automatic Transmission

Given the problems with the ZF automatic transmission, the County prefers that the affected vehicles be tested at their dealerships. Accordingly, if the technician enters an A (for automatic) for the transmission type, and if the vehicle make, model and model year match BMW/Peugeot/Volvo criteria, the EIS shall display the following message:

BECAUSE OF THE POSSIBILITY OF TRANSMISSION DAMAGE TO THIS VEHICLE, THE COUNTY PREFERS THAT IT BE INSPECTED AT ITS DEALERSHIP. IF YOU STILL WISH TO PERFORM THE INSPECTION, YOU MAY DO SO AT YOUR OWN RISK. PRESS "ENTER" TO CONTINUE. IF NOT, PRESS "ESC" TO ABORT THE TEST.

For all 1984-1987 BMWs with automatic, 1986-87 Peugeot 505s with automatic, and 1985-88 Volvo 740s with automatic transmission.

^{*}Emissions test RPM requirements may vary depending upon the test sequences.

If these vehicles fail the first chance, display the following message:

DUE TO POSSIBLE SERIOUS TRANSMISSION DAMAGE, DO NOT RAISE THE ENGINE SPEED ABOVE IDLE RPM WHILE THE TRANSMISSION IS IN NEUTRAL OR PARK. IF THE VEHICLE NEEDS TO BE PRECONDITIONED, DRIVE IT UNTIL IT HAS REACHED OPERATING TEMPERATURE.

The analyzer shall start the second chance test as soon as the EIS detects engine RPM within the idle RPM range. The EIS shall perform the emissions measurement at idle for 30 seconds. After the second chance, the EIS shall allow the technician to continue with the remainder of the inspection.

d) **Programming Criteria For All Procedures:**

- The manufacturer shall provide for the capability to utilize as many different preconditioning procedures as can be contained in the analyzer. The preconditioning procedure number shall be recorded on the test record in the *Preconditioning Procedure* field.
 - 1. <u>For all procedures</u>- The analyzer shall automatically instruct the technician to initiate the preconditioning procedure whenever a vehicle fails the emissions test before the test can proceed. The analyzer software shall select and display only the appropriate preconditioning procedure based on the vehicle make and model year information entered by the technician.
 - 2. For procedure # 1- A message shall be displayed instructing the technician to remove the exhaust probe and increase the engine RPM to 2500 (±300) and hold it there for 3 minutes. The analyzer shall detect a signal in the proper range for 3 minutes within a 3-minute and 15-second period, with no single excursion exceeding 5 seconds. A message shall be displayed instructing the technician to adjust the engine RPM, restart the test or abort the test as appropriate if the RPM is outside of the specified limits. The preconditioning period shall begin as soon as the engine RPM is stable (for a period of 1 second) and in the proper range. To avoid loading the sample system with vehicle exhaust during the preconditioning sequence or prevent preconditioning if the probe is in the tailpipe. Preconditioning prior to or during the preconditioning sequence.

When the preconditioning period is complete, the technician shall be instructed to allow the vehicle to return to idle and the analyzer shall ensure that the engine speed is reduced for at least 10 seconds, but no more than 30 seconds. If the engine speed is reduced for less than 10 seconds or more than 30 seconds, a message shall be displayed instructing the technician to either restart the preconditioning procedure or abort the test. Messages indicating the appropriate ignition key on/off and retest instructions shall be displayed at the end of the 10 second idle period. The technician shall be instructed to strike the ENTER key as soon as possible after 10 seconds of idling has occurred.

- 3. <u>For all procedures</u>- The analyzer shall display the engine speed and the time remaining during each stage of the preconditioning sequence. The number of the preconditioning procedure shall be recorded on the test record automatically by the analyzer. If no preconditioning procedure was used (vehicle passed the emissions portion of the test the first time), this record shall be filled with a space.
- 4. Error Messages:

(For all procedures)

NO RPM SIGNAL - MAKE SURE THE TACH LEAD IS CONNECTED. (For procedures 1 & 2) ENGINE RPM DROPPED BELOW 2200 RPM -RAISE THE ENGINE SPEED TO 2500 RPM AND HOLD IT THERE FOR 3 MINUTES.

(For procedures 1 & 2)

ENGINE RPM INCREASED ABOVE 2800 RPM-REDUCE THE ENGINE SPEED TO 2500 RPM AND HOLD IT THERE FOR 3 MINUTES.

5. <u>For all procedures</u>- If a vehicle subject to preconditioning receives a second-chance test, the emissions results of both tests shall be stored in the test record. The results for either or both tests shall not be written to the test record until the pass/fail decision has been made by the EIS.

3.4.4 **OBD II Inspection Procedures**

During this portion of the vehicle inspection the UTAH 2000 shall automatically test the status and condition of the vehicles' emission control components as reported by the OBD II system. The software shall adhere to the following flow chart for details of the OBD II procedure. The UTAH 2000 software shall display a message to the technician only where indicated by the word "Prompt:". The other portions of the flow chart are for the use of the software internally so as it is transparent to the technician.

During the OBD II inspection, the software shall collect and store the appropriate data as required by Section 3.5.1 Vehicle Inspection Data. Following the flow chart is a series of notes that are referenced for clarifications.







- 1

a) Notes on OBD II Flowchart:

<u>ہ</u>

- 1. <u>Note 1</u>: The purpose of this step is to verify the On-Board Diagnostic (OBD) system has control of the Malfunction Indicator Light (MIL) and the MIL is functional. Operation of the MIL varies between vehicle manufacturers. Key On Engine Off (KOEO) typically results in the MIL on steady, however, there are systems which will illuminate the MIL only briefly during KOEO. In either situation MIL presence and illumination capability has been established. If the vehicle fails the I/M test at this point, the vehicle inspection report should indicate the MIL problem should be repaired and also include information gathered during the remaining I/M test steps.
- 2. <u>Note 2</u>: It is important for the I/M testing personnel to verify proper diagnostic equipment operation before failing the vehicle. If the diagnostic equipment is functional then the vehicle's communication problem must be resolved.

Without communication between the OBD system and the test equipment the I/M test must be ended and the problem resolved before further interrogation of the vehicle can be performed. This step includes identification of Data Link Connector (DLC) tampering, serial data circuit problems and any other condition that would prevent the OBD system from communicating with the test equipment.

- 3. <u>Note 3</u>: I/M test failure is a result of MIL illumination even though the OBD system has not commanded the MIL on, or has stored any Diagnostic Trouble Codes (DTCs); e.g., a serial data line failure between the OBD computer and the Instrument Panel.
- 4. <u>Note 4</u>: I/M test failure is a result of both the actual and commanded state of the MIL. DTCs should be stored since the MIL is commanded on. A vehicle should not fail an I/M test when DTCs are stored but there is no MIL on; e.g., the DTC was stored by a loose gas cap which was subsequently tightened.
- 5. Note 5: Readiness Code status must be identified at this stage in the I/M test to determine whether or not all emission control systems have been tested by the OBD system. If any one (or more) Readiness Code(s) are not set ("ready") the OBD system has not yet completed testing of the system(s) and failures may be present but not yet identified. It is important to understand that the vehicle does not fail the I/M test at this point; no emission related faults have been identified. The current state of the vehicle's emission control system is undetermined. The software shall print the two page "OBD II Drive Cycle" (see Appendix III) to inform the vehicle owner how to set the Readiness Codes.

X Part .

The emission control systems and related components are tested under specific vehicle operating conditions. Therefore, to set the Readiness Codes the vehicle must be operated within these specific conditions (commonly referred to as "enable criteria") for the OBD system tests to be performed. Once testing of an emission control system is complete, the related Readiness Code will be set ("ready"). When all Readiness Codes are set, the vehicle is ready for further I/M testing.

3.4.5 Visual Inspection Procedures

The EIS shall prompt the technician to enter the condition for each of the following visual inspection items;

- 1. Air Injection System,
- 2. Fuel Evaporation Control System,
- 3. EGR System,
- 4. PCV System,
- 5. Catalytic Converter,

The possible conditions for the visual inspection are pass, fail and not applicable. A fail item shall be the only cause for failing the visual inspection.

3.4.6 Gas Cap Pressure Test Inspection Procedures

The EIS shall prompt the technician to answer the following questions in this same sequence:

- 1. "Is the gas cap accessible? (Y/N)",
- 2. "Can the gas cap be removed? (Y/N)",
- 3. "Does it fit a standard adapter? (Y/N)".

If the technician answers "no" to any of the first two questions the software shall terminate the inspection process and print a VIR with all of the information collected thus far. However, if the technician answers "yes" to those same questions, the software shall not terminate the test.

If the technician answers no to the third question the gas cap test shall not be conducted. However, if yes is answered to the final question then the pressure test shall be conducted. The software shall provide the appropriate prompts to facilitate the technician through the gas cap test process. At the conclusion of the gas cap test, the software shall evaluate whether the gas cap in question failed or passed the test.

a) If the gas cap failed the test, the software shall prompt the technician if the gas cap was replaced with a new one. If so, the software shall prompt the technician to place the new gas cap on the tester and the software shall test the new gas cap. If the new cap fails the test then the software shall record this result in the EIS record. However, if the technician does not replace the gas cap with a new one the software shall keep the result and store it in the EIS record. b) If the gas cap passes the test, the software shall keep the result and store it in the EIS record.

At the conclusion of the gas cap test the software shall prompt the technician to remove the gas cap from the tester and to place it back on the filler neck of the vehicle.

3.4.7 Safety Inspection Procedures

The EIS shall prompt the technician to enter the Inspector Number, Inspection Fee and condition for each of the following safety items;

- 1. Wheel Lugs,
- 2. Emergency Brake,
- 3. Steering and Suspension,
- 4. Exhaust System,
- 5. Windshield Wipers/Washer,
- 6. Windshield/Tinting,
- 7. Other Glass,
- 8. Mirrors,
- 9. Headlights,
- 10. Other Lights,
- 11. Turn Signals,
- 12. Tires,
- 13. Horn,
- 14. Fuel System,
- 15. Other (i.e., body, seat, belts, ...).

The possible conditions for the safety inspection are pass, fail, repair, advisory and not applicable. If a technician enters a repair condition for any item, the software must prompt to enter the cost associated with the repair of the item(s) in question. An entry of repair or advisory for any of the safety inspection items shall require the software to prompt the technician to enter comments for the item(s) in question.

During the course of the safety inspection the EIS shall also prompt the technician to remove a front and rear wheel and note which wheel was pulled for both, the left or right. Upon removal of the wheels, the technician shall be notified to inspect and measure the brake shoe thickness remaining on each set of brake shoe of those wheels pulled. The brake shoe thickness shall be measured and entered onto the system and the values shall range anywhere from 0 to a value of 32. There will be four (4) measurements taken, Front Wheel Primary, Front Wheel Secondary, Rear Wheel Primary and Rear Wheel Secondary. The lowest reading shall be used for pass/fail criteria. Values less than and equal to 1 shall be cause for failing the brake shoe safety inspection, thus failing the safety inspection. A fail for any safety inspection item shall be cause for failing the safety inspection. All data entered by the technician shall be stored to the Safety test record and sent to the HOST on the next HOST contact.

3.4.8 Repair Data Entry

11

The EIS shall prompt the technician to enter the date the repairs were made. The EIS shall not accept any repairs conducted before the initial test date. If the repairs were conducted prior to the initial test then the software shall inform the technician that the repairs are not valid to obtain a waiver. The EIS shall display a list of all repair categories and prompt the technician to select the category or categories of the system(s) which were repaired and/or service recommended. The codes for each of the two possible entries shall be stored in the repair record and sent to the HOST on the next call. The technician must be able to return to the list of major categories after each subcategory has been completed. Prior to exiting this function, the software shall summarize the information that was selected by the technician. This summary screen shall give the technician the option to accept/continue or modify the repair information. If the repair information is accepted then the software shall continue with the inspection process. Else, if the modify-option is selected then the software shall go back and allow the technician to change the previous selections. All prior selections shall not be erased when the technician selects to modify the information.

All repair actions shall be documented on the vehicle inspection report (VIR), and stored in the repair record. The technician shall be required to sign on the VIR to document the repairs that have been performed to reduce emissions. The vehicle repair cost shall be printed on the VIR and the parts cost and labor cost of the repairs shall be entered in the *PartsCost* and *LaborCost*, respectively, fields of the repair record.

Categories

Spark Plugs/Wires Other Ignition Timing/Spark Advance Air/Fuel Mixture Adjustment Idle Speed Adjustment Choke System Other Carburetor Repairs Vacuum Leaks Fuel Injection System Other Induction System Air Filter Oil Change Thermostatic Air Cleaner PCV System Air Injection/Reaction System EGR System Fuel Evaporative System Catalytic Converter(s) Oxygen (O2) Sensor(s) MAP Sensor Mass Air Flow Sensor Coolant Temp. Sensor Throttle Position Sensor

Other Sensors/Switches Computer/PROM Internal Engine Repair Other Repairs/Unknown

After entry of eligible repair information the software shall allow the test to proceed and conduct the inspection as normal.

3.4.9 Pass/Fail Determination

4

The final inspection results shall be determined as follows:

a) Emission Inspection Pass Criteria

If the OverallEmissionsResult, OverallGasCapTestResult, OBDIITestResult, and number 96 through 105 fields of the EIS test record all contain pass or N/A entries, then the vehicle shall pass the emissions portion of the inspection and a "P" shall be entered into the OverallTestResult field of the EIS test record. The vehicle shall pass the inspection and the EIS shall issue an emissions certificate.

b) Emission Inspection Fail Criteria

If the OverallEmissionsResult, OverallGasCapTestResult, OBDIITestResult, or number 96 through 105 fields of the EIS test record contain a fail or a tamper entry, then the vehicle shall fail the emissions portion of the inspection and a "F" shall be entered into the OverallTestResult field of the EIS test record. The vehicle shall fail the inspection and the EIS shall not issue an emissions certificate but may allow a waiver if the vehicle meets the waiver criteria.

c) Emission Inspection "Not Ready" Criteria

If the OBDIITestResult field of the EIS test record contains a "not ready" entry, then the vehicle shall not fail or pass the emissions portion of the inspection and a "R" shall be entered into the OverallTestResult field of the EIS test record. The vehicle shall not fail or pass the inspection and the EIS shall not issue an emissions certificate shall not allow a waiver.

d) Safety Inspection Pass Criteria

If field numbers 18 through 32 all contain either a pass, repair, advisory or N/A entry and both the *FWMeasured* and *RWMeasured* fields of the Safety test record each contain a value greater than 1, then the *SafetyTestResult* field of the Safety test record shall contain a "P" entry and the vehicle shall pass the safety portion of the inspection. The software shall issue a safety certificate.

e) Safety Inspection Fail Criteria

If any field numbers 18 through 32 contain a fail entry or either or both the *FWMeasured* and *RWMeasured* fields contain a value less than or equal to 1, then the *SafetyTestResult* field of the Safety test record shall contain a "F" entry and the vehicle shall fail the safety portion of the inspection. The software shall not issue a safety certificate.

3.4.10 Emissions and Safety Certificates of Compliance

The UTAH 2000 shall issue a certificate of compliance on pre-perforated paper provided by County. The certificate can then be used for registration of vehicle. The analyzer shall automatically keep track of the number of remaining certificates based on the total number purchased.

The certificate number shall be printed on the VIR in both alpha numeric and UPRC onedimensional barcode and shall be transmitted during the next HOST contact along with any other required inspection information.

The certificate number shall be put in the *CertificateNumber* field of the EIS test record. The 10 digits shall be used sequentially for each emissions test requiring a certificate number. The certificate numbers shall be purchased by the UTAH 2000 unit from the HOST during a certificate purchase transaction.

The UTAH 2000 software shall keep track of the unused certificate numbers issued to it by the HOST and notify the technicians when the certificate inventories are "running low".

3.4.11 Waivers

Upon completion of the inspection, if eligible repair information was entered and the vehicle failed for OBD II or TSI the software shall evaluate whether the vehicle is eligible for a waiver using all of the following criteria;

Repairs were conducted on or after the intial inspection date,

Failed OBD II inspection or TSI,

If a TSI inspection was conducted then current HC and CO readings shall not exceed those of the previous inspection, and

The software shall reference the Waiver Criteria Table for the corresponding county-specific limits and criteria.

The county-specific waiver criteria are provided in the Waiver Criteria Table.

3.4.12 Vehicle Inspection Report (VIR)

After display and review of the final test results, the EIS shall print the VIR. The VIRs shall adhere to the format provided in the sample VIRs in Appendix IV.

3.4.13 Training Mode

The UTAH 2000 shall have a TRAINING MODE feature that will allow a technician or student to go through the complete inspection procedure. This capability will be used by the manufacturers for training purchasers of the EIS, by EIS owners to train new employees, or for schools to train students. The training mode shall not require the use of a technician's access code or allow access to secured areas of hardware or software and will not communicate to the HOST. The display shall show a message throughout the

inspection that this is a training exercise and not an official test (no certificates shall be issued). The EIS shall print "NOT AN OFFICIAL INSPECTION" on the VIR.

The training mode test results shall be recorded and shall <u>not</u> be transmitted to the HOST at the next required communication session (i.e. next Inspection, data file refresh, etc.).

3.4.14 Dilution Correction Factor

The software shall apply a DCF to the HC and CO inspection emissions results. This dilution correction accounts for any exhaust sample dilution, intentional or unintentional, occurring during the emissions inspection process. The software shall calculate the DCF using the following procedure, and shall pre-select the formula appropriate to the vehicle's fuel type. If the calculated DCF exceeds 3.0, a default value of 3.0 shall be used. If the DCF falls below 1.0, then a default value of 1.0 shall be used. -----

بسيسانيان بالاسان والاسا

Calculate "x" using the EIS measurements of CO and CO₂: a)

$$x = \frac{[CO_2]_{meas.}}{[CO_2]_{meas.} + [CO]_{meas.}}$$

where $[CO_2]_{meas}$ and $[CO]_{meas}$ are the final readings of each mode of the inspection (for example, 2500 RPM and idle).

b) Calculate the $[CO_2]_{adj}$ using the following formulas.

For Gasoline

$$[CO_2]_{adj.} = \left[\frac{x}{4.644 + 1.88x}\right] 100$$

For Methanol or Ethanol:

$$\left[CO_{2}\right]_{adj} = \left[\frac{x}{4.73 + 1.88x}\right]100$$

For Compressed Natural Gas (CNG):

$$\left[CO_2\right]_{adj.} = \left[\frac{x}{6.64 + 1.88x}\right] 100$$

c) Calculate the corrected readings;

Corrected HC = Observed HC x DCF Corrected CO = Observed CO x DCF

3.4.15 Dilution

The software shall monitor the dilution in the exhaust throughout the duration of the TSI emissions sampling. The dilution (not to be mistaken for the DCF) is calculated summing the uncorrected CO and uncorrected CO2 emissions readings (CO + CO2). If the dilution value is below 6% then sample dilution is occurring and the emissions testing shall not proceed and cause a restart in the emissions testing procedure. However, CNG- and LPG-powered vehicles shall use a value of 4%.

3.4.16 Sample System Readiness (Zero & HC Hang-up)

- a) The analyzer shall be zeroed in accordance with Section 2.4.5 a) and b).
- b) The HC hang-up check will be done, according to Section 2.4.5 s), immediately after the analyzer is zeroed and the ambient air is sampled. The zeroing and HC Hang-up process' shall be initiated after an inspection has been initiated and the technician access code has been accepted by the analyzer. The whole zeroambient air-HC hang-up sequence <u>shall run in background</u> while the inspection process continues but before the TSI. If the hang-up check is not completed before the technician is ready to start the tailpipe test, the EIS shall display the following message:

DISPLAY PROMPT:

HC HANG-UP CHECK IN PROGRESS.

If the hang-up check is not successfully completed in 150 seconds from the start of the hang-up check, the EIS shall display the following message:

DISPLAY PROMPT:

POSSIBLE DIRTY PROBE, HOSE OR FILTER. VERIFY THAT THE PROBE TIP IS NOT NEAR AN EMISSIONS SOURCE AND THAT THE PARTICULATE FILTER IS NOT DIRTY. DO YOU WANT TO RETRY OR ABORT THE INSPECTION.

c) The software shall not allow the inspection to continue before the system passes the HC hang-up check <u>unless</u> it has been determined that the inspection will be an OBD II test.

3.4.17 Aborts

The software shall record the following types of aborts. Any information already gathered prior to the abort shall be written to the test record and stored on to the hard

drive. The technician shall have the option to abort any inspection at any time with the selection of the <Esc> or a function key. Aborted inspection records shall be sent to the HOST. If an abort occurs, then the software shall select the most appropriate abort code, write it to the *AbortCodeNumber* field in the EIS test record, and place an "A" to the *OverallTestResult* field of the EIS test record.

<u>Code</u>	<u>Description</u>
01	Inspector license expired
02	Incorrect VIN
03	After HOST contact
04	Gas cap not accessible
05	Gas cap can not be removed
06	After visual inspection but prior to gas cap test
07	During OBD II inspection
08	During TSI inspection
09	During Safety inspection

3.5 VEHICLE INSPECTION INFORMATION

3.5.1 Vehicle Inspection Data

a) Data Collection, format and Storage:

Following is a brief description describing all of the data, including TSI, OBD II, safety, gas cap, repair, calibration, lockout, Vehicle Reference Table (VRT), and emissions limits data that shall be collected and stored onto the hard drive. The data described in this section is divided into several different subsets referred to as records and/or tables. Each record contains data for each of the major tasks of the UTAH 2000 units. The records are sorted by alphabetical order in Appendix II. Some information will be generated by the UTAH 2000 units and the rest will be supplied by the HOST. Those fields designated by the fields "Sent to HOST" and "Sent by HOST" with a "X" signify that the data will be generated at the EIS unit or the HOST, respectively and transferred during the inspection's HOST communication session. Some fields may be generated from both sources and are designated by a "X" in both fields. The "RefreshAll" designates the data which shall be transmitted during a non-inspection HOST session. The "RefreshAll" process is used only for updating the individual EIS units with the latest information, as in the initialization process. All field formats shall be adhered to when storing on to any of the drives and during the HOST transmission.

Important Note: The manufacturer shall adhere to the possible entries for each field(s) when specified in this specification. Appendix II contains the possible entries for many fields. When possible entries are absent/omitted in this specification it shall be the responsibility of the manufacturer to develop the codes and possible entries upon consultation and approval of the County.

1. <u>Audit Record (Audit)</u>: During a County gas audit, the software shall record the following information in the prescribed format and store the information onto the hard drive and floppy drive. These records shall be sent to the HOST during the inspection HOST session. Once sent to the HOST the record shall be removed from the floppy drive but not from the hard drive. Only the last fifty (50) Audit records are required to be kept in storage on the hard drive.

- 2. <u>Calibration Record (Calibration)</u>: This record shall be generated during each UTAH 2000 unit calibration. All records must be sent to the HOST, one time only, on the initial call of the following inspection. Prior to being sent the test records must be stored on both the hard drive and floppy disk. Once sent to the HOST, the record shall be removed from the floppy disk. The EIS shall keep in storage on the hard drive, at the very least, the last one hundred (100) calibration records.
- 3. <u>Certified Emissions Repair Facility Record (CERF)</u>: This record will be generated and sent by the HOST. This record contains information on all repair facilities certified by the County to conduct repairs on emissions components. This record shall be referenced by the software before allowing repair information to be entered. Any repairs from facilities not found in this list or with an expired license will not count towards the inspection. The CERF records shall be sent by the HOST during a "RefreshAll" session.
- 4. <u>Certified Emissions Repair Technician Record (CERT)</u>: This record will be generated and sent by the HOST. This record contains information on all repair technicians certified by the County to conduct repairs on emissions components. This record shall be referenced by the software before allowing repair information to be entered. Any repairs from technicians not found in this list or with an expired license will not count towards the inspection. The CERT records shall be sent by the HOST during a "RefreshAll" session.
- 5. <u>City Zip Code Record (CityZip)</u>: This record will be generated and sent by the HOST. This record contains zip code related information. This record shall be used by the software for cross referencing county codes, county names, city names, and city zip codes. Included in each record is the field "GasCapTest" which shall be used by the software to determine whether a vehicle is required to undergo the gas cap pressure test. These records shall be sent by the HOST during a "RefreshAll" session.
- 6. <u>County Information Record (CountyInfo)</u>: This record will be generated and sent by the HOST. This record contains technical center phone numbers for each county. This record shall be used by the software for cross referencing county codes and the technical center phone numbers. These records shall be sent by the HOST during a "RefreshAll" session.
- 7. <u>Technical Service Bulletin (TSB)</u>: This record will be generated and sent by the HOST. This record contains technical service bulletins for the

V Danto A

. . .

. .

technicians. The TSB shall be displayed by the software upon receiving the TSB. The software shall allow the technicians to review on screen and print each TSB. The last five (5) TSB's received shall be stored on the hard drive so that technicians may print or review the last five (5) at a later time. The TSB records shall be sent by the HOST during an inspection HOST session.

8. <u>Vehicle Inspection Record (EIS)</u>: This record shall be generated and sent by the UTAH 2000 units to the HOST. This record shall contain information from the vehicle inspection. All records must be sent to the HOST, one time only, on the initial call of the following inspection HOST session. Prior to being sent the test records must be stored on both the hard drive and floppy disk. Once sent to the HOST, the record shall be removed from the floppy disk. The EIS shall keep in storage on the hard drive, at the very least, the last one thousand (1,000) vehicle inspection records.

, A

- 9. Vehicle Inspection Certificate Inventory Record (InspCertInventory): This record will be generated and sent by the HOST. This record contains information on the remaining vehicle inspection certificate inventory of the individual UTAH 2000 unit. The software shall reference the inventory records and select the lowest un-issued certificate number prior to issuing a certificate. Once a certificate has been issued it shall not be re-used for any other inspection. The certificate number shall be printed on the Certificate. The InspCertInventory records shall be sent by the HOST during a HOST "RefreshAll" session. The "UnitPrice" and "TotalPrice" fields will only be filled when a purchase is made using the "BuyCerts" HOST session.
- 10. Lock Out Record (LockOut): This record will be generated and sent by both the HOST and the UTAH 2000 units. This record shall contain lock out information for the individual EIS units. The LockOut record shall be sent to the HOST at the beginning of an inspection HOST session and returned by the HOST prior to terminating the same session. Any lock outs returned by the HOST shall be set by the software for that particular EIS unit. The LockOut record shall be stored on both the hard drive and floppy drives at all times. Any current lock outs set by the EIS or HOST shall be contained within this record.
- 11. <u>OBD II Fault Code and Readiness Record (OBDIIFC)</u>: These records will be generated and sent by the HOST. This record contains information used by the software for cross referencing the OBD II fault and readiness codes with their text translation. The *FaultReadinessCode* field contains information whether the fault code is cause for inspection failure or not. These records shall be sent by the HOST during a "RefreshAll" session.

- 12. <u>Repair Record (Repair)</u>: This record shall be generated and sent by the UTAH 2000 units to the HOST. This record shall contain information from the vehicle repairs. All records must be sent to the HOST, one time only, on the initial call of the following inspection HOST session. Prior to being sent the Repair records must be stored on both the hard drive and floppy disk. Once sent to the HOST, the record shall be removed from the floppy disk. The EIS shall keep in storage on the hard drive, at the very least, the last one thousand (1,000) Repair records.
- 13. <u>Safety Record (Safety)</u>: This record shall be generated and sent by the UTAH 2000 units to the HOST. This record shall contain information from the vehicle safety inspection. All records must be sent to the HOST, one time only, on the initial call of the following inspection HOST session. Prior to being sent the Safety records must be stored on both the hard drive and floppy disk. Once sent to the HOST, the record shall be removed from the floppy disk. The EIS shall keep in storage on the hard drive, at the very least, the last one thousand (1,000) Safety records.
- 14. <u>Vehicle Safety Inspection Certificate Inventory Record</u> (SftyCertInventory): This record will be generated and sent by the HOST. This record contains information on the remaining vehicle safety inspection certificate inventory of the individual UTAH 2000 unit. The software shall reference the inventory records and select the lowest unissued certificate number prior to issuing a certificate. Once a certificate has been issued it shall not be re-used for any other inspection. The certificate number shall be printed on the Certificate. The SftyCertInventory records shall be sent by the HOST during a HOST "RefreshAll" session. The "UnitPrice" and "TotalPrice" fields will only be filled when a purchase is made using the "BuyCerts" HOST session.
- 15. <u>State Information Record (States)</u>: These records will be generated and sent by the HOST. This record contains information used by the software for cross referencing the state codes with their text translation. These records shall be sent by the HOST during a "RefreshAll" session.
- 16. <u>System Record (System)</u>: This record will be generated and sent by the HOST. This record will contain system information for the individual EIS units. The record shall be sent by the HOST during an inspection HOST session.
- 17. <u>Emissions Limits Record (Table3)</u>: These records will be generated and sent by the HOST. This record contains the emissions limit information used by the software for evaluating the emissions test results. These records will be sent by the HOST during a "RefreshAll" session.

The emissions standards (Table3) table shall reside in the EIS and receive updates from the HOST. The EIS shall look into Table3 for emission cut

points. Table3 shall also have a version number. Upon implementing the new table, the old version shall be purged. Additional standards categories may be added at a future date. Based on the vehicle information entered, the EIS shall determine the emissions test standards for the vehicle being tested.

For each vehicle, the table will contain HC and CO Pass/Fail emissions for the two-speed idle tests. Print these emissions values on the VIR. Table3 values and the criteria for selecting categories shall be designed in a manner that allows for easy modification or addition.

The ESC for the vehicle under test shall be written to the Emission Standards Category field of the test record.

, ^

18. <u>Tamper History Record (Tamper)</u>: This record shall be generated and sent by the UTAH 2000 units to the HOST. This record shall contain tamper information from the each individual unit. All records must be sent to the HOST, one time only, on the initial call of the following inspection HOST session. Prior to being sent the Tamper records must be stored on both the hard drive and floppy disk. Once sent to the HOST, the record shall be removed from the floppy disk. The EIS shall keep in storage on the hard drive, at the very least, the last one hundred (100) Tamper records.

- 19. <u>Tamper Limits Record (TamperCriteria</u>): These records will be generated and sent by the HOST. The record contains information used by the software for determining the tampering criteria for each County. These records shall be sent by the HOST during a "RefreshAll" session.
- 20. <u>Station Technician Information Record (TechInformation</u>): This record will be generated and sent by the HOST. This record contains information on all inspection technicians certified by the County to conduct inspection at that individual inspection station. This record shall be referenced by the software before allowing the vehicle inspection from proceeding. Upon initiation of a vehicle safety or I/M inspection the software shall prompt the user for their access code designated in the "AccessCode" field of this record. If the access code is not found in any of these records the inspection shall be aborted. The CERT records shall be sent by the HOST during a "RefreshAll" session.
- 21. <u>Vehicle Reference Table Record (VRT)</u>: This table shall be generated by the HOST and sent during a "RefreshAll" session. This record must be stored onto the hard drive and referenced by the software for finding vehicle information during data input. This record will be updated on a yearly basis or as new vehicle model years are introduced into the fleet. Each record in the dataset will contain information for every particular make, model, year, engine size ... vehicle currently in circulation. The VRT shall be referenced by the software to determine whether OBD II

A 47 MLAN ALL 4 404

testing is applicable. The "OBDIITest" field provides information whether a vehicle is subject to an OBD II inspection.

.

22. <u>Waiver Cost Limits Record (WaiverCriteria)</u>: These records will be generated and sent by the HOST. The record contains information used by the software for determining the applicable waiver cost limits for each County. These records will be sent by the HOST during a "RefreshAll" session.

£-1

SECTION 4 CENTRAL COMPUTER SYSTEM SPECIFICATION

This section outlines the performance and functionality of the Central Computer System (CCS) for the UTAH 2000 I/M program. The Contractor shall be responsible for designing, implementing, operating and maintaining the hardware, software and peripheral equipment necessary to accept and store data created at UTAH 2000 inspection stations. The CCS shall have a dedicated, secure connection so that the UTAH 2000 units may contact from the testing facilities and to facilitate the flow of data between the inspection stations and the CCS data repository. All UTAH 2000 units shall connect to the CCS through an independent and secured means using the internally housed modems and dedicated telephone lines.

All requirements in the Central Computer System Specification section must be addressed in sufficient detail to allow the County to determine how the Contractor will meet each requirement.

4.1 GENERAL

All components including hard drives, memory, I/O devices and media subsystems used in the CCS shall be of types and brands that are presently in common usage. The host computer(s) and all CPU devices connected within the CCS must operate using a common operating system without modification. Multiple redundant systems or system clusters are an acceptable method of achieving desired performance and redundancy. The County may allow deviations upon approval.

a) CCS Capacity

The CCS must have the capacity to annually handle, manage and store one (1) million vehicle record files, approximately 1.2 million emissions tests which includes retests, and allow for an estimated growth rate of 6%. This system must be able to handle 4,000 transactions per hour.

 b) CCS Expandability The CCS must be expandable to allow for future expansion to support increased load, transactions, vehicle records, data storage requirements and/or application development.

c) CCS Reliability/Availability

The CCS shall be fault tolerant and shall be accessible 24 hours a day, 7 days a week excluding scheduled maintenance downtime(s).

- d) Backup System
 The CCS shall include a reliable data backup system capable of restoring the system and data to working order with minimal data loss and service interruption. The backup media must be stored in an off-site location.
- e) Security

Due to the nature of the data being transmitted, security is a very serious issue to be considered by the Contractor.

1. The CCS must be able to make secure connections between itself and the remotely located UTAH 2000 units, remotely located County personal computer (PC) or any other County authorized computers.

4-1 X, Part D, App 1.109

- 2. The CCS must have multiple levels of security to protect applications and data files from unauthorized access, modification or deletion. The multiple levels of access shall reflect different types of users.
- 3. The CCS must have physical security measures to protect applications and data files from intruders, electrical power fluctuations or failure, fire damage, water damage or other acts of God.

4.2 HARDWARE REQUIREMENTS

a) Architecture

An architectural design of the CCS and all major components is required. It must describe the design of the network between the inspection stations and the CCS, and it must include any local area network within the CCS.

b) Year 2000 Compliance

All computer systems, sub-systems, components, networks, and/or computer equipment including hardware, software, firmware and operating systems to be utilized by the Contractor for this project must be Year 2000 Compliant. All dates must be represented and interpreted by four (4) digit years.

c) Database Sub-System

The heart of the CCS is the Database subsystem. This is where the applications and data repository will be stored and accessed. This system or group of systems must be capable of responding to the many data requests from inspection stations and the remotely located County PC's. The minimum requirements are as follows:

- 1. Hardware
 - A. Processors

The Database Sub-System must be able to support multiple processors or distributed systems.

B. Storage Capacity

The Database Sub-System must be able to store one (1) million vehicle record files, approximately 1.2 million emissions tests which includes retests, and allow for an estimated growth rate of 6%. This system must be able to handle 4,000 transactions per hour. The minimum hard-drive capacity is expected to be 35 GB, and the hard-drive subsystem must employ some sort of hot swappable redundant array of disks.

C. Memory

The Database Sub-System must contain sufficient RAM and Cache Memory to quickly and efficiently handle the volume and flow of transactions listed above.

D. Operating System

The Database Sub-System must be controlled by a robust, multitasking operating system with built-in security features such as access-level control implementing user identification numbers and blind entry passwords.

4-2

E. Documentation

The Contractor shall supply documentation and/or user manuals for all hardware and software including manufacturer contact information of any outside or third party software integrated into the system.

2. Database Engine

Commercially supported product with technical support from the manufacturer available 24 hours a day, 7 days a week. The DataBase Management Systems (DBMS) must support parallel, symmetric server architecture with automatic load balancing across multiple processors.

3. Secure Transactions

All transactions with the inspection stations, the County's remote PC's and any County authorized computers must be secure. Therefore, as a minimum the data must be encrypted using a private key or other means.

Backup System

d)

The CCS shall include a reliable data backup system capable of restoring the system and data to working order with minimal data loss and service interruption. The system must be capable of backing up all data and applications. A Master Backup Plan must be supplied outlining the procedures for daily, weekly and monthly back-ups, media rotation, and off-site storage and retrieval procedures. The Master Backup Plan shall include, at a minimum, the procedures for daily backup, off-site storage of all applications and data files, and data retrieval procedures.

e) Un-interruptible Power Supply

The CCS must be protected by an un-interruptible power supply (UPS) to isolate the system from power spikes, surges, fluctuations, brown-outs and complete power loss situations. The UPS must be able to maintain system operation for a minimum of one (1) hour for proper shutdown of the system without loss of data. The UPS must have a communications link and communications software that is compatible with the CCS' operating system and application software. Through this link, the UPS shall notify the CCS when power has been interrupted so that it may initiate a safe system shutdown automatically.

f) Automated Emergency Notification

The CCS shall employ a method which automatically and immediately notifies through the use of a paging system or some other reliable method a local primary and secondary representative of any emergency or potential emergency situation. This feature shall have the capability to be activated by the CCS hardware and/or software in case of an emergency. The paging system shall contact the primary and secondary representatives when an emergency arrives. A plan shall be set into place whereby the primary, the secondary or both representatives shall respond to the emergency within a period of time acceptable to the County. The County must also be informed of such an emergency.

4.3 DISASTER RECOVERY

6
A disaster recovery plan shall be provided outlining the steps to be followed in the case of a disaster or failure of the CCS. In the event of an outage of the system, the Contractor must implement the disaster recovery plan approved and on file with the County. The Contractor must notify the County immediately of the disaster and the system must be operational again within one calendar day (24 hours). The Contractor's design shall address:

- a) Scope of the disaster recovery plan (i.e. list all likely events this plan would cover),
- b) Major steps of the disaster recovery plan,
- c) Precautions taken to prevent possible downtime and data loss,
- d) List of redundancies and fault tolerance built into the system,
- e) Ability of Contractor to implement plan immediately,
- f) Estimate of time required to make system operational.

4.4 SOFTWARE REQUIREMENTS

a) Automatic Report Generation

The HOST software shall provide a method by which authorized remote PCs may access periodic reports. The reports shall consist of;

- 1. Repeat of the "Station Evaluation Reports" defined in Section 3 County Menu with the addition of categories by county, station, and overall for any user selected year, month or day,
- 2. Repeat of the "Monthly Station Inspection Report" defined in Section 3 Additional Site Specific Software for a user selected station and any user selected year, month or day,
- b) User Access Levels

The HOST software shall provide a method by which only authorized UTAH 2000 units and remote PCs may access the HOST. Access Codes for the Remote County PCs shall require a double-blind entry while a method shall be developed by the manufacture to establish a secure login procedure for the UTAH 2000 units. The software shall offer the following user access levels;

- 1. User Level 1 Access:
 - Modification of all reference tables or records (i.e., all tables/records but the Audit, Calibration, EIS, Repair, and Safety information which is generated by the analyzers) sent to the analyzers by the HOST,
 - Ability to set and reset lockouts for any analyzer,
 - Downloading and printing of reports locally,
 - Downloading of data from time intervals specified by the user.
- 2. User Level 2 Access:
 - Ability to set and reset lockouts for any analyzer,
 - Downloading and printing of HOST-generated reports locally,
 - Downloading of data from time intervals specified by the user.

F-1

- 3. User Level 3 Access:
 - Downloading and printing of reports locally.
- 4. User Level 4 Access (UTAH 2000 units only):
 - Communication with authorized UTAH 2000 units for transferring test information and reference tables to and from the HOST.

c) Data Downloads For AD-Hoc Reporting

The HOST software shall provide a method by which authorized remote PCs may access the HOST and download different test data from user specified time periods. The software shall be designed in such a way as to allow the user the flexibility to specify which field from the selected test records to download. This last feature will limit the time needed to transfer data without transferring unwanted information.

d) UTAH 2000 HOST Communications

The HOST software shall provide a method by which authorized UTAH 2000 units may contact the HOST and download vehicle-specific and analyzer-specific information as defined in Appendix II. During a normal HOST contact the UTAH 2000 units will transfer the information marked by the *SendToHOST* field in Appendix II. Likewise, the HOST shall return to the UTAH 2000 units that information marked by the *SendByHOST* field in Appendix II. In a second type of HOST contact, the HOST shall send the UTAH 2000 units that information marked by the *RefreshAll* field in Appendix II. It shall be the responsibility of the manufacturer to establish the communications protocol between the HOST and UTAH 2000 and the remote County PCs described in the next section.

4.5 REMOTE COUNTY PERSONAL COMPUTER REQUIREMENTS

a) General

The Remote County Personal Computers (RCPC) will be located in various, but limited number of, County offices. Their main function is to enable restricted County employees access to the CCS to download and print reports on a local printer. In addition, these remote PCs will allow several different levels of access into the HOST computer. These levels are discussed in this section of the specification. The RCPCs must also be capable of downloading subsets of data from the CCS for Ad-Hoc analysis outside the scope of this work.

- b) Hardware Requirements
 - 1. Year 2000 Compliant
 - The RCPCs shall be year 2000 compliant.
 - 2. Processor

The RCPCs shall contain an Intel Pentium II or equivalent 400 MHz processor

or faster, minimum of 32KB Level 1 Cache and minimum of 512KB Level 2 Cache.

3. Operating System

The RCPCs shall have Windows 95 or Windows 98 as the operating system. 4. Memory

The RCPCs shall have, at a minimum, 128MB RAM and expandable to 256MB RAM.

5. Expansion Slots

The RCPCs shall have a minimum of 3 PCI and 2 ISA expansion slots for possible future expansion.

6. I/O Ports

The RCPCs shall have a minimum of

- -One (1) serial port
- -One (1) parallel port
- -One (1) PS/2 port

-Two (2) USB ports

7. Disk Drives

The RCPCs shall have the following disk drives:

- -6GB or larger internal hard disk
- -1.44MB 3.5" floppy drive

-16X or faster CD-ROM drive.

8. Video

The RCPCs shall have an SVGA video display capable of generating 1024x768 resolution with 256 colors. The low glare screen shall have a 16" or greater diagonal viewing area and a .28mm or smaller dot pitch.

9. Multimedia Accessories

The RCPCs shall be equipped with a sound card and speakers for multimedia presentations.

10. Modem

The RCPCs shall have a 56Kbps V.90 or Kflex compatible internal or external analog modem.

11. Backup

The RCPCs shall have a backup system capable of backing up the entire system in an unattended mode.

12. Power Supply

The RCPCs shall have a UL approved 200 watt or greater power supply. The power supply must be attached to a high quality surge suppressor with modem protection.

13. User Inputs

The RCPCs shall have an industry standard 104-key keyboard and a standard mouse with mouse pad.

14. Other Software

-Office productivity suite containing a word processor, spreadsheet and database program such as Microsoft Office Professional or Lotus SmartSuite.

-Virus scanning software

15. Printer

The RCPCs shall have a quality laser printer attached. The printer must be capable of printing 6 pages per minute on 8.5" X 11" paper at 96 characters per line, and six (6) lines per pitch.

c) Software Requirements

The software which communicates these remote PCs to the HOST shall be password protected based on the three different levels of user permissions defined in this section. Upon running the software the user shall be prompted to enter his/her access code and ID. Only authorized access codes and IDs shall be accepted by the software. These access codes and IDs shall be stored in the. remote PCs in such a way that prevents unauthorized modifications. The software shall run of the remote PCs yet shall allow use of the PC in its "normal" (allow usage of other software applications) capacity (i.e., Windows Office B, etc.). The software shall be designed in such a way that the user interface is easy to move about from feature to feature and must be approved by the County. This software shall also monitor the system to determine if a period of more than 20 minutes have gone by without any action. Upon detection of the 20 minute idle the software shall automatically logout or prompt for re-entry of the access code and ID.

SECTION 5 DOCUMENTATION, SERVICE AND WARRANTY

5.1 GENERAL

The following items shall be included with the delivery of each UTAH 2000 unit to the inspection stations:

- a) Instruction manual, securely held in a binder (or other suitable container) made to with stand normal use in the garage environment, and
- b) A copy of the warranty, and
- c) A copy of the annual service agreement, and
- d) A copy of the extended service agreement if purchased, and
- e) Four extra sets of particulate filter elements, and
- f) Special adjustment tools if needed for calibration of the analyzer, the fuel cap tester and any other integral device, if applicable.

5.2 INSTRUCTION MANUAL

The instruction manual accompanying each UTAH 2000 unit shall contain the following minimum information:

- a) Background information describing how vehicular emissions are formed during the combustion process, the general types of controls that are used on vehicles and what negative health impacts can result from vehicle emissions;
- b) Functional diagrams (mechanical and electrical);
- c) Accessories and options (included and/or available);
- d) Model number and identification markings and locations;
- e) Maintenance procedures and frequencies recommended by the manufacturer. The services that should be performed only by the manufacturer shall be clearly identified;
- f) Gas calibration/leak check procedures as well as calibration procedures for the fuel cap tester and any other integral device;
- g) Brief description with a subject index of the inspection/test procedures as they pertain to the EIS prompts;
- h) Brief description of emission analyzer operating principles;
- i) Information provided shall include a listing of warranty repair stations by name, address and phone number; and
- j) Name, address and phone number of the manufacturer's representative in charge of sales and service personnel for the company in Utah.

5.3 UTAH 2000 WARRANTY AND SERVICE MAINTENANCE CONTRACT

a) The cost of the UTAH 2000 shall include a one-year, transferable warranty covering parts and labor. In addition, at the time of original sale, the manufacturer shall offer optional yearly extended warranties to be in addition to cost of the EIS.

1

Warranty provisions protecting the interest of the buyer shall include:

- 1. Location, phone number and address of the repair centers throughout the state. These shall be an adequate number of qualified repair technicians and an adequate number of repair locations conveniently located to efficiently and promptly meet statewide service needs. The response time established by the manufacturer may be longer for a lower purchase price or shorter if the price is higher. All response time and cost provisions shall be clearly indicated in the warranty provisions.
- 2. Name of the manufacturer's representative closest to each franchised service center if not a factory service center.
- 3. Coverage of at least all of the hardware and software contained inside the tamper resistant analyzer cabinet, the computer keyboard and monitor and the fuel cap tester. A description of specific parts and labor covered by the provisions of the warranty shall be provided to the purchaser. In addition, the warranty shall itemize the cost of labor which are not covered by the warranty.

To ensure that purchasers are properly notified regarding the cost and provisions of the warranty, the UTAH 2000 units shall not be delivered until a copy of the warranty has been signed by the purchaser and a company representative. Service response time and loaner provisions shall be initialed by the purchaser. A copy of the signed warranty shall be provided to the purchaser and a copy filed by the company.

b) The manufacturer shall make available an annual service contract covering, as a minimum, all of the items provided with the analyzer and the fuel cap tester.

Service contract provisions protecting the interests of the buyer shall include:

- 1. The necessary level of service to ensure that the UTAH 2000 unit functions properly within the operating conditions listed in this specification. Such items as filters, disk drive cleaning and alignment, analyzer bench service, and pump maintenance are typical service maintenance items.
- 2. The manufacturer is responsible for specifying the frequency of performance.
- 3. The manufacturer shall include in the annual service/maintenance contract the cost of making the necessary software changes.

- 4. The manufacturer or his sales representative must notify the County of the cost for this service as a condition of certification and include projected increases.
- 5. The information in Items 1 4 above must also be made available to the potential buyer of a UTAH 2000 before purchase or lease.
- c) The following provisions apply to both the warranty and service maintenance contract:
 - 1. Any change to the warranty or service contract must be approved by the County.
 - 2. The UTAH 2000 owner shall be provided a cost estimate prior to the performance of any service or maintenance unless the work is covered by the warranty or service contract. Regardless of whether or not the work is covered by the warranty or service contract, the owner shall be provided a detailed description of the work performed when the job is completed. In addition, the manufacturer shall include a toll-free telephone number for the owner of the analyzer to call if he/she wants to complain about the work performed, the courtesy or competency of the manufacturer's technician or any other aspect of the warranty or service contract.
 - 3. Manufacturers shall provide a station with a loaner UTAH 2000 unit if the station's EIS is out of service for more than 3 "normal work days". Loaner units shall be calibrated, provided with new filters, and shall contain the latest version of I/M testing software.

5.4 SPARE PARTS

The UTAH 2000 manufacturer shall maintain an adequate supply of spare parts and accessories to fulfill the service requirements of the warranty or service contract. The manufacturer shall, at the time of delivery, supply the purchaser with four sets of filters, at least 500 sheets of paper and one calibration gas cylinder.

5.5 SERVICE CENTERS

The EIS manufacturer shall provide or contract for warranty or service contract repairs for all UTAH 2000 units. The service center(s) shall be located in or within a "reasonable driving distance" of the participating Counties. The term/word "reasonable driving distance" is defined here as the distance which allows all service repairs to be conducted in the time frame allowed in the specification.

5.6 WORKMANSHIP

Each manufacturer, or his agent, shall guarantee the repairs made for a period of 90 days. The manufacturer shall ultimately be held responsible, regardless if an agent performed the repairs.

5-3

1

5.7 NONCOMPLIANCE WITH ANY PORTION OF THE SPECIFICATIONS

The term of the UTAH 2000 Approval shall begin on the date of issuance and terminate upon completion of the UTAH 2000 I/M Program or may be conditionally revoked if the County determines at any time during the coarse of the UTAH 2000 I/M program that the EIS units do not fully comply with all portions of these specifications and/or any of the following conditions exist:

- a) Software updates are not performed within the time frame agreed upon by the County and manufacturer or do not meet the requirements specified by the County;
- b) UTAH 2000 units in the field are found to be in violation of the specification and the manufacturer is unwilling to resolve the matter either in the time frame requested by the County or in a way that is satisfactory to the County.

Revocation of the manufacturer's approval may be limited to future sales of UTAH 2000 units. In addition, existing EIS units which do not conform to specifications may be locked out until they are brought into compliance. If identified problems are not corrected within the time specified by the County, the UTAH 2000 Approval may be permanently revoked. If a approval is conditionally or permanently revoked, the County will notify all licensed stations and representatives of the repair industry that the County will no longer license new stations purchasing affected UTAH 2000 units.

. . .

APPENDIX B FEE SCHEDULE

The assessed fees for implementing the requirements of Section 6.10 of the Vehicle Emissions Inspection/Maintenance Program shall be:

Permitting of an Official I/M Program Station
Annual Renewal of Station Permit
. Annual Renewal of Expired Station Permit
Repermitting an I/M Station at a New Location
Temporary I/M Station Permit
Mechanic Certification Course
Tamper Detection Class Mandatory
Tamper Detection Class Voluntary
Permitting an Official I/M Emissions Mechanic
Annual Renewal of I/M Mechanic Permit
Annual Renewal of Expired Mechanic Permit
Emissions Certificate of Compliance/Number
Duplicate Certificate of Compliance
Duplicate Vehicle Inspection Report (VIR)
Emissions Inspection Fee

Effective Date June 10, 2003

(Applies to all Permit renewals and emissions tests that take place or are due to take place after June 30, 2003).

B-1

APPENDIX C

UTAH COUNTY EMISSION STANDARDS <u>CUTPOINTS</u>

MOTOR VEHICLE EMISSIONS INSPECTION/MAINTENANCE PROGRAM

The following schedule gives the maximum allowable concentrations for carbon monoxide (CO) and hydrocarbons (HC) for both cars and trucks as determined by an approved infrared gas analyzer using the prescribed procedures. The effective date for these cutpoints is JULY 1, 1991.

ALL PASSENGER VEHICLES 1978 AND OLDER LIGHT DUTY TRUCKS 6,000 POUNDS GVWR OR LESS 1979 TRUCKS AND NEWER 8,500 POUNDS GVWR OR LESS <u>MAXIMUM CONCENTRATION STANDARDS</u>

MODEL YEAR	PERCENT CARBON MONOXIDE	PARTS/MILLION HYDROCARBONS	
1968-1969	6.0	800	
1970-1974	5.0	700	
1975-1976	4.0	60 0	
1977-1979	3.0	500	
1980	2.0	300	
1981 and newer	1.2	220	

HEAVY DUTY TRUCKS AND VANS 1978 AND OLDER 6,001 AND OVER GVWR 1979 AND NEWER OVER 8,500 POUNDS GVWR MAXIMUM CONCENTRATION STANDARDS

1968-1969	7.0	1500
1970-1978	5.0	1200
1979-1980	4.0	1000
1981 and newer	3.5	800

The minimum dilution factor must also be reached as part of the testing requirement. The dilution factor is contained in the analyzer specifications in Appendix A and is updated as deemed necessary.

NOTE: These should be considered as "cutpoints" for maximum allowable emissions levels. Vehicles must never be reset to these emission levels when readjustments are made, but rather shall be adjusted using manufacturer's specifications. By using manufacturer's specifications, the emissions levels should be well below the "cutpoints."

APPENDIX D

PENALTY SCHEDULE

VIOLATION	1 st Occurrence	2 nd Occurrence	3 rd Occurrence	4 th Occurrence
Inaccurate incomplete data or performing unauthorized repairs	Formal warning (Station & Mechanic)	Up to 1-month suspension (Station & Mechanic)	Up to 3-month suspension (Station & Mechanic)	Revocation (Station & Mechanic)
Fail Passing Vehicle	15 day to 3-month suspension (Station & Mechanic)	30 day to 6-month suspension (Station & Mechanic)	60 day to 6- month suspension (Station & Mechanic)	Revocation (Station & Mechanic)
Pass failed vehicle recording pass for tampering on tampered vehicle	15 day to 3-month suspension (Station & Mechanic)	30 day to 6-month suspension (Station & Mechanic)	60 day to 6- month suspension (Station & Mechanic)	Revocation (Station & Mechanic)
Non-certified I/M Mechanic	30 day suspension (Station) 45 day suspension (Mechanic)	60 day suspension (Station) 90 day suspension (Mechanic)	90 day suspension (Station) 120 day suspension (Mechanic)	Revocation (Station)
Intentional pass of a failing vehicle	30 day suspension (Station) 180 day suspension (Mechanic)	90 day suspension (Station) Revocation (Mechanic)	Revocation (Station & Mechanic)	
Failure to Inspect	30 day to 6-month suspension (Station) Revocation (Mechanic)	Revocation (Station)		

Mechanic and station suspensions may be reduced in length by a Negotiated Consent Agreement which may substitute monetary penalties for part or all of the suspension time. Consent Agreements for stations are based on 50% of the testing revenue that could have been expected during the suspension time for a third occurrence, 35% for a second occurrence and 10% for a first occurrence, based on a minimum test fee of \$20.00. Consent Agreements for the technicians are based on \$100 increments for any 15 day period or portion thereof up to a maximum of 75 days.

Any finding of inspector incompetence associated with any violation in the Penalty Schedule shall result in mandatory training offered by the department or approved substitute before inspection privileges are restored.

APPENDIX E

UTAH COUNTY EMISSION STANDARDS WAIVER CUTPOINTS

In order for a waiver to be granted, the subject vehicle must first qualify by not exceeding the following maximum allowable concentrations for carbon monoxide (CO) for both cars and trucks as determined by an approved infrared gas analyzer using the prescribed procedures.

ALL PASSENGER VEHICLES 1978 AND OLDER LIGHT DUTY TRUCKS 6,000 POUNDS GVWR OR LESS 1979 TRUCKS AND NEWER 8,500 POUNDS GVWR OR LESS <u>MAXIMUM CONCENTRATION STANDARDS</u>

MODEL YEAR	PERCENT
	CARBON MONOXIDE
1969-OLDER	7.0
1970-1974	6.0
1975-1976	5.0
1977-1979	4.0
1980	3.0
1981-NEWER	2.0
HEAVY DUTY HEAVY DUTY 1978 AND OLDER 6,001 1979 AND NEWER O MAXIMUM CONCE	TRUCKS AND VANS POUNDS AND OVER GVWR VER 8,500 POUNDS GVWR INTRATION STANDARDS
198°B/149	4.05.07.08.0

The minimum dilution factor must also be reached as part of the testing requirement. The dilution factor is contained in the analyzer specifications in Appendix A and is updated as deemed necessary.

NOTE: These should be considered as "cutpoints" for maximum allowable emissions levels. Vehicles must never be reset to these emission levels when readjustments are made, but rather shall be adjusted using manufacturer's specifications. By using manufacturer's specifications, the emissions levels should be well below the "cutpoints".

E-1

- - . .

APPENDIX F

OBD TEST PROCEDURES

The following test procedure is to be followed for 1996 model year vehicles or newer:

1. Turn the ignition switch to the off position for at least 12 seconds.

2. Visually examine the instrument panel to determine if the malfunction indicator light (MIL) illuminates when the ignition key is turned to the key on/engine off position. If the MIL does not illuminate at all then the vehicle fails if tested after December 31, 2000, perform two speed idle test if tested prior to January 1, 2001 and advise the owner/operator to repair the MIL problem. Exception: OBD deficient vehicles will be given the two speed idle test.

3. Locate DLC and connect test equipment. If DLC is missing, has been tampered with, or is otherwise inoperable then the vehicle fails if tested after December 31, 2000, perform two speed idle test if tested prior to January 1, 2001 and advise the owner/operator to repair the DLC. Exception: OBD deficient vehicles will be given the two speed idle test.

4. Start and leave the engine running. Check for MIL illumination. If the MIL illuminates after the engine has been started, even if no fault codes are present, the vehicle fails if tested after December 31, 2000, perform two speed idle test if tested prior to January 1, 2001 and advise the owner/operator to repair the MIL problem. Exception: OBD deficient vehicles will be given the two speed idle test.

5. Vehicle status not ready; perform two speed idle test prior to January 1, 2001, after December 31, 2000 the vehicle must be preconditioned to a ready status. Exception: OBD deficient vehicles will be given the two speed idle test.

6. Vehicle status ready; check DTC's, determine pass or fail, and record results.

7. Turn off engine and disconnect test equipment.

UITH COUNTY

Page Z el S

Ton: Facily Services To: RALPH CLYDE

Den: 1/20105 Time: 11:10:27 ORDINANCE NUMBER 1994- 106

SHORT TITLE:

1

AN ORDINANCE ENACTING SECTION 9.32.180 IN THE PROVO CITY ORDINANCES MAKING IT UNLAWFUL TO OPERATE A MOTOR VEHICLE IN PROVO CITY FOR AN AGGREGATE OF MORE THAN SIXTY DAYS EACH YEAR WITHOUT FIRST OBTAINING AN ANNUAL VEHICLE EMISSION INSPECTION MAINTENANCE TEST.

I

PASSAGE BY MUNICIPAL COUNCIL

ROLL CALL

DISTRICT	NAME	MOTION	SECOND	FOR	AGAINST	OTHER
N WEST	DAVID L RAIL				_ L.	
NEAST .	JANE L. CARLILE		÷ .	~		
Tz/	JAMES H. DALEY				J.	
S WEST	GREGORY A. HUDNALL					
CU I	KARL J. THALMAN			••		
CN 11	SHARI C. HOLWEG				<u></u>	
CD 111	DENNIS R. HALL			-		
	· · ·		TOTALS.	4	3	

	This ordinance was	passed by the Munic	tipal Council of Prove City, on the	~ <u>v</u>	_day
of	la samelen	, 1994 on a roll c	all vote as described above. Signed this	ريام ا	_day
o[]	N. S. for her	1994.	Dovid L. Ros	· D	

Chairman '

П

APPROVAL BY MAYOR

This ordinance is approved by me this	12 the day of DEZmbar.	_1994.
X, Part D, App. 4.1	AD 515	•••
••••••••••••••••••••••••••••••••••••••	Mayor	X, Part D, App 3.1

: Padily Services To: RALPH CLYDE

Dana: 1/34/85 Time: 11:10:55

Page S of S

ORDINANCE NUMBER 1994 -106

CITY RECORDER'S CERTIFICATE AND ATTEST

This ordinance was recorded in the office of the Provo City Recorder on the <u>19</u> day of <u>fluenclus</u> 1994, with a short summary being published in the <u>16</u> day of <u>illeenclus</u> 1994, in <u>The Daily Herald</u>, a newspaper sublished in Provo, Utah. I hereby certify and attest that the foregoing constitute a true and accurate record of proceedings with respect to Ordinance Number <u>1994-106</u>.



Signed this 19th day of Accentur 1994.

Julio 7. Perry

X, Part D, App 3.2

. 01/2	26/95	13:24	B8 01 3	70 8709	UTAL	COUNTY	DAQ	Ø 005/006
Prom: Pari	iily Sardoo	To:RALPHC AN ORI ORDINA VEHICL DAYS EL	LIDE DINANCE NCES E IN PRO ACH YEA	E ENACTI MAKING IVO CITY AR WITHC	Dam: 1/20/06 Tam ING SECTION IT UNLAWF FOR AN AGG JUT FIRST OB	E 11:11:17 1 9.32.180 FUL TO C REGATE C TAINING A E TEST.	IN THE PROVO OPERATE A MO OF MORE THAN S AN ANNUAL VEH	CITY CITY TOR IXTY ICLE
8 9 10 11 12 13 14 15 16 17 18 19	genera who o an ani pursua employ	WHERE WHERE I health, s perate a ve nual state int to Unit WHERE yees with	AS, state : AS, Prove afery, and chicle in I or count ed States AS, to pro this provi	and federa o City desi I welfare a Provo City y motor v Environme mote comp sion of the	l laws impose res to comply nd improve air for more than chicle cmissio mtal Protection pliance by non-	air quality s with these r quality by an aggrega n~inspection Agency air resident store ity desires t	tandards on Provo equirements and to requiring motor ve te of sixty days per n maintenance test, r quality regulations lents, and long-term o require owners o	City; and promote the hicle owners year to pass , established ; and i visitors and f vehicles to
20 21 22 23 24 25 26	provid authori follow	e proof of WHEREA ity to enac NOW TH s:	complian AS, Utah a and enfo IEREFOR	ce with thi Code § 19 prec ordina E, be it o	is section; and 9-2-12] grants unces to contro rdained by the	political su air pollutio Municipal	bdivisions of the s on; Council of Provo C	tate of Uiah
29 30 31 32 33 34 35 36	PART shall b PART]: Section 9. e as descri]]:	.32.180 of ihed in the	the Provo) City Ordinand exhibit.	es is hereby	y enacted, the conte	nis of which
37 38 39 40 41	END C	This ordin	aanæ shal <u>IANCE</u> .	take effe	ct January 1. 1	995.		
5								

X, Part D, App 3.3

•

۰.

-

.

Page 2 of 2

43 9.32.180. Motor Vehicle Emissions Tests Required.

44 (1) It shall be unlawful, and a class C misdemeanor, for the owner or operator of a motor vehicle which has not passed an annual state or county motor vehicle emission inspection 45 maintenance test, established pursuant to United States Environmental Protection Agency air 46 47 quality regulations, within the last year, to operate or permit another to operate the motor 48 vehicle within the municipal limits of Provo City for an aggregate total of more than sixty (60) 49 days per any calendar year without obtaining a Utah County Vchicle Emission Inspection 50 Maintenance Program test certificate valid for the date the motor vehicle is being operated within the municipal limits of Provo City. 51

52 (2) For purposes of this Section, any motor vehicle expressly exempt from obtaining a 53 vehicle emission inspection maintenance test under the Utah County Vehicle Emission Inspection 54 Maintenance Program because of the vehicle's use, engine type, or age, etc., shall also be 55 exempt from the requirements of this Section.

(3) Upon request of a Provo City police officer, the operator of a motor vehicle shall
provide evidence (i) that the vehicle has passed a vehicle inspection maintenance test in
compliance with the requirements of subsection (1) above; (ii) is exempt pursuant to subsection
(2) above; or (iii) is not being operated in Provo City for an aggregate total of more than sixty
(60) days per any calendar year. Failure to provide such evidence shall constitute a prima facie
case that the operator is in violation of this Section, however, a charge brought pursuant to this
Section shall be dismissed by the court if a person charged with violating this Section provides
the court with proof of compliance with this Section.

(4) The evidence of compliance required by subsection (3) may include, but shall not be 64 limited to, a current emission test certificate or motor vehicle registration indicating the vehicle 65 is in compliance with a state or county motor vehicle emission inspection maintenance program 66 established pursuant to United States Environmental Protection Agency air quality regulations, 67 evidence that shows that due to of the age of the vehicle, its type of engine, etc., that the vehicle 68 is exempt from obtaining a vehicle emission inspection maintenance test under the Utah County 69 Vehicle Emission Inspection Maintenance Program, or evidence that the vehicle is registered in 70 71 another county or state and is not being operated in Provo for more than an aggregate total of 72 more than sixty (60) days each calendar year.



3

A Station Performance Report is to be filled out during each audit. Violations are to be clearly marked and the owner or responsible individual informed. A copy of the station performance report is submitted to the Director.

1.5

.

÷.,

If the analyzer is warmed up, a gas audit of the station's analyzer shall be performed by the station compliance officer using Department span gas. If the analyzer fails the gas audit, this is to be noted on the station performance report. The analyzer will be automatically locked out of operation until a successful gas calibration is performed.

All items covered on the station performance report are to be checked on each audit. Of particular concern is certificate accountability. All certificates must be appropriately accounted for. It is the responsibility of the station compliance officer to see that all certificates are properly accounted for. -- Any certificates unaccounted for are to be referred to the Director upon return to the office.

During each routine audit the analyzer should have the data disk exchanged for a blank disk provided by the Department.

Where possible, the compliance officer should attempt to observe an actual emissions test during the station audit. (this test may be used for the annual mechanic recertification hands-on requirement for the mechanic that performed the test) Also, wehicles that have already been tested or are waiting to be tested whould be checked for tampering violations.

Covert Audits:

1. To the extent possible a covert audit will be performed of each inspector and station at least once a year. The number of covert audits at least equals the number of permitted inspectors. Covert audits will be performed using a variety of vehicles that are representative of the subject fleet that are set to fail across a full range of malfunctions. Suspected problem stations and inspectors are targeted for earlier and more frequent audits. Complaints also trigger additional audits.

2. Remote visual observations of inspector performance will be conducted at least once per year per inspector in high volume stations.

3. At least one covert audit including the purchase of repairs and subsequent retesting if the vehicle initially fails the tailpipe emissions, on an FTE basis as indicated in the SIP.

4. Vehicles used for covert audits will be pre-tested and the emissions levels recorded prior to the vehicle being used for covert work.

5. The vehicle operator must follow the guidelines in memorandum dated 20 March 1986 from deputy County Attorney regarding __Guidelines to Avoiding Entrapment*. Reference Appendix A.

The audit should be performed in such a manner as to avoid giving employees at the station being audited any indication that an audit is being performed.

The Surveillance Report form should be filled out completely for each phase of the audit.

8. The vehicle will be retested for tail pipe emissions and tampering at the Technical Center any time the vehicle is passed at one of the I/M stations. The results of this post covert audit test will be recorded.

9. A complete covert audit includes the following phases:

- Pretesting and adjusting of vehicle. Audit/testing at designated station. a.
- b.
- Post testing and readjusting original configuration if C. needed:
- A briefing with the Director or his representative on the đ. results.
- Appropriate action taken on any discrepancies.

Complaints:

When a complaint is received that requires follow up by Air Quality Programs personnel, that complaint is to be recorded on a complaint record form by the person who receives the complaint. The complaint record form is then passed on to the appropriate person for response. The person responding to the complaint will record the results of the investigation on the complaint form.

Initial contacts to resolve the complaint should be made as soon as possible, preferably within 2 working days.

Following the resolution of the complaint to the satisfaction of the investigator, the complainant is to be contacted and notified of the results of the investigation.

Following notification of the complainant, the complaint record form is to be given to the program supervisor. Following his review, the complaint is to be filed by the secretary.

Whenever possible complainants shall be kept anonymous.

Analyzer Disks:

The analyzer information disks retrieved during the station audits are to be returned to the office and should be edited and transferred to the county mainframe on a timely basis. The disk should then be saved for future reference or until it has been determined that the information has been successfully loaded to the mainframe.

Disks that have been successfully loaded will then be erased and reformatted. Reformatted disks will be made available for personnel to reuse. When the disk is erased, previous label

SUKAETERINGE KERANT LAN

VEHICLE EMISSIONS INSPECTION/MAINTENANCE PROGRAM CITY-COUNTY HEALTH DEPARTMENT OF UTAH COUNTY

LN:		MAKE:	YR :	LIC	:	
VWR :	_ CUTPOIN	TS: CO		ŧ HC		PPM
ATALYTIC CONVERT	ER PASS	FAIL	NA	COMMENTS		
IR INJECTION SYS	. PASS	FAIL	NA	COMMENTS		_
JEL NECK RESTRIC	TOR PASS	FAIL	NA	COMMENTS		
EAD PLUMBTESMO	PASS	FAIL	NA	_ COMMENTS		·
ECHNICAL CENTER	READINGS	DATE	J/	TIME	MECH	
2500 RPM CO	н	с	IDLE (0	НС	
P	F	P 1	5	P	P P	F
TATION TEST RESU	LTS	DATE	_//_	TIME	MECH #	
IN	ITIAL TEST	STAT	MECI	H #		
2500 RPM CO	н	С	IDLE (CO	HC	
T T	F	P F		P	F P	F
AFTED ADTIC	TMENTE (TE	DENITOEN) CTAT #	MEX-11		
TI ICA AUJUS	112012 (1F	KEQUIKED.	JINI T	FIECH	₩	
2500 554 60		-				
2500 RPM CO	H		IDLE (CO		
	-			• ·		•
JKE	AFTER AD	TENTE	riming	TATES		
4041100	AFIER AD	J2112		INITIAL	AFTER AD	JSMTS
WELL	/					
INITIAL	AFTER AD	JSMTS	AIR/FUEL	ADJUSTMEN	T PROCEDURI	E
DLE RPM	1		ADJSMT CO	OST	ACTUAL CO	OST
INITIAL	AFTER AD	JSMTS				
ESTING PROCEDURE	(IF OPERD	VEDI				
	(11 OBSER	VED)				
. EXHAUST CHECK		7	PRECO	NDITION EN	GINE	ا الم تعرفي ا
. POLLUTION CON	TROL CHECK		OR 250	00 RPM TES	r	
A) LEAD PLUMB	TESMO TEST	B .	RECORI	CORRECT	READINGS	
• VERIFY VIN FR	OM VEHICLE	9	INFORM	VEHICLE	owner	<u>`</u>
. ENGINE TEMP.	CHECK	10). COMPLE	ETE FORMS	PROPERLY	
- ZERO/SPAN ANA	LYZER	12	L. USES R	REFERENCE	MANUALS	
. ACCESSORIES OF	FF	1:	. USES /	PPROPRIAT	E TUNE	
				U	P TOOLS	
ECHNICAL CENTER	READINGS ·	DATE	//_	TIME	MECH	
2500 RPM		нс <u></u>		ະ ແມ	 	7
		•	•	-	- 4	•
FMARKS /DICOPERAN	CIES					
Linkasj discrepan			•			
·						

X, Part D, App. 3.3

SIGNATURE

vehicle, whether or not for financial or other incentive, the Health Department Investigator is free to follow the lead of the test operator.

. . .

AUTHORITIES:

. . .

State_vs. Sprague, ±80 P.2d 404

Utah Code Annotated, Section 76-2-303



-2-

BOARD OF COUNTY COMMISSIONERS OF UTAH COUNTY

í

۰.

COUNTY ORDINANCE AMENDMENT

REMOTE SENSING PROGRAM

Adopted by the Board of County Commissioners of Utah County

April 30, 1997

Under Authority of Section 41-6-163.6 (a)(ii) Utah Code Annotated, 1953, as amended

CERTIFIED OFFICIAL COPY BOARD OF COUNTY COMMISSIONERS OF UTAH COUNTY

By

David J. Gardner, Chairman Utah County Board of Commissioners

Attest

Arlin V. Kuhni, County Clerk

REMOTE SENSING PROGRAM

۰.

INDEX

÷.

SECTIONS

1.0	Definitions
2.0	Purpose
3.0	Jurisdiction of the Department
4.0	Powers and Duties
5.0	Scope
6.0	General Provisions
7.0	Emission Standards
8.0	Penalty
9.0	Severability
10.0	Effective Date

APPENDICES

Appendix A Cutpoints

-i-

X, Part B, App. 4.2

1.0 DEFINITIONS.

For the purpose of this Ordinance, the following terms, phrases, and words shall have the following meanings, unless otherwise defined:

- 1.1 Certificate: A Certificate of Compliance with the Vehicle Emission Inspection Maintenance Program.
- 1.2 Certificate of Compliance: A document used in the Vehicle Emissions Inspection/Maintenance Program to certify that a vehicle meets all applicable requirements of the Program.
- 1.3 CO: Carbon Monoxide.
- 1.4 County: Utah County, Utah.
- 1.5 Cutpoints: The maximum allowable concentrations of carbon monoxide (CO) for a given weight class and model year of vehicle, as determined by the Department, using an approved infrared exhaust gas analyzer.
- 1.6 Department: The Utah County Health Department.
- 1.7 Director: The Director of the Utah County Health Department or his authorized representative.
 - 1.8 Gross Emitter: A vehicle with emission levels greater than those established in Appendix A of this Ordinance; those established by Ordinance 1996-19; or those established by the Department for I/M240, ASM, or other testing.
 - 1.9 I/M Test: An official vehicle emissions test performed for the purpose of determining if the vehicle meets prescribed emission levels and/or issuing a Certificate of Compliance or Waiver.
 - 1.10 Owner of Record: The individual, organization, corporation, or any other entity listed with the Utah State Tax Commission Department of Motor Vehicles as the most recent owner of a vehicle.
 - 1.11 Remote Sensing Device (RSD): Equipment capable of measuring vehicle exhaust emissions as the vehicle is driven past the equipment and can be utilized for on-highway testing.
 - 1.12 Vehicle: A self-propelled motorized vehicle with an internal combustion powered engine which is required to be licensed for operation on public roads and/or streets.

-1-

- 1.13 Vehicle Emissions Inspection/Maintenance Program: The program established by the Department pursuant to Section 41-6-163.6, Utah Code Annotated 1953, as amended.
- 1.14 Waiver: A document used to verify that a vehicle has met the repair or adjustment requirements of the I/M Program Ordinance and a specific emission standards has been met.

2.0 PURPOSE.

It is the purpose of this Ordinance to reduce air pollution levels by requiring periodic inspections of on-road, in-use motor vehicles and by requiring emission related repairs and/or adjustments for those vehicles that fail to meet prescribed standards so as to:

- 2.1 Protect and promote public health, safety and welfare;
- 2.2 Improve air quality;
- 2.3 Comply with federal regulations contained in the Clean Air Act Amendments of 1990, PL 101-549; and
- 2.4 Satisfy the requirements of the Section 348 National Highway System Designation Act of 1995 submittal of the Utah State Implementation Plan.

3.0 JURISDICTION OF THE DEPARTMENT.

All aspects of the Remote Sensing Ordinance within Utah County enumerated in Section 2.0 shall be subject to the direction and control of the Department.

4.0 POWERS AND DUTIES.

- 4.1 The Department, by the Director, shall be responsible for the enforcement and administration of this Ordinance and any other powers vested in it by law and shall:
 - 4.1.1 Make policies and procedures necessary to ensure that the provisions of this Ordinance are met and that the purposes of this Ordinance are accomplished.
 - 4.1.2 Randomly sample the vehicle fleet operating within Utah County for exhaust emissions by use of a Remote Sensing Device.
 - 4.1.3 Establish cutpoints for the vehicle fleet operating within Utah County that will identify gross emitters of vehicle exhaust pollutants.

- 4.1.4 Require follow-up inspections to vehicles in violation of established cutpoints in accordance with Section 6.0 of this Ordinance.
- 4.1.5 Require vehicles in violation of this Ordinance to be brought into compliance with this Ordinance by making the proper repairs and/or adjustments.

5.0 SCOPE.

It shall be unlawful for any person to fail to comply with any policy, procedure, or regulation promulgated by the Department, unless expressly waived by this Ordinance.

6.0 GENERAL PROVISIONS.

- 6.1 Beginning May 15, 1997, any vehicle that has been operating or is operated in Utah County and is identified with an RSD as a gross emitter twice in a running twelve month period, including vehicles which are otherwise exempt from emission testing, may voluntarily receive a standard two speed idle I/M test, an I/M 240 test, and other related emissions tests, from the Department.
 - 6.1.1 Vehicle owners will be encouraged to repair the vehicle if emissions problems are detected by the additional emissions testing specified in Section 6.1.
 - 6.1.2 Vehicles of 1967 model year or older will be tested to 1968 model year standards.
- 6.2 Beginning August 1, 1997, any vehicle that has been operating or is operated in Utah County and is identified with an RSD as a gross emitter at least twice in a running twelve month period, must receive a standard two speed idle I/M test, an I/M 240 test, and additional emissions testing as applicable from the Department. Vehicles determined by the additional testing to be gross emitters shall be repaired as required by the Department.

6.2.1 Vehicles of 1967 model year or older will be tested to 1968 model year standards.

- 6.3 It is the responsibility of the Department to notify in writing the owner of record of any vehicle identified as a gross emitter by the RSD.
- 6.4 Upon written notification from the Department, it will be the responsibility of the vehicle owner to complete the required I/M testing within 30 days of the date of the written notification.
- 6.5 The following vehicles are exempt from the requirements of this section:
 - a) Motorcycles

X. Part D. App. 4.5

b) Pickup trucks qualifying for registration as a farm truck under Section 41-1a-102, UCA 1953 as amended.

7.0 EMISSIONS STANDARDS FOR ON-ROAD MOTOR VEHICLES EXHAUST GASES.

- 7.1 Maximum concentration for on-road cutpoints shall be determined by the Director as needed, to meet the National Ambient Air Quality Standards established by the Environmental Protection Agency. The established on-road cutpoints shall remain in effect until changed by order of the Director. Any change in on-road cutpoints shall be effective upon the first day of any calendar month designated by the Director. The Director shall establish on-road cutpoints by considering the following factors:
 - 7.1.1 The existing ambient air quality;
 - 7.1.2 The requirements for air quality currently in effect as promulgated by the Environmental Protection Agency, the Utah State Department of Environmental Quality, and the Department. The on-road cutpoints established shall be part of an overall progress in accordance with EPA guidelines to achieve the required tailpipe reduction of CO from vehicles measured from the date this program is implemented;
 - 7.1.3 The general level of emission control technology on vehicles registered in Utah County;
 - 7.1.4 Population growth and other factors which may reasonably be expected to impact CO concentrations in the atmosphere;
 - 7.1.5 The likelihood of a particular cutpoint to achieve desired air quality goals.
- 7.2 Upon determining the appropriate cutpoints, the Director shall cause notice thereof to be issued to the public by publication at least once in a newspaper of general circulation in the county at least thirty days prior to the effective date. Such notice shall indicate that written comment on the proposed on-road cutpoint levels will be received by the Director until fifteen days prior to the effective date of the on-road cutpoints. The Director shall consider any written comment timely submitted and, should cause appear, may alter or suspend the proposed on-road cutpoints as appropriate. Otherwise, the proposed on-road cutpoints shall take effect on the date determined by the Director.
 - 7.2.1 The on-road cutpoints are referenced in Appendix A to this Ordinance and may be amended pursuant to the provisions of Section 8.0 of this Ordinance.

-4-

- 8.0 PENALTY.

....

، ۲

- 8.1 Any person who violates any of the provisions of this Ordinance, either by failing to do those acts required herein or by doing a prohibited act, shall be guilty of a Class B Misdemeanor, punishable by a fine of less than one thousand dollars (\$1,000.00) or by imprisonment not to exceed six (6) months or by both such fine and imprisonment at the discretion of the court.
- 8.2 Each day such violation is committed or permitted to continue shall constitute a separate violation.
- 8.3 The County Attorney may initiate legal action, civil or criminal, requested by the Department, to abate any condition that exists in violation of this Ordinance.
- 8.4 In addition to other penalties imposed by a court of competent jurisdiction, any person(s) found guilty of violating any part of this Ordinance shall be liable for all expenses incurred by the Department.
- 8.5 Failure to comply with this Ordinance may also result in the suspension or revocation of the vehicle registration.

9.0 SEVERABILITY.

If any provision, clause, sentence, or paragraph of this Ordinance or the application thereof to any person or circumstances shall be held to be invalid, such invalidity shall not affect the other provisions or applications of this Ordinance. The valid part of any clause, sentence, or paragraph of this Ordinance shall be given independence from the invalid provisions or application and to this end the provisions of this Ordinance are hereby declared to be severable.

10.0 EFFECTIVE DATE.

This ordinance shall become effective fifteen days after its passage and upon at least one publication in a newspaper published in and having general circulation in Utah County. The Appendices become effective on the specific date found on each of the appendices.

-5-

APPENDIX A

UTAH COUNTY EMISSION STANDARDS CUIPOINTS

REMOTE SENSING PROGRAM

The following schedule gives the maximum allowable concentrations for vehicle exhaust pollutants for both cars and trucks as determined by an approved infrared remote sensing device. The effective date for these cutpoints is July 1, 1997.

ALL PASSENGER VEHICLES 1978 AND OLDER LIGHT DUTY TRUCKS 6,000 POUNDS GVWR OR LESS 1979 TRUCKS AND NEWER 8,500 POUNDS GVWR OR LESS MAXIMUM CONCENTRATION STANDARDS

	PERCENT	PPM
MODEL YEAR	CARBON MONOXIDE	HYDROCARBONS
All	5.0%	N/A
	HEAVY DUTY TRUCKS AND V	ANS
	1978 AND OLDER 6,001 AND OVER	GVWR
	1979 AND NEWER OVER 8,500 POUNI	DS GVWR
	MAXIMUM CONCENTRATION STAN	NDARDS
ILA	5.0%	N/A

EFFECTIVE DATE: May 15, 1997

X, Part 2, App. 4.8

WEBER-MORGAN HEALTH DEPARTMENT

1

1

Regulation for

MOTOR VEHICLE INSPECTION AND MAINTENANCE PROGRAM

Adopted by the Weber-Morgan Board of Health

May 22, 2000

Amended May 12, 2003

Under Authority of Section 26A-1-121, 41-6-163.6 and 41-6-163.7 Utah Code Annotated, 1953, as amended

TABLE OF CONTENTS

1.0	Page Title and Definitions1
2.0	Purpose
3.0	Authority Jurisdiction
4.0	Powers and Duties9
5.0	Scope
6.0	General Provisions
7.0	Standards and Specifications for Exhaust Gas Analyzers and Span Gases17
8.0	Permit Requirements of the Vehicle Emissions Inspection and Maintenance Program Station20
9.0	Inspection Procedures25
10.0	Adjustment Procedures
11.0	Engine Switching
12.0	Emission Repair Technician / Emission Tester Certificate of Qualification
13.0	Emissions Standards for Motor Vehicles
14.0	Certificates of Compliance and Certificates of Compliance - Waiver
15.0	Right of Appeal41
16.0	Penalty41
17.0	Severability
18.0	Effective Date42
Appendix A - Utah 2000_Analyzer Specifications A-1	
Appendix B - Fee Schedule A-2	
Appendix C - Weber County Emission Standards (Cut points) A-3	
Appendix D - Penalty Schedule A-4	
Appendix E - Approved Calibration Gas and Calibration Gas Blenders A-5	
Appendix F - OBD IM Inspection Procedures	

1.0 TITLE AND DEFINITIONS

٢

These standards shall be known as the Motor Vehicle Inspection and Maintenance Program Regulation, hereinafter referred to as "this Regulation."

For the purpose of this Regulation, the following words and phrases, when used herein, except as otherwise required by the context, have the following meanings.

1.1 "Accreditation" means Certification that the Analyzer and Analyzer manufacturer meet the operating criteria, specifications and requirements of Weber County and the Department;

1.2 **"Accuracy"** means the degree by which an instrument is able to determine the true concentration of pollutants of interest-Also means freedom from error especially as a result of care ;

1.3 "Air Intake Systems" means systems that allow for the induction of ambient air, including preheated air into the engine combustion chamber for the purpose of mixing with a fuel for combustion;

1.4 **"A.I.R. (Air Injection Reaction) System"** means a system for providing supplementary air into a vehicle's exhaust system to promote further oxidation of hydrocarbons (HC) and carbon monoxide (CO) gases and to assist catalytic reaction;

1.5 "Analyzer" See definition for UTAH2000 Analyzer;

1.6 **"Audit"** means a procedure performed by Department personnel which includes but is not limited to, inspection of the I/M Program station, review of Station records, inspection of Analyzer and related I/M Program equipment, review of personnel working knowledge and records. The audit procedure is intended to ensure compliance with this Regulation and Department policies and procedures;

1.7 **"BAR 97"** Refers to California Bureau of Automotive Repair Exhaust Gas Analyzer Specifications, which became effective in 1997;

1.8 **"Basic Engine Systems"** means parts or assemblies that provide efficient conversion of a compressed air/fuel charge into useful power, including but not limited to valve train mechanisms, cylinder head to block integrity, pistonring-cylinder sealing integrity and post-combustion emissions control device integrity meeting OEM Standards;

1.9 **"Bench"** means the main sample processing assembly of the exhaust gas Analyzer including detectors, sampling tubes, processor boards, infrared sources and power supply; 1.10 "Board of Health" means the Weber-Morgan Board of Health;

1.11 "**Calibration**" means the process of establishing or verifying the accuracy of an exhaust gas Analyzer to perform an accurate and consistent evaluation of engine exhaust using calibration gases having precisely known concentrations;

1.12 "Calibration [Span] Gases" means gases of known concentration that are used as references for establishing or verifying the calibration curve of an exhaust gas Analyzer and which are traceable to the National Institute of Standards and Technology and are approved by the Department for use;

1.13 **"Carbon Monoxide"** A colorless, odorless, asphyxiating gas produced by the incomplete burning of fuels. Carbon monoxide may be referred to in these Regulations as CO;

1.14 "Catalytic Converter" A post-combustion device that oxidizes HC and CO gases and/or reduces oxides of nitrogen gases;

1.15 **"Certificate of Compliance"** means a serially numbered document issued to the owner of a motor vehicle upon passing an inspection or reinspection and is evidence that the motor vehicle complies with the standards and criteria of this Regulation and other requirements as adopted by the Board of Health;

1.16 **"Certificate of Compliance Numbers"** means numbers issued to I/M Program Stations, and entered into the approved Analyzer for the purpose of issuing Certificates of Compliance;

1.17 "**Certificate of Waiver or Waiver**" means a document, issued by the Department used to verify that the vehicle for which it was issued has met the waiver requirements of this Regulation;

1.18 **"Certification"** means assurance by an authorized source, whether it be a laboratory, the manufacturer, the state, or the Department, that a specific product or statement is in fact true and meets all requirements;

1.19 "**Certified Emission Tester or Tester**" means an individual who has successfully completed all certification requirements and has been issued a current, valid Emission Tester Certificate of Qualification by the Department;

1.20 **"Certified Emissions Repair Technician or Technician"** means an individual who has successfully completed all certification requirements and has been issued a current, valid Emission Repair Technician Certificate of Qualification by the Department. A person certified by the Department who inspects vehicles, diagnoses emission related faults, and supervises or performs emissions related repairs and adjustments to bring vehicles into compliance with

the requirements of this Regulation;

ć

1.21 "CO" see Carbon Monoxide;

1.22 **"Compliance"** means verification that certain data and hardware submitted by a manufacturer for accreditation consideration, meets all Department requirements; Also meeting the requirements of this Regulations;

1.23 "County" means Weber County, Utah;

1.24 "Curb Idle" means the manufacturer's specified idle speed for the specific motor vehicle being tested (See also "Idle Mode");

1.25 "Cut-Points" Same as Emission Standards;

1.26 "**Department**" means the Weber-Morgan Health Department, Division of Environmental Health;

1.27 **"Director"** means the Environmental Health Division Director of the Weber-Morgan Health Department or his authorized representative;

1.28 "Domiciled" means County in which primary residence is located;

1.29 "E.G.R. System (Exhaust Gas Recirculation System)" means an emissions control system that recycles or recirculates a portion of the exhaust gases back to the engine combustion chambers;

1.30 **"Emissions"** means substances expelled into the atmosphere from a motor vehicle; particularly, air contaminants produced by combustion and/or incomplete combustion hydrocarbon evaporation from the fuel system and/or the crankcase, and particulate matter from the crankcase;

1.31 **"Emissions control systems"** means any device or combination of parts, originally installed by the manufacturer to control the emissions of a motor vehicle;

1.32 **"Emission Inspection or Inspection"** means a motor vehicle inspection performed for the purpose of determining whether the vehicle qualifies for issuance of a Certificate of Compliance or Certificate of Waiver, carried out in compliance with this Regulation;

1.33 **"Emission Repair or Repair"** means repair of a motor vehicle for the purpose of such vehicle passing or attempting to pass an emission inspection;

1.34 "Emission Repair Technician Certificate of Qualification" means a certificate issued by the Department authorizing an individual to conduct

emission inspections, repair failed vehicles and issue Certificates of Compliance while under the auspices of a I/M Program Station;

1.35 **"Emissions Standards (Cut-Points)"** means the maximum allowable concentration of regulated emissions for a given weight class and model year of a motor vehicle, as determined by the Board of Health using an approved Analyzer;

1.36 **"Emission Test"** means that portion of the Emission Inspection procedures where the engine exhaust gasses, from the tailpipe of the vehicle being Inspected, are tested to determine whether a vehicle produces emissions in excess of the Emissions Standards and/or an OBD IM test procedure;

1.37 **"Emission Tester Certificate of Qualification"** means a certificate issued by the Department authorizing an individual to perform emission inspections and issue certificates of compliance while under the auspices of a I/M Program Station;

1.38 **"Engine Switching"** means an engine is removed from a vehicle and is replaced by an engine that is not identical to the original engine;

1.39 "Engine Verification" means a document issued by the Department for the use of Kit Cars and Replica Vehicles only, validating engine size and year, for the purpose of inspecting the vehicle for the year of the engine instead of the year of the vehicle;

1.40 "EPA" means the United States Environmental Protection Agency;

1.41 **"Evaporative control system"** means an emission control system that prevents the escape of fuel vapors from the fuel system and/or air cleaner and stores them to be burned in the combustion chamber;

1.42 **"Exhaust Gas Analyzer"** means an instrument that is capable of measuring the concentrations of certain air contaminants in the exhaust gases emanating from a motor vehicle when approved by the Department for use in accordance with this Regulation as an official test instrument;

1.43 "Federal Installation" means any property or facility subject to the

jurisdiction of any department, agency, or instrumentality of the executive, legislative, and judicial branches of the Federal government;

1.44 **"Fleet Facility"** means a cooperation or other business entity permitted by the Department to perform the functions of the inspection program for a privately owned fleet of ten or more motor vehicles, including emissions related repairs, as well as the inspection; 1.45 **"Fuel Control Systems"** means the mechanical, electro mechanical, galvanic or electronic parts or assemblies that regulate the air/fuel ratio in an engine to provide a combustible charge;

1.46 "Gaseous Fuel" means, but is not limited to, liquefied petroleum gases and natural gases in liquefied or gaseous forms;

1.47 "HC" means hydrocarbons;

1.48 "**Hangup**" means hydrocarbons that cling to the surface of the sampling and Analyzer systems in contact with the exhaust gas sample stream resulting in errors in HC readings;

1.49 "Heavy Duty Vehicles" means a vehicle 1978 and older with a weight of more than 6000 pounds or 1979 and newer with a weight of more than 8501 pounds GVW (gross vehicle weight);

1.50 "Hydrocarbons" means unbumed fuel;

1.51 "Idle Mode" means a condition where the vehicle's engine is at proper operating temperature and running at the rate specified by the manufacturer's cub idle, where the engine is not propelling the vehicle, and where the throttle is in the closed or idle stop position. This condition achieved without placing a load on the vehicle to decrease its RPM to the specified rate(See also **Curb Idle**);

1.52 "**Ignition Systems**" the means parts or assemblies that are designed to cause and time the ignition of a compressed air/fuel charge;

1.53 "I/M Clearance" means a stamp placed on the motor vehicle registration form by an employee of the Utah State Motor Vehicle Office or the Department indicating that the motor vehicle represented by the registration form is in compliance with the inspection program requirements in that the motorist has presented a valid Certificate of Compliance or Certificate of Waiver for the motor vehicle and paid applicable fees:

1.54 "**I/M Program**" means the Vehicle Emissions Inspection and Maintenance Program established by the County Board of Health and this Regulation;

1.55 "I/M Program Test and Repair Station" means a business permitted by the Department which engages in emissions testing and emissions related repairs to motor vehicles, and which meets the requirements of this Regulation for test and repair facilities;

1.56 **"I/M Program Test Only Station"** means a business permitted by the Department which engages only in emissions related inspections of motor vehicles, and which meets the requirements of this Regulation for test only facilities;
1.57 **"Inspection Area"** means the Department approved area that is occupied by the Analyzer, sample hose, and the vehicle being inspected;

1.58 **"Inspection Report"** means a document used to record information generated by the Tester/Technician during an emissions inspection other than a Certificate of Compliance;

1.59 **"Instrument"** means the complete UTAH2000 Analyzer system that samples and displays the concentration of emission gases and also performs OBD IM test procedures. The instrument includes the sample handling system, the exhaust gas Analyzer associated computer equipment and the enclosure cabinet;

1.60 **"Kit Car"** means a fully assembled custom motor vehicle containing all of the needed components for assembly (i.e., body, chassis, engine, and transmission);

1.61 "Light Duty Motor Vehicle" means all passenger vehicles, 1978 and older; light duty trucks 6000 GVW rating or less; 1979 trucks and newer 8500 pounds GVW rating or less;

1.62 **"Lock-Out"** means when the UTAH2000 Analyzer automatically prohibits access to the testing portion of the UTAH2000 Analyzer;

1.63 **"Motor Vehicle or Vehicle"** means any equipment or mechanical device propelled primarily on land by an internal combustion powered engine which is driven on public roads and/or streets. Motor vehicles exempted from the inspection requirements of this Regulation as listed in Section 6.6 of this Regulation;

1.64 **"Motorcycle"** means every motor vehicle having a saddle for the use of the rider and designed to travel with not more than three wheels in contact with the ground, but excluding a farm tractor;

1.65 "Non-Certified Tester/Technician" means any person who has not been certified by the Department to perform official emissions inspections;

1.66 "OBD" means Vehicle On-Board Diagnostics;

1.67 "**OBDII**" means Updated On-Board Diagnostics Standard effective in 1996 and newer light duty car and light duty trucks sold in the United States;

1.68 "OEM" means Original Equipment Manufacturer;

1.69 **"Off-Highway Vehicles"** means a vehicle licensed to operate exclusively off public highways and roads;

1.70 "Original Condition" means the condition of the emission control system(s) as installed by the manufacturer, but not necessarily to the original level of effectiveness;

í.

i. A 1.71 "**PCV System (Positive Crankcase Ventilation System)**" means and emissions control system which returns crankcase vapors and blowby gases to the combustion chamber to be burned;

1.72 **"Permit"** means the document issued by the Department that authorizes a person to operate an I/M Program Station;

1.73 **"Person"** means an individual, corporation, association, firm, partnership, joint stock company, public or municipal corporation, political subdivision, the state or any agency thereof, or the federal government or any agency thereof;

1.74 "**Prompts**" means instructions and/or data fields, requiring data input to the Analyzer from a Tester/Technician performing an emission inspection;

1.75 "**Publicly-Owned Vehicles**" means a motor vehicle owned by a government entity, including but not limited to the federal government or any agency thereof, the State of Utah or any agency or political subdivision thereof-;

1.76 **"Readiness"** means codes set by the OBD system that indicate a vehicle readiness to be OBD tested;

1.77 **"Registered or Registration"** means the process by which a motor vehicle receives a license so that it can be legally operated on public streets and highways;

1.78 "**Reinspection**" means any emission inspection performed on a motor vehicle after it has failed an emissions inspection and repair and/or adjustment has been attempted;

1.79 **"Repeatability"** means the instrument's capability to provide the same value, within specified tolerances, for successive measurements of the same sample;

1.80 **"Response Time"** means the period of time, in seconds, for an instrument to measure and display a pollutant concentration after a concentration of gases is introduced or removed from the sample probe;

1.81 **"Safety Inspection"** means an evaluation of a vehicle's relative safety as required by 41-6-158 U.C.A., 1953;

1.82 "**Smoker**" means a motor vehicle emitting visible emissions after the engine has reached normal operating temperature;

1.83 "**Stabilization**" means the process of bringing an instrument into equilibrium with the ambient environment and operative conditions;

1.84 "**Station**" means and I/M Program Station including all station personnel, employees, and owner(s);

1.85 **"Tampering"** means the intentional or accidental altering of or removal of emission control devices, and/or emissions-related equipment. Also, the use of fuels other than those required by the manufacturer's specification as found in the motor vehicles's owners manual. Also, engine modifications which may include, but is not limited to, exhaust systems, air intake systems, ignition systems, internal engine modifications, engine switching, etc;

1.86 **"Technical Bulletin"** means a document issued to Tester/Technicians and/or I/M Program Stations by the Department to update, clarify or establish policies and/or procedures for their implementation in the Vehicle Emission Inspection and Maintenance Program;

1.87 **"Tester/Technician"** means a Department Certified Emission Tester or Department Certified Emissions Repair Technician;

1.88 **"Training Program"** means a formal program administered, conducted, or approved by the Department for the education of Testers/Technicians in basic emission control technology, inspection procedures, diagnosis and repair of emission related problems, Vehicle Emissions Inspection and Maintenance Program policies, procedures and this Regulation; it may also include the promotion of training for all mechanics;

1.89 **"UTAH2000 Analyzer or Analyzer"** means the official computerized engine exhaust analyzer and associated test equipment approved by the Department for use in the areas of Utah requiring inspections as specified in Section 41-6-163.6 and 41-6-163.7, Utah Code Annotated, 1953, as amended;

1.90 "Vehicle Emissions Inspection and Maintenance Program" means the program established by the Board of Health pursuant to Section 41-6-163.6, Utah Code Annotated 1953, as amended and Weber County Ordinance 14-4-1;

1.91 **"Waiver Referral"** means a document prepared and signed by an Tester/Technician used in the Vehicle Emissions Inspection and Maintenance program to certify that the vehicle has qualified to seek a waiver which may be issued by the Department;

1.92 "Waiver" see Certificate of Waiver.

2.0 PURPOSE

It is the purpose of this Regulation to reduce air pollution levels in Weber County by requiring annual inspection of in-use motor vehicles and by requiring emission related repairs and adjustments for those vehicles that fail to meet prescribed standards so as to:

2.1 Protect and promote the public health, safety and welfare;

2.2 Improve air quality;

5.

2.3 Comply with Federal Regulations contained in the Clean Air Act of 1970, 42 USC 7401-7671: and the amendments to the Act, Amendments of 1977, PL 95-190; and Amendments of 1990, PL 101-549;

2.4 Comply with the law enacted by the Legislature of the State of Utah, Sections 41-6-163.6 and 41-6-163.7 Utah Code Annotated, 1953, as amended; and

2.5 Comply with Weber County Ordinance 14 Chapter 4, Implement Emissions Inspection, Sections 1 through 6.

3.0 AUTHORITY AND JURISDICTION OF THE DEPARTMENT

3.1 Under Section 14-4-3 of the Weber County Ordinance the Weber County Commission authorizes and directs the Weber-Morgan Board of Health and the Director of Health to adopt and promulgate rules and regulations to ensure compliance with EPA and State requirements with respect to Emission Standards and delegates its authority as an administrative body "under 41-6-163.6 U.C.A.,1953, as amended, to the Weber-Morgan Board of Health, to address all issues pertaining to the adoption and administration of the Vehicle Emission I&M Program. Authorizes a one dollar fee to be assessed upon every motorized vehicle registered in Weber County at the time of registration, to be known as the Air Pollution Control Fee.

3.2 Section 14-4-4 of the Weber County Ordinance, directs the Weber-Morgan Health Department to adopt regulations and set fees for I&M Stations and Mechanic Permits and Emission and Waiver Certificates as necessary to sustain and operate an I&M Program.

3.3 The Weber-Morgan Board of Health is authorized to make standards and regulations pursuant to Section 26A-1-121(1) of the Utah Code Annotated, 1953 as amended.

3.4 The Weber-Morgan Board of Health is authorized to establish and collect fees pursuant to Section 26A-1-114(1)(h)(4i) of the Utah Code Annotated, 1953 as amended.

3.5 All aspects of the Vehicle Emissions Inspections and Maintenance Program within Weber County enumerated in Section 2.0 shall be subject to the direction and control of the Weber-Morgan Health Department.

4.0 POWERS AND DUTIES

4.1 General Powers and duties. The Department shall be responsible for the enforcement and administration of this Regulation and any other powers vested in it by law and shall:

4.1.1 Require the submission of information reports, plans and Specifications from I/M Program Stations, and as necessary to implement the provisions and requirements of this Regulation;

4.1.2 Issue permits, certifications and charge fees as necessary to Implement this Regulation;

4.1.3 Perform audits of any I/M Program Station and Tester/Technician and issue orders and/or notices, hold hearings, levy administrative penalties and negotiate consent agreements as necessary to effect the purposes of this Regulation;

4.1.4 When necessary take samples and make analysis to ensure that the provisions of this Regulation are met; and

4.1.5 Adopt policies and procedures necessary to ensure that the provisions of this Regulation are met and that the purposes of this Regulation are accomplished.

4.2 Suspension, Revocation, or Denial of Permits. The Department may suspend, revoke or deny a Permit of an I/M Program Station and/or negotiate a monetary penalty in lieu of suspending a permit under a consent agreement, and/or require the surrender of the Permit and unused Certificates of Compliance and other official documents of such I/M Program Station upon showing that:

4.2.1 A vehicle was inspected and issued a Certificate of Compliance by I/M Program Station personnel who did not, at the time of inspection, comply with all applicable policies, procedures, Technical Bulletins, and this Regulation;

4.2.2 A vehicle was inspected and rejected by the station when, it can be proven, as determined by the Department that the vehicle was in such condition that it did comply with the requirements of this Regulation;

4.2.3 A vehicle was inspected and issued a Certificate of Compliance when it can be proven, as determined by the department that the vehicle did not at the time of inspection comply with the requirements of Section 9.10 regarding tampering inspection;

4.2.4 A vehicle was passed and issued a Certificate of Compliance without being present for inspection or another vehicle was probed for the gas analysis during the two speed idle test;

4.2.5 The station is not open and available to perform Inspections during a major portion of the normal business hours. of 8:00 a.m. to 5:00 p.m., Mondays through Fridays,(except I/M Program stations which test only their own vehicles);

4.2.6 The station has violated any provisions of this Regulation, or any Rule, Regulation, or Department policy properly promulgated for the operation of an I/M Program Station;

4.2.7 The station was or is not equipped as required by Section 8.0 of this Regulation;

4.2.8 The I/M Program Station is not operating within the property boundaries as of the location specified on the Permit;

4.2.9 An official inspection was done by a Non-Certified Tester/Technician or a Non-Certified Tester/Technician has gained access to the official testing portion of the Analyzer or a Non-Certified Tester/Technician has signed a Certificate of Compliance or other official testing document;

4.2.10 The approved Analyzer has been tampered with or altered in any way contrary to the certification and maintenance requirements of the Analyzer or the test vehicle has been altered or tampered with in any way so that it will either pass or fail the emissions test when it would not otherwise;

4.2.11 The I/M Program Station denies access to a representative of the Department to conduct an audit or other necessary business during regular business hours;

4.2.12 The I/M Program Station denies access to or conceals pertinent information from a representative of the Department during an audit or while conducting other necessary business during regular business hours;

4.2.13 The I/M Program Station performed unnecessary repairs not justified by the results of the inspection;

4.2.14 In accordance with 41-6-163.6 and 41-6-163.7 U.C.A., 1953, As amended, an emissions inspection for a Salt Lake, Utah, Davis or Weber County resident was performed but not as required by the Regulations adopted by the applicable county and/or the UTAH2000 Analyzer prompts; or

4.2.15 A Tester/Technician employed at an I/M Program Station, and authorized to conduct emission inspections, who violates any of the provisions of Section 4.3.

4.3 Suspension, Revocation, or Denial of Certificates. The Department may suspend, revoke, or deny the Certificate of Qualification of a Tester/Technician and require the surrender of the Tester/Technician Certificate of Qualification upon showing that:

4.3.1 The Tester/Technician caused any of the violations listed in section 4.2 to occur;

4.3.2 The Tester/Technician issued or caused a Certificate of Compliance to be issued to an owner/operator without an approved inspection being made;

4.3.3 The Tester/Technician denied the issuance of a Certificate of Compliance to the owner/operator of a vehicle that, at the time of the inspection, complied with the law for issuance of said certificate;

4.3.4 The Tester/Technician issued a Certificate of Compliance to a vehicle that, at the time of issuance, was in such condition that it did not comply with this Regulation;

4.3.5 The Tester/Technician inspected, recorded and passed the tampering inspection, for a vehicle that did not, at the time of inspection, comply with the tampering requirements of the tampering inspection detailed in Section 9.10, regardless of whether a Certificate of Compliance was issued or not;

4.3.6 Inspections were performed by the Tester/Technician but not in accordance with applicable policies, procedures, technical bulletins, and this Regulation;

4.3.7 The Tester/Technician allowed a Non-Certified Tester/Technician to perform an inspection or gain access to the official testing portion of the Analyzer;

4.3.8 The Tester/Technician signed an inspection or certificate stating that he had performed the emissions test when, in fact, he did not;

4.3.9 The Tester/Technician signed a certificate prior to a test being performed and the certificate printed by the dedicated printer;

4.3.10 The Tester/Technician falsified any inspection or official document

of the Vehicle Emissions Inspection and Maintenance Program;

).

بينغمانهم زي

4.3.11 The Tester/Technician performed unnecessary repairs not justified by the results of the inspection; or

4.3.12 In accordance with Sections 41-6-163.6 and 41-6-163.7 U.C.A., 1953, as amended, an emissions inspection for a Weber, Salt Lake, Davis or Utah County resident was performed but not as required by the Regulations/Ordinances adopted by the applicable county.

4.4 <u>Administrative Penalty</u>: in lieu of suspending a Permit under Sections 4.2 and 4.3 the Department may impose an administrative penalty such as a negotiated monetary penalty by means of a consent agreement.

4.5 The Department shall respond, according to the policies and procedures this regulation, to public complaints regarding the fairness and integrity of inspections they receive, shall provide a method that inspection results may be challenged if there is a reason to believe them to be inaccurate. To challenge the results of an inspection or reinspection, a motorist must present his or her vehicle within 2 days (excluding Saturday, Sunday and Holidays) of the inspection being challenged for another emissions inspection at the Weber County Technical Center.

4.6 The Department is authorized to take any and all necessary measures to ensure or facilitate a smooth transition from the UTAH91 Analyzer testing program to the new UTAH2000 Analyzer testing program required by this Regulation. Such measures may include, but shall not be limited to the following:

4.6.1 Extending the period of time that a UTAH91 Analyzer may be used to perform an official Inspection beyond the mandatory date for use of the UTAH2000 Analyzer as specified in Section 7.0 of this Regulation;

4.6.2 Permitting the use of the UTAH2000 Analyzer to perform official emissions tests prior to the mandatory date as specified in Section 7.0 of this Regulation;

4.6.3 Extending the period of time that the two speed idle test can be substituted for compliance when 1996 and newer motor vehicles fail the OBD IM test procedures;

4.6.4 Modify OBD IM test procedures when necessary and may include the temporary suspension of the OBD testing requirement; and

4.6.5 Only the Department may exempt vehicles from the OBD test requirements of this Regulation.

5.0 SCOPE

It shall be unlawful for any person not to comply with any policy, procedure, technical bulletin, regulation or ordinance promulgated by the County and/or the Department unless expressly waived by this Regulation.

6.0 GENERAL PROVISIONS

Subject to the exceptions described in Section 6.4 and pursuant to the schedule in Section 6.1, motor vehicles of model years 1968 and newer that are owned and/or operated by person domiciled in Weber County, shall be subject to an annual or biennial emission inspection performed by an I/M Program Station or other entity approved by the Department. Registration of a vehicle owned and operated by a Weber County resident in a County other than Weber is a violation of this Regulation. Fraudulent registration of a vehicle as a farm truck, diesel, or other exempted vehicle is also a violation of this Regulation.

6.1 Beginning 1 January 1992 a Certificate of Compliance, Certificate of Waiver or evidence that the motor vehicle is exempt from the Inspection and Maintenance Program requirements (as defined in section 6.4) shall be presented to the Weber County Assessor or the Utah State Tax Commission and the Air Pollution Control Fee paid (See section 6.8.2) as conditions precedent to annual registration or annual renewal of registration of a motor vehicle. Certificates of Compliance from other EPA approved I/M Programs may be accepted, if approved by the Director, provided those I/M Programs are equally effective in reducing emissions.

6.2 A Certificate of Compliance issued to a dealer licensed with the State of Utah and issued in the dealer's name, shall be valid for registration purposes for a period of six months as specified in Section 41-3-28.5, Utah Code Annotated, 1953, as amended. The purchaser's name, address, and phone number shall be recorded by the dealer on the back of the Certificate.

6.3 Publicly-Owned Vehicles. Owners of publicly-owned vehicles shall comply with the Vehicle Emissions Inspections/Maintenance program requirements in accordance with this Regulation on an annual or biennial basis pursuant to a schedule determined by the Department. Federally-owned vehicles and vehicles of employees regularly operated on a federal installation located in the county that do not require registration in the State of Utah shall comply with the emissions testing requirements on an annual or biennial basis pursuant to a schedule determined by the Department and as required by Section 118 of the Clean Air Act (1990 amendment).

6.4 Vehicle Exemption. The following vehicles are exempt from the annual or

biennial emissions inspection:

6.4.1 Any motor vehicle of model year 1967 or older;

6.4.2 All agricultural implements of husbandry and any motor vehicle that qualifies for an exemption Section 41-6-163.6, Utah Code Annotated, 1953, as amended;

6.4.3 Any vehicle used for maintenance or construction and not designed or licensed to operate on the highway;

6.4.4 Any motorcycle or motor driven cycle (including vehicles which operate with an engine normally used in a motorcycle);

6.4.5 Any vehicle that operates exclusively on diesel fuel or electricity;

6.4.6 Any new motor vehicle being sold for the first time that has a valid (Manufacturer's Statement of Origin) (MSO) form;

6.4.7 Any vehicle with an engine smaller than forty (40) cubic inch displacement (655 cc); and

6.4.8 Tactical military vehicles.

6.5 It shall be the responsibility of the Tester/Technician to determine if a motor vehicle is exempted from Section 6.6 of this Regulation when presented to the Tester/Technician for an inspection. It shall be the Tester/Technicians responsibility to inform the owner/operator of the vehicle that the vehicle is not required to have an inspection for vehicle registration purposes.

6.6 Official Signs.

6.6.1 All I/M Program Stations, except those stations authorized to inspect only their own motor vehicles such as fleet facilities, shall display in a conspicuous location on the premises an official sign provided and approved by the Department.

6.6.2 The emissions standards, as promulgated under authority of Section 12.0 and Appendix C of this Regulation shall be posted in a conspicuous place on the station's premises.

6.6.3 The station shall post on a clear and legible sign and in a conspicuous place at the station, the fees charged by that station for the performance of the emissions inspection.

6.6.4 The signs required by Sections 6.8.1, 6.8.2, and 6.8.3 shall be

located so as to be easily in the public view.

6.7 Equipment Available for Inspection.

6.7.1 Required tools, materials, publications (see Section 8.1.4.3) supplies, records, unused Certificates of Compliance, other required forms, records of completed inspections, a complete copy of this Regulation, and duplicate copies of Certificates of Compliance issued shall be kept at the station at all times and shall be available for inspection and collection by the Department at any time the station is open for business.

6.7.2 A periodic inspection and audit shall be made by a representative of the Department to verify compliance with this Regulation for each I/M Program Station. As part of the periodic inspection and audit of the I/M Program Station the Department representative shall, as applicable, perform a gas calibration and leak check of each certified Analyzer, examine leak check and gas calibration records, and examine inspection records and Certificates of Compliance, as well as other required reports, forms, or records to see that the use of these items is in compliance with this Regulation and the policies and procedures of the Department.

6.7.2.1 During the time of the inspection and audit by the Department, the Department representative shall have exclusive access to the approved testing Analyzer(s).

6.7.2.2 The Department representative may check the accuracy of the Analyzer using Department gas to verify that the Analyzer is reading within the tolerances established by the Department. Analyzers not reading within the acceptable tolerances shall be calibrated to acceptable tolerances or placed "out of service".

6.8 Fees

6.8.1 The fees assessed I/M Program Stations and Testers/Technicians shall be determined according to a fee schedule adopted by the Board of Health. The fee schedule is referenced in Appendix B of this Regulation.

6.8.2 The following fee is hereby assessed upon every motor vehicle registered in Weber County annually at the time of registration of the vehicle:

6.8.2.1 Air Pollution Control Fee -- one dollar (\$1.00).

6.8.2.2 This fee assessment is included upon all motorized vehicles including those that are exempted from the inspection requirements of this Regulation by Section 6.4 unless a separate

fee is assessed on other motor vehicles by other Health Regulations.

6.8.3 I/M Program Stations may charge a fee for the required service. The fee may not exceed, for each vehicle inspected, the amount set by the Board of Health and referenced in Appendix B of this Regulation:

6.8.3.1 The inspection fee pays for a complete inspection leading to a Certificate of Compliance or a failure. If a vehicle fails the inspection, the owner is entitled to one free reinspection if he returns to the station that performed the original inspection within fifteen (15) days from the date of the initial inspection. The station shall extend the fifteen day free reinspection time to accommodate the vehicle owner if the station is unable to schedule the retest of the vehicle within the fifteen day time period. The emissions inspection fee shall be the same whether the vehicle passes or fails the emission test.

6.8.3.2 At the request of the Department, a station shall extend the free retest time for vehicle owners who are unable to complete repairs because of the unavailability of parts to make the necessary repairs. In no case shall this extended time exceed the data storage capacity time of the approved emission Analyzer.

6.8.3.3 Duplicate Certificates of Compliance issued to a vehicle owner/operator shall not exceed that set by the Board of Health and referenced in Appendix B of this Regulation.

6.9 If a vehicle fails the inspection and is within the time and mileage requirements of the federal emissions warranty contained in the Federal Clean Air Act, the Tester/Technician shall inform the owner/operator that he may qualify for warranty coverage of emission related repairs as provided by the vehicle manufacturer and mandated by the Federal Environmental Protection Agency.

6.10 These fees are subject to change and may be amended as deemed necessary by the Board of Health to accomplish the purposes of this Regulation.

7.0 STANDARDS AND SPECIFICATIONS FOR EXHAUST GAS ANALYZERS AND CALIBRATION GASES

- 7.1 Approval of Exhaust Gas Analyzers
 - 7.1.1 No emission inspection or emission test required by this Regulation

shall be performed after May 31,2000 unless the type of instrument used for determining compliance with this Regulation is the UTAH2000 Analyzer (except for provisions noted in Section 4.6). The Analyzer shall meet the requirements of the Analyzer specifications referenced in Appendix A of this Regulation. The Analyzer shall also be certified by the manufacturer as meeting the criteria of Section 207 (b) warranty provisions of the Clean Air Act. The instrument shall be in good working condition, capable of meeting calibration requirements of the Department and operated according to manufacturer's specifications and operating procedures and capable of operating to the standard required by the Utah2000 Analyzer Specification.

7.1.2 Analyzer Registration

Any Analyzer used by an I/M Program Station shall be registered with and approved by the Department and shall be issued an analyzer registration number and, if required, shall bear a registration sticker issued by the Department. Registration stickers are not transferable or assignable. Any new or used Analyzer put in use after station approval must be approved by the Department before use. Analyzers used temporarily during times of breakdown or repair of the registered Analyzer do not require a registration sticker but shall meet all other requirements of this section including the approval of the Department before use.

7.1.3 Running Changes

Any changes to the design characteristics or component specifications that may affect the performance of an Analyzer to be used as an official test instrument in the Vehicle Emissions Inspection and Maintenance Program shall be approved by the Department. It shall be the Analyzer manufacturer's responsibility to verify that the changes have no detrimental effect on the performance of the Analyzer.

7.1.3.1 It shall be unlawful for any person to alter or modify the hardware or software of an approved emissions Analyzer without written application and formal written approval by the Department.

7.1.3.2 It shall be unlawful for any person to gain access to any Department controlled portions of an approved Analyzer without approval by the Department.

7.1.4 Calibration/Span Gases

7.1.4.1 General

The Analyzer manufacturer and/or manufacturer designated

marketing vendor shall, supply at a reasonable cost calibration gases approved by the Department to any ultimate purchaser of the Analyzer. Each new or used Analyzer sold by the manufacturer or marketing vendor shall have when deemed necessary by the Department, approved full calibration gas containers installed and operational at the time of delivery. The Department shall establish necessary procedures for approving calibration/span gases.

7.1.4.2 Calibration/Span Gas Blends

The calibration/Span gases supplied to any I/M Program Station shall conform to the specifications of the Department. All calibration gases shall meet all requirements for emissions warranty coverage. Only gas blends supplied by Department approved venders shall be used to calibrate Analyzers. (Approved gas blend and Department approved vendors are referenced in Appendix E of this Regulation.)

7.1.5 Documentation, Logistics, and Warranty Requirements

7.1.5.1 Instrument Manual

An instrument manual shall be provided by the Analyzer manufacture. The instruction manual shall be conveyed to the purchaser at the time of sale and shall contain at least the following information for the Analyzer:

(a) A complete technical description;

(b) The functional mechanical and electrical schematics;

(c) The accessories and options that are included and/or available;

(d) The model number, identification marking and location;

(e) Operating maintenance schedule including daily, weekly, and monthly accommodations and procedures for maintaining sample system integrity including, but not limited to, leaks, hangup, calibration and filters. The services to be performed only by the manufacturer shall be clearly identified;

(f) Field Calibration Procedures (i.e., Department inspection procedure with separate gas supply);

(g) Cal-port gas inlet calibration, zero, and span instruction;

(h) Information concerning the nearest service facility where equipment can be serviced; and

(i) The warranty provisions for the Analyzer, including a list of warranty repair stations by name, address and telephone number.

7.1.5.2 Analyzer Maintenance.

The Analyzer shall be maintained in accordance with the manufacturer's recommended maintenance schedule and records of this maintenance service shall be maintained for examination by the Department.

7.1.5.3 Analyzer printers shall be maintained in such a manner that the printing of the Certificates, inspection reports and documents are accurate and legible. If any printer fails to properly function, then the Station shall discontinue testing until the required repairs have been performed or a replacement printer is installed.

7.1.6 Propane Equivalency Factor (P.E.F.).

Each instrument shall be labeled with a valid propane equivalency factor, shown with an accuracy of at least two decimal places, (i.e., 0.52). P.E.F. confirmation shall be made on each assembled Analyzer by measuring both N-hexane and propane values on assembly line quality checks. If the Analyzer bench is replaced, then a new P.E.F. label applicable to the replacement bench shall be appropriately attached to the Analyzer.

7.2 Gas Calibration and Leak Check.

7.2.1 A Tester/Technician shall perform a gas calibration of the exhaust gas Analyzer, with an approved calibration gas, within 72 hours prior to performing any emission test. A leak test must be performed every 24 hours. The gas calibration and leak test must be performed in accordance with the Analyzer specifications as contained in Appendix A.

7.2.2 The Analyzer instruction manual and other Department approved information shall be reviewed by the Tester/Technician to ensure that proper procedures are being used for performing the gas calibration.

7.2.3 The Analyzer shall lock-out when calibrations and leak tests are not performed within prescribed time frames.

7.2.4 The Department shall use and require for use in the calibration of Analyzers, calibration and span gases and containers meeting the guidelines contained in Section 7.1.5.

8.0 PERMIT REQUIREMENTS OF THE VEHICLE EMISSIONS I/M PROGRAM STATION

8.1 Permit Required.

8.1.1 No person shall operate a I/M Program Station without a valid Permit to operate issued by the Department. A person desiring to operate a I/M Program Station shall submit to the Department a written application for a Permit on a form provided by the Department. To qualify for a Permit, an applicant shall:

8.1.1.1 Be an owner of the proposed I/M Program Station or an officer of the legal ownership;

8.1.1.2 Comply with the requirements of this Regulation;

8.1.1.3 Agree to allow Department access to the I/M Program Station and to provide required information;

8.1.1.4 Pay the application permit fee at the time the application is submitted;

8.1.1.5 Present a copy of a current business license relating to the I/M Program Station; and

8.1.1.6 Other information required by the Department.

8.1.2 The application shall include:

8.1.2.1 The name, mailing address, telephone number, and signature of the person applying for the Permit and the name, mailing address, and location of the I/M Program Station;

8.1.2.2 Information specifying whether the Station is owned by an association, corporation, individual, partnership, or other legal entity;

8.1.2.3 The name, title, address, and telephone number of the

person directly responsible for the Station;

8.1.2.4 The name, title, address, and telephone number of the person who functions as the immediate supervisor of the person specified under 8.1.2.3 of this section such as zone, district, or regional supervisor;

8.1.2.5 A statement signed by the applicant that attests to the accuracy of the information provided in the application, and affirms that the applicant will comply with this Regulation, and allow the Department access to the Station; and

8.1.2.6 Other information required by the Department.

8.1.3 No person shall in any way represent any place as an I/M Program Station unless the station is operated under a valid Permit issued by the Department.

8.1.4 The Department is authorized to issue or deny Permits for I/M Program Stations.

8.1.5 A Permit may not be transferred from one person to another person, from one I/M Program Station to another I/M Program Station or from one type of operation to another, i.e., test and repair to test only, unless approved in writing by the Department. The Permit shall be posted in a conspicuous place within public view on the premises.

8.1.6 The Department may renew a Permit for an existing I/M Program Station or may issue a Permit to a new owner of an existing Station after a properly completed renewal form is submitted, reviewed, and approved, the fees are paid, and an inspection shows that the Station is in compliance with this Regulation.

8.1.7 No Permit shall be issued unless the Department finds that the facilities, tools, and equipment of the applicant comply with the requirements of this Regulation and that competent personnel, certified under the provisions of Section 12.0, are employed and available to make inspections and adjustments, and the operation thereof will be properly conducted in accordance with this Regulation.

8.1.7.1 An I/M Program Station shall immediately notify the Department if the station does not have a Tester/Technician employed.

8.1.7.2 An I/M Program Station shall comply with all terms stated in the permit application and all the requirements of this Regulation.

8.1.7.3 As a condition for permitting all I/M Program Stations, the following tools and materials shall be available for performance of the inspection and maintenance of motor vehicles unless specifically exempted by the Department:

(a) A Department approved Analyzer;

(b) An accurate dwell meter (Optional);

(c) An ignition timing light;

(d) A propane enrichment kit;

(e) Reference manuals (printed or electronic) approved by the Department that contain idle speed, idle mixture, timing, dwell, fast idle speed specifications, and information covering the emissions control systems for the model years an makes of vehicles involved in the Vehicle Emissions Inspection and Maintenance Program;

(f) Sufficient hand tools for proper performance of the inspection and minimum repairs and maintenance as required by the Department;

(g) Department approved calibration/ gases;

 (h) The Analyzer manufacturer's maintenance and calibration manual, which must be retained in the inspection area;

(i) All forms, technical bulletins, a copy of this Regulation, and other information materials provided by the Department;

(j) A suitable non-reactive tailpipe extender or suitable probe adapter for inspecting vehicles with screened or baffled exhaust systems; and

(k) Suitable tools to interface with onboard vehicle computers for computer controlled vehicles which are intended to receive official repairs. (For example, General Motors compatible can tools are required if the I/M Program Station wishes to repair computer controlled GM vehicles failing the I/M test.)

8.2 Duties of Testers/Technicians Working in Permitted Program Stations:

8.2.1 All facets of the Vehicle Emissions Inspection and Maintenance Program shall be performed by the Tester/Technicians including, but not limited to:

8.2.1.1 Analyzer preparation, calibration checks, and leak checks;

8.2.1.2 Exhaust gas sampling and analysis for purposes of an official emissions test for issuance of a Certificate of Compliance;

8.2.1.3 Preparation of reports, forms, and certificates.;

8.2.1.4 Accessing the official emissions testing section of the Analyzer; and

8.2.1.5 All other aspects of the official emissions test, including but not limited to, the tampering inspection, inserting the exhaust probe, hooking up the tachometer, hooking up the OBDII data link connector, entering data into the Analyzer, verifying that the engine is at normal operating temperature, ensuring that accessories are off, preconditioning the vehicle, and signing Certificates of Compliance and inspection forms, etc., unless otherwise approved in writing by the Director.

8.3 Safety.

An I/M Program Station facility shall be kept in good repair, free of obstructions and hazards and in a safe condition for inspection purpose. No inspection shall be conducted if unsafe conditions exist.

8.3.1 At no time shall carbon monoxide (CO) readings in the ambient air within the station exceed a peak hourly level of 35 parts per million.

8.3.2 All applicable Occupation Safety and Health Administration (OSHA), and other applicable health and safety rules and regulations must be followed in the station.

8.4 Fleet Facility.

A person may establish a fleet facility that is exempt from conducting business at regular hours, or displaying program signs as long as only vehicles owned or controlled by the fleet facility owner are inspected at the station. All other requirements of this Regulation apply. 8.5 Permit Duration and Renewal.

8.5.1 The Permit for I/M Program Stations shall be issued annually and shall expire one (1) year from the date of issuance. The Permit is renewable within sixty (60) days prior to the date of expiration.

8.5.2 It is the responsibility of the owner/operator of the I/M program station to pursue the Permit renewal through appropriate channels,

8.5.3 The Station Permit fee shall be paid annually to the Department by the billing due date set by the Department.

8.5.4 Prior to the date on which the Station Permit fee is due the Department shall attempt to notify each regulated Station of the amount of the fee. Fees unpaid after the billing due date will be assessed a late fee which shall be added to the original fee amount.

8.6 I/M Program Station Permit Revocation and Suspension.

8.6.1 I/M Program Station Permits may be suspended by the Department for violations of this Regulation.

8.6.2 I/M Program Station Permits may be revoked by the Department for severe and/or repeated violations of this Regulation.

8.6.3 Suspension of I/M Program Station Permits shall follow the provisions of Appendix D of this Regulation.

8.6.4 I/M Program Station Permits are and remain the property of the Department, only their use and the license they represent is tendered.

8.6.5 Station Permit may be suspended or revoked by the Department because of returned checks and may not be reinstated until repayment is confirmed. All returned checks will be charged a returned check handling fee (referenced in Appendix B).

8.6.6 Failure to pay the Station Permit fee and any additional charges after the due date may result in suspension and/or revocation of the Permit and the right to operate as an I/M Program Station.

8.7 The I/M Station Shall Hold the Department harmless.

In making application for a Permit or for its renewal, such action shall constitute a declaration by the applicant that the Department shall be held harmless from liability incurred due to action or inaction of I/M Program Station owner or their

employees.

9.0 INSPECTION PROCEDURE

9.1 The official emission inspection shall be solely performed by a Tester/Technician who has been certified at the station where the inspection is being performed and Department approved inspection procedures are being followed.

9.2 If the Tester/Technician is unable, unqualified, or unwilling to make the required repairs or adjustments, should the vehicle fail the inspection, he shall notify the owner/operator of the vehicle before the inspection is administered.
9.3 The entire inspection shall take place within the reach of the Analyzer hose and tachometer lead.

9.4 The temperature of the inspection area shall be between 41 Fahrenheit and 110 Fahrenheit (2 Celsius and 43 Celsius) during the inspection.

9.5 The Analyzer shall be kept in an area that provides adequate protection from the weather, wind, moisture, and extreme temperatures or any other damaging environmental exposure.

9.6 The electrical supply to the Analyzer shall be able to meet the Analyzer manufacturer's requirements for voltage and frequency stability.

9.7 The Tester/Technician shall not inspect or test any motor vehicle with a mechanical condition which may cause injury to personnel or damage to the station or test equipment or which may affect the validity of the inspection, until such condition is corrected. Such conditions included but are not limited to: coolant, oil, or fuel leaks, low oil or low fluid levels, carburetor gas overflow, vehicle electronic instrument panel malfunction, and visible emissions (smoker).

9.8 Any time an engine stalls during an Emission Test, the Emission Test shall be restarted. If a Tester/Technician cannot complete an Emission Test because of continuous stalling, then the stalling problem shall be corrected before the test is performed.

9.9 The Tester/Technician shall verify the vehicle license plate and vehicle identification numbers by comparing the information on the vehicle's registration with those on the vehicle and shall enter them in the Analyzer at the appropriate prompt.

9.9.1 The Tester/Technician shall verify the owner's name and correct address and enter this information into the Analyzer at the appropriate prompt.

9.9.2 The Tester/Technician shall determine and enter the county in which the vehicle is registered at the appropriate prompt.

9.9.3 The Tester/Technician shall enter completely and accurately all the information required as part of the data entry procedure for the official vehicle Emission Test on the approved Analyzer at the appropriate prompt sequence.

9.9.4 All data entries to the Analyzer during the inspection shall be true and factual.

9.10 The Tester/Technician shall:

9.10.1 Conduct the inspection in accordance with the prompts from the Analyzer and the requirements of this Regulation.

9.10.2 Examine the emissions/tune-up specification decal (sticker) under the hood and/or check an approved reference manual to determine if the vehicle was manufactured with a catalytic converter, air injection reaction (AIR) system, PCV System, EGR System, and Fuel Evaporate Control System, etc, as prompted by the Analyzer.

9.10.3 On 1996 and newer model year vehicles follow the OBD IM test procedures in accordance with Appendix F.

9.10.4 On 1990 through 1995 vehicles, visually inspect for the presence and apparent operability of the AIR system, catalytic converter, EGR system, Fuel Evaporative Control system, PCV system, and gas tank cap in accordance with Department procedures and record the information in the Analyzer. If these parts or systems have been removed, or are inoperable, the vehicle fails and the owner shall repair or replace the parts or systems before the emissions test may be continued.

9.10.5 On 1968 through 1989 vehicles, visually inspect for the presence and apparent operability of the AIR system, PCV system, EGR system, Fuel Evaporate Control System, catalytic converter and gas tank cap etc. in accordance with Department procedures and record the information on the emissions Analyzer.

9.11 Prior to performing the emission test each vehicle shall be checked to determine that it is at normal operating temperature by feeling the top radiator hose or by checking the temperature gauge.

9.12 The inspection shall be performed with the transmission in 'park' or 'neutral' and with all accessories off and the emergency brake applied (the vehicle may not be placed in gear to drop idle speed, headlights may be turned on).

9.13 The Analyzer probe shall be inserted into the exhaust pipe at least twelve inches (12") or as recommended by the Analyzer manufacturer, whichever is greater.

9.14 If a baffle or screen prevents probe insertion of at least twelve inches, a suitable probe adapter or snug fitting, non-reactive hose which effectively lengthens the exhaust pipe shall be used.

9.15 For all vehicles equipped with a multiple exhaust system that does not originate from a common point, both sides shall be tested simultaneously with an approved adaptor.

9.16 When inspecting a vehicle under windy conditions, the tailpipe shall be shielded from the wind with a suitable cover.

9.17 For 1995 Model Year Vehicles and Older:

9.17.1 With the tachometer properly attached to the vehicle being tested;

9.17.2 The vehicle shall be tested according to the testing sequence as programmed into the Analyzer and as detailed in the Analyzer

- specifications referenced in Appendix A. Vehicles failing because of
- excessive exhaust dilution shall repair the dilution problem prior to
 - continuing the emission test. The dilution standard shall be contained in the Analyzer specifications as referenced in Appendix A and adjusted when the Department determines by analysis that an adjustment is necessary to yield a more accurate level of emissions readings.

9.18 A Certificate of Compliance shall be issued if:

9.18.1 The vehicle emissions levels are the same as or less than the applicable emissions standards; and

9.18.2 For 1990 through 1995 model year vehicles, the vehicle passes the visual inspection described in Section 9.10;

9.18.3 Beginning January 1, 2001 for 1996 and newer model year vehicles, the vehicle passes the On-Board Diagnostics (OBD) test requirements as specified in Appendix F of this Regulation.

9.19 If the vehicle fails the initial Inspection, the owner shall have fifteen (15) days to have repairs or adjustments made and return the vehicle to the I/ M program station that performed the initial Inspection for one (1) free reinspection. The vehicle that failed the initial inspection shall then be issued a Certificate of Compliance only when all of the following are met:

9.19.1 The vehicle is re-inspected;

9.19.2 The vehicle's emission levels are the same or less than the applicable Emission Standards and;

9.19.3 For 1990 through 1995 model year vehicles, the vehicle passes the visual Inspection as provided for in Section 9.10.

9.19.4 Beginning January 1, 2001 for 1996 and newer model year vehicles, the vehicle passes the On-Board Diagnostics (OBD) test requirements as specified in Appendix F of this Regulation.

9.20 A Certificate of Waiver shall be issued only under the following conditions:

9.20.1 For all vehicles, air pollution control devices applicable and specified for the make, model and year of the vehicle as specified in Section 9.10 of this Regulation are in place and operable on the vehicle. If the devices have been removed or rendered inoperable, they shall be replaced or repaired before a Waiver is granted.

9.20.2 For 1968 to 1980 model year motor vehicles, if the vehicle continues to exceed applicable emissions standards after one hundred dollars (\$100) of acceptable emissions related repairs have been performed and the adjustments required by Section 10.0 have been performed by a Certified Emissions Repair Technician as part of the one hundred dollars (\$100) in emissions related repairs. Proof of repair costs, for that specific vehicle, shall be provided to the Department in the form of an itemized bill, invoice, work order, manifest or statement in which emission related parts and labor are specifically identified. If repairs are made by the vehicle owner or by someone who does not possess a valid Emission Repair Technician Certificate, the cost of labor may not be included in the one hundred dollars (\$100).

9.20.3 For 1981 to 1995 model year motor vehicles, at least two hundred dollars (\$200) of acceptable emissions related repairs have been performed and the adjustments (were applicable) required by Section 10.0 have been performed by a Certified Emissions Repair Technician as part of the two hundred dollars (\$200) in emissions related repairs. Proof of repair costs, for that specific vehicle, shall be provided to the Department in the form of an itemized bill, invoice, work order, manifest or statement in which emission related parts and labor are specifically identified. If repairs are made by the vehicle owner or by someone who does not possess a valid Emission Repair Technician Certificate, the cost of labor may not be included in the two hundred dollars (\$200).

9.20.4 For 1996 and newer model year vehicles, at least four hundred dollars (\$400) of acceptable emissions related repairs have been performed by a Certified Emissions Repair Technician as part of the four hundred dollars (\$400) in emissions related repairs. Proof of repair costs, for that specific vehicle, shall be provided to the Department in the form of an itemized bill, invoice, work order, manifest or statement in which emission related parts and labor are specifically identified. If repairs are made by the vehicle owner or by someone who does not possess a valid Emission Repair Technician Certificate, the cost of labor may not be included in the four hundred dollars (\$400). Any repair costs eligible under the federal emissions warranties shall not be eligible to be applied to the repair cost waiver limits.

9.20.5 Any vehicle that experiences an increase in any emissions levels shall not be eligible for a certificate of waiver regardless of the amount spent in attempting to repair the vehicle.

9.20.6 As used in this section acceptable emissions related repairs:

9.20.6.1 Refers to those expenditures and costs associated with the adjustment, maintenance, and repair of the motor vehicle which are directly related to reduction of exhaust emissions necessary to comply with the applicable emissions standards, cut-points, and procedures.

9.20.6.2 Refers to repairs and maintenance of the following systems, if done according to manufacturer's specifications, to the extent that the purpose is to reduce emissions:

(a) Air Intake Systems;

- (b) Ignition Systems;
- (c) Fuel Control Systems;

(d) Emission Control Systems except as noted in Section 9.22.4.4;

(e) Basic Engine Systems; and

(f) Repair of problems identified by On-Board Diagnostic (OBD) fault codes.

9.20.6.3 Does not include adjustments, maintenance, or repairs performed prior to the official emissions inspection.

9.20.6.4 Does not include the fee paid for the inspection.

9.20.6.5 Does not include costs associated with the repairs or replacement required by Section 9.10 or the replacement, and/or repair of air pollution control equipment on the vehicle if the need for such adjustment, maintenance, replacement, or repair is due to disconnection of, tampering with, or abuse of the emissions control systems, or costs incurred due to engine switching and/or modifications.

9.20.6.6 Does not include repairs performed to the vehicle's exhaust system to correct problems with excessive exhaust dilution.9.20.6.7 Does not include any diagnostics performed or any chemical additives.

9.21 Information regarding all performed repairs shall be entered into the appropriate data base of the Analyzer prior to the vehicle being reinspected.

ţ

9.22 Certificate of Waiver shall only be issued by the Department unless the Department determines other acceptable methods of issuance. A Waiver shall only be issued after determining that the vehicle complies with the requirements of Section 9.20. A Waiver shall not be issued to a vehicle with an inoperable or glowing check engine light.

9.23 Prior to referring the vehicle owner/operator to the Department for waiver eligibility, the Tester/Technician or Station shall verify that the repair and eligibility requirements of this Section have been met.

9.24 The inspection records shall be completed accurately, signed immediately, filed, and distributed, as required by the Department. The customer shall be given the appropriate copies.

9.25 After a passing Inspection customers shall be given the Certificate of Compliance along and appropriate copy of the Inspection form.

9.26 Vehicles capable of being operated on both gaseous and liquid petroleum fuels shall be tested for both fuels in accordance with the Analyzer specifications as referenced in Appendix A of this Regulation.

9.27 When a vehicle owner requests an Inspection, the Tester/Technician shall perform the inspection in the testing mode of the approved Analyzer. Performing a screening test (or pre-test) in the manual mode of the approved Analyzer or on a non-approved analyzer shall be a violation of this Regulation if the vehicle owner requested an emissions inspection. Adjustments or repairs shall not be made prior to a requested inspection.

9.28 At the end of each business day the UTAH2000 Analyzer shall be placed in a standby mode and be connected to the appropriate telecommunications line in order for the Department to collect data, load certificates, update station and Tester/Technician information or any other administrative procedures.

10.0 ADJUSTMENT PROCEDURES (Vehicles without computer Controlled Engine Systems)

10.1 The following adjustments should be performed on all 1981 and older vehicles (where applicable) that failed the I/M test. These adjustments must be performed by a emission repair technician before a vehicle will be eligible for an emissions waiver.

10.2 The manufacturer's or high altitude specifications, if available, for idle speed, idle air/fuel mixture, ignition timing, and dwell, shall be determined for the purpose of adjustment. The emissions repair technician shall refer to the emissions tune-up specifications. Fuel control systems designed with sealed tamper-resistant adjustment screws for air/fuel mixture shall be adjusted according to manufacturer's specifications and resealed. On vehicles that have limiter caps on the fuel control systems, the limiter caps shall be removed and the air/fuel ratio adjusted to meet manufacturer's specifications and the proper limiter caps shall be reinstalled. The adjustment procedures shall be as follows:

10.2.1 The dwell ,if applicable, shall be checked with a dwell meter to determine if it is within the recommended tolerance of 2 degrees of specifications. The dwell shall be reset if it exceeds this tolerance;

10.2.2 The idle speed shall be checked with a tachometer to determine if it is within 50 rpm of the manufacturer's specifications. If it is not, it shall be set to within 50 rpm of the manufacturer's specifications;

10.2.3 The ignition timing shall be checked, using a timing light or engine analyzer, to determine if it is within +4 degrees to -2 degrees of the recommended settings while the engine is idling at the specified speed. If the timing exceeds this tolerance, it shall be adjusted until it falls within +4 degrees to -2 degrees of the recommended setting;

10.2.4 The idle air/fuel ratio shall be adjusted according to manufacturer's suggested procedures and/or specifications using an infrared analyzer, propane enrichment kit, or tachometer;

10.2.5 The choke shall be checked for normal operation and, if appropriate, adjusted according to manufacturer's suggested procedures and/or specifications;

10.2.6 After completing the preceding steps, the idle speed shall be readjusted to manufacturer's specifications; and

10.2.7 The performed adjustments shall be entered in the required data base of the Analyzer.

11.0 ENGINE SWITCHING

11.1 All vehicles which qualify for testing under this section shall be tested by the Department only.

11.2 Vehicles qualifying for testing under this Section shall not be eligible for a Waiver.

11.3 Engine switching shall be allowed only in accordance with E.P.A. policy.

11.4 Vehicles not meeting the requirements of Section 11.0 shall be deemed as tampered and dealt with in accordance with the tampering provisions of this Regulation.

11.5 All 1990 and newer vehicles with switched engines shall be verified, to meet E.P.A. requirements, by the Department prior to issuance of a Certificate of Compliance.

11.6 For 1968 to 1989 vehicles, having an engine other than the original engine and emission control configuration are deemed as tampered. These vehicles must meet the HC and CO standards for the Model Year of the vehicle in order to receive a Certificate of Compliance, and are not eligible for a Certificate of Waiver, unless they are restored to the original engine and emission control configuration or a configuration approved by the Department.

12.0 EMISSION REPAIR TECHNICIAN / EMISSION TESTER CERTIFICATE OF QUALIFICATION

12.1 Emission Repair Technician Certification Required.

12.1.1 No person shall perform any part of the Inspection for the issuance of a Certificate of Compliance unless the person possesses a valid Emission Repair Technician Certificate of Qualification issued by the Department.

12.1.2 Applications for an Emission Repair Technician Certificate of Qualification shall be made upon an application form prescribed by the

Department. No Certificate of Qualification shall be issued unless:

12.1.2.1 The applicant has shown evidence of at least an associate degree in automotive technology or similar, or at least two (2) years work experience as an automotive mechanic, or other Department approved prerequisites.

12.1.2.2 The applicant has shown adequate competence by successfully completing the written and practical portions of the Emission Repair Technician Certification requirements as specified in this Regulation.

12.1.2.3 The applicant has paid the required permit fees as set by the Board of Health (reference in Appendix B).

12.1.3 An applicant shall comply with all of the terms stated in the Certificate of Qualification Application and with all the requirements of this Regulation.

12.1.4 An applicant shall complete a Department approved training course and shall demonstrate knowledge and skill in the performance of Inspections, use of the approved Analyze; and adjustment and repair of vehicles to manufacturer's specifications. Such knowledge and skill shall be shown by passing:

12.1.4.1 A written qualification test including but not limited to the following:

(a) Operation and purposes of emission control systems;

(b) Relationship of HC and CO to timing and air/fuel ratio adjustments;

(c) Adjustment to manufacturer's and high altitude specifications;

(d) Function and operation of computer controlled emission control system including, but not limited to the following:

Oxygen sensor; Engine control module (ECM); Other sensors; Three way catalytic convertor; Fuel injection system (type and discussion of differences); (e) Inspection procedures as outlined in this Regulation andprompted by the Analyzer;

(f) Operation of the Analyzer including the performance of gas calibration and leak check;

(g) The provisions of Section 207(b) Warranty provisions of the Federal Clean Air Act; and

(h) The provisions of this Regulation and other applicable Department policies and procedures.

12.1.4.2 A performance qualification test including but not limited to the following:

(a) Visual inspection and knowledge of the required emission control equipment;

(b) Demonstration of skill in proper use, care, maintenance, calibration and leak testing of the Analyzer;

(c) Demonstration of ability to conduct the Inspection;

(d) Demonstration of ability to adjust the engine systems to manufacturer's and high altitude specifications; and

(e) Demonstration of ability to accurately enter date in the Analyzer; and legibly, accurately complete the required reports and forms.

12.1.5 A signed "Hands-on Performance" check sheet shall be necessary for successful completion of the performance qualification test. The "hands-on Performance" check sheet shall be signed by an instructor or other person approved by the Department.

12.1.6 The Department shall issue Emission Technician Certificate to an applicant upon successful completion of the requirements of this Section .

12.1.7 The Emission Repair Technician Certificate shall be valid only at the station where the Emission Technician is presently employed. If the Emission Technician is later employed at another station, he shall notify the Department of the employment change. He shall also be required to be certified there prior to performing any Inspections. That certification will expire on the same date as the original. A transfer fee or duplicate fee will be charged, as set by the Board of Health and referenced in Appendix B of this Regulation. 12.1.8 Emission Repair Technician Certificates are and remain the property of the Department, only their use and the license they represent is tendered.

12.2 Emission Tester Certification Required

12.2.1 No person shall perform any part of the Inspection for the issuance of a Certificate of Compliance unless the person possesses a valid Emission Tester Certificate of Qualification issued by the Department.

12.2.2 Applications for an Emission Tester Certificate of Qualification shall be made upon an application form prescribed by the Department. No Certificate of Qualification shall be issued unless:

12.2.2.1 The applicant has shown adequate competence by successfully completing the written and practical portions of the Emission Tester Certification requirements as specified in this Regulation; and

12.2.2.2 The applicant has paid the required permit fees as set by the Board of Health and reference in Appendix B.

12.2.3 An applicant shall comply with all of the terms stated in the Certificate of Qualification Application and with all the requirements of this Regulation.

12.2.4 An applicant shall complete a Department approved training course and shall demonstrate knowledge and skill in the performance of emission testing and use of the approved Analyzer. Such knowledge and skill shall be shown by passing:

12.2.4.1 Operation and purposes of emission control systems;

12.2.4.2 Inspection procedures as outlined in this Regulation and prompted by the Analyzer;

12.2.4.3 Operation of the Analyzer including the performance of gas calibration and leak check;

12.2.4.4 The provisions of Section 207(b) Warranty provisions of the Federal Clean Air Act;

12.2.4.5 The provisions of this Regulation and other applicable Department policies and procedures; and

12.2.4.6 A performance qualification test including but not limited to the following:

(a) Visual inspection and knowledge of the required emission control equipment;

(b) Demonstration of skill in proper use, care, maintenance, calibration and leak testing of the Analyzer;

(c) Demonstration of ability to conduct the Inspection; and

(d) Demonstration of ability to accurately enter date in the Analyzer; and legibly, accurately complete the required reports and forms.

12.2.5 A signed "Hands-on Performance" check sheet shall be necessary for successful completion of the performance qualification test. The "hands-on Performance" check sheet shall be signed by an instructor or other person approved by the Department.

12.2.6 The Department shall issue an Emission Tester Certificate to an applicant upon successful completion of the requirements of this Section.

12.2.7 The Emission Tester Certificate shall be valid only at the station where the Tester is presently employed. If the Tester is later employed at another station, he shall notify the Department of the employment change. He shall also be required to be certified there prior to performing any Inspections. That certification will expire on the same date as the original. A transfer fee or duplicate fee will be charged, as set by the Board of Health and referenced in Appendix B of this Regulation.

12.2.8 Emission Tester Certificates are and remain the property of the Department, only their use and the license they represent is tendered.

12.3 Tester/Technician Certificate of Qualification Suspension and Revocation.

12.3.1 Tester/Technician Certificate of Qualification may be suspended by the Department for violations of this Regulation.

12.3.2 Tester/Technician Certificates of Qualification may be revoked by the Department for Severe and/or repeated violations of this Regulation.

12.3.3 Suspension or revocation of Tester/Technician Certification shall follow the provisions of Appendix D of this Regulation.

12.3.4 Tester/Technician Certification may be suspended or revoked by the Department because of returned checks and may not be reinstated until repayment is confirmed. All returned checks will be charged a returned check handling fee (referenced in Appendix B).

12.4 Re-Qualification Requirements for all Tester/Technician Certification.

12.4.1 Tester/Technician Certificates shall not be transferred from one person to another person. Tester/Technician Certificates may not be transferred from one I/M Program Station to another or from one status to another, i.e., from test and repair to test only, without a written request and Department approval.

12.4.2 The Department may renew Certification for an existing Tester/Technician after a properly completed renewal form is submitted, reviewed, and approved, the re-certification requirements have been completed, the fees are paid and the Tester/Technician has complied with this Regulation.

12.4.3 Upon determination, by the Department, of the necessity of updating the qualification for Tester/Technician, they shall be required to re-qualify.

12.4.4 Tester/Technicians shall be required to re-qualify within a specified time period determined by the Department (from the date of written notification by the Department). The notice shall be mailed to the address of record in the office of the Department. Failure to re-qualify within the required period of time shall result in suspension or revocation of the Tester/Technician certification as described in this Regulation.

12.5 Certification Expiration.

12.5.1 The Tester/Technician Certificate shall be issued annually and shall expire one year form the date of issuance. The Permit shall be renewable within sixty (60) days prior to the date of expiration.

12.5.2 It is the responsibility of the Tester/Technician to pursue the renewal of the Tester/Technician Certificate.

12.5.2.1 Tester/Technicians who are lacking in training hours may be allowed to have a single 30 day extension past the expiration of their Certificate provided they make a request to the Department in person prior the expiration date of their Certificate and:

(a) Pay the extension fee as referenced in Appendix B; and

(b) Show proof that they are registered for training that qualifies for re-certification.

12.5.2.2 When an extension has been granted and after the training is finished the Tester/Technician shall come back and apply for their renewal approval and pay the renewal fee (reference in Appendix B). If the Tester/Technician comes in after the extension expires with the proper training requirements he will be charged the late renewal fee.

13.0 EMISSIONS STANDARDS FOR MOTOR VEHICLES

13.1 In order to obtain a valid Certificate of Compliance, a motor vehicle subject to an Emission Inspection shall not exceed the maximum concentrations for carbon monoxide (CO), and Hydrocarbons (HC) as established by the Board of Health (referenced in Appendix C) and/or pass a valid OBD test.

13.2 Maximum concentration of Cut-Points shall be adopted by the Board of Health to meet the National Ambient Air Quality Standards established by the United States Environmental Protection Agency (USEPA). The adopted Cut-Points shall remain in effect until changed by the Board of Health. Any change in Cut-Points shall be effective upon the first day of any calendar month designated by the Board of Health. The Board of Health shall adopt Cut-Points by considering the following factors:

13.2.1 The existing ambient air quality;

13.2.2 To provide for the required stringency necessary to meet air Quality Standards;

13.2.3 The requirements for air quality programs currently in effect as promulgated by the EPA, the Utah Department of Environmental Quality, the County and the Board of Health; The Cut-Points established shall be part of an overall program, in accordance with EPA guidelines, to achieve the required tailpipe reductions, of CO and HC from motor vehicles measured from the date this program is implemented;

13.2.4 The general level of emission control technology on vehicles registered in the County;

13.2.5 Population growth and other factors which may reasonably be expected to impact CO and HC concentrations in the atmosphere;

13.2.6 The likelihood of a particular Cut-Point to achieve desired air quality goals; and

13.2.7 The ability to ensure compliance with the requirements of Section 41-6-163.6 and Section 41-6-163.7, Utah Code Annotated, 1953, as amended.

14.0 CERTIFICATE OF COMPLIANCE, CERTIFICATE OF COMPLIANCE NUMBERS AND CERTIFICATE OF WAIVER

14.1 No person shall make, issue or knowingly use any imitation or counterfeit of a Certificate of Compliance, Certificate of Compliance Numbers or Certificate of Waiver.

14.1.1 No person shall use a stolen certificate or certificate number.

14.2 Certificate of Compliance Numbers shall be obtained only from the Department.

14.3 No refund or credit shall be allowed for unused certificates/numbers, except as provided in Section 13.10.

14.4 Obtaining Certificate of Compliance Numbers.

· . (

- at 14.4.1 Certificate of Compliance Numbers may be obtained in person by
- a an authorized representative, of the I/M Program Station, possessing an acceptable form of identification.

14.4.2 Certificate of Compliance Numbers shall be issued in lots to be determined by the Department. The Department may limit the number of Certificate of Compliance Numbers issued to the number that the Department feels can be secured and stored safely.

14.4.3 Certificates of Compliance shall not be sold, loaned, transferred, or given to any other I/M Program Station, or any unauthorized individual. The I/M Program Station shall at all times account for all certificates/numbers that have been issued to the station. Failure to properly safeguard and/or account for Certificate of Compliance Numbers may lead to immediate suspension of Station Permit.

14.5 Certificates of Compliance shall only be issued to the vehicle owner/operator after being printed by the Analyzer. Completion of Certificates by other means than the Analyzer by any person or station other than the Department is strictly prohibited. The certificate shall be signed immediately after printing by the Tester/Technician who inspected the vehicle.

14.6 Certificates of Compliance shall not be issued until an inspection has been

performed as required by this Regulation.

14.7 All unused Certificate of Compliance Numbers shall be kept in a secure place at all times to prevent loss or theft.

14.8 Certificate of Compliance Numbers found to be missing, stolen, or unaccounted for, shall be reported to the Department within twenty-four hours and the Station shall cease performing Inspections until an investigation by the Department has been completed and the Department re-authorizes the Station to resume Inspections.

14.9 I/M Program Stations shall have Department issued Certificate of Compliance Numbers on hand at all times.

14.10 Upon final cancellation, suspension or revocation of the I/M Station Permit, the Station owner, manager or other responsible person shall immediately surrender all unused Certificates of Compliance Numbers to the Department. The Department may receipt and refund the fee paid for unused certificates of Compliance Numbers to the Station owner according to the Weber County Clerk/Auditor's procedures. Upon transfer or termination of business ownership, the Station Permit and all Certificate of Compliance Numbers shall be immediately surrendered to the Department. Any person acquiring a business that has been permitted as an official I/M Program Station, is prohibited from using any Permit, Certificate of Compliance Numbers or emissions documents issued to the former business without authorization of the Department; and

14.10.1 Any Analyzer manufacturer or their authorized representative who repossesses or otherwise removes an Analyzer from an I/M Program Station shall immediately notify the Department and shall immediately surrender any Certificate of Compliance Numbers and the data disk that may still be in the Analyzer to the Department.

14.11 No person may alter computer software or electronic data associated with the Inspection without written permission by the Department.

14.12 No person may engage in repair of analyzer unless approved by the Department.

15.0 RIGHT TO APPEAL

Within ten (10) calendar days after the Department has issued a notice of violation, Permits denial, warning, suspension or revocation, any person(s) aggrieved may request in writing, a hearing before the Department. The written request for hearing shall be made on a form provided by the Department. The hearing shall take place within ten (10) calendar days, or other time mutually
agreed upon, after there request is received. A written notice of the Department's final determination shall be given within ten (10) calendar days after adjournment of the hearing. The Department may sustain, modify, or reverse the action or order, and/or negotiate a consent agreement but shall not require less than the minimum requirements of this Regulation and other applicable law.

16.0 PENALTY

ì

16.1 Any person who is found guilty of violating any of the provisions of this Regulation, either by failing to do those acts required herein or by doing a prohibited act, shall be guilty of a class B misdemeanor pursuant to Section 26A-1-123, Utah Code Annotated, 1953, as amended. If a person is found guilty of a subsequent similar violation within two years, he shall be guilty of a class A misdemeanor pursuant to Section 26A-1-123, Utah Code annotated, 1953, as amended.

16.2 Each day such a violation is committed or permitted to continues shall constitute a separate violation. Also, each improperly issued Certificate of Compliance constitutes a separate violation.

16.3 The County Attorney, may initiate legal action, civil or criminal, requested by the Department to abate any condition that exists in violation of this Regulation.

16.4 In addition to other penalties imposed by a court of competent jurisdiction, any person(s) found guilty of violating any of this Regulation shall be liable for all expenses incurred by the Department in prosecuting and/or abating the violation.

16.5 The Penalty Schedule for Permits warning, Permits suspension, Permits revocation, and/or negotiated settlement agreements as adopted by the Board of Health shall be referenced in Appendix D of this Regulation and may be changed and updated by the Board of Health as deemed necessary to accomplish the purposes of this Regulation.

16.6 Enforcement of any criminal penalties does not preclude imposition of administrative or civil penalties and visa-versa.

17.0 SEVERABILITY

If any provision, clause, sentence, or paragraph of this Regulation or the application thereof to any person or circumstances shall be held to be invalid, such invalidity shall not affect the other provisions or applications of this Regulation. The valid part of any clause, sentence, or paragraph of this Regulation shall be given independence from the invalid provisions or application

and to this end the provisions of this Regulation are hereby declared to be severable.

18.0 EFFECTIVE DATE

.

This Regulation including Appendix A through F shall become effective the day of its adoption by the Board of Health. Appendices may be modified by the Board of Health without affecting the rest of this Regulation. Appendices when amended by the Board shall become effective on the day of adoption of amendments by the Board of Health.

Adopted by the Weber-Morgan Board of Health - May 22, 2000. Amendments to Regulation and Appendix B shall become effective July 1, 2003.

APPENDIX A

UTAH2000 Analyzer Specifications

Available upon request from the Department

:

All proposed options for test procedures, equipment specifications and program design shall meet emission reduction required by Section 9.18. The option to be selected is that which is most cost effective to the consumer as determined by the Board of Health.

The inspection for light duty vehicles (0-8500 lbs. GVWR) will consist of a stationary test at low and high speed idle for concentrations of hydrocarbons (HC), carbon monoxide (CO) and/or an OBD IM test, a functional inspection of the gas cap and a visual/tampering inspection of the fuel filler neck Restrictor, PCV, EGR, A.I.R. and catalytic converter systems.

All test equipment must meet specifications established by the Department. The Department may require Analyzers to meet portions of specifications established by the State of California Bureau of Automotive Repair termed Bar 97.

APPENDIX B

FEE SCHEDULE

The assessed fees for implementing the requirements of Section 6.10 of the Vehicles Emissions Inspection and Maintenance Program shall be:

PERMITTING OF AN OFFICIAL I/M PROGRAM STATION	\$250.00
ANNUAL RENEWAL OF STATION PERMIT	\$25.00
ANNUAL RENEWAL OF EXPIRED STATION PERMIT	\$50.00
RE-PERMITTING AN I/M STATION AT A NEW LOCATION	\$50.00
CERTIFICATE OF QUALIFICATION	\$25.00
TEMPORARY STATION PERMIT	\$50.00
ANNUAL RENEWAL OF TESTER/TECHNICIAN CERTIFICATION	\$10.00
TRANSFER OF TESTER/TECHNICIAN CERTIFICATE TO NEW STATION	\$10.00
DUPLICATE TESTER/TECHNICIAN CERTIFICATE	\$10.00
ANNUAL RENEWAL OF EXPIRED TESTER/TECHNICIAN CERTIFICATION	\$30.00
30 DAY EXTENSION OF TESTER/TECHNICIAN CERTIFICATE EXPIRATION DATE	\$30.00
AIR POLLUTION CONTROL FEE (PAID AT REGISTRATION ALL VEHICLES)	\$1.00
DIESEL CONTROL FEE (PAID AT REGISTRATION ALL DIESEL VEHICLES)	\$9.00
EMISSIONS CERTIFICATE OF COMPLIANCE/NUMBER (EACH)	\$3.00
DUPLICATE CERTIFICATE OF COMPLIANCE (MAXIMUM)	\$3.00
COST FOR CHECKING A CATALYTIC CONVERTER (DEPARTMENT ONLY)	\$25.00
MAXIMUM EMISSION INSPECTION FEE	\$25.00
CLEAR A TAMPERING LOCK OUT THIRD TIME WITHIN ONE YEAR AND EACH TIME AFTER	\$10.00
RETURNED CHECK FEE	\$25.00
COPY OF I/M PROGRAM REGULATION	\$2.50

APPENDIX C

MOTOR VEHICLE EMISSIONS Inspection and Maintenance PROGRAM

The following schedule gives the maximum allowable concentration for carbon monoxide (CO) and hydrocarbons (HC) for both cars and trucks as determined by an approved Analyzer using the prescribed procedures. The effective date for these cut-points is 1 November 1991.

ALL PASSENGER VEHICLES 1978 AND OLDER LIGHT DUTY TRUCKS 6,000 POUNDS GVWR OR LESS 1979 TRUCKS AND NEWER 8,500 POUNDS GVWR OR LESS		
MAXIMUM CONCENTRATION STANDARDS		
<u>MODEL YEAR</u> 1968 - 1969 1970 - 1974 1975 - 1976 1977 - 1979 1980 1981 AND NEWER	PERCENT CARBON <u>MONOXIDE</u> 6.0 5.0 4.0 3.0 2.0 1.2	PARTS/MILLION <u>HYDROCARBONS</u> 800 700 600 500 300 220
HEAVY DUTY TRUCKS AND VANS 1978 AND OLDER 6,001 OR GREATER 1979 AND NEWER OVER 8,500 POUNDS GVWR <u>MAXIMUM CONCENTRATION STANDARDS</u>		
<u>MODEL YEAR</u> 1968 - 1969 1970 - 1978 1979 - 1980 1981 AND NEWER	PERCENT CARBON MONOXIDE 7.0 5.0 4.0 3.5	PARTS/MILLION HYDROCARBONS 1500 1200 1000 800

Note: These should be considered as "cut-points" for maximum allowable emissions levels. Vehicles must never be reset to these emission levels when readjustments are made, but rather shall be adjusted using manufacturer's specifications. By using manufacturer's specifications, the emission levels should be well below the "cut-points".

APPENDIX D

PENALTY SCHEDULE

VIOLATION	1 st OCCURRENCE		3 RD OCCURRENCE	4 TH OCCURRENCE
Failure to Inspect or probing a vehicle other than vehicle entered in test record.	Up to 6 Month Suspension (Station & Tester/Technician)	Revocation (Station & Tester/Technician)		
Non-certified Tester/Technician performing inspection	Up to 6 Month Suspension (Station & Tester/Technician)	6 Month Suspension (Station & Tester/Technician)	Revocation (Station & Tester/Technician)	
Fail a Passing Vehicle/Pass a Failing Vehicle	Up to 3month Suspension (Station & Tester/Technician)	Up to 6 Month Suspension (Station & Tester/Technician)	Revocation (Station & Tester/Technician)	
Failure to Comply with Proper Test Procedures	Formal Warning (Station & Tester/Technician)	Up to 3 Month Suspension (Station & Tester/Technician)	Up to 6 Month Suspension (Station & Tester/Technician)	Revocation (Station & Tester/Technician)
Performing Unnecessary or Unrelated Repairs	1 Month Suspension (Station & Tester/Technician)	3 Month Suspension (Station & Tester/Technician)	Revocation (Station & Tester/Technician)	
Inaccurate or Incomplete Data	Formal Warning (Station & Tester/Technician).	Up to 30 Day Suspension (Station & Tester/Technician)	Up to 3 Month Suspension (Station & Tester/Technician)	Revocation (Station & Tester/Technician)

OTHER APPROPRIATE WARNINGS, SUSPENSIONS, NEGOTIATED CONSENT AGREEMENT, AND/OR REVOCATIONS AS DEEMED NECESSARY AND PRUDENT BY THE DEPARTMENT.

All Tester/Technician and Station suspensions may be reduced in length by a Negotiated Consent Agreement which may substitute monetary penalties for part of all of the suspension time. Consent Agreements for stations are based on 50% of the testing revenue that could have been expected during the suspension time based on a minimum test fee of \$20.00. Consent agreements for the technicians is based on \$100 increments for any 15 day period or portion thereof up to a maximum of 90 days. Negotiated Consent Agreements are only applicable in relation to suspension.

APPENDIX E

Department Approved Calibration Gas and Calibration Gas Blenders Available at:

List available upon request from Department

APPENDIX F

OBD IM TEST PROCEDURES

The following test procedure is to be followed for 1996 model year vehicles or newer:

1. Turn the ignition switch to the off position for at least 12 seconds;

2. Visually examine the instrument panel to determine if the Malfunction Indicator Light (MIL) illuminates when the ignition key is turned to the key on/ engine off position. (Note: MIL may only light briefly, watch closely.) If the MIL does not illuminate at all then the vehicle fails if tested after December 31, 2000, perform two speed idle test if tested prior to January 1, 2001 and advise the owner/operator to repair the MIL problem. Exception: OBDII deficient vehicles will be given the two speed idle test.

3. Locate the Data Link Connector (DLC) in the vehicle and connect the OBDII test equipment (Analyzer). If DLC is missing, has been tampered with, or is otherwise inoperable then the vehicle fails if tested after December 31, 2000, perform two speed idle test if tested prior to January 1, 2001 and advise the owner/operator to repair the MIL problem. Exception: OBDII deficient vehicles will be given the two speed idle test.

4. Start and leave the engine running. Check for MIL illumination. If the MIL illuminates after the engine has been started, even if no fault codes are present, the vehicle fails if tested after December 31, 2000, perform two speed idle test if tested prior to January 1, 2001 and advise the owner/operator to repair the MIL problem. Exception: OBDII deficient vehicles will be given the two speed idle test.

5. Vehicle status not ready; perform two speed idle test prior to January 1, 2001, after December 31, 2000 the vehicle must be preconditioned to a ready status. Exception: OBDII deficient vehicle will be given the two speed idle test. (Note: Certain vehicles may be exempted by the Weber-Morgan Health Department from OBD IM testing. Exemption can only be given by the Weber-Morgan Health Department.)

6. Vehicle status ready; when MIL light is on check Diagnostic Trouble Codes (DTC=s); determine pass or fail, and record results. (Note: Do not check DTCs if MIL is not commanded on.)

7. Turn off engine and disconnect test equipment.

WEBER-MORGAN DISTRICT HEALTH DEPARTMENT

POLICIES AND PROCEDURES OF THE WEBER COUNTY I/M PROGRAM

STATION AUDITS

- 1. I/M Program stations are to be audited on a quarterly basis with the high volume and high risk stations done on a more frequent schedule. Audit frequency should be altered to a degree to help prevent the predictability of the audit date by station personnel.
- 2. Monthly station/mechanic evaluation reports: Each auditor is routinely given a monthly report listing the activities of each station and mechanic in his district. He is instructed to review this printout and identify problem areas, i.e., high fail rates, low fail rates, tampered vehicle, etc. He then discusses these problems with the station manager and mechanic during the audit. These reports are then filed in the station folder for future reference and used as a means of determining if the problem has really been taken care of. Repeat violators are sent notices; and then suspended if the problem persists. Monthly evaluation printouts are also used to select stations for undercover audits (low-high fail rates).
- 3. One copy of the station evaluation and one copy of the station performance report will be obtained during each audit.
- 4. The station performance report will be completed during each audit. Any violation(s) will be noted and station manager or responsible individual informed regarding the violation(s).
- 5. Gas audits of station analyzer shall be performed on a quarterly basis by the station auditor. Using span gas provided by the Department, the analyzer readings must be within a tolerance of +/-5% for co and +/7% with hang-up correction for HC. If analyzer fails the gas audit, the auditor shall perform analyzer maintenance and attempt another gas audit. If this attempt fails the auditor shall place the analyzer out of service, in accordance with the Weber County I/M Regulations, Section 6.9, Subsection 6.9.2.1. If analyzer is locked out due to a gas audit failure, it shall be noted on the station performance report and station manager shall read and initial and date this statement. Once notification of repair is received from station personnel, the station auditor shall make every effort to remove lockout within the same date of notification. If same date lockout removal cannot be achieved, then next day will be acceptable.
- 6. If a station auditor notes a consistent violation (two consecutive violations), that the I/M station does not correct in spite of violation notices on performance reports, and has been given a reasonable amount of time for any corrections, then

this information is to be given to the I/M Program Director for administrative action.

- 7. All items covered on the station performance report are to be checked on each audit. Of particular concern is the certificate accountability. All unaccounted certificates will constitute a station to cease inspection operations until the Department authorizes the station to resume inspections, as per Weber County I/M Regulations, Section 13.0, Subsection 13.8.
- 8. Scheduling: Auditors routinely vary the date that they visit each station. This reduced the possibility that the station will prepare for the audit just before expected visit. This has caught some stations off guard and resulted in the discovery of improprieties that otherwise would go undetected. The policy of using the supervisor to perform random audits has also helped to take away a predicted routine that stations and mechanics could abuse.
- 9. During regular audits, the analyzer's data disk will be removed from the analyzer and exchanged for a new, formatted disk provided by the Department. Old disks will be discarded after the information has been downloaded to the system.
- 10. The analyzer information disk is to be returned to the office and should be transferred to the county mainframe on a timely basis. The disk should then be saved for future reference or until it has been determined that the information has been successfully uploaded.
- 11. Where possible, the station auditor should attempt to observe an actual emissions test during the station audit.
- 12. In the event that a station comes under investigation for alleged misconduct, the station's auditor will conduct a station audit immediately and turn over all information to the Department Director for any administrative action as deemed by the Weber County I/M Regulations and the Department Director.

AUDIT PROCEDURES - UTAH 91 ANALYZERS

- I. Prior to the audit:
 - A. Review station folder to determine if there is any follow-up corrective action required from the previous audit.
 - B. View the performance report to determine if there are nay missing certificates.
 - C. Obtain calibration cylinder, vehicle identification reference manual. (Or use station reference manual)
- II During the audit:

- A. Obtain station performance report. Verify that the station permit, mechanic's certificates, station sign, fee chart, reference manuals, and other required equipment is available. Note any discrepancies for the station performance report.
- B. Access the station analyzer audit menu of the analyzer computer. <u>Select</u> <u>the station evaluation report</u> (see attachment no. 1) and print out the information. Review this information to see if there are any unusual entries, i.e., a significant increase/decrease in the fail rate. Note and question the station manager on any unusual information. Verify that the time and date entries are correct.
 - 1. <u>Select the station performance report</u> (see attachment no. 2) and program in the appropriate information. Note on this report the number of the next unused certificate.
 - 2. Select gas audit and introduce your audit calibration gas into sample line probe. Obtain printout of results (see attachment no. 8). Calculate the reading that you are supposed to observe and compare this reading with that on the printout. If it is determined that the analyzer is out of calibration, select the calibration history and review previous calibration reports (attachment no. 9). Determine if there is any indication of drift beyond the maximum allowed since the last calibration. If so, schedule follow up checks to determine how long the analyzer will remain in calibration. If it is determined that the analyzer cannot remain in calibration for three days, it will be locked out until repairs are completed.
 - 3. Select the mechanic evaluation report (see attachment no. 3) and program in each mechanic. Obtain printouts and review for unusual entries, i.e., <u>significant fail or pass rates</u>. Any significant discrepancies should be noted and discussed with the station manager.
 - a. Select multiple repairs report (see attachment no. 4) for suspect mechanics and obtain printout. May refer back to station/mechanic evaluation report. Unusual entries will be discussed with station manager.
 - b. Select fast testing report (attachment no. 5) and review for unusual entries. If observed, obtain printout for documentation of problem.
 - c. Select emissions reductions report (attachment no. 6) and review for 50% reductions in HC and CO if a significant number of repairs have been performed. If not, obtain

printout for follow up action or to discuss with I/M mechanic or other responsible individual.

d. Select consecutive test comparison report and observe to see if there are any repeat tests with the same readings. If so, obtain a printout and discuss with the station manager or notify the Director to do an undercover audit there. You may also want to printout test records to verify the same readings indicating that the same vehicle was probably used to each test.

4. Select auditor's notes (see attachment no. 10) and review to determine if there are any unresolved discrepancies that require follow up. Also, use this opportunity to record any information that you want to pass on to the next audit.

5. Select analyzer tampering/access report (see attachment no. 11) and review for lockouts. Insure that county lockouts involving everything but service are cleared by the auditor.

- C. Review station files:
 - 1. Using the vehicle identification encyclopedia, randomly select a few vehicle inspection reports (VIRs) and check for proper entries for engine displacement. Then using the application guide, verify that the device check is properly programmed on the report. If a significant number for a particular mechanic is observed, check more thoroughly to determine the extent of the problem. Bring this information to the attention of the station manager and mechanic.
 - 2. Insure that the VIRs are filed in numerical sequence and that they have the yellow copy of the certificate of compliance attached.
 - 3. Insure that separate files are available for fail and pending vehicles awaiting retest.
- III After the audit:
 - A. Advise supervisor of any discrepancy involving a warning or suspension letter and provide documentation to support it.
 - B. Compile all information and file in the station folder.
 - C. Complete I/M station audit report indicating when the audit was performed.

COMPLAINTS

1. When and emission related complaint is received, the secretary will retrieve the following information on an approved complaint form: complainant's name, address, home and/or work phone, station at which incident occurred, inspectors name, dated and time of the incident and a description of complaint. Complaint will then be assigned to a station auditor who in a timely manner will investigate the complaint, take appropriate action, file a report with the Director, and notify the complainant as to the results of the investigation.

POLICIES AND PROCEDURES COVERT AUDITS

- 1. Covert audits. The counties to the extent possible perform a covert audit of each inspector and station at least once a year. The number of covert audits at least equals the number of permitted inspectors. Covert audits are performed using a variety of vehicles that are representative of the subject malfunctions. Suspected problem stations and inspectors are targeted for earlier and more frequent audits. Complaints also trigger additional audits.
- Covert performance audits shall include: Remote visual observation of inspector performance, which may include the use of aids such as binoculars or video cameras, at least once per year per inspector in high-volume stations (i.e., those performing more than 4000 test per year);

Site visits at least once per year per number of permitted inspectors (per inspector FTE) using covert vehicles set to fail (this requirement sets a minimum level of activity not a requirement that each inspector be involved in a covert audit); and

For stations that conduct both testing and repairs, at least one covert vehicle visit per station per year including purchase of repairs and subsequent retesting if the vehicle is initially failed for tailpipe emissions, at least once per year per station.

- 3. Vehicles used for undercover audits will be pre-tested and adjusted for use at the Technical Center.
- 4. The vehicle operator must follow the guidelines in memorandum dated 20 March 1986 from deputy County Attorney regarding "Guidelines to Avoiding Entrapment".
- 5. The audit should be performed in such a manner as to avoid giving employees at the station being audited any indication that an audit is being performed.
- 6. The surveillance report form should be filled out completely for each phase of the audit.

- 7. Covert auditors will complete a training course provided by the Department which is designed to thoroughly familiarize them with all aspects of the emission inspection procedures.
- 8. Covert auditors will be rotated to avoid becoming recognizable to inspection station personnel.
- 9. In all cases, vehicles shall be adjusted to fail either the emissions or tampering or both aspects of the test.
- 10. On covert audit failures, administrative actions will follow the penalty schedule of the Weber County I/M Regulations.
- 11. A complete audit includes the following phases:
 - a. Pre-testing and adjusting of vehicle.
 - b. Audit/testing at designated station.
 - c. Post-testing and readjusting original configuration if needed. Completed inspection forms are to be returned to the Program Director
 - d. A briefing with Director of his representative on test results.
 - e. Appropriate action taken on any discrepancies.

VEHICLE TAMPERING INSPECTION PROTOCOL

In accordance with Weber County Inspection/Maintenance Rules and Regulations, Section 1.77 Tampering Definitions, Section 3.0 Jurisdiction of Counties, Section 4.1.5 Powers and Duties, and Section 4.2.3 Powers and Duties.

All Weber County Inspection/Maintenance inspector/mechanics when conducting a tampering inspection, will determine the following on all vehicle model years included in the program:

1. Through the use of the vehicle decal or an approved vehicle reference manual, determine those emission control devices that should be present on a particular vehicle.

2. Conduct a visual inspection of said vehicle to determine if those items are indeed present.

3. Determine if those items have been intentionally or accidentally removed or altered from the original configuration.

4. Enter pass/fail information correctly into the analyzer.

5. Inspectors will be held accountable for their tampering inspections. If an inspection is found to be incorrect, and inspector may be subject to penalties found in the Inspection/Maintenance Penalty Schedule.

ENFORCEMENT AGAINST STATIONS AND INSPECTORS

1. A penalty schedule to be used in the case of violations by individual inspectors and stations. The penalties escalate based upon the frequency

of commission and relative severity in terms of a violation's direct impact on the emission reduction potential of the program. Penalties shall take the form of suspension of license revocation with annual competency re-certification requirements.

- 2. Auditors are granted the authority to immediately suspend station and inspector licenses upon finding a violation. A formal hearing is conducted to evaluate evidence and determine penalty. The I/M program management shall record all enforcement activities, including all warnings, fines, suspensions, revocations and other notices of violation. At least once a year, the program shall compile summary statistics on its enforcement activities and report this information to the public and EPA.
- 3. In the case of inspector incompetence, the inspector is required to be retrained and must successfully demonstrate the ability to perform the test procedure prior to the restoration of testing privileges. For cases of inspector violations resulting from factors other than incompetence, the inspector shall be suspended from testing for a minimum of one month, with such suspension increasing with the severity and frequency of violation, leading ultimately to permanent license revocation.

VIOLATION	1ST OCCURRENCE	2ND	3RD	4TH
		OCCURRENCE	OCCURRENCE	OCCURRENCE
INTENTIONALLY	6 MONTH	REVOCATION		
PASSING A	SUSPENSION			
FAILING VEHICLE				
FAILURE TO	UP TO 6 MONTH	REVOCATION		
INSPECT	SUSPENSION			
NON-CERTIFIED	UP TO 6 MONTH	REVOCATION		
INSPECTOR	SUSPENSION			
PASS A FAILING	UP TO 3 MONTH	UP TO 6 MONTH	REVOCATION	
VEHICLE	SUSPENSION	SUSPENSION		
FAIL A PASSING	UP TO 3 MONTH	UP TO 6 MONTH	REVOCATION	
VEHICLE	SUSPENSION	SUSPENSION		
FAILURE TO	FORMAL	UP TO 3 MONTH	UP TO 6 MONTH	REVOCATION
CALIBRATE OR	WARNING	SUSPENSION	SUSPENSION	
OTHER			l	
PROCEDURES				
PERFORMING	1 MONTH	3 MONTH	REVOCATION	
UNNECESSARY	SUSPENSION	SUSPENSION		
OR UNRELATED				
REPAIRS				
INACCURATE OR	FORMAL	UP TO 30 DAY	UP TO 3 MONTH	REVOCATION
INCOMPLETE	WARNING	SUSPENSION	SUSPENSION	
DATA				
OTHER APPROPR	RIATE WARNINGS,	SUSPENSIONS, NEC	GOTIATED CONSEN	T AGREEMENTS,
AND/OR REV	OCATIONS AS DEE	MED NECESSARY A	ND PRUDENT BY 1	HE DIVISION

APPENDIX D PENALTY SCHEDULE

All suspensions may be reduced in length by a negotiated consent agreement which may substitute monetary penalties for part or all of the suspension time. Negotiated Consent Agreements are only applicable in relation to suspension.

UTAH STATE IMPLEMENTATION PLAN

SECTION X

VEHICLE INSPECTION AND MAINTENANCE PROGRAM

PART E

WEBER COUNTY

.

Adopted by the Utah Air Quality Board November 3, 2004

Section X, Part E, page i

Table of Contents

1. I/M performance standard	1
2. Network type	2
3. Tools and resources	2
4. Test convenience	2
5. Vehicle Coverage	2
6. Test procedures and standards	
7. Test Equipment	5
8. Quality Control	6
9. Waivers	7
10. Motorist compliance enforcement	
11. Motorist compliance enforcement program oversight	9
12. I/M Program quality assurance	13
13. Enforcement against stations and inspectors	11
14. Data collection	12
15. Data analysis and reporting	13
16. Inspector training and permitting	13
17. Public information and consumer protection	14
18. Improving repair effectiveness	15
19. I/M SIP implementation	15

SECTION X, PART E WEBER COUNTY Appendices

1 Vehicle Emissions Inspection/Maintenance Program regulation revised May 12, 2003.

2 Audit Policies

.

•

.

UTAH STATE IMPLEMENTATION PLAN SECTION X AUTOMOTIVE INSPECTION AND MAINTENANCE (I/M) PROGRAM PART E WEBER COUNTY

1. I/M performance standard

Federal requirements EPA's I/M regulation, 40 CFR Part 51, Inspection and Maintenance Program Requirements last amended at 66 FR 18156, April 5, 2001, specifies a model Basic I/M program. Utah is required by Section 182 of the Clean Air Act to implement an I/M program in Weber County that is at least as effective as the EPA's Basic Performance Standard. The Basic I/M performance standard is specified in 40 CFR 51.352. While local governments have flexibility to implement programs best suited for their area, EPA's regulations require a performance demonstration that local I/M programs result in automotive emissions equal to or less than predicted for the EPA model I/M program. State and local governments may choose options best suited for their area to meet the performance standard.

UM Program MOBILE modeling Mobile source emission factors are determined by use of the most recent release of EPA's MOBILE model. The MOBILE model is able to calculate emission factors, grams of a particular pollutant per vehicle mile traveled across the fleet in an area (G/VMT), given information about the fleet, climate, fuel characteristics and I/M programs in a local area. The latest approved release of the MOBILE model was used for the I/M performance standard demonstration analysis and the documentation is included in the respective criteria pollutant section of the SIP.

I/M Program Performance Standard Weber County's I/M program exceeds the Basic I/M performance standard for all pollutants, although the EPA only requires the demonstration for each pollutant which caused an area to be subject to an I/M program. The Ogden City is a Particulate Matter (PM10) non-attainment area and a carbon monoxide (CO) National Ambient Air Quality Standard (NAAQS) maintenance area.

I/M Program Improvements Weber County is committed to implementing improvements to its Basic I/M Program, if necessary to achieve air quality standards. The Weber-Morgan Health Department Board of Health, within one year of written notification by the Executive Secretary that Weber County has violated the CO NAAQS, will enact Basic I/M program improvements. These contingency measures are identified in the Ogden City Carbon Monoxide (CO) Maintenance Plan. These include improved detection and elimination of illegal registration of vehicles, lowered emission cut points in all tested vehicles, application of the vehicle tampering requirements to additional years and improved technician training and certification.

2. Network type

Weber County's I/M program is a basic, decentralized, test-and-repair network. On-Board Diagnostic (OBD) compliant 1996 and newer model year vehicles undergo an OBD inspection. The Weber-Morgan Health Department regulation administering the program is provided in Appendix E.1.

3. Tools and resources

Funding mechanisms Weber County's I/M program is funded through several mechanisms including a \$1 air pollution control fee for each vehicle registered in the county. I/M Certificates are sold to I/M test stations for \$3.25 each. The county also charges fees for various permitting activities. The fees are dedicated to the I/M program. A fee schedule can be found in an Appendix to Weber County's I/M Program ordinance.

I/M program funding requirements Weber County will allocate funding as needed to comply with the relevant requirements specified in Utah's SIP; Utah statutes; county ordinances, regulations and policies; and the federal I/M program regulation.

4. Test convenience

There are approximately 110 permitted Basic I/M stations within Weber County. Specific operating hours are not specified by the county. Some stations that test and/or service only one type of vehicle are permitted. There are also government and private fleet permitted stations that are not open to the public.

5. Vehicle coverage

Subject fleet All model year 1968 and newer model year light duty vehicles, light duty trucks, and heavy duty trucks registered or principally-operated in Weber County are subject to the I/M programs except for exempt vehicles as specified in the Weber County I/M Ordinance. All vehicles six years old and newer on January 1 are tested every other year; older vehicles are tested annually as per Utah Code 41-6-163.6(6) amended in 2002.

Alternative fuels Vehicles operated on alternative fuels such as propane, alcohol, and natural gas are also subject to the program. Dual-fueled vehicles are tested twice, once on each fuel.

Government fleet Section 41-6-163.6(1)(b) of the Utah Code requires that all vehicles owned or operated in the county by federal, state, or local government entities comply with the I/M programs.

Vehicles owned by students and federal employees Section 41-6-163.3(5) of the Utah Code requires universities and colleges located in Utah's I/M areas to require proof of compliance with the I/M program for vehicles that are permitted to park on campus regardless of where the vehicle is registered. Vehicles operated by federal employees and

Section X, Part E, page 2

operated on a federal installation located within an I/M program area are also subject to the I/M program regardless of where they are registered. Proof of compliance consists of a current vehicle registration in an I/M program area or an I/M certificate of compliance or waiver, or evidence of exempt vehicle status as specified in this section.

Farm truck exemption Eligibility for the farm truck exemption from the I/M programs is specified in Section 41-6-163.6(4) of the Utah Code and must be verified in writing. The owner must sign an affidavit on Utah State Tax Commission form TC-838 that vehicle use will be limited to agricultural activities.

Diesel vehicle exemption Persons registering a diesel vehicle that was registered as a gasoline vehicle are required to present the vehicle to the Weber County I/M Technical Center for verification of fuel type..

New vehicle exemption Proof that a vehicle is new and being registered for the first time is established by presentation of a Manufacturer's Statement of Origin (MSO) at the time of registration.

Out-of-state exemption Vehicles registered in an I/M county but operated out-of-state are eligible for an exemption. The owner must complete Utah State Tax Commission form TC-810 in order to be registered without inspection documentation from Weber County. The owner must explain why the vehicle is unavailable for inspection in Utah. Common situations include Utah citizens that are military personnel stationed outside of the state, students attending institutions of higher education elsewhere, and people serving missions. If the temporary address of the owner must submit proof of compliance with that I/M program at the time of, and as a condition precedent to, registration or renewal of registration. The vehicle owner must identify their anticipated date of return to the state and is required to have the vehicle inspected within 10 days after the vehicle is back in Utah, unless they can demonstrate that the vehicle had passed an I/M inspection in another area. Weber County maintains a record of such exemptions and requires submission of an I/M inspection certificate or waiver at the indicated time.

Exempt vehicle The following vehicles are exempt from inspection: motorcycles, electric powered vehicles, new vehicles registered for the first time, model year 1968 and older vehicles, farm vehicles and equipment, construction equipment, and other off-road vehicles.

Unregistered vehicles I/M ordinances and regulations require that vehicles available for rent or use in Weber County be subject to its I/M program. To the extent practicable, all vehicles principally operated within the county are subject to the I/M program.

6. Test procedures and standards

Specifications Detailed specifications for the I/M test procedures and standards are described in the Weber County I/M ordinance provided in Section X, Part E, Appendix

E.1. Specifications for the test procedure and equipment were developed according to good engineering practices to ensure test accuracy.

Test procedure and analyzer The Basic I/M program is compatible with EPA's PRECONDITIONED TWO SPEED IDLE TEST as specified in EPA-AA-TSS-I/M-90-3 March 1990, Technical Report, "Recommended I/M Short Test Procedures for the 1990's: Six Alternatives. 1996 and newer vehicles are tested using OBD II test procedures. All Basic emissions inspections are performed using the UTAH2000 Analyzer, a BAR97type emissions analyzer. The UTAH2000 Analyzer calibration specifications and emissions test procedures meet the minimum standards established in Appendix A of the EPA's I/M Guidance Program Requirements, 40 CFR Part 51 Subpart S.

Covered vehicles are defined in Section 5 above. All covered vehicles in Weber County are subject to the Basic test procedure and inspected using the UTAH2000 analyzer as specified in this section.

Pre-inspection emissions-related repairs Inspectors in the county's test-and-repair networks are required to perform the emissions test prior to making any emissions-related repairs when a vehicle is presented for an emissions inspection. All inspectors who conduct test-only inspections, are required to ask the vehicle owner or operator whether a tune-up or other emissions-related repairs have been performed within 6 weeks prior to the emissions inspection and to document the owner's response in the UTAH2000 computer database.

Safety issues Vehicles presented in unsafe condition must be repaired before inspection. Vehicles are also subject to an annual safety inspection administered by the Highway Patrol. Submission of proof of compliance with the safety program is also required as a condition for registration or renewal of registration. Most owners in Utah's test-andrepair networks have the safety and emissions inspection performed at the same time. Data relative to the safety inspection can be recorded in the UTAH2000 Analyzer. Weber County's I/M program is administered with close cooperation with the Utah Highway Patrol Safety Program.

Emission standards Vehicles must pass both the hydrocarbon and carbon monoxide emission standard regardless of the NAAQS attainment status of the county. The emission standard for the Basic I/M program was used in the MOBILE modeling that was conducted to demonstrate compliance with the Basic I/M performance standard.

Stringency The Weber County I/M program will adjust tailpipe emission standards as necessary to maintain a stringency rate of at least 22% for pre-81 model year vehicles, the stringency rate used in the Basic I/M performance standard modeling demonstration.

Re-test standards The same test procedure and emission standards are used for initial tests and retests, regardless of which part a vehicle may have failed during an initial test. Weber County's I/M test procedure requires an official test, once initiated, to be

performed in its entirety regardless of intermediate outcomes, except in the case of invalid test conditions, unsafe conditions or the fast pass/fail algorithms.

Anti-tampering provisions Regardless of the vehicle model year, Weber County does not allow waivers for tampered vehicles or money spent to repair tampered or missing emission control devices to be applied towards a minimum waiver cost. The county requires repair of any tampering of the air system, catalyst, exhaust gas recirculation (EGR) valve, evaporative system, positive pressure crankcase valve (PCV), and gas cap on model year 1990 and newer vehicles.

Engine changes Weber County's I/M ordinance has a section that addresses engine changes. After an engine change, vehicles are tested to the tailpipe emission standards and anti-tampering requirements applicable to vehicles of the engine model year. Mixing vehicle classes (e.g., light-duty with heavy-duty) and certification types (e.g. California with federal) within a single vehicle is considered tampering.

Fuel switching Vehicles that are switched to a fuel type for which there is no certified configuration are tested according to the most stringent emission standards for that vehicle model year and vehicle type.

7. Test Equipment

Specifications The UTAH2000 Analyzer is a BAR97-type computerized emissions analyzer. Additional written technical specifications for Weber County's I/M test equipment are specified in Weber County's I/M ordinance.

Analyzer access restrictions An inspector access code is required to use the UTAH2000 analyzer for official tests, a service access code to repair or service the analyzer, and an auditor access code to access the audit functions. DOS functions are not accessible to station owners or inspectors. County I/M auditors make programming changes from disks supplied by the analyzer manufacturer.

Data security provisions Manual data entry is minimized. For initial inspections, the inspector enters vehicle registration and vehicle information from the keyboard. Data elements are described in the UTAH2000 analyzer specifications. For retests, the inspector calls up the initial test file, compares the vehicle and owner data, and confirms the VIN/license plate data. Data regarding inspections, analyzer calibration and service, lockout activities, and audit information are downloaded to the county vehicle identification database daily; data from each analyzer is downloaded once or twice weekly.

Automated test procedure The UTAH2000 analyzer automatically reads all test measurements, records test results in the computer database, determines whether the vehicle has passed or failed a test, and prints vehicle inspection reports and inspection certificates for all subject vehicles. The analyzers are capable of simultaneously sampling dual exhaust vehicles. The analyzer bench includes two non-dispersive infrared (NDIR)

Section X, Part E, page 5

analyzers for carbon monoxide, carbon dioxide, and hydrocarbon measurements (one low range and one high range), and one NDIR analyzer for carbon dioxide measurement. The test procedure is automated to the highest degree practical to minimize the potential for intentional fraud and/or human error.

Security lockouts The analyzers are programmed to trigger lockouts when abuse or tampering occur. Lockouts occur after any security system is tampered, failure to conduct or pass periodic calibration tests, or the data recording medium is full. The analyzer cannot be used until a Weber County I/M auditor has cleared the lockout. The analyzer automatically keeps an electronic record of all lockouts including the date of the lockout, the reason for the lockout, and the date and person that cleared the lockout.

Certified analyzer use restriction Since March 1, 2000, the Weber County Basic I/M program requires that official emissions tests be conducted only on registered, jointly certified UTAH2000 analyzers. There have been several updates of the UTAH2000 Analyzer specifications to date and more will follow, as necessary, to accommodate new technology vehicles and changes to the program.

8. Quality Control

General quality control specifications Weber County's I/M Program, the UTAH2000 Analyzer specifications, and current I/M program ordinances and regulations were carefully designed to insure that emission measurement equipment is calibrated and maintained properly, and that inspection, calibration records, and maintenance records are accurately created, recorded, and maintained. The specifications meet the test equipment quality assurance practices described in 40 CFR 51 Subpart S Sec. 51.359 and Section X, Appendix A.

Automatic electronic quality assurance features Operational analyzer quality assurance measures such as analyzer calibration, zero and span check, hydrocarbon hang-up check, and leak check are mandatory automatic analyzer capabilities. Gas accuracy tolerances, dilution limits, analyzer warm up requirements, system response time requirements, optical correction factors, and interference effects are also addressed in the analyzer specifications. If the checks are not performed on schedule or identify measurements outside of acceptable limits established in the specifications, a lockout occurs preventing use of the analyzer until such problems are corrected. See Sections 2.12, 2.13, and 2.18 of the UTAH2000 Analyzer specifications. Records of all quality assurance activities with respect to the analyzer are automatically recorded in the analyzer's electronic database and evaluated by Weber County I/M auditors on a regular basis. Section 1.7 discusses requirements for assurance that unauthorized access to the I/M database in the analyzer is secure. Attempts to deliberately avoid or defeat analyzer or inspection quality assurance provisions result in disciplinary action against the I/M mechanic and/or station.

Analyzer maintenance Section 1.8 of the UTAH2000 Analyzer specifications describes required services, warranty provisions, and documentation that analyzer manufacturers must provide to customers. It includes ensuring that the analyzer meets the quality

assurance specifications at the time of delivery, that routine quarterly preventative maintenance is performed, training on how to use, maintain, and operate the analyzer is provided by the manufacturer, and that if repair of defects can not be made promptly a temporary analyzer replacement is provided. Service activities are recorded in the analyzer's electronic database. Weber County has conducted a survey of analyzer owners to determine compliance with these provisions. Failure of an analyzer manufacturer to meet quality assurance specifications could result in de-certification of that manufacturer's product for use in Utah.

Document security Document security was a high priority during the UTAH2000 analyzer design phase. The analyzer tracks the unique certificate numbers and ensures that the printed certificate matches the test number. Missing certificate numbers are stored in the analyzer database for auditor review. The blank certificates are commercially printed on counterfeit-resistant security paper.

Analyzer certification Sound engineering practices were followed during the design and certification of the UTAH2000 analyzer to insure accurate and repeatable inspections under a range of environmental conditions. Manufacturer owner's manuals, operating instructions, and warranty provisions were also reviewed during the certification process. Comprehensive records of the certification process have been maintained.

Analyzer security provisions Weber County's I/M ordinance requires use of a certified and registered UTAH2000 analyzer for official inspections. Inspection records include the analyzer registration number. The ordinances and regulations make it illegal to alter analyzer software or hardware without written approval. Analyzer calibration requirements, maintenance, and warranty provisions are also specified in the Weber County I/M ordinance.

9. Waivers

Waiver rate Weber County will take corrective action as needed to maintain a maximum waiver rate of 5% of the initially failed vehicles or the Utah Air Quality Board will revise the SIP and emission reductions claimed based on the actual waiver rate. The conditions for issuing waivers legally authorized and specified in the Weber County I/M ordinance meets the minimum waiver issuance criteria specified in 40 CFR Subpart S 51.360.

Waiver procedures A waiver document may be issued only by Weber County I/M technical center staff and only after verification of required documentation. Any tampered, missing, or inoperable emission control devices must have been replaced or repaired. At least \$100 for 1968 through 1980 model year vehicles, \$200 for 1981 through 1995 and \$400.00 for 1996 and newer model year vehicles must have been spent on acceptable emission repairs as verified by a Weber County I/M program auditor by physical examination of the vehicle and review of the repair documentation. Repair documentation, such as receipts, are copied and retained by the auditor to prevent reuse. Weber County requires that emissions-related repairs be performed by a licensed auto repair business in order to count the labor costs. Any vehicle that experiences an increase

in all emissions levels is not eligible for an emissions repair waiver regardless of the amount spent to repair the vehicle. In the state of Utah, vehicles still under the federal emissions warranty are not eligible for a waiver until all warranties are exhausted. Warranted repair and tampering repair may not be applied to the repair cost waiver limits. Waivers are only valid for one test cycle. The vehicle owner surrenders the original waiver document at the time of registration; copies are not accepted for registration purposes. Specific provisions regarding waivers may be found in Weber County's I/M ordinance and the Utah State Tax Commission Division of Motor Vehicle policy manual that is available upon request. The I/M program in Weber County does not provide for time extensions to relieve economic hardships in obtaining emission-related repairs.

10. Motorist compliance enforcement

Registration denial Weber County's I/M program is enforced by means of registration denial. Vehicle owners must present proof of compliance with the I/M program, a waiver, or evidence of exemption from the I/M program as a condition precedent to vehicle registration or registration renewal. See sections 4 and 6 above for a more detailed discussion of inspection frequency, inspection scheduling, license plate requirements, and enforcement of the registration requirements. Citations are routinely issued to operators of vehicles with expired or missing license plates during routine traffic stops, parking lot inspections, and roadblocks. As specified in Section 41-1a-1303 of the Utah Code, driving without registration is a Class C misdemeanor. The penalty for a Class C misdemeanor is imprisonment of no more than 90 days and \$750 for persons or up to \$1000 for corporations, associations, partnerships, or government instrumentalities. In addition to paying a fine the motorist must register the vehicle. It is currently a Class B misdemeanor to violate a county I/M regulation or ordinance. The penalty for a Class B misdemeanor is imprisonment not exceeding six months and, for persons, a fine of up to \$1000 or, for corporations, associations, partnerships, or government instrumentalities, a fine of up to \$5000. In Utah, the magnitude of such penalties is a judicial rather than an administrative decision. Per Section 41-1a-1315 falsification of evidences of title and registration is a second degree felony.

Certificate of Compliance The Certificate of Compliance is dated by the UTAH2000 analyzer in Weber County immediately after a passing inspection is completed. The certificate is only valid for registration purposes for two months. At the same time the analyzer also prints the following information on the certificate to ensure unambiguous vehicle identification: the vehicle identification number (VIN), license number, model year, make, and model. A sample of the Certificate of Compliance is in Appendix C of the UTAH2000 specifications. The certificates are only printed in the event that the vehicle passed the emissions inspection. Separate documentation, including the same vehicle information, is used for waivers.

Fuel changes to non-subject status Vehicle changes that would result in registration changes from a subject to exempt status require physical confirmation by Weber County I/M program personnel at the I/M technical center. Falsification of registration or title information is a felony offense.

Title transfers Proof of compliance with the I/M program is required for a title transfer. The system ensures that owners are not able to avoid the program by extending the inspection date through manipulation of the title and registration system.

Weber County I/M program staff, peace officers, and the Utah State Tax Commission Motor Vehicle Customer Service Division routinely work together to ensure that motor vehicle owners that move into an I/M program area complete registration transfer including compliance with the I/M program. Except for higher education students and active duty military personnel, people are required to register their vehicles in the county in which they are domiciled. As discussed in the Vehicle Coverage section, although these two exempted classes of vehicle owners do not have to register their vehicles in Utah, they do have to comply with the I/M programs. Employment status, maintenance of a residence, enrollment of children in local schools, and voting districts are considered when identifying persons in violation of this requirement.

The Weber County I/M program staff work with citizens, the Motor Vehicle Customer Service Division and county attorneys to identify and prosecute people that illegally transfer registration to a non-subject area to avoid the I/M program. The process is very labor intensive. There are many legitimate reasons to be operating a vehicle in an I/M program area that is registered elsewhere. Violators must be dealt with on a case-by-case basis. Persons caught are subject to fines. Those prosecuted and convicted could end up with a criminal record and actual jail time. Fraudulent registration of a motor vehicle is a felony offense. To date, most people promptly comply when confronted with evidence of their guilt and the seriousness of their offense. The involved agencies are developing more efficient methods of dealing with illegal registrations that result in exemption from the I/M programs.

Weber County is committed to a cooperative aggressive effort to ensure that vehicles operated in the county comply with the I/M program to ensure a compliance rate of at least 95%.

11. Motorist compliance enforcement program oversight

Utah State Tax Commission, tax assessors, and county roles The Utah State Tax Commission Motor Vehicle Customer Service Division and county tax assessors deny application for vehicle registration or renewal of registration without submission of a valid certificate of compliance, waiver, or verified evidence of exemption. Proof is retained by the tax clerk, micro-photo-copied, and then destroyed. Altered or handwritten documents are not accepted. All certificate data is collected by Weber County I/M program auditors and subjected to scrutiny for evidence of any improprieties.

Database quality assurance The vehicle registration database is maintained and quality assured by the Motor Vehicle Customer Service Division. The I/M inspection database is maintained and quality assured by Weber County I/M program staff. See Appendix F of the UTAH2000 analyzer specifications for a file layout description. The Weber County I/M program staff have access to the Motor Vehicle Customer Service Division database and use it on a regular basis for quality assurance purposes. The databases are subject to regular auditing, cross-referencing, and analysis.

Oversight provisions The oversight program includes verification of exempt vehicle status through inspection, data accuracy through automatic and redundant data entry for most data elements, an audit trail for program documentation to ensure control and tracking of enforcement documents, identification and verification of exemptiontriggering changes in registration data, and regular audits of I/M inspection records, I/M program databases, and the Motor Vehicle Customer Service Division database.

Enforcement staff quality assurance I/M program auditors and tax clerks involved in vehicle registration are subject to regular performance audits by their supervisors. All enforcement personnel (direct and indirect) involved in the motorist enforcement program are subject to disciplinary action, additional training, and termination for deviation from procedures. Specific provisions are outlined in the Motor Vehicle Customer Service Division procedures manual, the county I/M audit policy documents contained in the Weber County I/M ordinances, and Section 3.9 of the UTAH2000 analyzer specifications.

Co-operative enforcement oversight effort Motor Vehicle Customer Service Division, Utah Division of Air Quality, Utah highway patrol, and Weber County I/M program staff meet as needed to ensure on-going high quality oversight of joint motorist compliance program. EPA audit of this process is authorized if measures to protect taxpayer confidentiality acceptable to Motor Vehicle Customer Service Division are exercised.

12. I/M Program quality assurance

Station/inspector audits Weber County regularly audits all permitted I/M inspectors and stations to ensure compliance with the Weber County I/M ordinance. Particular attention is given to identifying and correcting any fraud or incompetence with respect to vehicle emissions inspections. Compliance with record keeping, document security, analyzer maintenance, and program security requirements are scrutinized. The inspector's skill level is also evaluated during audits. Another major purpose of the audits is to retrain inspectors, as necessary, as soon as problems are identified. Documentation sufficient to support a legal case to suspend or revoke a permit is also collected in the event of serious and/or repeated violations. Stations and inspectors are audited quarterly.

Covert audits Weber County, to the extent possible, performs a covert audit of each inspector and station at least once a year. The number of covert audits at least equals the number of permitted inspectors. Covert audits are performed using a variety of vehicles that are representative of the subject fleet that are set to fail across a full range of malfunctions. Suspected problem stations and inspectors are targeted for earlier and more frequent audits. Complaints also trigger additional audits.

Covert performance audits shall include:

Remote visual observation of inspector performance, which may include the use of aids such as binoculars or video cameras, at least once per year per inspector in high-volume stations (i.e., those performing more than 4000 tests per year);

Site visits at least once per year per number of permitted inspectors (per inspector FTE) using covert vehicles set to fail (this requirement sets a minimum level of activity, not a requirement that each inspector be involved in a covert audit); and

For stations that conduct both testing and repairs, at least one covert vehicle visit per station per year including purchase of repairs and subsequent retesting if the vehicle is initially failed for tailpipe emissions.

Electronic audit capabilities The UTAH2000 performs various analyses to identify statistically inconsistent data indicative of problem stations and inspectors. Overt audit records are maintained electronically in the UTAH2000. After overt audits the auditor retrieves the data containing the audit, vehicle inspection, and analyzer service, maintenance, and calibration records dating back to the previous audit. The data from each audit is added to the comprehensive central county I/M database. Further analysis of the central database results in identification of stations and inspectors for which additional audits are performed.

Auditor quality assurance Auditors receive on-the-job training in: the use of the UTAH2000 analyzer; the I/M program regulations; basic air pollution control; basic principles of emissions-related motor vehicle engine repair; emission control systems; evidence gathering; administrative procedures and laws; quality assurance practices; and covert audit procedure. Weber County sends auditors to additional automotive emissions-related training and meetings on a regular basis. Auditor supervisors audit the I/M program auditors by reviewing their documentation and also auditing a number of their stations at least once every year.

Written audit procedures The Weber County I/M program overt and covert audit procedures are contained in the Weber County I/M ordinances. A detailed description of the audit capabilities of the UTAH2000 analyzer are found in Section 3.9 of the UTAH2000 analyzer specifications.

13. Enforcement against stations and inspectors

General enforcement provisions The Weber County I/M program is responsible for enforcement action against incompetent or dishonest stations and inspectors. The Weber County I/M ordinance includes a penalty schedule. For serious or repeated offenses, auditors are authorized to immediately suspend the station or inspector by locking out their UTAH2000 analyzer(s). The County does not have legal authority to impose direct fines on stations or inspectors, but suspension or revocation of a station permit results in a substantial loss of income that is far in excess of \$100 fine suggested by the EPA

Section X, Part E, page 11

guidance. Station fee settlements are based on 50% of the expected revenue from I/M testing during the suspension, up to a maximum of \$3,000. Fee settlements for the inspectors are \$100 for any portion of a 15-day period, up to a maximum of \$500. A station permit may be suspended or revoked even if the owner/operator had no direct knowledge of the violation. In the case of incompetence, re-training is required before the permit is restored.

The County revised its penalty schedule to comply with the more stringent specifications included in 40 CFR 51.364; it is found in Appendix D of Weber County's regulation. Inspector suspensions may not be reduced by more than 75 days through a negotiated fee settlement.

Suspension and revocation Suspension or revocation effectively bars an individual from further inspections because the auditor removes the inspector's authorization code from the UTAH2000 analyzer. Evidence of indirect participation in emissions inspections by an individual while suspended or revoked could result in legal action against the station. If the station is suspended or revoked the analyzer is totally locked-out. The analyzers are initialized by an auditor for use at a single permitted station and only by inspectors permitted for that station. A record of the serial numbers of all registered analyzers and their locations is maintained by Weber County.

Enforcement records Weber County keeps comprehensive records on all audit activities, warnings, suspensions, and revocations and report enforcement activity statistics to the EPA and the executive secretary on an annual basis.

14. Data collection

I/M data collection Weber County maintains records regarding inspections, equipment maintenance, and the required quality assurance activities.

Analyzer inspection data The UTAH2000 analyzer creates a detailed record of each emissions inspection performed including, but not limited to the following data, for each vehicle tested: test record number; inspection station number; inspector number; test system number; date of the test; emission test start time; the time final emission scores are determined; vehicle identification number (VIN); license plate number; test certificate number, gross vehicle weight rating (GVWR); model year, make, and type of vehicle; number of cylinders or engine displacement; transmission type; odometer reading; category of test performed (i.e., initial, first retest, or subsequent retest); fuel type of the vehicle; emission scores for HC, CO, and CO₂ at idle and 2500 RPM; and results (pass/fail/not applicable) for visual inspection of the catalytic converter, air system, gas cap, evaporative system, and positive crankcase (PCV) valve. The tailpipe emission standards for each vehicle type are included in a look-up table in the UTAH2000 analyzer. The UTAH2000 analyzer automatically uses appropriate standards for the type of vehicle being tested and makes a pass/fail determination. The UTAH2000 analyzer records the inspection data during the inspection procedure. Analyzer quality assurance data Quality assurance data including a detailed history of all calibration (including the concentration values of the calibration gases), service, lockout, and document security events are also recorded and maintained by the UTAH2000 analyzer. Each UTAH2000 record includes, as applicable, station number, mechanic access number, auditor access number, service access number, analyzer serial number, date, and activity time.

Analyzer database specifications The programming criteria for the analyzer database are described in Section 3 of the UTAH2000 analyzer specifications. Appendix A of the UTAH2000 analyzer specifications contains a complete description of the electronic data records. The data containing inspection and quality assurance information is transferred electronically nightly and maintained permanently in the county's central I/M database.

15. Data analysis and reporting

Annual Reports Weber County shall analyze I/M program data and submit annual reports to the U.S. Environmental Protection Agency and the executive secretary upon request. Weber County will submit to EPA and the executive secretary an annual report, for January through December of the previous year, which provides statistics on the testing, quality assurance, and enforcement activities of each I/M program. At a minimum the annual reports will include all of the data elements listed 40 CFR Subpart S 51.366.

Data link Weber County requires all certified station owners to provide a computer data link between their station(s) and the Weber County health department in a manner approved by the health department and consistent with the requirements of 40 CFR 51 Subpart S.

16. Inspector training and permitting

Inspector permitting and initial training No person may conduct an official I/M inspection unless they are certified and subsequently permitted. Weber County requires formal training prior to certifying inspectors. Each class includes at least the following information: the causes and effects of air pollution; the purpose, function, and goal of the I/M program; I/M inspection ordinances, policies, and procedures; technical details of the test procedures and the rational for their design; emission control device function, configuration, and maintenance; quality control procedures and their purposes; public relations; and safety and health issues related to the I/M inspection process. Inspector candidates will not be issued a permit unless they have passed a written test with at least 70% correct responses and a hands-on test during which the trainee demonstrates the ability to properly conduct all test procedures, calibrate the UTAH2000 analyzer, properly utilize equipment, and to follow other I/M program requirements. Weber County takes appropriate steps to insure the security of the testing process.

Inspector Training The Weber County I/M ordinance requires an inspector training program, to include both classroom and hands-on training, with provisions for initial and periodic in-service training. Weber County requires in house training for each inspector

before the inspector may perform inspections periodic in-service training, over a period established by the health department.

Inspector permit renewal Inspector permits are valid for a period of one year, at which point refresher testing is required prior to permit renewal. An auditor enters the inspector's permit expiration date in the UTAH2000 analyzer(s) that the inspector is authorized to use. Starting 60 days prior to the inspector's permit expiration date the analyzer displays the message "Your mechanic permit expires MM/DD/YY". The analyzer locks-out inspectors that attempt to use the UTAH2000 analyzer after their permit expires and displays the following message. "Your mechanic permit expired (date). You are not authorized to perform any emissions inspections at this time. Please contact your local I/M office." Auditors will not clear the lock-out until the inspector has renewed the permit. Weber County may require evidence of more comprehensive emissions-related automotive training as a prerequisite to inspector permit renewal.

Inspector permit suspension and revocation A determination of inspector incompetence or failure to comply with I/M program requirements may result in suspension or revocation of an inspector's permit prior to the annual expiration date. A permit to conduct I/M inspections is not a legal right but rather a privilege bestowed by Weber County conditional upon adherence to its I/M program requirements.

Inspector training authority and materials Authority to require mandatory I/M inspector training is established and described in the Weber County I/M ordinances.

17. Public information and consumer protection

General public information Weber County, along with the Utah Department of Environmental Quality, provides a comprehensive public education and protection program including strategies to educate the public on: Utah's air quality problems; ways that people can reduce emissions; the requirements of state and federal law; the role of motor vehicles in the air quality problem; the need for and benefits of a vehicle emissions inspection program; ways to operate and maintain a vehicle in a low-emission condition; how to find a qualified repair technician; and the requirements of the I/M program. Information is provided via direct response to inquiries for information, reports, classes, pamphlets, fairs, school presentations, workshops, news releases, posters, signs, and public meetings.

County I/M Technical Center Weber County operates an I/M Technical Center staffed with trained auditors and capable of performing emissions tests. A major function of the I/M technical center is to serve as a referee station to resolve conflicts between permitted I/M inspectors, stations, and motorists. Auditors actively protect consumers against fraud and abuse by inspectors, mechanics, and others involved in the I/M program. Complaints made on a confidential basis are investigated and resolved in a manner that conceals the person's identity to ensure protection of whistle blowers. Auditors advise motorists regarding emissions warranty provisions and assist the owners in obtaining warrantycovered repairs for eligible vehicles. Applications for waivers are evaluated by auditors at the I/M technical center and issued only after visual verification that all the requirements for a waiver have been met, including retest of the vehicle. The I/M technical centers also provide motorists with information regarding the I/M program, general air pollution issues, and emissions-related automotive repairs.

Vehicle inspection report A vehicle inspection report (VIR) is printed and provided to the motorist after each vehicle inspection. A description of the VIR is included in the UTAH2000 analyzer specifications.

I/M county co-operative public education tools A variety of pamphlets and radio, television, and newspaper advertisements about automotive air pollution issues are developed and distributed by the Weber County I/M program in cooperation with other I/M counties and the Utah Division of Air Quality.

18. Improving repair effectiveness

High priority Weber County (along with other I/M counties) and the Utah Division of Air Quality staff jointly identified improvement of repair effectiveness as a high priority action item. The Governor's Clean Air Commission also recommended making affordable additional emissions-related training available. Full emission reductions will only be realized if the repair industry is able to competently diagnose and repair emissions-related defects.

Continuing education I/M program managers have worked with Utah's higher education institutions to develop and provide emissions-related automotive technology classes to mechanics. Inspectors are also encouraged to take classes offered by trade organizations, automobile manufacturers, and dealers. The permit renewal tests are difficult enough to make this provision a good incentive. The classes are advertised in the Weber County I/M technical bulletins.

UM program repair support activities In initiating improved automotive educational opportunities, Weber County works on a day-to-day basis to ensure that repair information is available. *I/M* stations are required to have available up-to-date relevant automotive diagnostic references and tools as a condition for obtaining a permit. Weber County maintains a hot line to its *I/M* technical center so that any mechanic can call for technical assistance related to vehicle inspection, diagnosis, and repair. Technical bulletins are regularly mailed to each permitted station with information regarding training schedules, common problems found with particular engine families, and diagnostic tips.

19. I/M SIP implementation

The I/M program ordinances or regulations, policies, procedures, and activities specified this I/M SIP revision have been implemented and shall continue until a maintenance plan without an I/M program is approved by EPA in accordance with Section 175 of the Clean Air Act as amended.

WEBER-MORGAN HEALTH DEPARTMENT

1

1

Regulation for

MOTOR VEHICLE INSPECTION AND MAINTENANCE PROGRAM

Adopted by the Weber-Morgan Board of Health

May 22, 2000

Amended May 12, 2003

Under Authority of Section 26A-1-121, 41-6-163.6 and 41-6-163.7 Utah Code Annotated, 1953, as amended

TABLE OF CONTENTS

1.0	Page Title and Definitions1	
2.0	Purpose	
3.0	Authority Jurisdiction	
4.0	Powers and Duties9	
5.0	Scope	
6.0	General Provisions	
7.0	Standards and Specifications for Exhaust Gas Analyzers and Span Gases17	
8.0	Permit Requirements of the Vehicle Emissions Inspection and Maintenance Program Station20	
9.0	Inspection Procedures25	
10.0	Adjustment Procedures	
11.0	Engine Switching	
12.0	Emission Repair Technician / Emission Tester Certificate of Qualification	
13.0	Emissions Standards for Motor Vehicles	
14.0	Certificates of Compliance and Certificates of Compliance - Waiver	
15.0	Right of Appeal41	
16.0	Penalty41	
17.0	Severability	
18.0	Effective Date42	
Appendix A - Utah 2000_Analyzer Specifications A-1		
Appendix B - Fee Schedule A-2		
Appendix C - Weber County Emission Standards (Cut points) A-3		
Appendix D - Penalty Schedule A-4		
Appendix E - Approved Calibration Gas and Calibration Gas Blenders A-5		
Apper	ndix F - OBD IM Inspection ProceduresA-6	
1.0 TITLE AND DEFINITIONS

٢

These standards shall be known as the Motor Vehicle Inspection and Maintenance Program Regulation, hereinafter referred to as "this Regulation."

For the purpose of this Regulation, the following words and phrases, when used herein, except as otherwise required by the context, have the following meanings.

1.1 "Accreditation" means Certification that the Analyzer and Analyzer manufacturer meet the operating criteria, specifications and requirements of Weber County and the Department;

1.2 **"Accuracy"** means the degree by which an instrument is able to determine the true concentration of pollutants of interest-Also means freedom from error especially as a result of care ;

1.3 "Air Intake Systems" means systems that allow for the induction of ambient air, including preheated air into the engine combustion chamber for the purpose of mixing with a fuel for combustion;

1.4 **"A.I.R. (Air Injection Reaction) System"** means a system for providing supplementary air into a vehicle's exhaust system to promote further oxidation of hydrocarbons (HC) and carbon monoxide (CO) gases and to assist catalytic reaction;

1.5 "Analyzer" See definition for UTAH2000 Analyzer;

1.6 **"Audit"** means a procedure performed by Department personnel which includes but is not limited to, inspection of the I/M Program station, review of Station records, inspection of Analyzer and related I/M Program equipment, review of personnel working knowledge and records. The audit procedure is intended to ensure compliance with this Regulation and Department policies and procedures;

1.7 **"BAR 97"** Refers to California Bureau of Automotive Repair Exhaust Gas Analyzer Specifications, which became effective in 1997;

1.8 **"Basic Engine Systems"** means parts or assemblies that provide efficient conversion of a compressed air/fuel charge into useful power, including but not limited to valve train mechanisms, cylinder head to block integrity, pistonring-cylinder sealing integrity and post-combustion emissions control device integrity meeting OEM Standards;

1.9 **"Bench"** means the main sample processing assembly of the exhaust gas Analyzer including detectors, sampling tubes, processor boards, infrared sources and power supply; 1.10 "Board of Health" means the Weber-Morgan Board of Health;

1.11 "**Calibration**" means the process of establishing or verifying the accuracy of an exhaust gas Analyzer to perform an accurate and consistent evaluation of engine exhaust using calibration gases having precisely known concentrations;

1.12 "Calibration [Span] Gases" means gases of known concentration that are used as references for establishing or verifying the calibration curve of an exhaust gas Analyzer and which are traceable to the National Institute of Standards and Technology and are approved by the Department for use;

1.13 **"Carbon Monoxide"** A colorless, odorless, asphyxiating gas produced by the incomplete burning of fuels. Carbon monoxide may be referred to in these Regulations as CO;

1.14 "Catalytic Converter" A post-combustion device that oxidizes HC and CO gases and/or reduces oxides of nitrogen gases;

1.15 **"Certificate of Compliance"** means a serially numbered document issued to the owner of a motor vehicle upon passing an inspection or reinspection and is evidence that the motor vehicle complies with the standards and criteria of this Regulation and other requirements as adopted by the Board of Health;

1.16 **"Certificate of Compliance Numbers"** means numbers issued to I/M Program Stations, and entered into the approved Analyzer for the purpose of issuing Certificates of Compliance;

1.17 "**Certificate of Waiver or Waiver**" means a document, issued by the Department used to verify that the vehicle for which it was issued has met the waiver requirements of this Regulation;

1.18 "**Certification**" means assurance by an authorized source, whether it be a laboratory, the manufacturer, the state, or the Department, that a specific product or statement is in fact true and meets all requirements;

1.19 "**Certified Emission Tester or Tester**" means an individual who has successfully completed all certification requirements and has been issued a current, valid Emission Tester Certificate of Qualification by the Department;

1.20 **"Certified Emissions Repair Technician or Technician"** means an individual who has successfully completed all certification requirements and has been issued a current, valid Emission Repair Technician Certificate of Qualification by the Department. A person certified by the Department who inspects vehicles, diagnoses emission related faults, and supervises or performs emissions related repairs and adjustments to bring vehicles into compliance with

the requirements of this Regulation;

ć

1.21 "CO" see Carbon Monoxide;

1.22 **"Compliance"** means verification that certain data and hardware submitted by a manufacturer for accreditation consideration, meets all Department requirements; Also meeting the requirements of this Regulations;

1.23 "County" means Weber County, Utah;

1.24 "Curb Idle" means the manufacturer's specified idle speed for the specific motor vehicle being tested (See also "Idle Mode");

1.25 "Cut-Points" Same as Emission Standards;

1.26 "**Department**" means the Weber-Morgan Health Department, Division of Environmental Health;

1.27 **"Director"** means the Environmental Health Division Director of the Weber-Morgan Health Department or his authorized representative;

1.28 "Domiciled" means County in which primary residence is located;

1.29 "E.G.R. System (Exhaust Gas Recirculation System)" means an emissions control system that recycles or recirculates a portion of the exhaust gases back to the engine combustion chambers;

1.30 **"Emissions"** means substances expelled into the atmosphere from a motor vehicle; particularly, air contaminants produced by combustion and/or incomplete combustion hydrocarbon evaporation from the fuel system and/or the crankcase, and particulate matter from the crankcase;

1.31 **"Emissions control systems"** means any device or combination of parts, originally installed by the manufacturer to control the emissions of a motor vehicle;

1.32 **"Emission Inspection or Inspection"** means a motor vehicle inspection performed for the purpose of determining whether the vehicle qualifies for issuance of a Certificate of Compliance or Certificate of Waiver, carried out in compliance with this Regulation;

1.33 **"Emission Repair or Repair"** means repair of a motor vehicle for the purpose of such vehicle passing or attempting to pass an emission inspection;

1.34 "Emission Repair Technician Certificate of Qualification" means a certificate issued by the Department authorizing an individual to conduct

emission inspections, repair failed vehicles and issue Certificates of Compliance while under the auspices of a I/M Program Station;

1.35 **"Emissions Standards (Cut-Points)"** means the maximum allowable concentration of regulated emissions for a given weight class and model year of a motor vehicle, as determined by the Board of Health using an approved Analyzer;

1.36 **"Emission Test"** means that portion of the Emission Inspection procedures where the engine exhaust gasses, from the tailpipe of the vehicle being Inspected, are tested to determine whether a vehicle produces emissions in excess of the Emissions Standards and/or an OBD IM test procedure;

1.37 **"Emission Tester Certificate of Qualification"** means a certificate issued by the Department authorizing an individual to perform emission inspections and issue certificates of compliance while under the auspices of a I/M Program Station;

1.38 **"Engine Switching"** means an engine is removed from a vehicle and is replaced by an engine that is not identical to the original engine;

1.39 "Engine Verification" means a document issued by the Department for the use of Kit Cars and Replica Vehicles only, validating engine size and year, for the purpose of inspecting the vehicle for the year of the engine instead of the year of the vehicle;

1.40 "EPA" means the United States Environmental Protection Agency;

1.41 **"Evaporative control system"** means an emission control system that prevents the escape of fuel vapors from the fuel system and/or air cleaner and stores them to be burned in the combustion chamber;

1.42 **"Exhaust Gas Analyzer"** means an instrument that is capable of measuring the concentrations of certain air contaminants in the exhaust gases emanating from a motor vehicle when approved by the Department for use in accordance with this Regulation as an official test instrument;

1.43 "Federal Installation" means any property or facility subject to the

jurisdiction of any department, agency, or instrumentality of the executive, legislative, and judicial branches of the Federal government;

1.44 **"Fleet Facility"** means a cooperation or other business entity permitted by the Department to perform the functions of the inspection program for a privately owned fleet of ten or more motor vehicles, including emissions related repairs, as well as the inspection;

1.45 **"Fuel Control Systems"** means the mechanical, electro mechanical, galvanic or electronic parts or assemblies that regulate the air/fuel ratio in an engine to provide a combustible charge;

1.46 **"Gaseous Fuel"** means, but is not limited to, liquefied petroleum gases and natural gases in liquefied or gaseous forms;

1.47 "HC" means hydrocarbons;

1.48 "**Hangup**" means hydrocarbons that cling to the surface of the sampling and Analyzer systems in contact with the exhaust gas sample stream resulting in errors in HC readings;

1.49 "Heavy Duty Vehicles" means a vehicle 1978 and older with a weight of more than 6000 pounds or 1979 and newer with a weight of more than 8501 pounds GVW (gross vehicle weight);

1.50 "Hydrocarbons" means unbumed fuel;

1.51 "Idle Mode" means a condition where the vehicle's engine is at proper operating temperature and running at the rate specified by the manufacturer's cub idle, where the engine is not propelling the vehicle, and where the throttle is in the closed or idle stop position. This condition achieved without placing a load on the vehicle to decrease its RPM to the specified rate(See also **Curb Idle**);

1.52 "**Ignition Systems**" the means parts or assemblies that are designed to cause and time the ignition of a compressed air/fuel charge;

1.53 "I/M Clearance" means a stamp placed on the motor vehicle registration form by an employee of the Utah State Motor Vehicle Office or the Department indicating that the motor vehicle represented by the registration form is in compliance with the inspection program requirements in that the motorist has presented a valid Certificate of Compliance or Certificate of Waiver for the motor vehicle and paid applicable fees:

1.54 "**I/M Program**" means the Vehicle Emissions Inspection and Maintenance Program established by the County Board of Health and this Regulation;

1.55 "I/M Program Test and Repair Station" means a business permitted by the Department which engages in emissions testing and emissions related repairs to motor vehicles, and which meets the requirements of this Regulation for test and repair facilities;

1.56 **"I/M Program Test Only Station"** means a business permitted by the Department which engages only in emissions related inspections of motor vehicles, and which meets the requirements of this Regulation for test only facilities;

1.57 **"Inspection Area"** means the Department approved area that is occupied by the Analyzer, sample hose, and the vehicle being inspected;

1.58 **"Inspection Report"** means a document used to record information generated by the Tester/Technician during an emissions inspection other than a Certificate of Compliance;

1.59 **"Instrument"** means the complete UTAH2000 Analyzer system that samples and displays the concentration of emission gases and also performs OBD IM test procedures. The instrument includes the sample handling system, the exhaust gas Analyzer associated computer equipment and the enclosure cabinet;

1.60 **"Kit Car"** means a fully assembled custom motor vehicle containing all of the needed components for assembly (i.e., body, chassis, engine, and transmission);

1.61 "Light Duty Motor Vehicle" means all passenger vehicles, 1978 and older; light duty trucks 6000 GVW rating or less; 1979 trucks and newer 8500 pounds GVW rating or less;

1.62 **"Lock-Out"** means when the UTAH2000 Analyzer automatically prohibits access to the testing portion of the UTAH2000 Analyzer;

1.63 **"Motor Vehicle or Vehicle"** means any equipment or mechanical device propelled primarily on land by an internal combustion powered engine which is driven on public roads and/or streets. Motor vehicles exempted from the inspection requirements of this Regulation as listed in Section 6.6 of this Regulation;

1.64 **"Motorcycle"** means every motor vehicle having a saddle for the use of the rider and designed to travel with not more than three wheels in contact with the ground, but excluding a farm tractor;

1.65 "Non-Certified Tester/Technician" means any person who has not been certified by the Department to perform official emissions inspections;

1.66 "OBD" means Vehicle On-Board Diagnostics;

1.67 "**OBDII**" means Updated On-Board Diagnostics Standard effective in 1996 and newer light duty car and light duty trucks sold in the United States;

1.68 "OEM" means Original Equipment Manufacturer;

1.69 **"Off-Highway Vehicles"** means a vehicle licensed to operate exclusively off public highways and roads;

1.70 "Original Condition" means the condition of the emission control system(s) as installed by the manufacturer, but not necessarily to the original level of effectiveness;

í.

i. A 1.71 "**PCV System (Positive Crankcase Ventilation System)**" means and emissions control system which returns crankcase vapors and blowby gases to the combustion chamber to be burned;

1.72 **"Permit"** means the document issued by the Department that authorizes a person to operate an I/M Program Station;

1.73 **"Person"** means an individual, corporation, association, firm, partnership, joint stock company, public or municipal corporation, political subdivision, the state or any agency thereof, or the federal government or any agency thereof;

1.74 "**Prompts**" means instructions and/or data fields, requiring data input to the Analyzer from a Tester/Technician performing an emission inspection;

1.75 "**Publicly-Owned Vehicles**" means a motor vehicle owned by a government entity, including but not limited to the federal government or any agency thereof, the State of Utah or any agency or political subdivision thereof-;

1.76 **"Readiness"** means codes set by the OBD system that indicate a vehicle readiness to be OBD tested;

1.77 **"Registered or Registration"** means the process by which a motor vehicle receives a license so that it can be legally operated on public streets and highways;

1.78 "**Reinspection**" means any emission inspection performed on a motor vehicle after it has failed an emissions inspection and repair and/or adjustment has been attempted;

1.79 **"Repeatability"** means the instrument's capability to provide the same value, within specified tolerances, for successive measurements of the same sample;

1.80 **"Response Time"** means the period of time, in seconds, for an instrument to measure and display a pollutant concentration after a concentration of gases is introduced or removed from the sample probe;

1.81 **"Safety Inspection"** means an evaluation of a vehicle's relative safety as required by 41-6-158 U.C.A., 1953;

1.82 "**Smoker**" means a motor vehicle emitting visible emissions after the engine has reached normal operating temperature;

1.83 "**Stabilization**" means the process of bringing an instrument into equilibrium with the ambient environment and operative conditions;

1.84 "**Station**" means and I/M Program Station including all station personnel, employees, and owner(s);

1.85 **"Tampering"** means the intentional or accidental altering of or removal of emission control devices, and/or emissions-related equipment. Also, the use of fuels other than those required by the manufacturer's specification as found in the motor vehicles's owners manual. Also, engine modifications which may include, but is not limited to, exhaust systems, air intake systems, ignition systems, internal engine modifications, engine switching, etc;

1.86 **"Technical Bulletin"** means a document issued to Tester/Technicians and/or I/M Program Stations by the Department to update, clarify or establish policies and/or procedures for their implementation in the Vehicle Emission Inspection and Maintenance Program;

1.87 **"Tester/Technician"** means a Department Certified Emission Tester or Department Certified Emissions Repair Technician;

1.88 **"Training Program"** means a formal program administered, conducted, or approved by the Department for the education of Testers/Technicians in basic emission control technology, inspection procedures, diagnosis and repair of emission related problems, Vehicle Emissions Inspection and Maintenance Program policies, procedures and this Regulation; it may also include the promotion of training for all mechanics;

1.89 **"UTAH2000 Analyzer or Analyzer"** means the official computerized engine exhaust analyzer and associated test equipment approved by the Department for use in the areas of Utah requiring inspections as specified in Section 41-6-163.6 and 41-6-163.7, Utah Code Annotated, 1953, as amended;

1.90 "Vehicle Emissions Inspection and Maintenance Program" means the program established by the Board of Health pursuant to Section 41-6-163.6, Utah Code Annotated 1953, as amended and Weber County Ordinance 14-4-1;

1.91 **"Waiver Referral"** means a document prepared and signed by an Tester/Technician used in the Vehicle Emissions Inspection and Maintenance program to certify that the vehicle has qualified to seek a waiver which may be issued by the Department;

1.92 "Waiver" see Certificate of Waiver.

2.0 PURPOSE

It is the purpose of this Regulation to reduce air pollution levels in Weber County by requiring annual inspection of in-use motor vehicles and by requiring emission related repairs and adjustments for those vehicles that fail to meet prescribed standards so as to:

2.1 Protect and promote the public health, safety and welfare;

2.2 Improve air quality;

5.

2.3 Comply with Federal Regulations contained in the Clean Air Act of 1970, 42 USC 7401-7671: and the amendments to the Act, Amendments of 1977, PL 95-190; and Amendments of 1990, PL 101-549;

2.4 Comply with the law enacted by the Legislature of the State of Utah, Sections 41-6-163.6 and 41-6-163.7 Utah Code Annotated, 1953, as amended; and

2.5 Comply with Weber County Ordinance 14 Chapter 4, Implement Emissions Inspection, Sections 1 through 6.

3.0 AUTHORITY AND JURISDICTION OF THE DEPARTMENT

3.1 Under Section 14-4-3 of the Weber County Ordinance the Weber County Commission authorizes and directs the Weber-Morgan Board of Health and the Director of Health to adopt and promulgate rules and regulations to ensure compliance with EPA and State requirements with respect to Emission Standards and delegates its authority as an administrative body "under 41-6-163.6 U.C.A.,1953, as amended, to the Weber-Morgan Board of Health, to address all issues pertaining to the adoption and administration of the Vehicle Emission I&M Program. Authorizes a one dollar fee to be assessed upon every motorized vehicle registered in Weber County at the time of registration, to be known as the Air Pollution Control Fee.

3.2 Section 14-4-4 of the Weber County Ordinance, directs the Weber-Morgan Health Department to adopt regulations and set fees for I&M Stations and Mechanic Permits and Emission and Waiver Certificates as necessary to sustain and operate an I&M Program.

3.3 The Weber-Morgan Board of Health is authorized to make standards and regulations pursuant to Section 26A-1-121(1) of the Utah Code Annotated, 1953 as amended.

3.4 The Weber-Morgan Board of Health is authorized to establish and collect fees pursuant to Section 26A-1-114(1)(h)(4i) of the Utah Code Annotated, 1953 as amended.

3.5 All aspects of the Vehicle Emissions Inspections and Maintenance Program within Weber County enumerated in Section 2.0 shall be subject to the direction and control of the Weber-Morgan Health Department.

4.0 POWERS AND DUTIES

4.1 General Powers and duties. The Department shall be responsible for the enforcement and administration of this Regulation and any other powers vested in it by law and shall:

4.1.1 Require the submission of information reports, plans and Specifications from I/M Program Stations, and as necessary to implement the provisions and requirements of this Regulation;

4.1.2 Issue permits, certifications and charge fees as necessary to Implement this Regulation;

4.1.3 Perform audits of any I/M Program Station and Tester/Technician and issue orders and/or notices, hold hearings, levy administrative penalties and negotiate consent agreements as necessary to effect the purposes of this Regulation;

4.1.4 When necessary take samples and make analysis to ensure that the provisions of this Regulation are met; and

4.1.5 Adopt policies and procedures necessary to ensure that the provisions of this Regulation are met and that the purposes of this Regulation are accomplished.

4.2 Suspension, Revocation, or Denial of Permits. The Department may suspend, revoke or deny a Permit of an I/M Program Station and/or negotiate a monetary penalty in lieu of suspending a permit under a consent agreement, and/or require the surrender of the Permit and unused Certificates of Compliance and other official documents of such I/M Program Station upon showing that:

4.2.1 A vehicle was inspected and issued a Certificate of Compliance by I/M Program Station personnel who did not, at the time of inspection, comply with all applicable policies, procedures, Technical Bulletins, and this Regulation;

4.2.2 A vehicle was inspected and rejected by the station when, it can be proven, as determined by the Department that the vehicle was in such condition that it did comply with the requirements of this Regulation;

4.2.3 A vehicle was inspected and issued a Certificate of Compliance when it can be proven, as determined by the department that the vehicle did not at the time of inspection comply with the requirements of Section 9.10 regarding tampering inspection;

4.2.4 A vehicle was passed and issued a Certificate of Compliance without being present for inspection or another vehicle was probed for the gas analysis during the two speed idle test;

4.2.5 The station is not open and available to perform Inspections during a major portion of the normal business hours. of 8:00 a.m. to 5:00 p.m., Mondays through Fridays,(except I/M Program stations which test only their own vehicles);

4.2.6 The station has violated any provisions of this Regulation, or any Rule, Regulation, or Department policy properly promulgated for the operation of an I/M Program Station;

4.2.7 The station was or is not equipped as required by Section 8.0 of this Regulation;

4.2.8 The I/M Program Station is not operating within the property boundaries as of the location specified on the Permit;

4.2.9 An official inspection was done by a Non-Certified Tester/Technician or a Non-Certified Tester/Technician has gained access to the official testing portion of the Analyzer or a Non-Certified Tester/Technician has signed a Certificate of Compliance or other official testing document;

4.2.10 The approved Analyzer has been tampered with or altered in any way contrary to the certification and maintenance requirements of the Analyzer or the test vehicle has been altered or tampered with in any way so that it will either pass or fail the emissions test when it would not otherwise;

4.2.11 The I/M Program Station denies access to a representative of the Department to conduct an audit or other necessary business during regular business hours;

4.2.12 The I/M Program Station denies access to or conceals pertinent information from a representative of the Department during an audit or while conducting other necessary business during regular business hours;

4.2.13 The I/M Program Station performed unnecessary repairs not justified by the results of the inspection;

4.2.14 In accordance with 41-6-163.6 and 41-6-163.7 U.C.A., 1953, As amended, an emissions inspection for a Salt Lake, Utah, Davis or Weber County resident was performed but not as required by the Regulations adopted by the applicable county and/or the UTAH2000 Analyzer prompts; or

4.2.15 A Tester/Technician employed at an I/M Program Station, and authorized to conduct emission inspections, who violates any of the provisions of Section 4.3.

4.3 Suspension, Revocation, or Denial of Certificates. The Department may suspend, revoke, or deny the Certificate of Qualification of a Tester/Technician and require the surrender of the Tester/Technician Certificate of Qualification upon showing that:

4.3.1 The Tester/Technician caused any of the violations listed in section 4.2 to occur;

4.3.2 The Tester/Technician issued or caused a Certificate of Compliance to be issued to an owner/operator without an approved inspection being made;

4.3.3 The Tester/Technician denied the issuance of a Certificate of Compliance to the owner/operator of a vehicle that, at the time of the inspection, complied with the law for issuance of said certificate;

4.3.4 The Tester/Technician issued a Certificate of Compliance to a vehicle that, at the time of issuance, was in such condition that it did not comply with this Regulation;

4.3.5 The Tester/Technician inspected, recorded and passed the tampering inspection, for a vehicle that did not, at the time of inspection, comply with the tampering requirements of the tampering inspection detailed in Section 9.10, regardless of whether a Certificate of Compliance was issued or not;

4.3.6 Inspections were performed by the Tester/Technician but not in accordance with applicable policies, procedures, technical bulletins, and this Regulation;

4.3.7 The Tester/Technician allowed a Non-Certified Tester/Technician to perform an inspection or gain access to the official testing portion of the Analyzer;

4.3.8 The Tester/Technician signed an inspection or certificate stating that he had performed the emissions test when, in fact, he did not;

4.3.9 The Tester/Technician signed a certificate prior to a test being performed and the certificate printed by the dedicated printer;

4.3.10 The Tester/Technician falsified any inspection or official document

of the Vehicle Emissions Inspection and Maintenance Program;

).

بينغمانهم زي

4.3.11 The Tester/Technician performed unnecessary repairs not justified by the results of the inspection; or

4.3.12 In accordance with Sections 41-6-163.6 and 41-6-163.7 U.C.A., 1953, as amended, an emissions inspection for a Weber, Salt Lake, Davis or Utah County resident was performed but not as required by the Regulations/Ordinances adopted by the applicable county.

4.4 <u>Administrative Penalty</u>: in lieu of suspending a Permit under Sections 4.2 and 4.3 the Department may impose an administrative penalty such as a negotiated monetary penalty by means of a consent agreement.

4.5 The Department shall respond, according to the policies and procedures this regulation, to public complaints regarding the fairness and integrity of inspections they receive, shall provide a method that inspection results may be challenged if there is a reason to believe them to be inaccurate. To challenge the results of an inspection or reinspection, a motorist must present his or her vehicle within 2 days (excluding Saturday, Sunday and Holidays) of the inspection being challenged for another emissions inspection at the Weber County Technical Center.

4.6 The Department is authorized to take any and all necessary measures to ensure or facilitate a smooth transition from the UTAH91 Analyzer testing program to the new UTAH2000 Analyzer testing program required by this Regulation. Such measures may include, but shall not be limited to the following:

4.6.1 Extending the period of time that a UTAH91 Analyzer may be used to perform an official Inspection beyond the mandatory date for use of the UTAH2000 Analyzer as specified in Section 7.0 of this Regulation;

4.6.2 Permitting the use of the UTAH2000 Analyzer to perform official emissions tests prior to the mandatory date as specified in Section 7.0 of this Regulation;

4.6.3 Extending the period of time that the two speed idle test can be substituted for compliance when 1996 and newer motor vehicles fail the OBD IM test procedures;

4.6.4 Modify OBD IM test procedures when necessary and may include the temporary suspension of the OBD testing requirement; and

4.6.5 Only the Department may exempt vehicles from the OBD test requirements of this Regulation.

5.0 SCOPE

It shall be unlawful for any person not to comply with any policy, procedure, technical bulletin, regulation or ordinance promulgated by the County and/or the Department unless expressly waived by this Regulation.

6.0 GENERAL PROVISIONS

Subject to the exceptions described in Section 6.4 and pursuant to the schedule in Section 6.1, motor vehicles of model years 1968 and newer that are owned and/or operated by person domiciled in Weber County, shall be subject to an annual or biennial emission inspection performed by an I/M Program Station or other entity approved by the Department. Registration of a vehicle owned and operated by a Weber County resident in a County other than Weber is a violation of this Regulation. Fraudulent registration of a vehicle as a farm truck, diesel, or other exempted vehicle is also a violation of this Regulation.

6.1 Beginning 1 January 1992 a Certificate of Compliance, Certificate of Waiver or evidence that the motor vehicle is exempt from the Inspection and Maintenance Program requirements (as defined in section 6.4) shall be presented to the Weber County Assessor or the Utah State Tax Commission and the Air Pollution Control Fee paid (See section 6.8.2) as conditions precedent to annual registration or annual renewal of registration of a motor vehicle. Certificates of Compliance from other EPA approved I/M Programs may be accepted, if approved by the Director, provided those I/M Programs are equally effective in reducing emissions.

6.2 A Certificate of Compliance issued to a dealer licensed with the State of Utah and issued in the dealer's name, shall be valid for registration purposes for a period of six months as specified in Section 41-3-28.5, Utah Code Annotated, 1953, as amended. The purchaser's name, address, and phone number shall be recorded by the dealer on the back of the Certificate.

6.3 Publicly-Owned Vehicles. Owners of publicly-owned vehicles shall comply with the Vehicle Emissions Inspections/Maintenance program requirements in accordance with this Regulation on an annual or biennial basis pursuant to a schedule determined by the Department. Federally-owned vehicles and vehicles of employees regularly operated on a federal installation located in the county that do not require registration in the State of Utah shall comply with the emissions testing requirements on an annual or biennial basis pursuant to a schedule determined by the Department and as required by Section 118 of the Clean Air Act (1990 amendment).

6.4 Vehicle Exemption. The following vehicles are exempt from the annual or

biennial emissions inspection:

6.4.1 Any motor vehicle of model year 1967 or older;

6.4.2 All agricultural implements of husbandry and any motor vehicle that qualifies for an exemption Section 41-6-163.6, Utah Code Annotated, 1953, as amended;

6.4.3 Any vehicle used for maintenance or construction and not designed or licensed to operate on the highway;

6.4.4 Any motorcycle or motor driven cycle (including vehicles which operate with an engine normally used in a motorcycle);

6.4.5 Any vehicle that operates exclusively on diesel fuel or electricity;

6.4.6 Any new motor vehicle being sold for the first time that has a valid (Manufacturer's Statement of Origin) (MSO) form;

6.4.7 Any vehicle with an engine smaller than forty (40) cubic inch displacement (655 cc); and

6.4.8 Tactical military vehicles.

6.5 It shall be the responsibility of the Tester/Technician to determine if a motor vehicle is exempted from Section 6.6 of this Regulation when presented to the Tester/Technician for an inspection. It shall be the Tester/Technicians responsibility to inform the owner/operator of the vehicle that the vehicle is not required to have an inspection for vehicle registration purposes.

6.6 Official Signs.

6.6.1 All I/M Program Stations, except those stations authorized to inspect only their own motor vehicles such as fleet facilities, shall display in a conspicuous location on the premises an official sign provided and approved by the Department.

6.6.2 The emissions standards, as promulgated under authority of Section 12.0 and Appendix C of this Regulation shall be posted in a conspicuous place on the station's premises.

6.6.3 The station shall post on a clear and legible sign and in a conspicuous place at the station, the fees charged by that station for the performance of the emissions inspection.

6.6.4 The signs required by Sections 6.8.1, 6.8.2, and 6.8.3 shall be

located so as to be easily in the public view.

6.7 Equipment Available for Inspection.

6.7.1 Required tools, materials, publications (see Section 8.1.4.3) supplies, records, unused Certificates of Compliance, other required forms, records of completed inspections, a complete copy of this Regulation, and duplicate copies of Certificates of Compliance issued shall be kept at the station at all times and shall be available for inspection and collection by the Department at any time the station is open for business.

6.7.2 A periodic inspection and audit shall be made by a representative of the Department to verify compliance with this Regulation for each I/M Program Station. As part of the periodic inspection and audit of the I/M Program Station the Department representative shall, as applicable, perform a gas calibration and leak check of each certified Analyzer, examine leak check and gas calibration records, and examine inspection records and Certificates of Compliance, as well as other required reports, forms, or records to see that the use of these items is in compliance with this Regulation and the policies and procedures of the Department.

6.7.2.1 During the time of the inspection and audit by the Department, the Department representative shall have exclusive access to the approved testing Analyzer(s).

6.7.2.2 The Department representative may check the accuracy of the Analyzer using Department gas to verify that the Analyzer is reading within the tolerances established by the Department. Analyzers not reading within the acceptable tolerances shall be calibrated to acceptable tolerances or placed "out of service".

6.8 Fees

6.8.1 The fees assessed I/M Program Stations and Testers/Technicians shall be determined according to a fee schedule adopted by the Board of Health. The fee schedule is referenced in Appendix B of this Regulation.

6.8.2 The following fee is hereby assessed upon every motor vehicle registered in Weber County annually at the time of registration of the vehicle:

6.8.2.1 Air Pollution Control Fee -- one dollar (\$1.00).

6.8.2.2 This fee assessment is included upon all motorized vehicles including those that are exempted from the inspection requirements of this Regulation by Section 6.4 unless a separate

fee is assessed on other motor vehicles by other Health Regulations.

6.8.3 I/M Program Stations may charge a fee for the required service. The fee may not exceed, for each vehicle inspected, the amount set by the Board of Health and referenced in Appendix B of this Regulation:

6.8.3.1 The inspection fee pays for a complete inspection leading to a Certificate of Compliance or a failure. If a vehicle fails the inspection, the owner is entitled to one free reinspection if he returns to the station that performed the original inspection within fifteen (15) days from the date of the initial inspection. The station shall extend the fifteen day free reinspection time to accommodate the vehicle owner if the station is unable to schedule the retest of the vehicle within the fifteen day time period. The emissions inspection fee shall be the same whether the vehicle passes or fails the emission test.

6.8.3.2 At the request of the Department, a station shall extend the free retest time for vehicle owners who are unable to complete repairs because of the unavailability of parts to make the necessary repairs. In no case shall this extended time exceed the data storage capacity time of the approved emission Analyzer.

6.8.3.3 Duplicate Certificates of Compliance issued to a vehicle owner/operator shall not exceed that set by the Board of Health and referenced in Appendix B of this Regulation.

6.9 If a vehicle fails the inspection and is within the time and mileage requirements of the federal emissions warranty contained in the Federal Clean Air Act, the Tester/Technician shall inform the owner/operator that he may qualify for warranty coverage of emission related repairs as provided by the vehicle manufacturer and mandated by the Federal Environmental Protection Agency.

6.10 These fees are subject to change and may be amended as deemed necessary by the Board of Health to accomplish the purposes of this Regulation.

7.0 STANDARDS AND SPECIFICATIONS FOR EXHAUST GAS ANALYZERS AND CALIBRATION GASES

- 7.1 Approval of Exhaust Gas Analyzers
 - 7.1.1 No emission inspection or emission test required by this Regulation

shall be performed after May 31,2000 unless the type of instrument used for determining compliance with this Regulation is the UTAH2000 Analyzer (except for provisions noted in Section 4.6). The Analyzer shall meet the requirements of the Analyzer specifications referenced in Appendix A of this Regulation. The Analyzer shall also be certified by the manufacturer as meeting the criteria of Section 207 (b) warranty provisions of the Clean Air Act. The instrument shall be in good working condition, capable of meeting calibration requirements of the Department and operated according to manufacturer's specifications and operating procedures and capable of operating to the standard required by the Utah2000 Analyzer Specification.

7.1.2 Analyzer Registration

Any Analyzer used by an I/M Program Station shall be registered with and approved by the Department and shall be issued an analyzer registration number and, if required, shall bear a registration sticker issued by the Department. Registration stickers are not transferable or assignable. Any new or used Analyzer put in use after station approval must be approved by the Department before use. Analyzers used temporarily during times of breakdown or repair of the registered Analyzer do not require a registration sticker but shall meet all other requirements of this section including the approval of the Department before use.

7.1.3 Running Changes

Any changes to the design characteristics or component specifications that may affect the performance of an Analyzer to be used as an official test instrument in the Vehicle Emissions Inspection and Maintenance Program shall be approved by the Department. It shall be the Analyzer manufacturer's responsibility to verify that the changes have no detrimental effect on the performance of the Analyzer.

7.1.3.1 It shall be unlawful for any person to alter or modify the hardware or software of an approved emissions Analyzer without written application and formal written approval by the Department.

7.1.3.2 It shall be unlawful for any person to gain access to any Department controlled portions of an approved Analyzer without approval by the Department.

7.1.4 Calibration/Span Gases

7.1.4.1 General

The Analyzer manufacturer and/or manufacturer designated

marketing vendor shall, supply at a reasonable cost calibration gases approved by the Department to any ultimate purchaser of the Analyzer. Each new or used Analyzer sold by the manufacturer or marketing vendor shall have when deemed necessary by the Department, approved full calibration gas containers installed and operational at the time of delivery. The Department shall establish necessary procedures for approving calibration/span gases.

7.1.4.2 Calibration/Span Gas Blends

The calibration/Span gases supplied to any I/M Program Station shall conform to the specifications of the Department. All calibration gases shall meet all requirements for emissions warranty coverage. Only gas blends supplied by Department approved venders shall be used to calibrate Analyzers. (Approved gas blend and Department approved vendors are referenced in Appendix E of this Regulation.)

7.1.5 Documentation, Logistics, and Warranty Requirements

7.1.5.1 Instrument Manual

An instrument manual shall be provided by the Analyzer manufacture. The instruction manual shall be conveyed to the purchaser at the time of sale and shall contain at least the following information for the Analyzer:

(a) A complete technical description;

(b) The functional mechanical and electrical schematics;

(c) The accessories and options that are included and/or available;

(d) The model number, identification marking and location;

(e) Operating maintenance schedule including daily, weekly, and monthly accommodations and procedures for maintaining sample system integrity including, but not limited to, leaks, hangup, calibration and filters. The services to be performed only by the manufacturer shall be clearly identified;

(f) Field Calibration Procedures (i.e., Department inspection procedure with separate gas supply);

(g) Cal-port gas inlet calibration, zero, and span instruction;

(h) Information concerning the nearest service facility where equipment can be serviced; and

(i) The warranty provisions for the Analyzer, including a list of warranty repair stations by name, address and telephone number.

7.1.5.2 Analyzer Maintenance.

The Analyzer shall be maintained in accordance with the manufacturer's recommended maintenance schedule and records of this maintenance service shall be maintained for examination by the Department.

7.1.5.3 Analyzer printers shall be maintained in such a manner that the printing of the Certificates, inspection reports and documents are accurate and legible. If any printer fails to properly function, then the Station shall discontinue testing until the required repairs have been performed or a replacement printer is installed.

7.1.6 Propane Equivalency Factor (P.E.F.).

Each instrument shall be labeled with a valid propane equivalency factor, shown with an accuracy of at least two decimal places, (i.e., 0.52). P.E.F. confirmation shall be made on each assembled Analyzer by measuring both N-hexane and propane values on assembly line quality checks. If the Analyzer bench is replaced, then a new P.E.F. label applicable to the replacement bench shall be appropriately attached to the Analyzer.

7.2 Gas Calibration and Leak Check.

7.2.1 A Tester/Technician shall perform a gas calibration of the exhaust gas Analyzer, with an approved calibration gas, within 72 hours prior to performing any emission test. A leak test must be performed every 24 hours. The gas calibration and leak test must be performed in accordance with the Analyzer specifications as contained in Appendix A.

7.2.2 The Analyzer instruction manual and other Department approved information shall be reviewed by the Tester/Technician to ensure that proper procedures are being used for performing the gas calibration.

7.2.3 The Analyzer shall lock-out when calibrations and leak tests are not performed within prescribed time frames.

7.2.4 The Department shall use and require for use in the calibration of Analyzers, calibration and span gases and containers meeting the guidelines contained in Section 7.1.5.

8.0 PERMIT REQUIREMENTS OF THE VEHICLE EMISSIONS I/M PROGRAM STATION

8.1 Permit Required.

8.1.1 No person shall operate a I/M Program Station without a valid Permit to operate issued by the Department. A person desiring to operate a I/M Program Station shall submit to the Department a written application for a Permit on a form provided by the Department. To qualify for a Permit, an applicant shall:

8.1.1.1 Be an owner of the proposed I/M Program Station or an officer of the legal ownership;

8.1.1.2 Comply with the requirements of this Regulation;

8.1.1.3 Agree to allow Department access to the I/M Program Station and to provide required information;

8.1.1.4 Pay the application permit fee at the time the application is submitted;

8.1.1.5 Present a copy of a current business license relating to the I/M Program Station; and

8.1.1.6 Other information required by the Department.

8.1.2 The application shall include:

8.1.2.1 The name, mailing address, telephone number, and signature of the person applying for the Permit and the name, mailing address, and location of the I/M Program Station;

8.1.2.2 Information specifying whether the Station is owned by an association, corporation, individual, partnership, or other legal entity;

8.1.2.3 The name, title, address, and telephone number of the

person directly responsible for the Station;

8.1.2.4 The name, title, address, and telephone number of the person who functions as the immediate supervisor of the person specified under 8.1.2.3 of this section such as zone, district, or regional supervisor;

8.1.2.5 A statement signed by the applicant that attests to the accuracy of the information provided in the application, and affirms that the applicant will comply with this Regulation, and allow the Department access to the Station; and

8.1.2.6 Other information required by the Department.

8.1.3 No person shall in any way represent any place as an I/M Program Station unless the station is operated under a valid Permit issued by the Department.

8.1.4 The Department is authorized to issue or deny Permits for I/M Program Stations.

8.1.5 A Permit may not be transferred from one person to another person, from one I/M Program Station to another I/M Program Station or from one type of operation to another, i.e., test and repair to test only, unless approved in writing by the Department. The Permit shall be posted in a conspicuous place within public view on the premises.

8.1.6 The Department may renew a Permit for an existing I/M Program Station or may issue a Permit to a new owner of an existing Station after a properly completed renewal form is submitted, reviewed, and approved, the fees are paid, and an inspection shows that the Station is in compliance with this Regulation.

8.1.7 No Permit shall be issued unless the Department finds that the facilities, tools, and equipment of the applicant comply with the requirements of this Regulation and that competent personnel, certified under the provisions of Section 12.0, are employed and available to make inspections and adjustments, and the operation thereof will be properly conducted in accordance with this Regulation.

8.1.7.1 An I/M Program Station shall immediately notify the Department if the station does not have a Tester/Technician employed.

8.1.7.2 An I/M Program Station shall comply with all terms stated in the permit application and all the requirements of this Regulation.

8.1.7.3 As a condition for permitting all I/M Program Stations, the following tools and materials shall be available for performance of the inspection and maintenance of motor vehicles unless specifically exempted by the Department:

(a) A Department approved Analyzer;

(b) An accurate dwell meter (Optional);

(c) An ignition timing light;

(d) A propane enrichment kit;

(e) Reference manuals (printed or electronic) approved by the Department that contain idle speed, idle mixture, timing, dwell, fast idle speed specifications, and information covering the emissions control systems for the model years an makes of vehicles involved in the Vehicle Emissions Inspection and Maintenance Program;

(f) Sufficient hand tools for proper performance of the inspection and minimum repairs and maintenance as required by the Department;

(g) Department approved calibration/ gases;

 (h) The Analyzer manufacturer's maintenance and calibration manual, which must be retained in the inspection area;

(i) All forms, technical bulletins, a copy of this Regulation, and other information materials provided by the Department;

(j) A suitable non-reactive tailpipe extender or suitable probe adapter for inspecting vehicles with screened or baffled exhaust systems; and

(k) Suitable tools to interface with onboard vehicle computers for computer controlled vehicles which are intended to receive official repairs. (For example, General Motors compatible can tools are required if the I/M Program Station wishes to repair computer controlled GM vehicles failing the I/M test.)

8.2 Duties of Testers/Technicians Working in Permitted Program Stations:

8.2.1 All facets of the Vehicle Emissions Inspection and Maintenance Program shall be performed by the Tester/Technicians including, but not limited to:

8.2.1.1 Analyzer preparation, calibration checks, and leak checks;

8.2.1.2 Exhaust gas sampling and analysis for purposes of an official emissions test for issuance of a Certificate of Compliance;

8.2.1.3 Preparation of reports, forms, and certificates.;

8.2.1.4 Accessing the official emissions testing section of the Analyzer; and

8.2.1.5 All other aspects of the official emissions test, including but not limited to, the tampering inspection, inserting the exhaust probe, hooking up the tachometer, hooking up the OBDII data link connector, entering data into the Analyzer, verifying that the engine is at normal operating temperature, ensuring that accessories are off, preconditioning the vehicle, and signing Certificates of Compliance and inspection forms, etc., unless otherwise approved in writing by the Director.

8.3 Safety.

An I/M Program Station facility shall be kept in good repair, free of obstructions and hazards and in a safe condition for inspection purpose. No inspection shall be conducted if unsafe conditions exist.

8.3.1 At no time shall carbon monoxide (CO) readings in the ambient air within the station exceed a peak hourly level of 35 parts per million.

8.3.2 All applicable Occupation Safety and Health Administration (OSHA), and other applicable health and safety rules and regulations must be followed in the station.

8.4 Fleet Facility.

A person may establish a fleet facility that is exempt from conducting business at regular hours, or displaying program signs as long as only vehicles owned or controlled by the fleet facility owner are inspected at the station. All other requirements of this Regulation apply. 8.5 Permit Duration and Renewal.

8.5.1 The Permit for I/M Program Stations shall be issued annually and shall expire one (1) year from the date of issuance. The Permit is renewable within sixty (60) days prior to the date of expiration.

8.5.2 It is the responsibility of the owner/operator of the I/M program station to pursue the Permit renewal through appropriate channels,

8.5.3 The Station Permit fee shall be paid annually to the Department by the billing due date set by the Department.

8.5.4 Prior to the date on which the Station Permit fee is due the Department shall attempt to notify each regulated Station of the amount of the fee. Fees unpaid after the billing due date will be assessed a late fee which shall be added to the original fee amount.

8.6 I/M Program Station Permit Revocation and Suspension.

8.6.1 I/M Program Station Permits may be suspended by the Department for violations of this Regulation.

8.6.2 I/M Program Station Permits may be revoked by the Department for severe and/or repeated violations of this Regulation.

8.6.3 Suspension of I/M Program Station Permits shall follow the provisions of Appendix D of this Regulation.

8.6.4 I/M Program Station Permits are and remain the property of the Department, only their use and the license they represent is tendered.

8.6.5 Station Permit may be suspended or revoked by the Department because of returned checks and may not be reinstated until repayment is confirmed. All returned checks will be charged a returned check handling fee (referenced in Appendix B).

8.6.6 Failure to pay the Station Permit fee and any additional charges after the due date may result in suspension and/or revocation of the Permit and the right to operate as an I/M Program Station.

8.7 The I/M Station Shall Hold the Department harmless.

In making application for a Permit or for its renewal, such action shall constitute a declaration by the applicant that the Department shall be held harmless from liability incurred due to action or inaction of I/M Program Station owner or their

employees.

9.0 INSPECTION PROCEDURE

9.1 The official emission inspection shall be solely performed by a Tester/Technician who has been certified at the station where the inspection is being performed and Department approved inspection procedures are being followed.

9.2 If the Tester/Technician is unable, unqualified, or unwilling to make the required repairs or adjustments, should the vehicle fail the inspection, he shall notify the owner/operator of the vehicle before the inspection is administered.
9.3 The entire inspection shall take place within the reach of the Analyzer hose and tachometer lead.

9.4 The temperature of the inspection area shall be between 41 Fahrenheit and 110 Fahrenheit (2 Celsius and 43 Celsius) during the inspection.

9.5 The Analyzer shall be kept in an area that provides adequate protection from the weather, wind, moisture, and extreme temperatures or any other damaging environmental exposure.

9.6 The electrical supply to the Analyzer shall be able to meet the Analyzer manufacturer's requirements for voltage and frequency stability.

9.7 The Tester/Technician shall not inspect or test any motor vehicle with a mechanical condition which may cause injury to personnel or damage to the station or test equipment or which may affect the validity of the inspection, until such condition is corrected. Such conditions included but are not limited to: coolant, oil, or fuel leaks, low oil or low fluid levels, carburetor gas overflow, vehicle electronic instrument panel malfunction, and visible emissions (smoker).

9.8 Any time an engine stalls during an Emission Test, the Emission Test shall be restarted. If a Tester/Technician cannot complete an Emission Test because of continuous stalling, then the stalling problem shall be corrected before the test is performed.

9.9 The Tester/Technician shall verify the vehicle license plate and vehicle identification numbers by comparing the information on the vehicle's registration with those on the vehicle and shall enter them in the Analyzer at the appropriate prompt.

9.9.1 The Tester/Technician shall verify the owner's name and correct address and enter this information into the Analyzer at the appropriate prompt.

9.9.2 The Tester/Technician shall determine and enter the county in which the vehicle is registered at the appropriate prompt.

9.9.3 The Tester/Technician shall enter completely and accurately all the information required as part of the data entry procedure for the official vehicle Emission Test on the approved Analyzer at the appropriate prompt sequence.

9.9.4 All data entries to the Analyzer during the inspection shall be true and factual.

9.10 The Tester/Technician shall:

9.10.1 Conduct the inspection in accordance with the prompts from the Analyzer and the requirements of this Regulation.

9.10.2 Examine the emissions/tune-up specification decal (sticker) under the hood and/or check an approved reference manual to determine if the vehicle was manufactured with a catalytic converter, air injection reaction (AIR) system, PCV System, EGR System, and Fuel Evaporate Control System, etc, as prompted by the Analyzer.

9.10.3 On 1996 and newer model year vehicles follow the OBD IM test procedures in accordance with Appendix F.

9.10.4 On 1990 through 1995 vehicles, visually inspect for the presence and apparent operability of the AIR system, catalytic converter, EGR system, Fuel Evaporative Control system, PCV system, and gas tank cap in accordance with Department procedures and record the information in the Analyzer. If these parts or systems have been removed, or are inoperable, the vehicle fails and the owner shall repair or replace the parts or systems before the emissions test may be continued.

9.10.5 On 1968 through 1989 vehicles, visually inspect for the presence and apparent operability of the AIR system, PCV system, EGR system, Fuel Evaporate Control System, catalytic converter and gas tank cap etc. in accordance with Department procedures and record the information on the emissions Analyzer.

9.11 Prior to performing the emission test each vehicle shall be checked to determine that it is at normal operating temperature by feeling the top radiator hose or by checking the temperature gauge.

9.12 The inspection shall be performed with the transmission in 'park' or 'neutral' and with all accessories off and the emergency brake applied (the vehicle may not be placed in gear to drop idle speed, headlights may be turned on).

9.13 The Analyzer probe shall be inserted into the exhaust pipe at least twelve inches (12") or as recommended by the Analyzer manufacturer, whichever is greater.

9.14 If a baffle or screen prevents probe insertion of at least twelve inches, a suitable probe adapter or snug fitting, non-reactive hose which effectively lengthens the exhaust pipe shall be used.

9.15 For all vehicles equipped with a multiple exhaust system that does not originate from a common point, both sides shall be tested simultaneously with an approved adaptor.

9.16 When inspecting a vehicle under windy conditions, the tailpipe shall be shielded from the wind with a suitable cover.

9.17 For 1995 Model Year Vehicles and Older:

9.17.1 With the tachometer properly attached to the vehicle being tested;

9.17.2 The vehicle shall be tested according to the testing sequence as programmed into the Analyzer and as detailed in the Analyzer

- specifications referenced in Appendix A. Vehicles failing because of
- excessive exhaust dilution shall repair the dilution problem prior to
 - continuing the emission test. The dilution standard shall be contained in the Analyzer specifications as referenced in Appendix A and adjusted when the Department determines by analysis that an adjustment is necessary to yield a more accurate level of emissions readings.

9.18 A Certificate of Compliance shall be issued if:

9.18.1 The vehicle emissions levels are the same as or less than the applicable emissions standards; and

9.18.2 For 1990 through 1995 model year vehicles, the vehicle passes the visual inspection described in Section 9.10;

9.18.3 Beginning January 1, 2001 for 1996 and newer model year vehicles, the vehicle passes the On-Board Diagnostics (OBD) test requirements as specified in Appendix F of this Regulation.

9.19 If the vehicle fails the initial Inspection, the owner shall have fifteen (15) days to have repairs or adjustments made and return the vehicle to the I/ M program station that performed the initial Inspection for one (1) free reinspection. The vehicle that failed the initial inspection shall then be issued a Certificate of Compliance only when all of the following are met:

9.19.1 The vehicle is re-inspected;

9.19.2 The vehicle's emission levels are the same or less than the applicable Emission Standards and;

9.19.3 For 1990 through 1995 model year vehicles, the vehicle passes the visual Inspection as provided for in Section 9.10.

9.19.4 Beginning January 1, 2001 for 1996 and newer model year vehicles, the vehicle passes the On-Board Diagnostics (OBD) test requirements as specified in Appendix F of this Regulation.

9.20 A Certificate of Waiver shall be issued only under the following conditions:

9.20.1 For all vehicles, air pollution control devices applicable and specified for the make, model and year of the vehicle as specified in Section 9.10 of this Regulation are in place and operable on the vehicle. If the devices have been removed or rendered inoperable, they shall be replaced or repaired before a Waiver is granted.

9.20.2 For 1968 to 1980 model year motor vehicles, if the vehicle continues to exceed applicable emissions standards after one hundred dollars (\$100) of acceptable emissions related repairs have been performed and the adjustments required by Section 10.0 have been performed by a Certified Emissions Repair Technician as part of the one hundred dollars (\$100) in emissions related repairs. Proof of repair costs, for that specific vehicle, shall be provided to the Department in the form of an itemized bill, invoice, work order, manifest or statement in which emission related parts and labor are specifically identified. If repairs are made by the vehicle owner or by someone who does not possess a valid Emission Repair Technician Certificate, the cost of labor may not be included in the one hundred dollars (\$100).

9.20.3 For 1981 to 1995 model year motor vehicles, at least two hundred dollars (\$200) of acceptable emissions related repairs have been performed and the adjustments (were applicable) required by Section 10.0 have been performed by a Certified Emissions Repair Technician as part of the two hundred dollars (\$200) in emissions related repairs. Proof of repair costs, for that specific vehicle, shall be provided to the Department in the form of an itemized bill, invoice, work order, manifest or statement in which emission related parts and labor are specifically identified. If repairs are made by the vehicle owner or by someone who does not possess a valid Emission Repair Technician Certificate, the cost of labor may not be included in the two hundred dollars (\$200).

9.20.4 For 1996 and newer model year vehicles, at least four hundred dollars (\$400) of acceptable emissions related repairs have been performed by a Certified Emissions Repair Technician as part of the four hundred dollars (\$400) in emissions related repairs. Proof of repair costs, for that specific vehicle, shall be provided to the Department in the form of an itemized bill, invoice, work order, manifest or statement in which emission related parts and labor are specifically identified. If repairs are made by the vehicle owner or by someone who does not possess a valid Emission Repair Technician Certificate, the cost of labor may not be included in the four hundred dollars (\$400). Any repair costs eligible under the federal emissions warranties shall not be eligible to be applied to the repair cost waiver limits.

9.20.5 Any vehicle that experiences an increase in any emissions levels shall not be eligible for a certificate of waiver regardless of the amount spent in attempting to repair the vehicle.

9.20.6 As used in this section acceptable emissions related repairs:

9.20.6.1 Refers to those expenditures and costs associated with the adjustment, maintenance, and repair of the motor vehicle which are directly related to reduction of exhaust emissions necessary to comply with the applicable emissions standards, cut-points, and procedures.

9.20.6.2 Refers to repairs and maintenance of the following systems, if done according to manufacturer's specifications, to the extent that the purpose is to reduce emissions:

(a) Air Intake Systems;

- (b) Ignition Systems;
- (c) Fuel Control Systems;

(d) Emission Control Systems except as noted in Section 9.22.4.4;

(e) Basic Engine Systems; and

(f) Repair of problems identified by On-Board Diagnostic (OBD) fault codes.

9.20.6.3 Does not include adjustments, maintenance, or repairs performed prior to the official emissions inspection.

9.20.6.4 Does not include the fee paid for the inspection.

9.20.6.5 Does not include costs associated with the repairs or replacement required by Section 9.10 or the replacement, and/or repair of air pollution control equipment on the vehicle if the need for such adjustment, maintenance, replacement, or repair is due to disconnection of, tampering with, or abuse of the emissions control systems, or costs incurred due to engine switching and/or modifications.

9.20.6.6 Does not include repairs performed to the vehicle's exhaust system to correct problems with excessive exhaust dilution.9.20.6.7 Does not include any diagnostics performed or any chemical additives.

9.21 Information regarding all performed repairs shall be entered into the appropriate data base of the Analyzer prior to the vehicle being reinspected.

ţ

9.22 Certificate of Waiver shall only be issued by the Department unless the Department determines other acceptable methods of issuance. A Waiver shall only be issued after determining that the vehicle complies with the requirements of Section 9.20. A Waiver shall not be issued to a vehicle with an inoperable or glowing check engine light.

9.23 Prior to referring the vehicle owner/operator to the Department for waiver eligibility, the Tester/Technician or Station shall verify that the repair and eligibility requirements of this Section have been met.

9.24 The inspection records shall be completed accurately, signed immediately, filed, and distributed, as required by the Department. The customer shall be given the appropriate copies.

9.25 After a passing Inspection customers shall be given the Certificate of Compliance along and appropriate copy of the Inspection form.

9.26 Vehicles capable of being operated on both gaseous and liquid petroleum fuels shall be tested for both fuels in accordance with the Analyzer specifications as referenced in Appendix A of this Regulation.

9.27 When a vehicle owner requests an Inspection, the Tester/Technician shall perform the inspection in the testing mode of the approved Analyzer. Performing a screening test (or pre-test) in the manual mode of the approved Analyzer or on a non-approved analyzer shall be a violation of this Regulation if the vehicle owner requested an emissions inspection. Adjustments or repairs shall not be made prior to a requested inspection.

9.28 At the end of each business day the UTAH2000 Analyzer shall be placed in a standby mode and be connected to the appropriate telecommunications line in order for the Department to collect data, load certificates, update station and Tester/Technician information or any other administrative procedures.

10.0 ADJUSTMENT PROCEDURES (Vehicles without computer Controlled Engine Systems)

10.1 The following adjustments should be performed on all 1981 and older vehicles (where applicable) that failed the I/M test. These adjustments must be performed by a emission repair technician before a vehicle will be eligible for an emissions waiver.

10.2 The manufacturer's or high altitude specifications, if available, for idle speed, idle air/fuel mixture, ignition timing, and dwell, shall be determined for the purpose of adjustment. The emissions repair technician shall refer to the emissions tune-up specifications. Fuel control systems designed with sealed tamper-resistant adjustment screws for air/fuel mixture shall be adjusted according to manufacturer's specifications and resealed. On vehicles that have limiter caps on the fuel control systems, the limiter caps shall be removed and the air/fuel ratio adjusted to meet manufacturer's specifications and the proper limiter caps shall be reinstalled. The adjustment procedures shall be as follows:

10.2.1 The dwell ,if applicable, shall be checked with a dwell meter to determine if it is within the recommended tolerance of 2 degrees of specifications. The dwell shall be reset if it exceeds this tolerance;

10.2.2 The idle speed shall be checked with a tachometer to determine if it is within 50 rpm of the manufacturer's specifications. If it is not, it shall be set to within 50 rpm of the manufacturer's specifications;

10.2.3 The ignition timing shall be checked, using a timing light or engine analyzer, to determine if it is within +4 degrees to -2 degrees of the recommended settings while the engine is idling at the specified speed. If the timing exceeds this tolerance, it shall be adjusted until it falls within +4 degrees to -2 degrees of the recommended setting;

10.2.4 The idle air/fuel ratio shall be adjusted according to manufacturer's suggested procedures and/or specifications using an infrared analyzer, propane enrichment kit, or tachometer;

10.2.5 The choke shall be checked for normal operation and, if appropriate, adjusted according to manufacturer's suggested procedures and/or specifications;

10.2.6 After completing the preceding steps, the idle speed shall be readjusted to manufacturer's specifications; and

10.2.7 The performed adjustments shall be entered in the required data base of the Analyzer.

11.0 ENGINE SWITCHING

11.1 All vehicles which qualify for testing under this section shall be tested by the Department only.

11.2 Vehicles qualifying for testing under this Section shall not be eligible for a Waiver.

11.3 Engine switching shall be allowed only in accordance with E.P.A. policy.

11.4 Vehicles not meeting the requirements of Section 11.0 shall be deemed as tampered and dealt with in accordance with the tampering provisions of this Regulation.

11.5 All 1990 and newer vehicles with switched engines shall be verified, to meet E.P.A. requirements, by the Department prior to issuance of a Certificate of Compliance.

11.6 For 1968 to 1989 vehicles, having an engine other than the original engine and emission control configuration are deemed as tampered. These vehicles must meet the HC and CO standards for the Model Year of the vehicle in order to receive a Certificate of Compliance, and are not eligible for a Certificate of Waiver, unless they are restored to the original engine and emission control configuration or a configuration approved by the Department.

12.0 EMISSION REPAIR TECHNICIAN / EMISSION TESTER CERTIFICATE OF QUALIFICATION

12.1 Emission Repair Technician Certification Required.

12.1.1 No person shall perform any part of the Inspection for the issuance of a Certificate of Compliance unless the person possesses a valid Emission Repair Technician Certificate of Qualification issued by the Department.

12.1.2 Applications for an Emission Repair Technician Certificate of Qualification shall be made upon an application form prescribed by the

Department. No Certificate of Qualification shall be issued unless:

12.1.2.1 The applicant has shown evidence of at least an associate degree in automotive technology or similar, or at least two (2) years work experience as an automotive mechanic, or other Department approved prerequisites.

12.1.2.2 The applicant has shown adequate competence by successfully completing the written and practical portions of the Emission Repair Technician Certification requirements as specified in this Regulation.

12.1.2.3 The applicant has paid the required permit fees as set by the Board of Health (reference in Appendix B).

12.1.3 An applicant shall comply with all of the terms stated in the Certificate of Qualification Application and with all the requirements of this Regulation.

12.1.4 An applicant shall complete a Department approved training course and shall demonstrate knowledge and skill in the performance of Inspections, use of the approved Analyze; and adjustment and repair of vehicles to manufacturer's specifications. Such knowledge and skill shall be shown by passing:

12.1.4.1 A written qualification test including but not limited to the following:

(a) Operation and purposes of emission control systems;

(b) Relationship of HC and CO to timing and air/fuel ratio adjustments;

(c) Adjustment to manufacturer's and high altitude specifications;

(d) Function and operation of computer controlled emission control system including, but not limited to the following:

Oxygen sensor; Engine control module (ECM); Other sensors; Three way catalytic convertor; Fuel injection system (type and discussion of differences); (e) Inspection procedures as outlined in this Regulation andprompted by the Analyzer;

(f) Operation of the Analyzer including the performance of gas calibration and leak check;

(g) The provisions of Section 207(b) Warranty provisions of the Federal Clean Air Act; and

(h) The provisions of this Regulation and other applicable Department policies and procedures.

12.1.4.2 A performance qualification test including but not limited to the following:

(a) Visual inspection and knowledge of the required emission control equipment;

(b) Demonstration of skill in proper use, care, maintenance, calibration and leak testing of the Analyzer;

(c) Demonstration of ability to conduct the Inspection;

(d) Demonstration of ability to adjust the engine systems to manufacturer's and high altitude specifications; and

(e) Demonstration of ability to accurately enter date in the Analyzer; and legibly, accurately complete the required reports and forms.

12.1.5 A signed "Hands-on Performance" check sheet shall be necessary for successful completion of the performance qualification test. The "hands-on Performance" check sheet shall be signed by an instructor or other person approved by the Department.

12.1.6 The Department shall issue Emission Technician Certificate to an applicant upon successful completion of the requirements of this Section .

12.1.7 The Emission Repair Technician Certificate shall be valid only at the station where the Emission Technician is presently employed. If the Emission Technician is later employed at another station, he shall notify the Department of the employment change. He shall also be required to be certified there prior to performing any Inspections. That certification will expire on the same date as the original. A transfer fee or duplicate fee will be charged, as set by the Board of Health and referenced in Appendix B of this Regulation. 12.1.8 Emission Repair Technician Certificates are and remain the property of the Department, only their use and the license they represent is tendered.

12.2 Emission Tester Certification Required

12.2.1 No person shall perform any part of the Inspection for the issuance of a Certificate of Compliance unless the person possesses a valid Emission Tester Certificate of Qualification issued by the Department.

12.2.2 Applications for an Emission Tester Certificate of Qualification shall be made upon an application form prescribed by the Department. No Certificate of Qualification shall be issued unless:

12.2.2.1 The applicant has shown adequate competence by successfully completing the written and practical portions of the Emission Tester Certification requirements as specified in this Regulation; and

12.2.2.2 The applicant has paid the required permit fees as set by the Board of Health and reference in Appendix B.

12.2.3 An applicant shall comply with all of the terms stated in the Certificate of Qualification Application and with all the requirements of this Regulation.

12.2.4 An applicant shall complete a Department approved training course and shall demonstrate knowledge and skill in the performance of emission testing and use of the approved Analyzer. Such knowledge and skill shall be shown by passing:

12.2.4.1 Operation and purposes of emission control systems;

12.2.4.2 Inspection procedures as outlined in this Regulation and prompted by the Analyzer;

12.2.4.3 Operation of the Analyzer including the performance of gas calibration and leak check;

12.2.4.4 The provisions of Section 207(b) Warranty provisions of the Federal Clean Air Act;

12.2.4.5 The provisions of this Regulation and other applicable Department policies and procedures; and
12.2.4.6 A performance qualification test including but not limited to the following:

(a) Visual inspection and knowledge of the required emission control equipment;

(b) Demonstration of skill in proper use, care, maintenance, calibration and leak testing of the Analyzer;

(c) Demonstration of ability to conduct the Inspection; and

(d) Demonstration of ability to accurately enter date in the Analyzer; and legibly, accurately complete the required reports and forms.

12.2.5 A signed "Hands-on Performance" check sheet shall be necessary for successful completion of the performance qualification test. The "hands-on Performance" check sheet shall be signed by an instructor or other person approved by the Department.

12.2.6 The Department shall issue an Emission Tester Certificate to an applicant upon successful completion of the requirements of this Section.

12.2.7 The Emission Tester Certificate shall be valid only at the station where the Tester is presently employed. If the Tester is later employed at another station, he shall notify the Department of the employment change. He shall also be required to be certified there prior to performing any Inspections. That certification will expire on the same date as the original. A transfer fee or duplicate fee will be charged, as set by the Board of Health and referenced in Appendix B of this Regulation.

12.2.8 Emission Tester Certificates are and remain the property of the Department, only their use and the license they represent is tendered.

12.3 Tester/Technician Certificate of Qualification Suspension and Revocation.

12.3.1 Tester/Technician Certificate of Qualification may be suspended by the Department for violations of this Regulation.

12.3.2 Tester/Technician Certificates of Qualification may be revoked by the Department for Severe and/or repeated violations of this Regulation.

12.3.3 Suspension or revocation of Tester/Technician Certification shall follow the provisions of Appendix D of this Regulation.

12.3.4 Tester/Technician Certification may be suspended or revoked by the Department because of returned checks and may not be reinstated until repayment is confirmed. All returned checks will be charged a returned check handling fee (referenced in Appendix B).

12.4 Re-Qualification Requirements for all Tester/Technician Certification.

12.4.1 Tester/Technician Certificates shall not be transferred from one person to another person. Tester/Technician Certificates may not be transferred from one I/M Program Station to another or from one status to another, i.e., from test and repair to test only, without a written request and Department approval.

12.4.2 The Department may renew Certification for an existing Tester/Technician after a properly completed renewal form is submitted, reviewed, and approved, the re-certification requirements have been completed, the fees are paid and the Tester/Technician has complied with this Regulation.

12.4.3 Upon determination, by the Department, of the necessity of updating the qualification for Tester/Technician, they shall be required to re-qualify.

12.4.4 Tester/Technicians shall be required to re-qualify within a specified time period determined by the Department (from the date of written notification by the Department). The notice shall be mailed to the address of record in the office of the Department. Failure to re-qualify within the required period of time shall result in suspension or revocation of the Tester/Technician certification as described in this Regulation.

12.5 Certification Expiration.

12.5.1 The Tester/Technician Certificate shall be issued annually and shall expire one year form the date of issuance. The Permit shall be renewable within sixty (60) days prior to the date of expiration.

12.5.2 It is the responsibility of the Tester/Technician to pursue the renewal of the Tester/Technician Certificate.

12.5.2.1 Tester/Technicians who are lacking in training hours may be allowed to have a single 30 day extension past the expiration of their Certificate provided they make a request to the Department in person prior the expiration date of their Certificate and:

(a) Pay the extension fee as referenced in Appendix B; and

(b) Show proof that they are registered for training that qualifies for re-certification.

12.5.2.2 When an extension has been granted and after the training is finished the Tester/Technician shall come back and apply for their renewal approval and pay the renewal fee (reference in Appendix B). If the Tester/Technician comes in after the extension expires with the proper training requirements he will be charged the late renewal fee.

13.0 EMISSIONS STANDARDS FOR MOTOR VEHICLES

13.1 In order to obtain a valid Certificate of Compliance, a motor vehicle subject to an Emission Inspection shall not exceed the maximum concentrations for carbon monoxide (CO), and Hydrocarbons (HC) as established by the Board of Health (referenced in Appendix C) and/or pass a valid OBD test.

13.2 Maximum concentration of Cut-Points shall be adopted by the Board of Health to meet the National Ambient Air Quality Standards established by the United States Environmental Protection Agency (USEPA). The adopted Cut-Points shall remain in effect until changed by the Board of Health. Any change in Cut-Points shall be effective upon the first day of any calendar month designated by the Board of Health. The Board of Health shall adopt Cut-Points by considering the following factors:

13.2.1 The existing ambient air quality;

13.2.2 To provide for the required stringency necessary to meet air Quality Standards;

13.2.3 The requirements for air quality programs currently in effect as promulgated by the EPA, the Utah Department of Environmental Quality, the County and the Board of Health; The Cut-Points established shall be part of an overall program, in accordance with EPA guidelines, to achieve the required tailpipe reductions, of CO and HC from motor vehicles measured from the date this program is implemented;

13.2.4 The general level of emission control technology on vehicles registered in the County;

13.2.5 Population growth and other factors which may reasonably be expected to impact CO and HC concentrations in the atmosphere;

13.2.6 The likelihood of a particular Cut-Point to achieve desired air quality goals; and

13.2.7 The ability to ensure compliance with the requirements of Section 41-6-163.6 and Section 41-6-163.7, Utah Code Annotated, 1953, as amended.

14.0 CERTIFICATE OF COMPLIANCE, CERTIFICATE OF COMPLIANCE NUMBERS AND CERTIFICATE OF WAIVER

14.1 No person shall make, issue or knowingly use any imitation or counterfeit of a Certificate of Compliance, Certificate of Compliance Numbers or Certificate of Waiver.

14.1.1 No person shall use a stolen certificate or certificate number.

14.2 Certificate of Compliance Numbers shall be obtained only from the Department.

14.3 No refund or credit shall be allowed for unused certificates/numbers, except as provided in Section 13.10.

14.4 Obtaining Certificate of Compliance Numbers.

· . (

- at 14.4.1 Certificate of Compliance Numbers may be obtained in person by
- a an authorized representative, of the I/M Program Station, possessing an acceptable form of identification.

14.4.2 Certificate of Compliance Numbers shall be issued in lots to be determined by the Department. The Department may limit the number of Certificate of Compliance Numbers issued to the number that the Department feels can be secured and stored safely.

14.4.3 Certificates of Compliance shall not be sold, loaned, transferred, or given to any other I/M Program Station, or any unauthorized individual. The I/M Program Station shall at all times account for all certificates/numbers that have been issued to the station. Failure to properly safeguard and/or account for Certificate of Compliance Numbers may lead to immediate suspension of Station Permit.

14.5 Certificates of Compliance shall only be issued to the vehicle owner/operator after being printed by the Analyzer. Completion of Certificates by other means than the Analyzer by any person or station other than the Department is strictly prohibited. The certificate shall be signed immediately after printing by the Tester/Technician who inspected the vehicle.

14.6 Certificates of Compliance shall not be issued until an inspection has been

performed as required by this Regulation.

14.7 All unused Certificate of Compliance Numbers shall be kept in a secure place at all times to prevent loss or theft.

14.8 Certificate of Compliance Numbers found to be missing, stolen, or unaccounted for, shall be reported to the Department within twenty-four hours and the Station shall cease performing Inspections until an investigation by the Department has been completed and the Department re-authorizes the Station to resume Inspections.

14.9 I/M Program Stations shall have Department issued Certificate of Compliance Numbers on hand at all times.

14.10 Upon final cancellation, suspension or revocation of the I/M Station Permit, the Station owner, manager or other responsible person shall immediately surrender all unused Certificates of Compliance Numbers to the Department. The Department may receipt and refund the fee paid for unused certificates of Compliance Numbers to the Station owner according to the Weber County Clerk/Auditor's procedures. Upon transfer or termination of business ownership, the Station Permit and all Certificate of Compliance Numbers shall be immediately surrendered to the Department. Any person acquiring a business that has been permitted as an official I/M Program Station, is prohibited from using any Permit, Certificate of Compliance Numbers or emissions documents issued to the former business without authorization of the Department; and

14.10.1 Any Analyzer manufacturer or their authorized representative who repossesses or otherwise removes an Analyzer from an I/M Program Station shall immediately notify the Department and shall immediately surrender any Certificate of Compliance Numbers and the data disk that may still be in the Analyzer to the Department.

14.11 No person may alter computer software or electronic data associated with the Inspection without written permission by the Department.

14.12 No person may engage in repair of analyzer unless approved by the Department.

15.0 RIGHT TO APPEAL

Within ten (10) calendar days after the Department has issued a notice of violation, Permits denial, warning, suspension or revocation, any person(s) aggrieved may request in writing, a hearing before the Department. The written request for hearing shall be made on a form provided by the Department. The hearing shall take place within ten (10) calendar days, or other time mutually

agreed upon, after there request is received. A written notice of the Department's final determination shall be given within ten (10) calendar days after adjournment of the hearing. The Department may sustain, modify, or reverse the action or order, and/or negotiate a consent agreement but shall not require less than the minimum requirements of this Regulation and other applicable law.

16.0 PENALTY

ì

16.1 Any person who is found guilty of violating any of the provisions of this Regulation, either by failing to do those acts required herein or by doing a prohibited act, shall be guilty of a class B misdemeanor pursuant to Section 26A-1-123, Utah Code Annotated, 1953, as amended. If a person is found guilty of a subsequent similar violation within two years, he shall be guilty of a class A misdemeanor pursuant to Section 26A-1-123, Utah Code annotated, 1953, as amended.

16.2 Each day such a violation is committed or permitted to continues shall constitute a separate violation. Also, each improperly issued Certificate of Compliance constitutes a separate violation.

16.3 The County Attorney, may initiate legal action, civil or criminal, requested by the Department to abate any condition that exists in violation of this Regulation.

16.4 In addition to other penalties imposed by a court of competent jurisdiction, any person(s) found guilty of violating any of this Regulation shall be liable for all expenses incurred by the Department in prosecuting and/or abating the violation.

16.5 The Penalty Schedule for Permits warning, Permits suspension, Permits revocation, and/or negotiated settlement agreements as adopted by the Board of Health shall be referenced in Appendix D of this Regulation and may be changed and updated by the Board of Health as deemed necessary to accomplish the purposes of this Regulation.

16.6 Enforcement of any criminal penalties does not preclude imposition of administrative or civil penalties and visa-versa.

17.0 SEVERABILITY

If any provision, clause, sentence, or paragraph of this Regulation or the application thereof to any person or circumstances shall be held to be invalid, such invalidity shall not affect the other provisions or applications of this Regulation. The valid part of any clause, sentence, or paragraph of this Regulation shall be given independence from the invalid provisions or application and to this end the provisions of this Regulation are hereby declared to be severable.

18.0 EFFECTIVE DATE

.

This Regulation including Appendix A through F shall become effective the day of its adoption by the Board of Health. Appendices may be modified by the Board of Health without affecting the rest of this Regulation. Appendices when amended by the Board shall become effective on the day of adoption of amendments by the Board of Health.

Adopted by the Weber-Morgan Board of Health - May 22, 2000. Amendments to Regulation and Appendix B shall become effective July 1, 2003.

APPENDIX A

UTAH2000 Analyzer Specifications

Available upon request from the Department

:

All proposed options for test procedures, equipment specifications and program design shall meet emission reduction required by Section 9.18. The option to be selected is that which is most cost effective to the consumer as determined by the Board of Health.

The inspection for light duty vehicles (0-8500 lbs. GVWR) will consist of a stationary test at low and high speed idle for concentrations of hydrocarbons (HC), carbon monoxide (CO) and/or an OBD IM test, a functional inspection of the gas cap and a visual/tampering inspection of the fuel filler neck Restrictor, PCV, EGR, A.I.R. and catalytic converter systems.

All test equipment must meet specifications established by the Department. The Department may require Analyzers to meet portions of specifications established by the State of California Bureau of Automotive Repair termed Bar 97.

APPENDIX B

FEE SCHEDULE

The assessed fees for implementing the requirements of Section 6.10 of the Vehicles Emissions Inspection and Maintenance Program shall be:

PERMITTING OF AN OFFICIAL I/M PROGRAM STATION	\$250.00
ANNUAL RENEWAL OF STATION PERMIT	\$25.00
ANNUAL RENEWAL OF EXPIRED STATION PERMIT	\$50.00
RE-PERMITTING AN I/M STATION AT A NEW LOCATION	\$50.00
CERTIFICATE OF QUALIFICATION	\$25.00
TEMPORARY STATION PERMIT	\$50.00
ANNUAL RENEWAL OF TESTER/TECHNICIAN CERTIFICATION	\$10.00
TRANSFER OF TESTER/TECHNICIAN CERTIFICATE TO NEW STATION	\$10.00
DUPLICATE TESTER/TECHNICIAN CERTIFICATE	\$10.00
ANNUAL RENEWAL OF EXPIRED TESTER/TECHNICIAN CERTIFICATION	\$30.00
30 DAY EXTENSION OF TESTER/TECHNICIAN CERTIFICATE EXPIRATION DATE	\$30.00
AIR POLLUTION CONTROL FEE (PAID AT REGISTRATION ALL VEHICLES)	\$1.00
DIESEL CONTROL FEE (PAID AT REGISTRATION ALL DIESEL VEHICLES)	\$9.00
EMISSIONS CERTIFICATE OF COMPLIANCE/NUMBER (EACH)	\$3.00
DUPLICATE CERTIFICATE OF COMPLIANCE (MAXIMUM)	\$3.00
COST FOR CHECKING A CATALYTIC CONVERTER (DEPARTMENT ONLY)	\$25.00
MAXIMUM EMISSION INSPECTION FEE	\$25.00
CLEAR A TAMPERING LOCK OUT THIRD TIME WITHIN ONE YEAR AND EACH TIME AFTER	\$10.00
RETURNED CHECK FEE	\$25.00
COPY OF I/M PROGRAM REGULATION	\$2.50

APPENDIX C

MOTOR VEHICLE EMISSIONS Inspection and Maintenance PROGRAM

The following schedule gives the maximum allowable concentration for carbon monoxide (CO) and hydrocarbons (HC) for both cars and trucks as determined by an approved Analyzer using the prescribed procedures. The effective date for these cut-points is 1 November 1991.

ALL PASSENGER VEHICLES 1978 AND OLDER LIGHT DUTY TRUCKS 6,000 POUNDS GVWR OR LESS 1979 TRUCKS AND NEWER 8,500 POUNDS GVWR OR LESS				
MAXIM	UM CONCENTRATION STAN	DARDS		
<u>MODEL YEAR</u> 1968 - 1969 1970 - 1974 1975 - 1976 1977 - 1979 1980 1981 AND NEWER	PERCENT CARBON <u>MONOXIDE</u> 6.0 5.0 4.0 3.0 2.0 1.2	PARTS/MILLION <u>HYDROCARBONS</u> 800 700 600 500 300 220		
HEAVY DUTY TRUCKS AND VANS 1978 AND OLDER 6,001 OR GREATER 1979 AND NEWER OVER 8,500 POUNDS GVWR <u>MAXIMUM CONCENTRATION STANDARDS</u>				
<u>MODEL YEAR</u> 1968 - 1969 1970 - 1978 1979 - 1980 1981 AND NEWER	PERCENT CARBON MONOXIDE 7.0 5.0 4.0 3.5	PARTS/MILLION HYDROCARBONS 1500 1200 1000 800		

Note: These should be considered as "cut-points" for maximum allowable emissions levels. Vehicles must never be reset to these emission levels when readjustments are made, but rather shall be adjusted using manufacturer's specifications. By using manufacturer's specifications, the emission levels should be well below the "cut-points".

APPENDIX D

PENALTY SCHEDULE

VIOLATION	1 st OCCURRENCE		3 RD OCCURRENCE	4 TH OCCURRENCE
Failure to Inspect or probing a vehicle other than vehicle entered in test record.	Up to 6 Month Suspension (Station & Tester/Technician)	Revocation (Station & Tester/Technician)		
Non-certified Tester/Technician performing inspection	Up to 6 Month Suspension (Station & Tester/Technician)	6 Month Suspension (Station & Tester/Technician)	Revocation (Station & Tester/Technician)	
Fail a Passing Vehicle/Pass a Failing Vehicle	Up to 3month Suspension (Station & Tester/Technician)	Up to 6 Month Suspension (Station & Tester/Technician)	Revocation (Station & Tester/Technician)	
Failure to Comply with Proper Test Procedures	Formal Warning (Station & Tester/Technician)	Up to 3 Month Suspension (Station & Tester/Technician)	Up to 6 Month Suspension (Station & Tester/Technician)	Revocation (Station & Tester/Technician)
Performing Unnecessary or Unrelated Repairs	1 Month Suspension (Station & Tester/Technician)	3 Month Suspension (Station & Tester/Technician)	Revocation (Station & Tester/Technician)	
Inaccurate or Incomplete Data	Formal Warning (Station & Tester/Technician).	Up to 30 Day Suspension (Station & Tester/Technician)	Up to 3 Month Suspension (Station & Tester/Technician)	Revocation (Station & Tester/Technician)

OTHER APPROPRIATE WARNINGS, SUSPENSIONS, NEGOTIATED CONSENT AGREEMENT, AND/OR REVOCATIONS AS DEEMED NECESSARY AND PRUDENT BY THE DEPARTMENT.

All Tester/Technician and Station suspensions may be reduced in length by a Negotiated Consent Agreement which may substitute monetary penalties for part of all of the suspension time. Consent Agreements for stations are based on 50% of the testing revenue that could have been expected during the suspension time based on a minimum test fee of \$20.00. Consent agreements for the technicians is based on \$100 increments for any 15 day period or portion thereof up to a maximum of 90 days. Negotiated Consent Agreements are only applicable in relation to suspension.

APPENDIX E

Department Approved Calibration Gas and Calibration Gas Blenders Available at:

List available upon request from Department

APPENDIX F

OBD IM TEST PROCEDURES

The following test procedure is to be followed for 1996 model year vehicles or newer:

1. Turn the ignition switch to the off position for at least 12 seconds;

2. Visually examine the instrument panel to determine if the Malfunction Indicator Light (MIL) illuminates when the ignition key is turned to the key on/ engine off position. (Note: MIL may only light briefly, watch closely.) If the MIL does not illuminate at all then the vehicle fails if tested after December 31, 2000, perform two speed idle test if tested prior to January 1, 2001 and advise the owner/operator to repair the MIL problem. Exception: OBDII deficient vehicles will be given the two speed idle test.

3. Locate the Data Link Connector (DLC) in the vehicle and connect the OBDII test equipment (Analyzer). If DLC is missing, has been tampered with, or is otherwise inoperable then the vehicle fails if tested after December 31, 2000, perform two speed idle test if tested prior to January 1, 2001 and advise the owner/operator to repair the MIL problem. Exception: OBDII deficient vehicles will be given the two speed idle test.

4. Start and leave the engine running. Check for MIL illumination. If the MIL illuminates after the engine has been started, even if no fault codes are present, the vehicle fails if tested after December 31, 2000, perform two speed idle test if tested prior to January 1, 2001 and advise the owner/operator to repair the MIL problem. Exception: OBDII deficient vehicles will be given the two speed idle test.

5. Vehicle status not ready; perform two speed idle test prior to January 1, 2001, after December 31, 2000 the vehicle must be preconditioned to a ready status. Exception: OBDII deficient vehicle will be given the two speed idle test. (Note: Certain vehicles may be exempted by the Weber-Morgan Health Department from OBD IM testing. Exemption can only be given by the Weber-Morgan Health Department.)

6. Vehicle status ready; when MIL light is on check Diagnostic Trouble Codes (DTC=s); determine pass or fail, and record results. (Note: Do not check DTCs if MIL is not commanded on.)

7. Turn off engine and disconnect test equipment.

WEBER-MORGAN DISTRICT HEALTH DEPARTMENT

POLICIES AND PROCEDURES OF THE WEBER COUNTY I/M PROGRAM

STATION AUDITS

- 1. I/M Program stations are to be audited on a quarterly basis with the high volume and high risk stations done on a more frequent schedule. Audit frequency should be altered to a degree to help prevent the predictability of the audit date by station personnel.
- 2. Monthly station/mechanic evaluation reports: Each auditor is routinely given a monthly report listing the activities of each station and mechanic in his district. He is instructed to review this printout and identify problem areas, i.e., high fail rates, low fail rates, tampered vehicle, etc. He then discusses these problems with the station manager and mechanic during the audit. These reports are then filed in the station folder for future reference and used as a means of determining if the problem has really been taken care of. Repeat violators are sent notices; and then suspended if the problem persists. Monthly evaluation printouts are also used to select stations for undercover audits (low-high fail rates).
- 3. One copy of the station evaluation and one copy of the station performance report will be obtained during each audit.
- 4. The station performance report will be completed during each audit. Any violation(s) will be noted and station manager or responsible individual informed regarding the violation(s).
- 5. Gas audits of station analyzer shall be performed on a quarterly basis by the station auditor. Using span gas provided by the Department, the analyzer readings must be within a tolerance of +/-5% for co and +/7% with hang-up correction for HC. If analyzer fails the gas audit, the auditor shall perform analyzer maintenance and attempt another gas audit. If this attempt fails the auditor shall place the analyzer out of service, in accordance with the Weber County I/M Regulations, Section 6.9, Subsection 6.9.2.1. If analyzer is locked out due to a gas audit failure, it shall be noted on the station performance report and station manager shall read and initial and date this statement. Once notification of repair is received from station personnel, the station auditor shall make every effort to remove lockout within the same date of notification. If same date lockout removal cannot be achieved, then next day will be acceptable.
- 6. If a station auditor notes a consistent violation (two consecutive violations), that the I/M station does not correct in spite of violation notices on performance reports, and has been given a reasonable amount of time for any corrections, then

this information is to be given to the I/M Program Director for administrative action.

- 7. All items covered on the station performance report are to be checked on each audit. Of particular concern is the certificate accountability. All unaccounted certificates will constitute a station to cease inspection operations until the Department authorizes the station to resume inspections, as per Weber County I/M Regulations, Section 13.0, Subsection 13.8.
- 8. Scheduling: Auditors routinely vary the date that they visit each station. This reduced the possibility that the station will prepare for the audit just before expected visit. This has caught some stations off guard and resulted in the discovery of improprieties that otherwise would go undetected. The policy of using the supervisor to perform random audits has also helped to take away a predicted routine that stations and mechanics could abuse.
- 9. During regular audits, the analyzer's data disk will be removed from the analyzer and exchanged for a new, formatted disk provided by the Department. Old disks will be discarded after the information has been downloaded to the system.
- 10. The analyzer information disk is to be returned to the office and should be transferred to the county mainframe on a timely basis. The disk should then be saved for future reference or until it has been determined that the information has been successfully uploaded.
- 11. Where possible, the station auditor should attempt to observe an actual emissions test during the station audit.
- 12. In the event that a station comes under investigation for alleged misconduct, the station's auditor will conduct a station audit immediately and turn over all information to the Department Director for any administrative action as deemed by the Weber County I/M Regulations and the Department Director.

AUDIT PROCEDURES - UTAH 91 ANALYZERS

- I. Prior to the audit:
 - A. Review station folder to determine if there is any follow-up corrective action required from the previous audit.
 - B. View the performance report to determine if there are nay missing certificates.
 - C. Obtain calibration cylinder, vehicle identification reference manual. (Or use station reference manual)
- II During the audit:

- A. Obtain station performance report. Verify that the station permit, mechanic's certificates, station sign, fee chart, reference manuals, and other required equipment is available. Note any discrepancies for the station performance report.
- B. Access the station analyzer audit menu of the analyzer computer. <u>Select</u> <u>the station evaluation report</u> (see attachment no. 1) and print out the information. Review this information to see if there are any unusual entries, i.e., a significant increase/decrease in the fail rate. Note and question the station manager on any unusual information. Verify that the time and date entries are correct.
 - 1. <u>Select the station performance report</u> (see attachment no. 2) and program in the appropriate information. Note on this report the number of the next unused certificate.
 - 2. Select gas audit and introduce your audit calibration gas into sample line probe. Obtain printout of results (see attachment no. 8). Calculate the reading that you are supposed to observe and compare this reading with that on the printout. If it is determined that the analyzer is out of calibration, select the calibration history and review previous calibration reports (attachment no. 9). Determine if there is any indication of drift beyond the maximum allowed since the last calibration. If so, schedule follow up checks to determine how long the analyzer will remain in calibration. If it is determined that the analyzer cannot remain in calibration for three days, it will be locked out until repairs are completed.
 - 3. Select the mechanic evaluation report (see attachment no. 3) and program in each mechanic. Obtain printouts and review for unusual entries, i.e., <u>significant fail or pass rates</u>. Any significant discrepancies should be noted and discussed with the station manager.
 - a. Select multiple repairs report (see attachment no. 4) for suspect mechanics and obtain printout. May refer back to station/mechanic evaluation report. Unusual entries will be discussed with station manager.
 - b. Select fast testing report (attachment no. 5) and review for unusual entries. If observed, obtain printout for documentation of problem.
 - c. Select emissions reductions report (attachment no. 6) and review for 50% reductions in HC and CO if a significant number of repairs have been performed. If not, obtain

printout for follow up action or to discuss with I/M mechanic or other responsible individual.

d. Select consecutive test comparison report and observe to see if there are any repeat tests with the same readings. If so, obtain a printout and discuss with the station manager or notify the Director to do an undercover audit there. You may also want to printout test records to verify the same readings indicating that the same vehicle was probably used to each test.

4. Select auditor's notes (see attachment no. 10) and review to determine if there are any unresolved discrepancies that require follow up. Also, use this opportunity to record any information that you want to pass on to the next audit.

5. Select analyzer tampering/access report (see attachment no. 11) and review for lockouts. Insure that county lockouts involving everything but service are cleared by the auditor.

- C. Review station files:
 - 1. Using the vehicle identification encyclopedia, randomly select a few vehicle inspection reports (VIRs) and check for proper entries for engine displacement. Then using the application guide, verify that the device check is properly programmed on the report. If a significant number for a particular mechanic is observed, check more thoroughly to determine the extent of the problem. Bring this information to the attention of the station manager and mechanic.
 - 2. Insure that the VIRs are filed in numerical sequence and that they have the yellow copy of the certificate of compliance attached.
 - 3. Insure that separate files are available for fail and pending vehicles awaiting retest.
- III After the audit:
 - A. Advise supervisor of any discrepancy involving a warning or suspension letter and provide documentation to support it.
 - B. Compile all information and file in the station folder.
 - C. Complete I/M station audit report indicating when the audit was performed.

COMPLAINTS

1. When and emission related complaint is received, the secretary will retrieve the following information on an approved complaint form: complainant's name, address, home and/or work phone, station at which incident occurred, inspectors name, dated and time of the incident and a description of complaint. Complaint will then be assigned to a station auditor who in a timely manner will investigate the complaint, take appropriate action, file a report with the Director, and notify the complainant as to the results of the investigation.

POLICIES AND PROCEDURES COVERT AUDITS

- 1. Covert audits. The counties to the extent possible perform a covert audit of each inspector and station at least once a year. The number of covert audits at least equals the number of permitted inspectors. Covert audits are performed using a variety of vehicles that are representative of the subject malfunctions. Suspected problem stations and inspectors are targeted for earlier and more frequent audits. Complaints also trigger additional audits.
- Covert performance audits shall include: Remote visual observation of inspector performance, which may include the use of aids such as binoculars or video cameras, at least once per year per inspector in high-volume stations (i.e., those performing more than 4000 test per year);

Site visits at least once per year per number of permitted inspectors (per inspector FTE) using covert vehicles set to fail (this requirement sets a minimum level of activity not a requirement that each inspector be involved in a covert audit); and

For stations that conduct both testing and repairs, at least one covert vehicle visit per station per year including purchase of repairs and subsequent retesting if the vehicle is initially failed for tailpipe emissions, at least once per year per station.

- 3. Vehicles used for undercover audits will be pre-tested and adjusted for use at the Technical Center.
- 4. The vehicle operator must follow the guidelines in memorandum dated 20 March 1986 from deputy County Attorney regarding "Guidelines to Avoiding Entrapment".
- 5. The audit should be performed in such a manner as to avoid giving employees at the station being audited any indication that an audit is being performed.
- 6. The surveillance report form should be filled out completely for each phase of the audit.

- 7. Covert auditors will complete a training course provided by the Department which is designed to thoroughly familiarize them with all aspects of the emission inspection procedures.
- 8. Covert auditors will be rotated to avoid becoming recognizable to inspection station personnel.
- 9. In all cases, vehicles shall be adjusted to fail either the emissions or tampering or both aspects of the test.
- 10. On covert audit failures, administrative actions will follow the penalty schedule of the Weber County I/M Regulations.
- 11. A complete audit includes the following phases:
 - a. Pre-testing and adjusting of vehicle.
 - b. Audit/testing at designated station.
 - c. Post-testing and readjusting original configuration if needed. Completed inspection forms are to be returned to the Program Director
 - d. A briefing with Director of his representative on test results.
 - e. Appropriate action taken on any discrepancies.

VEHICLE TAMPERING INSPECTION PROTOCOL

In accordance with Weber County Inspection/Maintenance Rules and Regulations, Section 1.77 Tampering Definitions, Section 3.0 Jurisdiction of Counties, Section 4.1.5 Powers and Duties, and Section 4.2.3 Powers and Duties.

All Weber County Inspection/Maintenance inspector/mechanics when conducting a tampering inspection, will determine the following on all vehicle model years included in the program:

1. Through the use of the vehicle decal or an approved vehicle reference manual, determine those emission control devices that should be present on a particular vehicle.

2. Conduct a visual inspection of said vehicle to determine if those items are indeed present.

3. Determine if those items have been intentionally or accidentally removed or altered from the original configuration.

4. Enter pass/fail information correctly into the analyzer.

5. Inspectors will be held accountable for their tampering inspections. If an inspection is found to be incorrect, and inspector may be subject to penalties found in the Inspection/Maintenance Penalty Schedule.

ENFORCEMENT AGAINST STATIONS AND INSPECTORS

1. A penalty schedule to be used in the case of violations by individual inspectors and stations. The penalties escalate based upon the frequency

of commission and relative severity in terms of a violation's direct impact on the emission reduction potential of the program. Penalties shall take the form of suspension of license revocation with annual competency re-certification requirements.

- 2. Auditors are granted the authority to immediately suspend station and inspector licenses upon finding a violation. A formal hearing is conducted to evaluate evidence and determine penalty. The I/M program management shall record all enforcement activities, including all warnings, fines, suspensions, revocations and other notices of violation. At least once a year, the program shall compile summary statistics on its enforcement activities and report this information to the public and EPA.
- 3. In the case of inspector incompetence, the inspector is required to be retrained and must successfully demonstrate the ability to perform the test procedure prior to the restoration of testing privileges. For cases of inspector violations resulting from factors other than incompetence, the inspector shall be suspended from testing for a minimum of one month, with such suspension increasing with the severity and frequency of violation, leading ultimately to permanent license revocation.

VIOLATION	1ST OCCURRENCE	2ND	3RD	4TH
		OCCURRENCE	OCCURRENCE	OCCURRENCE
INTENTIONALLY	6 MONTH	REVOCATION		
PASSING A	SUSPENSION			
FAILING VEHICLE				
FAILURE TO	UP TO 6 MONTH	REVOCATION		
INSPECT	SUSPENSION			
NON-CERTIFIED	UP TO 6 MONTH	REVOCATION		
INSPECTOR	SUSPENSION			
PASS A FAILING	UP TO 3 MONTH	UP TO 6 MONTH	REVOCATION	
VEHICLE	SUSPENSION	SUSPENSION		
FAIL A PASSING	UP TO 3 MONTH	UP TO 6 MONTH	REVOCATION	
VEHICLE	SUSPENSION	SUSPENSION		
FAILURE TO	FORMAL	UP TO 3 MONTH	UP TO 6 MONTH	REVOCATION
CALIBRATE OR	WARNING	SUSPENSION	SUSPENSION	
OTHER			l	
PROCEDURES				
PERFORMING	1 MONTH	3 MONTH	REVOCATION	
UNNECESSARY	SUSPENSION	SUSPENSION		
OR UNRELATED				
REPAIRS				
INACCURATE OR	FORMAL	UP TO 30 DAY	UP TO 3 MONTH	REVOCATION
INCOMPLETE	WARNING	SUSPENSION	SUSPENSION	
DATA				
OTHER APPROPRIATE WARNINGS, SUSPENSIONS, NEGOTIATED CONSENT AGREEMENTS,				
AND/OR REV	OCATIONS AS DEE	MED NECESSARY A	ND PRUDENT BY 1	HE DIVISION

APPENDIX D PENALTY SCHEDULE

All suspensions may be reduced in length by a negotiated consent agreement which may substitute monetary penalties for part or all of the suspension time. Negotiated Consent Agreements are only applicable in relation to suspension.

UTAH STATE IMPLEMENTATION PLAN

SECTION X

VEHICLE INSPECTION AND MAINTENANCE PROGRAM

Part F

Cache County

Adopted by the Utah Air Quality Board November 6, 2013

Section X Part B, Page i

Table of Contents

1. Applicability	1
2. Description of Cache I/M Programs	1
3. I/M SIP Implementation	3

Section X Part B, Page ii

SECTION X Part F Cache County Emission Inspection/ Maintenance Program APPENDICES

1. Cache County Emission Inspection/ Maintenance Program Ordinance 2013-04

2. Bear River Health Department Regulation 2013-1

Section X Part B, Page iii

UTAH STATE IMPLEMENTATION PLAN SECTION X, PART F VEHICLE INSPECTION AND MAINTENANCE (I/M) PROGRAM

5 1. Applicability

6

1

2

3

4

Cache County I/M program requirements: Cache County was designated nonattainment for the 7 PM_{2.5} National Ambient Air Quality Standard (NAAQS) on December 14, 2009 (74 FR 58688, 8 November 13, 2009). Accordingly, Cache County must implement control strategies to attain the 9 PM2.5 NAAQS. A motor vehicle emission inspection and maintenance (I/M) program has been 10 identified by the PM2.5 State Implementation Plan (SIP) as a necessary control strategy to attain 11 the PM2.5 NAAQS as expeditiously as practicable. Therefore, pursuant to Utah Code Annotated 12 41-6a-1642, Cache County must implement an I/M program that complies with the minimum 13 requirements of 40 CFR Part 51 Subpart S. Cache County will implement its I/M program 14 county-wide. Parts A and F of Section X demonstrate compliance with 40 CFR Part 51 Subpart 15 S for Cache County. 16

17

18 2. Description of Cache I/M Programs

19

Below is a summary of Cache County's I/M program. Section X, Part F Appendices 1 and 2 contain the essential documents for Cache County's I/M program.

22 23

> 24 25

> 31

34

Network Type: Cache County's I/M program will comprise of a decentralized test-and-repair network.

Test Convenience: Cache County will make every effort to ensure that its citizens will
 have stations conveniently located throughout Cache County. Specific operating hours
 are not specified by the county; however, its Regulation requires that stations be open and
 available to perform inspections during a major portion of normal business hours of 8:00 a.m.
 to 5:00 pm Mondays through Fridays.

Subject fleet: All model year 1969 and newer vehicles registered or principally-operated in Cache County are subject to the I/M program except for exempt vehicles.

Station/inspector Audits: Cache County's I/M program will regularly audit all permitted
 I/M inspectors and stations to ensure compliance with county I/M ordinances, regulations,
 and policies. Particular attention will be given to identifying and correcting any fraud or
 incompetence with respect to vehicle emissions inspections. Compliance with
 recordkeeping, document security, analyzer maintenance, and program security
 requirements will be scrutinized. The Cache County I/M program will have an active
 covert compliance program to minimize potential fraudulent testing.

Section X Part B, Page 1

1 2 3	<i>Waivers:</i> Cache County's I/M program allows for the issuance of waivers under limited circumstances. The procedure for issuing waivers is specified in Cache County's I/M regulation provided in Section 9.6 of Appendix 2 of this part of the SIP and meets the
4	minimum waiver issuance criteria specified in 40 CFR Subparts 51.360.
5	
6	Test frequency: Vehicles less than six years old as of January 1 on any given year will be
7	exempt from an emissions inspection. All model year 1969 and newer vehicles are
8	subject to a biennial test.
9	
10	Test Equipment: Specifications for the I/M test procedures, standards and analyzers are
11	described in Cache County's I/M regulation provided in Appendix 2. Specifications for
12	the test procedure and equipment were developed according to good engineering practices
13	to ensure test accuracy. Analyzer calibration specifications and emissions test procedures
14	meet the minimum standards established in Appendix A of the EPA's I/M Guidance
15	Program Requirements, 40 CFR Part 51 Subpart S.
16	
17	Test Procedures:
18	
19	• The following vehicles are subject to an on-board diagnostic (OBD) II inspection:
20	
21	\circ 1996 and newer light duty vehicles' and
22	\circ 2008 and newer medium duty vehicles ²
23	
24	• The following vehicles are subject to a two-speed idle test that is compatible with
25	Section VI (Preconditioned Two Speed Idle Test) in Appendix B of the EPA I/M
26	Guidance Program Requirements, 40 CFR 51, Subpart S:
27	
28	\circ 1995 and older vehicles,
29	o 1996 to 2007 medium and heavy duty vehicles and
30	\circ 2008 and newer heavy duty vehicles.
31	The second
32	Test procedures are outlined in Appendix 2 of this part of the SIP
33	
34	
35	
36	
37	

- Light duty vehicles have a Gross Vehicle Weight of 8500 lbs or less.
 Medium duty vehicles have a Gross Vehicle Weight greater than 8500 lbs but less than 14,000 lbs
 Heavy Duty vehicles have a Gross Vehicle Weight greater 14,000 lbs

1 3. I/M SIP Implementation

 The I/M program ordinance, regulations, policies, procedures, and activities specified in this I/M SIP revision shall be implemented by January 1, 2014 and shall continue until a maintenance plan without an I/M program is approved by EPA in accordance with Section 175 of the Clean Air Act.

Section X Part B, Page 3

APPENDIX 1

Cache County Emission Inspection/ Maintenance Program Ordinance 2013-04

Plenning

CACHE COUNTY ATTORNEY

James M. Swink Cache County Attorney

Donald G. Linton Chief Deputy

Tony C. Baird Chief Prosecutor

June 12, 2013

A LANGE CONTRACTOR

UTAH DEPARTMENT OF Barbara K. Lachmar Spencer D. Walsh Andrew E. McAdams JUN 1 4 2013 JUN 1 4 2013 JUN 1 4 2013

DIVISION OF AIR QUALITY Deputy County Attorneys

Terryl Warner Jenny Nielsen Joan Liquin Victim Advocates

199 North Main Street Logan, Utah 84321 (435) 755-1860 Fax: (435) 755-1969 or (435)755-1970 E-mail: (first initial)(last name)@cacheattorney.org

Bryce Bird Utah Division of Air Quality PO Box 144820 Salt Lake City, UT 84114-4820

RE: Cache County Motor Vehicle Inspection/Maintenance Ordinance

Dear Mr. Bird,

The Cache County Motor Vehicle Inspection/Maintenance Ordinance (hereinafter "Ordinance") is being submitted to the Air Quality Board to be adopted as part of the Utah State Air Quality Plan. This Ordinance was adopted by the Cache County Council on March 12, 2013, with an effective date of March 27, 2013. (See Ordinance attached hereto for reference).

The Cache County Attorney's Office assisted the Bear River Health Department and the Cache County Council in the creation of the Ordinance. In doing so, the Cache County Attorney's Office referred to, *inter alia*, Utah Code Ann. § 41-6a-1642, as amended, and the Ordinance was adopted, in part, in accordance with this statute.

Pursuant to the provisions of Utah Code Ann. § 41-6a-1642, it is the opinion of the Cache County Attorney's Office that the Ordinance is applicable to the entire County – including incorporated as well as non-incorporated areas, and that the Ordinance is in full compliance with Utah Code Ann. § 41-6a-1642. If you have any further questions or concerns, please do not hesitate to call our office at the telephone number indicated above.

Sincerely,

Donald G. Linton Chief Deputy Cache County Attorney

attachment cc: Amanda Smith Lloyd Berentzen

ORDINANCE 2013-04

IMPLEMENTATION OF A VEHICLE EMISSIONS AND MAINTENANCE PROGRAM IN CACHE COUNTY

1.0 PURPOSE

The purpose of this ordinance to reduce air pollution levels in Cache County by requiring emission inspections of on-road motor vehicles and by requiring emission related repairs and/or adjustments for those vehicles that fail to meet the prescribed standards so as to:

1.1 Protect and promote the public health, safety, and welfare;

1.2 Improve air quality;

1.3 Comply with the federal regulations contained in 40 CFR part 51 subpart S;

1.4 Comply with the law enacted by the Legislature of the State of Utah, Section 41-6a-1642 Utah Code Annotated, 1953, as amended.

2.0 POWERS AND DUTIES

2.1 The Cache County Council (hereafter, "Council") has authority to implement a vehicle emission inspection and maintenance program under Section 41-6a-1642, Utah Code Annotated, 1953, as amended.

2.2 The Council is presently required by the EPA and the State of Utah to implement a vehicle emission inspection and maintenance program.

2.3 The Council hereby delegates its authority as an administrative body under Section 41-6a-1642, Utah Code Annotated, 1953, as amended, to the Bear River District Board of Health (hereafter "Board"), to address all issues pertaining to the adoption and administration of the vehicle emission inspection and maintenance program.

2.4 The Council authorizes and directs the Board to adopt and promulgate rules and regulations to ensure compliance with EPA and State Implementation Plan requirements with respect to an emission inspection and maintenance program.

3.0 GENERAL PROVISIONS

3.1 The Board, in conjunction with its staff, will administer and enforce this ordinance.

3.2 The Board shall adopt vehicle emission and inspection rules and regulations which meet EPA and State Implementation Plan requirements.

3.3 The Council shall approve the initial Rules and Regulations established by the Board and all changes in Rules and Regulations.

4.0 GUIDELINES TO BE FOLLOWED BY THE BEAR RIVER BOARD OF HEALTH IN IMPLEMENTATING A VEHICLE EMISSION INSPECTION AND MAINTENANCE PROGRAM IN CACHE COUNTY

4.1 Vehicles registered in Cache County that are not exempt from the program (see 41-6a-1642(3), Utah Code Annotated) will be inspected on the following schedule:

4.1.1 Gasoline and natural gas powered vehicles model year 1969 and newer: No emissions inspection will be required for gasoline and natural gas powered vehicles that are six model years and newer as of January 1 of any given year. An emissions inspection will be required every other year for vehicles that are seven model years and older as of January 1 of any given year. Emissions testing on odd-numbered years for vehicles with odd-numbered model years and on even-numbered years for vehicles with even-numbered model years. No emissions inspection for vehicles model year 1968 and older.

4.1.2 Diesel powered vehicles model year 1998 and newer: No emissions inspection will be required for diesel powered vehicles that are six model years and newer as of January 1 of any given year. An emissions inspection will be required every other year for vehicles that are seven model years and older as of January 1 of any given year. Emissions testing on odd-numbered years for vehicles with odd-numbered model years and on even-numbered years for vehicles with even-numbered model years.

Diesel powered vehicles 2007 and newer will be tested using OBD technology. Diesel powered vehicles 1998-2006 will have a visual inspection to verify that proper emissions control devices are in place. Diesel powered vehicles 1997 and older are subject to the Bear River Health Department Smoking Vehicle Program.

4.2 A maximum fee for inspection shall be set by the Board and approved by the Council. Part of this fee will be retained by the entity which performs the test and part may be remitted to the Board as reimbursement for administering the program. The intent of the Council is that this fee be as low as possible, while still maintaining the financial viability of the program.

4.3. If a vehicle fails an emissions inspection, a waiver may be granted that will allow the vehicle to be registered that year. In order to qualify for a waiver, the vehicle owner/ operator must spend a minimum of \$200.00 on emissions related repairs and meet any other requirements established by the Board. A waiver will be issued once during the lifetime of the vehicle. Any changes to the minimum required repair expenditure to qualify for the waiver shall be approved by the Council.

4.4 Emission inspections in Cache County will be conducted by private firms. The Board shall establish criteria to be used to identify how many and which firms are allowed to conduct inspections and the training that is required for certification.

4.5 To fund the administration of the emissions inspection and maintenance program and other air quality improvement programs, the Council authorizes an Air Pollution Control fee to be assessed upon every motorized vehicle registered in Cache County at the time of registration as provide by Section 41-1a-1223, Utah Code Annotated, 1953, amended. 4.5.1 The fee is set at \$3.00 for each vehicle registration within the County under section 41-1a-215, Utah Code Annotated, 1953, as amended and at \$2.25 for each vehicle registration within the county for a six month registration period under Section 41-1a-215.5, Utah Code, 1953 as amended.

4.5.2 Motor vehicle that are exempt from the registration fee, and commercial vehicles with an apportioned registration shall be exempt from this fee as per Section 41-1a-1223, Utah Code Annotated, 1953 as amended.

4.5.3 The fee shall be assessed beginning January 1, 2014.

5.0 REVIEW OF NEED FOR PROGRAM

The Council shall review the vehicle emissions and maintenance program at least every five years to evaluate the continuing need for the program.

6.0 EFFECTIVE DATE

This ordinance takes effect on March 27, 2013. Following its passage, but prior to the effective date, a copy of the Ordinance shall be deposited with the County Clerk and a short summary of the ordinance shall be published in a newspaper of general circulation within the County as required by law.

PASSED BY THE COUNTY COUNCIL OF CACHE COUNTY, UTAH THIS MARCH 12, 2013.

	In Favor	Against	Abstained	Absent
Potter		X		
Buttars		X		
White	X			
Petersen	X			
Robison	X			
Yeates		. X		
Zilles	X			
Total	4	3		

CACHE COUNTY

Val K. Potter, Chairman

ATTEST: Jill N. Zollinger, Cound

Publication Date: March 27, 2013

BEAR RIVER HEALTH DEPARTMENT

REGULATION NO. 2013-1

A REGULATION OF THE BEAR RIVER HEALTH DEPARTMENT FOR A VEHICLE EMISSIONS INSPECTION AND MAINTENANCE PROGRAM

Adopted by the Bear River Board of Health

May 9, 2013

Table of Contents

Sections		
1.0	Definitions	3
2.0	Purpose	7
3.0	Authority and Jurisdiction of the Department	7
4.0	Powers and Duties	7
5.0	Scope	10
6.0	General Provisions	10
7.0	Permit Requirements of the I/M Program Stations	14
8.0	Training and Certification of Inspectors	15
9.0	Inspection Procedure	18
10.0	Engine Switching	22
11.0	Specifications for Certified Testing Equipment and Calibration Gases	22
12.0	Quality Assurance	23
13.0	Cutpoint Standards for Motor Vehicle Exhaust Gases	24
14.0	Disciplinary Penalties and Right to Appeal	24
15.0	Penalty	25
16.0	Severability	26
17.0	Effective Date	26

- Appendix A Fee Schedule
- Appendix B Emission Cutpoints
- Appendix C Penalty Schedule
- Appendix D Test Procedures
- Appendix E Technical Specifications and Calibration Gas

1.0 DEFINITIONS

For the purpose of this Regulation, the following terms, phrases, and words shall have the following meanings, unless otherwise defined:

Air Intake Systems: Systems that allow for the induction of ambient air, including preheated air into the engine combustion chamber for the purpose of mixing with a fuel for combustion;

AIR System: (Air Injection Reaction) A system for providing supplementary air into a vehicle's exhaust system to promote further oxidation of HC and CO gases and to assist catalytic reaction;

Analyzer: See Exhaust Emissions Gas Analyzer;

Board: See Board of Health;

Board of Health: The Bear River Board of Health;

Cache County Council: The elected Cache County Council representatives;

Calibration: The process of establishing or verifying the accuracy of an Exhaust Emissions Gas Analyzer to perform a consistent evaluation of engine exhaust by using different calibration gases having precisely known concentrations;

Calibration Gases: Gases of accurately known concentration that are used as references for establishing or verifying the calibration curve and accuracy of an Exhaust Emissions Gas Analyzer and are approved by the Department for use.

Catalytic Converter: A post-combustion device that oxidizes HC and CO gases and/or reduces oxides of nitrogen gases;

Certificate of Compliance: A document used in the I/M Program to certify that a vehicle meets all applicable requirements of the program;

Certificate of Waiver: A document used to verify that a vehicle has met the repair or adjustment requirements of the I/M Program Rules and Regulations even though specific emission standards have not been met;

Certification: Assurance by an authorized source, whether it be a laboratory, the manufacturer, the State, or the Department, that a specific product or statement is in fact true and meets all required requirements;

Certified Emissions Inspector: A person who has successfully completed all certification requirements and has been issued a current, valid Certified Emissions Inspector Certification by the Department;

Certified Testing Equipment: An official test instrument that has been approved by the Department to test motor vehicles for compliance with this Regulation; this includes the Analyzer as well as the OBD testing portion of the machine;

CO: Carbon monoxide;

Compliance: Verification that certain submission data and hardware submitted by a manufacturer for accreditation consideration, meets all required accreditation requirements;

Council: See Cache County Council;

County: Cache County, Utah;

Custom Vehicle: A motor vehicle that meets the requirements of Section 41-6a-1507, Utah Code Annotated, 1953 as amended;

Cutpoints: The maximum allowable concentration of carbon monoxide (CO) and hydrocarbons (HC) for a given weight class and model year of a motor vehicle, as provided by this Regulation, using an approved infrared Exhaust Emissions Gas Analyzer;

Department: The Bear River Health Department;

Director: The Director of the Bear River Health Department or his authorized representative;

DLC: Data Link Connector used in OBD applications is a 16 pin connector used by scan tools and other emission diagnostic equipment to communicate with the vehicle's computer for the purpose of collecting emissions related data;

DTC: Diagnostic Trouble Code is a standardized 5 digit code that is used to identify a specific fault that has occurred or is occurring in a vehicle;

EGR System: The Exhaust Gas Recirculation System – An emissions control system that recycles or recirculates a portion of the exhaust gases back to the engine combustion chambers;

Emissions Control Systems: Parts, assemblies or systems originally installed by the manufacturer in or on a vehicle for the sole or primary purpose of reducing emissions;

Exhaust Emissions Gas Analyzer: An instrument that is capable of measuring the concentrations of certain air contaminants in the exhaust gas emanating from a
motor vehicle which is approved by the Department for this use in accordance with this Regulation as an official test instrument;

Evaporative Control System: An emissions control system that prevents the escape of fuel vapors from the fuel tank or air cleaner and stores them in a charcoal canister to be burned in the combustion chamber;

Gas Calibration Check: A procedure using known concentrations of HC and CO calibration gases to verify the accuracy of an Analyzer in measuring HC and CO;

HC: Hydrocarbons;

Idle: A condition where the vehicle engine is warm and running at the rate specified by the manufacturer's curb idle, where the engine is not propelling the vehicle, and where the throttle is in the closed or idle stop position. This condition must be achieved without placing a load on the vehicle to decrease the RPM to the specified rate;

I/M Program: See Vehicle Emissions Inspection and Maintenance Program;

I/M Program Station: A stationary Vehicle Emissions Inspection and Maintenance Station that qualifies and has a valid permit, issued by the Department, to operate as an emissions inspection and maintenance station in the I/M Program;

Inspection: An official vehicle emissions test performed for the purpose of issuing a Certificate of Compliance or Certificate of Waiver;

Inspector: A Certified Emissions Inspector;

MIL: Malfunction Indicator Light is an indicator located on the instrument panel that notifies the operator of an emissions fault;

Motor Vehicle: A self-propelled motorized vehicle with an internal combustion powered engine which is licensed for operation on public roads and/or streets. Motor Vehicles exempted from the inspection requirements of this Regulation are listed in Section 6.4 of this Regulation;

Non-certified Inspector: Any person who has not been certified by the Department to perform official emissions tests;

OBD: On Board Diagnostic refers to a vehicle's monitoring and diagnostic capabilities of its emissions systems;

PCV System: Positive Crankcase Ventilation System – an emissions control system which returns crankcase vapors and blowby gases to the combustion chamber to be burned;

Primary Residence: Is the place where an individual intends to permanently reside, maintains a permanent residence more than six (6) months during a calendar year, or where an individual lives more than six (6) months during a calendar year;

Publicly-owned Vehicles: A motor vehicle owned by a government entity, including but not limited to the federal government or any agency thereof, the State of Utah or any agency or political subdivision thereof;

Readiness: Readiness is used to identify the state of a vehicle's emissions monitors as they are tested. Readiness does not indicate whether the monitors passed or failed the test, it only indicates whether or not the test has been run for any particular monitor;

Station: An I/M Program Station;

Technical Bulletin: A document, issued to Certified Emissions Inspectors and/or I/M Program Stations by the Department to update, clarify or establish policies and/or procedures for their implementation in the I/M Program;

Training Program: A formal program administered, conducted, or approved by the Department for the education of emission inspectors in basic emission control technology, inspection procedures, diagnosis and repair of emissions related problems, I/M Program policies, procedures, and this Regulation;

Two-Speed Idle: A condition where the vehicle engine is warm and running at a high speed rate of 2200-2800 RPMs and then a low rate of 350-1200 RPMs;

Vehicle Emissions Inspection and Maintenance Program: The program established by the Department pursuant to Section 41-6a-1642 Utah Code Annotated, 1953, as amended, and Cache County Ordinance 2013-04;

Vintage Vehicle: A motor vehicle that meets the requirements of Section 41-21-1 Utah Code Annotated, 1953 as amended;

2.0 PURPOSE

It is the purpose of this Regulation to reduce air pollution levels in Cache County by requiring inspections of in-use motor vehicles and by requiring emission related repairs and/or adjustments for those vehicles that fail to meet the prescribed standards so as to:

2.1 Protect and promote the public health, safety, and welfare;

2.2 Improve air quality;

2.3 Meet or exceed the minimum design and performance requirements for I/M Programs as defined in 40 CFR Part 51, Subpart S.

2.4 Comply with the law enacted by the Legislature of the State of Utah, Sections 41-6a-1642 Utah Code Annotated, 1953, as amended.

2.5 Comply with Cache County Ordinance 2013-04.

3.0 AUTHORITY AND JURISDICTION OF THE DEPARTMENT

3.1 Under Section 2.3 of Cache County Ordinance 2013-04, the Cache County Council delegates its authority as an administrative body under Section 41-6a-1642, Utah Code Annotated, 1953, as amended, to the Bear River Board of Health (hereafter Board), to address all issues pertaining to the adoption and administration of the Vehicle Emissions Inspection and Maintenance Program (hereafter I/M Program).

3.2 Under Section 2.4 of Cache County Ordinance 2013-04, the Council directs the Board to adopt and promulgate rules to ensure compliance with State Implementation Plan requirements with respect to an I/M Program.

3.3 The Board is authorized to make standards and regulations pursuant to Section 26A-1-121(1) of the Utah Code Annotated, 1953, as amended.

3.4 The Board is authorized to establish and collect fees pursuant to Section 26A-1-114(1)(h)(i) of the Utah Code Annotated, 1953, as amended.

3.5 All aspects of the I/M Program within Cache County enumerated in Section 2.0 of this Regulation shall be subject to the direction and control of the Bear River Health Department (hereafter Department).

4.0 POWERS AND DUTIES

4.1 The Department shall be responsible for the enforcement and administration of this Regulation and any other powers vested in it by law and shall:

4.1.1 Make policies and procedures necessary to ensure that the provisions of this Regulation are met and that the purposes of this Regulation are accomplished;

4.1.2 Require the submission of information, reports, plans, and specifications from I/M Program Stations as necessary to implement the provisions, requirements, and standards of this Regulation;

4.1.3 Issue permits, certifications, and charge fees as necessary to implement the provisions, requirements, and standards of this Regulation; and

4.1.4 Perform audits of any I/M Program Station, issue orders and/or notices, hold hearings, and levy administrative penalties, as necessary to effect the purposes of this Regulation.

4.2 The Department may suspend, revoke, or deny a permit, subject to the Penalty Schedule in Appendix C, of an I/M Program Station and/or require the surrender of the permit of such I/M Program Station upon showing that:

4.2.1 A vehicle was inspected and issued a Certificate of Compliance by the station personnel that did not, at the time of inspection, comply with all applicable policies, procedures, Technical Bulletins, and this Regulation;

4.2.2 A vehicle was inspected and rejected by the I/M Program Station when, in fact, the vehicle was determined by the Department to be in such condition that it did comply with the requirements of this Regulation;

4.2.3 The I/M Program Station is not open and available to perform inspections during a major portion of the normal business hours of 8:00 AM to 5:00 PM Mondays through Fridays (except I/M Program Stations which only test their own vehicles);

4.2.4 The I/M Program Station has violated any provisions of this Regulation, or any Rule, Regulation, or Department policy properly promulgated for the operation of an I/M Program Station;

4.2.5 The I/M Program Station was not equipped as required by Section 7.0 of this Regulation;

4.2.6 The I/M Program Station is not operating from a location specified on the permit;

4.2.7 An official inspection was done by a non-certified inspector or a non-certified inspector has gained access to the official testing portion of the test equipment or a non-certified inspector signed a Certificate of Compliance;

4.2.8 The computerized test equipment has been tampered with or altered in any way contrary to the certification and maintenance requirements of the test equipment;

4.2.9 The I/M Program Station denies access to a representative of the Department to conduct an audit or other necessary business during regular business hours;

4.2.10 An engine change verification form was completed and signed when, in fact, the engine block number was not verified by a Certified Emissions Inspector or other authorized personnel approved by the Department;

4.2.11 The I/M fee signage procedures are not followed as specified in Section 6.6; or

4.2.12 The I/M fee has been determined by the Department to be discriminatory in that different fees are assessed dependent upon vehicle ownership, vehicle make or model, owner residence, etc.

4.3 The Department may suspend, revoke, or deny the certificate of a Certified Emissions Inspector, subject to the Penalty Schedule in Appendix C, and require the surrender of this certificate upon showing that:

4.3.1 The Certified Emissions Inspector caused a Certificate of Compliance to be issued without an approved inspection being made;

4.3.2 The Certified Emissions Inspector denied the issuance of a Certificate of Compliance to a vehicle that, at the time of inspection, complied with the law for issuance of said certificate;

4.3.3 The Certified Emissions Inspector issued a Certificate of Compliance to a vehicle that, at the time of issuance, was in such a condition that it did not comply with this Regulation;

4.3.4 Inspections were performed by the Certified Emissions Inspector, but not in accordance with applicable policies, procedures, Technical Bulletins, and this Regulation;

4.3.5 The Certified Emissions Inspector allowed a non-certified inspector to perform an official I/M test or gain access to the official testing portion of the test equipment;

4.3.6 The Certified Emissions Inspector signed an inspection form or certificate stating that he had performed the emissions test when, in fact, he did not; or

4.3.7 The Certified Emissions Inspector completed and signed an engine change verification form when in fact the engine block number was not verified.

4.4 The Department shall respond, according to the policies and procedures of the Department, to public complaints regarding the fairness and integrity of the inspections they receive and shall provide a method that inspection results may be challenged if there is a reason to believe them to be inaccurate.

5.0 SCOPE

It shall be unlawful for any person to fail to comply with any policy, procedure, Technical Bulletin, or regulation promulgated by the Department, unless expressly waived by this Regulation.

6.0 GENERAL PROVISIONS

Subject to the exceptions in Section 6.4 and pursuant to the schedule in Section 6.1, individuals with their primary residence in Cache County must register their motor vehicles in Cache County and motor vehicles (of model years 1969 and newer) that are or will be registered in Cache County, or operated from a facility within Cache County shall be subject to an emission inspection performed by an I/M Program Station or other entity approved by the Director. Owners of vehicles that meet the requirements of Section 6.2 or 6.3 shall comply with the inspection requirements regardless of the county of registration.

6.1 Beginning January 1, 2014 motor vehicles are subject to a biennial emissions inspection. Emissions inspections will be required in odd-numbered years for a vehicle with an odd-numbered model year. Emissions inspections will be required in even-numbered years for a vehicle with an even-numbered model year

6.1.1 A Certificate of Compliance, Certificate of Waiver, or evidence that the motor vehicle is exempt from the I/M Program requirements (as defined in Section 6.4) shall be presented to the Cache County Assessor or the Utah State Tax Commission as conditions precedent to registration or renewal of registration of a motor vehicle in odd-numbered years for a vehicle with an odd-numbered model year.

6.1.2 A Certificate of Compliance, Certificate of Waiver, or evidence that the motor vehicle is exempt from the I/M Program requirements (as defined in Section 6.4) shall be presented to the Cache County Assessor or the Utah State Tax Commission as conditions precedent to registration or renewal of registration of a motor vehicle in even-numbered years for a vehicle with an even-numbered model year.

6.1.3 The Air Pollution Control Fee shall be paid annually, as per Section 4.5 of Cache County Ordinance 2013-04, (see also Section 6.7 of this Regulation) as conditions precedent to registration or renewal of registration of a motor vehicle.

6.1.4 A Certificate of Compliance shall be valid for a period of time in accordance with 41-1a-205 Utah Code Annotated, 1953, as amended.

6.2 Publicly-Owned Vehicles. Owners of publicly-owned vehicles shall comply with the inspection program requirements. Federally-owned vehicles and vehicles of employees operated on a federal installation that do not require registration in the State of Utah shall comply with the emissions testing requirements.

6.3 Vehicles of employees and/or students parked at a college or university that do not require registration in Cache County shall comply with the emissions testing requirements as authorized by 41-6a-1642(5)(a) Utah Code Annotated, 1953, as amended.

6.3.1 College or university parking areas that are metered or for which payment is required per use are not subject to the requirements in Section 6.3.

6.4 Vehicle Exemption. The following vehicles are exempt from these emissions testing requirements:

6.4.1 Any vehicle of model year 1968 or older;

6.4.2 All agricultural implements of husbandry and any motor vehicle that qualifies for an exemption as provided by 41-6a-1642(3) and 41-6a-1642(4) Utah Code Annotated, 1953, as amended;

6.4.3 Any vehicle used for maintenance or construction and not designed or licensed to operate on the highway;

6.4.4 Any motorcycle or motor driven cycle (including vehicles which operate with an engine normally used in a motorcycle);

6.4.5 Any vehicle that operates exclusively on electricity;

6.4.6 Any motor vehicle which qualifies for legislative exemptions;

6.4.7 Tactical military vehicles;

6.4.8 Any vintage vehicle as provided by 41-6a-1642(3) Utah Code Annotated, 1953, as amended;

6.4.9 Any custom vehicle as provided by 41-6a-1642(3) Utah Code Annotated, 1953, as amended;

6.4.10 Any vehicle that is less than six years old on January 1 based on the age of the vehicle as determined by the model year identified by the manufacturer;

6.4.11 Any diesel powered vehicle 1997 and older. These vehicles will be subject to a smoking vehicle program established by the Board; and

6.4.12 Any diesel powered vehicle with a GVW greater than 14,000 lbs. These vehicles will be subject to a smoking vehicle program established by the Board.

6.5 It shall be the responsibility of the Certified Emissions Inspector if a vehicle exempted from this Regulation by Section 6.4 of this Regulation is brought to the Certified Emissions Inspector for an official emission test to inform the owner/operator of the vehicle that the vehicle is not required to have an official emission inspection for vehicle registration purposes.

6.6 Official Signs.

6.6.1 All I/M Program Stations, except those stations authorized to inspect only their own motor vehicles as a fleet inspection station, shall display in a conspicuous location on the premises an official sign provided and approved by the Department;

6.6.2 The emission cutpoints, as referenced in Appendix B shall be posted in a conspicuous place on the station's premises;

6.6.3 The readiness requirements for an OBD test as referenced in Appendix D shall be posted in a conspicuous place on the station's premises;

6.6.4 The station shall post on a clear and legible sign and in a conspicuous place at the station, the fees charged by that station for the performance of the emissions inspection;

6.6.5 The free re-inspection policy as referenced in Section 9.6 shall be posted in a conspicuous place on the station's premises;

6.6.6 The signs required by Sections 6.6.1 through 6.6.5 shall be located so as to be easily in the public view.

6.7 Fees.

6.7.1 The fees assessed upon I/M Program Stations and Certified Emissions Inspectors shall be determined according to a fee schedule adopted by the Board. The fee schedule is referenced in Appendix A to this Regulation and may be amended by the Board as necessary. 6.7.2 An Air Pollution Control Fee is hereby assessed upon every motor vehicle registered in Cache County as per Section 4.5 of Cache County Ordinance 2013-04. The fee will be assessed annually at the time of registration of the vehicle.

- 6.7.2.1 This fee assessment is included upon all motorized vehicles including those that are exempted from the inspection requirements of this Regulation by Section 6.4 unless a separate fee is assessed on other motor vehicles by other Board of Health Regulations.
- 6.7.2.2 A motor vehicle that is exempt from the registration fee, and a commercial vehicle with an apportioned registration shall be exempt from this fee as per Section 41-1a-1223, Utah Code Annotated, 1953, as amended and Section 4.5.2 of Cache County Ordinance 2013-04.

6.7.3 I/M Program Stations may charge a fee for the required service. The fee may not exceed, for each vehicle inspected, the amount set by the Board and referenced in Appendix A of this Regulation.

- 6.7.3.1 The inspection fee pays for a complete inspection leading to a Certificate of Compliance or a failure. If a vehicle fails the inspection, the owner/operator is entitled to one free reinspection if he returns to the I/M Program Station that performed the original inspection within fifteen (15) calendar days from the date of the initial inspection. The I/M Program Station shall extend the fifteen day free reinspection to accommodate the vehicle owner/operator if the I/M Program Station is unable to schedule the retest of the vehicle within the fifteen day time period. The inspection fee shall be the same whether the vehicle passes or fails the emission test.
- 6.7.3.2 At the request of the Department, an I/M Program Station shall extend the free retest time for vehicle owners/operators who are unable to complete repairs because of the unavailability of parts to make the necessary repairs.

6.7.4 If a vehicle fails the inspection and is within the time and mileage requirements of the federal emissions warranty contained in section 207 of the Federal Clean Air Act, the Certified Emissions Inspector shall inform the owner/operator that he may qualify for warranty coverage of emission related repairs as provided by the vehicle manufacturer and mandated by the Federal Environmental Protection Agency (see 40 CFR Part 85, Subpart V).

7.0 PERMIT REQUIREMENTS OF THE VEHICLE EMISSIONS I/M PROGRAM STATION

7.1 Permit Required.

7.1.1 No person shall in any way represent any place as an official I/M Program Station unless the station is operated under a valid permit issued by the Department.

7.1.2 The Department is authorized to issue or deny permits for I/M Program Stations.

7.1.3 No permit for any official I/M Program Station may be assigned, transferred, or used by any person other than the original owner identified on the permit application for that specific I/M Program Station.

7.1.4 The permit shall be posted in a conspicuous place within public view on the premises.

7.1.5 Application for an I/M Program Station permit shall be made to the Department upon a form provided by the Department. No permit shall be issued unless the Department finds that the facilities, tools, and equipment of the applicant comply with the requirements of this Regulation and that competent personnel, certified under the provisions of Section 8.0, are employed and will be available to make inspections, and the operation thereof will be properly conducted in accordance with this Regulation.

- 7.1.5.1 An I/M Program Station shall notify the Department and cease any emission testing if the station does not have a Certified Emissions Inspector employed;
- 7.1.5.2 An I/M Program Station shall notify the Department upon termination and/or resignation of any Certified Emissions Inspector employed by the station;
- 7.1.5.3 An I/M Program Station shall comply with all the terms stated in the permit application and all the requirements of this Regulation;
- 7.1.5.4 As a condition for permitting test and repair I/M Program Stations, the station will keep and maintain all necessary tools and resources needed to effectively repair vehicles that fail an emissions test;

- 7.1.5.5 As a condition for permitting test only I/M Program Stations, the station will notify the vehicle owner/operator that the facility is a test only facility and will not provide repairs, prior to any official emissions test;
- 7.1.5.6 An I/M Program Station shall have a building with a suitable exhaust extraction system; and
- 7.1.5.7 An I/M Program Station shall provide a dedicated internet connection for the Certified Testing Equipment.
- 7.2 Permit Duration and Renewal

7.2.1 The permit for I/M Program Stations shall be issued annually and shall expire on the last day of the month, one year from the month of issue. The permit shall be renewable sixty days prior to the date of expiration.

7.2.2 It is the responsibility of the owner/operator of the I/M Program Station to pursue the permit renewal through appropriate channels.

7.3 I/M Program Station to hold Department Harmless

7.3.1 In making application for a permit or for its renewal, such action shall constitute a declaration by the applicant that the Department shall be held harmless from liability incurred due to action or inaction of I/M Program Station's owners or their employees.

7.4 An I/M Program Station shall be kept in good repair and in a safe condition for inspection purposes free of obstructions and hazards.

8.0 TRAINING AND CERTIFICATION OF INSPECTORS

8.1 Certified Emissions Inspector Certification Required.

8.1.1 No person shall perform any part of the inspection for the issuance of a Certificate of Compliance unless the person possesses a valid Certified Emissions Inspector Certification issued by the Department.

8.1.2 Applications for a Certified Emissions Inspector Certification shall be made upon an application form prescribed by the Department. No certification shall be issued unless:

8.1.2.1 The applicant has shown adequate competence by successfully completing the written and practical portions

of the Certified Emissions Inspector Certification requirements as specified in this Regulation; and

8.1.2.2 The applicant has paid the required permit fees as set by the Board and referenced in Appendix A of this Regulation.

8.1.3 An applicant shall comply with all of the terms stated in the application and with all the requirements of this Regulation.

8.1.4 An applicant shall complete a Department approved training course and shall demonstrate knowledge and skill in the performance of emission testing and use of the test equipment. Such knowledge and skill shall be shown by passing at minimum:

8.1.4.1	Operation and purposes of emission control systems;
8.1.4.2	Inspection procedures as outlined in this Regulation and prompted by the test equipment;
8.1.4.3	Operation of the Certified Testing Equipment including the performance of gas calibration and leak check;
8.1.4.4	The provisions of Section 207(b) warranty provisions of the Federal Clean Air Act, and other federal warranties;
8.1.4.5	The provisions of this Regulation and other applicable Department policies and procedures; and
8.1.4.6	A performance qualification test including but not limited to the following:
	(a) Visual inspection and knowledge of the required emission control equipment;
	(b) Demonstration of skill in proper use, care, maintenance, calibration, and leak testing of the Certified Testing Equipment;
	(c) Demonstration of ability to conduct the inspection; and
	(d) Demonstration of ability to accurately enter data in the test equipment.

8.1.5 A signed hands-on performance check sheet shall be necessary for successful completion of the performance qualification test. The hands-on

performance check sheet shall be signed by an instructor or other equally qualified person approved by the Department.

8.1.6 The Department shall issue a Certified Emissions Inspector Certificate to an applicant upon successful completion of the requirements of this section.

8.1.7 The Certified Emissions Inspector Certificates are and remain the property of the Department, only their use and the license they represent is tendered.

8.1.8 Certified Emissions Inspector Certifications shall not be transferred from one person to another person.

8.2 Recertification Requirements for Certified Emissions Inspectors

8.2.1 The Department may renew certifications for an existing Certified Emissions Inspector after a properly completed renewal form is submitted, reviewed, and approved, the recertification requirements have been completed, the fees are paid and the Certified Emissions Inspector has complied with this Regulation.

8.2.2 Certified Emissions Inspectors shall be required to recertify annually. Failure to recertify shall result in suspension or revocation of the Certification as described in this Regulation.

8.2.3 Certified Emissions Inspectors shall complete a Department approved refresher course every 2 years. Applicants for recertification shall complete a Department approved refresher course no more than sixty days prior to the date of expiration. Applicants shall demonstrate knowledge and skill in the performance of emission testing and use of the test equipment.

8.3 Certification Expiration

8.3.1 The Certified Emissions Inspector Certification shall be issued annually and shall expire on the last day of the month one year from the month of issue. The certification shall be renewable sixty days prior to the date of expiration.

8.3.2 It is the responsibility of the Certified Emissions Inspector to pursue the renewal of the Certification.

8.4 Certified Emissions Inspector Certification Suspension and Revocation

8.4.1 Certified Emissions Inspector Certifications may be suspended or revoked by the Department for violations of this Regulation.

8.4.2 Suspension or revocation of Certified Emissions Inspector Certifications shall follow the provisions of Appendix C of this Regulation.

9.0 INSPECTION PROCEDURE

9.1 The official emissions inspection shall be solely performed by a Certified Emissions Inspector at an I/M Program Station and Department approved inspection procedures are to be followed.

9.2 The Certified Emissions Inspector shall verify the vehicle license plate and vehicle identification numbers by comparing the information on the vehicle's registration with those on the vehicle and shall accurately record them on the inspection test equipment.

9.2.1 The Certified Emissions Inspector shall verify the owner's name and address and enter this information into the test equipment. The Certified Emissions Inspector shall determine and enter the county in which the vehicle is registered.

9.2.2 The Certified Emissions Inspector shall enter completely and accurately all the information required as part of the data entry procedure for the official vehicle emissions test on the approved test equipment.

9.3 A complete official test must be performed any time an inspection is requested. Do not perform any part of the inspection without initiating an official test on the test equipment.

9.4 The Certified Emissions Inspector shall perform the official vehicle emissions test using the proper testing procedure:

9.4.1 All gasoline, and natural gas powered light-duty (8,500 lbs or less) OBDII compliant vehicles, model year 1996 and newer shall be tested as specified in Appendix D, OBDII Test Procedures.

9.4.2 All gasoline and natural gas powered vehicles model year 1995 and older shall be tested as specified in Appendix D, Two-Speed Idle Test Procedures.

9.4.3 All gasoline and natural gas powered vehicles model year 1996 to 2007 with a GVW greater than 8,500 lbs shall be tested as specified in Appendix D, Two-Speed Idle Test Procedures.

9.4.4 All gasoline and natural gas powered vehicles model year 2008 and newer with a GVW greater than 8,500 lbs and GVW less than 14,000 lbs shall be tested as specified in Appendix D, OBDII Test Procedures.

9.4.5 All gasoline and natural gas powered vehicles model year 2008 and newer with a GVW greater than 14,000 lbs shall be tested as specified in Appendix D, Two-Speed Idle Test Procedures.

9.4.6 All diesel powered vehicles model year 1998 and newer shall be tested as specified in Appendix D, Diesel Test Procedures.

9.5 Retesting Procedures

9.5.1 If the vehicle fails the initial emissions inspection, the owner/operator shall have fifteen calendar days in which to have repairs or adjustments made and return the vehicle to the I/M Program Station that performed the initial inspection for one (1) free re-inspection. In order to be in compliance, the vehicle that failed the initial test shall meet the following conditions:

- 9.5.1.1 The vehicle is re-tested; and
- 9.5.1.2 The vehicle emissions levels are the same or less than the applicable cutpoint standards.

9.6 Certificate of Waiver

9.6.1 A Certificate of Waiver may be issued for 1969 to 1989 model year vehicles if all of the following requirements are met:

- 9.6.1.1 Air pollution control devices identified in the emission decal are in place and operable on the vehicle. If the decal is missing, at a minimum, the catalytic converter, PCV System, and AIR system are in place and operable on the vehicle. If the devices have been removed or rendered inoperable, they shall be replaced or repaired before a Certificate of Waiver is granted; and
- 9.6.1.2 The vehicle continues to exceed applicable cutpoint standards after \$200.00 of acceptable emissions related repairs have been performed. Proof of repair costs shall be provided for the vehicle to the Department in the form of an itemized bill, invoice, work order, manifest, or statement in which emissions related parts are specifically identified. If repairs are made by someone with ASE L1, ASE A8, or another certification approved by the Department, the cost of labor may be included in the \$200.00
- 9.6.2 A Certificate of Waiver may be issued for 1990 through 1995 model year vehicles if all of the following requirements are met:

- 9.6.2.1 Air pollution control devices identified in the emission decal are in place and operable on the vehicle. If the decal is missing, at a minimum, the AIR System, catalytic converter, EGR System, Evaporative Control System, PCV System, and gas tank cap are in place and operable on the vehicle. If the devices have been removed or rendered inoperable, they shall be replaced or repaired before a Certificate of Waiver is granted; and
- 9.6.2.2 The vehicle continues to exceed applicable cutpoint standards after \$200.00 of acceptable emissions related repairs have been performed. Proof of repair costs shall be provided for the vehicle to the Department in the form of an itemized bill, invoice, work order, manifest, or statement in which emissions related parts are specifically identified. If repairs are made by someone with ASE L1, ASE A8, or another certification approved by the Department, the cost of labor may be included in the \$200.00

9.6.3 A Certificate of Waiver may be issued for 1996 and newer model year vehicles if all of the following requirements are met:

- 9.6.3.1 Air pollution control devices identified in the emission decal are in place and operable on the vehicle. If the devices have been removed or rendered inoperable, they shall be replaced or repaired before a Certificate of Waiver is granted; and
- 9.6.3.2 At least \$200.00 has been spent on acceptable emissions related repair costs for that specific vehicle, and if proof of repair costs for that specific vehicle have been provided to the Department in the form of an itemized bill, invoice, work order, manifest, or statement in which emissions related parts are specifically identified. If repairs are made by someone with ASE L1, ASE A8, or another certification approved by the Department, the cost of labor may be included in the \$200.00
- 9.6.3.3 The vehicle is not within the time and mileage requirements of the federal emissions warranties. Any vehicle that is within time and mileage requirements of the federal emissions warranties shall not be eligible for an emissions repair waiver, but shall be repaired to pass the testing requirements.

9.6.4 As used in Sections 9.6.1, 9.6.2, and 9.6.3, acceptable emissions related repairs:

9.6.4.1	Refers to those expenditures and costs associated with the
	adjustment, maintenance, and repair of the motor vehicle
	which are directly related to reduction of exhaust emissions
	necessary to comply with the applicable emissions
	standards, and procedures, and/or repairs to the evaporation
	vapor recovery system;

- 9.6.4.2 Does not include adjustments, maintenance, or repairs performed prior to the official emissions test;
- 9.6.4.3 Does not include the fee paid for the test;
- 9.6.4.4 Does not include costs associated with the repairs or replacements of air pollution control equipment on the vehicle if the need for such adjustment, maintenance, replacement, or repair is due to disconnection of, tampering with, or abuse of the emissions control systems;
- 9.6.4.5 Does not include repairs performed to the vehicle's exhaust system to correct problems with excessive exhaust dilution;
- 9.6.4.6 Refers to repairs, maintenance, and diagnostic evaluations done in accordance with manufacturer's specifications, to the extent that the purpose is to reduce emissions.

9.6.5 Information regarding all performed repairs shall be entered into the appropriate data base of the test equipment prior to the vehicle being retested.

9.6.6 Certificates of Waiver shall only be issued by the Department unless the Department determines other acceptable methods of issuing the waivers. A waiver shall only be issued after determining that the vehicle complies with the requirements of this Section for waiver issuance.

9.6.7 Prior to referring the owner/operator to the Department for determining waiver eligibility, the I/M Program Station and the Certified Emissions Inspector shall verify that the repair and eligibility requirements of this Section have been met.

9.6.8 A Certificate of Waiver shall only be issued once to any vehicle that qualifies, throughout the lifetime of the vehicle.

10.0 ENGINE SWITCHING

10.1 Engine switching shall be allowed only in accordance with EPA policy.

10.2 Vehicles not meeting the requirements of Section 10.0 shall be deemed as tampered and are not eligible for a Certificate of Waiver, unless they are restored to the original engine and emission control configuration.

11.0 SPECIFICATIONS FOR CERTIFIED TESTING EQUIPMENT AND CALIBRATION GASES

11.1 Approval of Certified Testing Equipment

11.1.1 Certified Testing Equipment shall meet the specifications as detailed in Appendix E.

11.1.2 It shall be illegal for any person to modify the hardware or software of approved emissions test equipment without written application and formal approval by the Department.

11.1.3 It shall be illegal for any person to gain access to any Department controlled portions of an approved test equipment without approval by the Department.

11.2 Calibration Gases

11.2.1 General: The approved vendor shall, on request, supply at a reasonable cost to the I/M Program Station, calibration gases approved by the Department. The approved vendor shall have approved, full calibration gas containers installed and operational at the time of delivery. The Department shall establish necessary procedures for approving calibration gases.

11.2.2 Calibration Gas Blends: The calibration gases supplied to any I/M Program Station shall conform to the specifications of the Department as specified in Appendix E. All calibration gases shall meet all Federal requirements for the emissions warranty coverage. Only gas blends supplied by Department approved blenders shall be used to calibrate official Analyzers.

11.3 Warranty and Maintenance Requirements

11.3.1 It shall be the responsibility of the I/M Program Station to obtain warranty coverage for testing equipment supplied by the approved vendor. Coverage requirements will be determined by the Department.

11.3.2 The testing equipment shall be maintained in accordance with the manufacturer's recommended maintenance schedule and records of this maintenance service shall be maintained for examination by the Department.

11.4 Gas Calibration and Leak Check:

Gas calibrations and leak checks shall be performed in accordance with the schedule referenced in Appendix E.

12.0 QUALITY ASSURANCE

12.1 A quarterly inspection and audit shall be made by a representative of the Department to verify compliance with this Regulation for each I/M Program Station.

12.1.2 During the time of the inspection and audit by the Department, the Department representative shall have exclusive access to the test equipment.

12.1.3 Required tools and equipment as noted in Section 7.1.5, shall be kept at the I/M Program Station at all times and shall be available for inspection by the Department at any time the inspection station is open for business.

- 12.2 An annual covert inspection and audit shall be made by a representative of the Department to verify compliance with this Regulation for each I/M Program Station.
- 12.3 The Department may increase the frequency of inspections and audits for I/M Program Stations and/or Certified Emissions Inspectors if the Department receives information of a violation of this Regulation.
- 12.4 The Department shall regularly monitor I/M Program Stations and/or Certified Emissions Inspectors through inspection records and/or technology integrated into the Certified Test Equipment.

13.0 CUTPOINT STANDARDS FOR MOTOR VEHICLES EXHAUST GASES

In order to obtain a valid emissions Certificate of Compliance, exhaust emissions from a motor vehicle subject to an biennial exhaust gas emission inspection shall not exceed the

maximum concentrations for carbon monoxide (CO) and hydrocarbons (HC) as specified in Appendix B.

14.0 DISCIPLINARY PENALTIES AND RIGHT TO APPEAL

14.1 When the Department, or its representative(s), receives information of a violation of any regulation contained herein which may result in a permit denial, revocation, or suspension, the Department shall notify the affected entity, in writing, informing the entity of the violation and penalties to be enforced. The affected entity may request a hearing within ten calendar days of the Department giving notice of the potential permit denial, revocation, or suspension. Only a written request for a hearing shall be honored by the Department. No appeal may be made on a formal warning.

14.1.1 In considering the appropriate administrative action to be taken as indicated in Appendix C, the Director shall consider the following:

14.1.1.1	whether the violation was unintentional or careless;
14.1.1.2	the frequency of the violation or violations;
14.1.1.3	the audit and covert audit history of the I/M Program Station and the Certified Emissions Inspector;
14.1.1.4	whether the fault lies with the I/M Program Station or the Certified Emissions Inspector.

14.1.2 After consideration of the factors in Section 14.1.1 the Director may take appropriate administrative action as indicated in Appendix C against either the I/M Program Station, the Certified Emissions Inspector, or both.

14.2 Appeals Hearing Procedure:

14.2.1 An appeals hearing shall be held at the request of the affected entity in order to determine the accuracy of information obtained by the Department and whether there are mitigating factors which would justify a reduction of the imposed penalties.

14.2.2 The requesting party may bring to the hearing any witnesses and any evidence believed to be pertinent to the disciplinary action.

14.2.3 The appeal shall be heard by the Vehicle Inspection and Maintenance Appeal Board, hereafter I/M Board, consisting of at least three persons, who are not employees of Bear River Health Department, appointed by the Board. The I/M Board shall have the discretion to determine which witnesses shall be heard and what evidence is relevant. 14.2.4 Violations determined to be intentional or flagrant shall result in the maximum enforcement of the penalty schedule pursuant to Appendix C.

14.2.5 In considering whether to reduce a penalty indicated by Appendix C, the I/M Board and the Department shall consider the following:

14.2.5.1	whether the violation was unintentional or careless;
14.2.5.2	the frequency of the violation or violations;
14.2.5.3	the audit and covert audit history of the I/M Program Station and the Certified Emissions Inspector;
14.2.5.4	whether the fault lies with the I/M Program Station, the Certified Emissions Inspector, or both.

14.3 Written notice of the final determination of the I/M Board, including the I/M Board's finding under Section 14.2.5, shall be made within ten calendar days after the conclusion of the appeals hearing.

15.0 PENALTY

15.1 Any person who is found guilty of violating any of the provisions of this Regulation, either by failing to do those acts required herein or by doing a prohibited act, shall be guilty of a class B misdemeanor pursuant to Section 26A-1-123, Utah Code Annotated, 1953, as amended. If a person is found guilty of a subsequent similar violation within two years, he shall be guilty of a class A misdemeanor pursuant to Section 26A-1-123, Utah Code Annotated, 1953, as amended. If a person is found guilty of a subsequent similar violation within two years, he shall be guilty of a class A misdemeanor pursuant to Section 26A-1-123, Utah Code Annotated, 1953, as amended.

15.2 Each day such violation is committed or permitted to continue shall constitute a separate violation.

15.3 The county attorney may initiate legal action, civil or criminal, requested by the Department to abate any condition that exists in violation of this Regulation.

15.4 In addition to other penalties imposed by a court of competent jurisdictions, any person(s) found guilty of violating any of this Regulation shall be liable for all expenses incurred by the Department.

15.5 A Penalty Schedule for permit warning, suspension, or revocation is adopted as Appendix A and may be amended by the Board as the Board deems necessary to accomplish the purposes of this Regulation.

16.0 SEVERABILITY

If any provision, clause, sentence, or paragraph of this Regulation or the application thereof to any person or circumstances shall be held to be invalid, such invalidity shall not affect the other provisions or applications of this Regulation. The valid part of any clause, sentence, or paragraph of this Regulation shall be given independence from the invalid provisions or application and to this end the provisions of this Regulation are hereby declared to be severable.

17.0EFFECTIVE DATE

This Regulation shall become effective on $M_{agg}9,2013$ as adopted by the Bear River Board of Health. M. hypothetical Marguest Margues

Appendix A

Fee Schedule

\$250.00
\$50.00
\$75.00
\$75.00
\$25.00
\$15.00
\$25.00
Cost
\$3.00
\$2.25
\$15.00
\$20.00

APPENDIX B

BEAR RIVER HEALTH DEPARTMENT EMISSION STANDARDS CUTPOINTS

MOTOR VEHICLE EMISSIONS INSPECTION/MAINTENANCE PROGRAM

The following schedule gives the maximum allowable concentrations for carbon monoxide (CO) and hydrocarbons (HC) for both cars and trucks as determined by an approved infrared gas analyzer using the prescribed procedures. The effective date for these cutpoints is January 1, 2014.

ALL PASSENGER VEHICLES 1978 AND OLDER LIGHT DUTY TRUCKS 6,000 POUNDS GVWR OR LESS 1979 TRUCKS AND NEWER 8,500 POUNDS GVWR OR LESS <u>MAXIMUM CONCENTRATION STANDARDS</u>

MODEL YEAR	PERCENT <u>CARBON MONOXIDE</u>	PARTS/MILLION <u>HYDROCARBONS</u>
1968-1969	6.0	800
1970-1974	5.0	700
1975-1976	4.0	600
1977-1979	3.0	500
1980	2.0	300
1981-1995	1.2	220
1996 and newer	N/A – OBD II	N/A – OBD II

HEAVY DUTY TRUCKS AND VANS 1978 AND OLDER 6,001 AND OVER GVWR 1979 AND NEWER OVER 8,500 GVWR MAXIMUM CONCENTRATION STANDARDS

1968-1969	7.0	1500
1970-1978	5.0	1200
1979-1980	4.0	1000
1981 and newer	3.5	800

The minimum dilution factor must also be reached as part of the testing requirement. The dilution factor determination is contained in the analyzer specifications provided by the approved vendor.

NOTE: These should be considered as "cutpoints" for maximum allowable emissions levels. Vehicles must never be reset to these emission levels when readjustments are made, but rather shall be adjusted using manufacturer's specifications. By using manufacturer's specifications, the emissions levels should be well below the "cutpoints."

APPENDIX C – PENALTY SCHEDULE

Violation (resets after 2 years of no similar violations unless revoked)	1 st Occurrence	2 nd Occurrence	3 rd Occurrence	4 th Occurrence
Failure to inspect or substituting a vehicle other than the vehicle on the test record (intentional pass)	Tech: 180 day suspension and mandatory retraining	Tech: Revocation of permit for up to 5 years		
	Station: 180 day suspension	Station: 270 day suspension	Station: Revocation of inspection station permit for up to 5 years	
Passing a failing vehicle or recording pass for tampering on a tampered vehicle (gross negligence)	Tech: 30 day suspension and mandatory retraining	Tech: 60 day suspension and mandatory retraining	Tech: Revocation of permit for up to 5 years	
	Station: 15 day suspension	Station: 30 day suspension	Station: 60 day suspension	Station: Revocation of permit for up to 5 years
Falsifying an inspection record	Tech: 180 day suspension and mandatory retraining	Tech: Revocation of permit for up to 5 years		
or emissions certificate or Failing a passing vehicle (intentional)	Station: 180 day suspension	Station: 270 day suspension	Station: Revocation of inspection station permit for up to 5 years	
Non-certified	Tech: 60 day suspension	Tech: 180 day suspension	Tech: Revocation of permit for up to 5 years	
test (gross negligence table)	Station: 60 day suspension	Station: 180 day suspension	Station: Revocation of inspection station permit for up to 5 years	
Inaccurate or	Tech: Formal warning and mandatory retraining	Tech: 30 day suspension and mandatory retraining	Tech: 90 day suspension and mandatory retraining	Tech: Revocation of permit for up to 5 years
entry (incompetence)	Station: Formal warning	Station: 15 day suspension	Station: 45 day suspension	Station: Revocation of inspection station permit for up to 5 years
Failure to follow	Tech: Formal warning and mandatory retraining	Tech: 30 day suspension and mandatory retraining	Tech: 90 day suspension and mandatory retraining	Tech: Revocation of permit for up to 5 years
procedures (incompentence)	Station: Formal warning	Station: 15 day suspension	Station: 45 day suspension	Station: Revocation of inspection station permit for up to 5 years

Appendix D – Test Procedures

OBDII Test Procedures

On-Board Diagnostics (OBD) is the monitoring and fault detection/notification process of the Powertrain Control Module (PCM) related to the vehicle's emission control system and powertrain operation on 1996 and newer model year vehicles. When an emissions control malfunction is detected, a dashboard light illuminates, displaying one of the following: "Check Engine," "Service Engine Soon," or the international engine symbol. If the OBD system detects a problem that may cause vehicle emission to exceed applicable federal standards, the Malfunction Indicator Light (MIL) is illuminated and the appropriate diagnostic trouble code (DTC) and engine operating conditions will be stored in PCM memory.

- 1.0 Locate the Diagnostic Link Connector (DLC) on the vehicle being tested. Connect the vehicle to the test equipment.
 - 1.1 If the DLC is missing, has been tampered with, or is otherwise inoperable, the vehicle fails the test and shall be repaired.
 - 1.2 If the DLC is unaccessible, the problem must be remedied before the test can continue.
- 2.0 Turn the ignition switch to the off position for at least 30 seconds.
- 3.0 Visually examine the instrument panel to determine if the malfunction indicator light (MIL) illuminates, at least briefly, when the ignition key is turned to the "key on, engine off" (KOEO) position. Enter your visual inspection result into the test equipment.
 - 3.1 If the MIL does not illuminate, the vehicle fails the test and must be repaired.
- 4.0 Turn the ignition switch to the off position for at least 30 seconds.
- 5.0 Start the engine so the vehicle is in the "key on, engine running" (KOER) condition and follow the test equipment screen prompts until the test is complete.
- 6.0 For 1996-2000 model year vehicles two (2) not ready flags are allowed for a passing test. For 2001 and newer vehicles one (1) not ready flag is allowed. If the not ready status exceeds these numbers the vehicle must be driven additional miles until readiness monitors are set "ready" or repairs have been made allowing readiness flags to set ready.
- 7.0 If the MIL is commanded on while the engine is running, regardless of Diagnostic Trouble Codes (DTC's), the vehicle will fail the test and will require repairs.

- 8.0 Certain vehicles have been determined by the EPA to be OBDII deficient. The test equipment software will maintain a list of these vehicles and perform a modified OBDII test.
- 9.0 A Certificate of Compliance will be issued if the vehicle meets the requirements established in this section.

Two-Speed Idle (TSI) Test Procedures

During a two-speed idle test, the Analyzer measures the tailpipe exhaust emissions of a vehicle while the vehicle idles at both high and low speed. The Analyzer tests vehicles for carbon dioxide in addition to hydrocarbons and carbon monoxide. The two-speed idle test comprises two phases: (1) high speed test (2200-2800 RPMs) for the first phase of the emissions test; then, (2) tested at idle (350-1100 RPMs).

- 1.0 The Certified Emissions Inspector shall not inspect or test any motor vehicle with a mechanical condition which may cause injury to inspection personnel or damage to the inspection station or test equipment or which may affect the validity of the test, until such condition is corrected. Such conditions include, but are not limited to: coolant, oil, or fuel leaks; low oil or low fluid levels; and high visible emissions.
- 2.0 Prepare the Analyzer for testing as specified by the manufacturer.
- 3.0 Each vehicle shall be checked to determine that it is at normal operating temperature by feeling the top radiator hose or by checking the temperature gauge. Each vehicle shall be at normal operating temperature before performing the emissions inspection.
- 4.0 The inspection shall be performed with the transmission in "park" or "neutral" and with all accessories off and the emergency brake applied.
- 5.0 The Analyzer probe shall be inserted into the exhaust pipe at least twelve inches or as recommended by the Analyzer manufacturer, whichever is greater.
- 6.0 If a baffle or screen prevents probe insertion of at least twelve inches, a suitable probe adapter or snug fitting, non-reactive hose which effectively lengthens the exhaust pipe shall be used.
- 7.0 For all vehicles equipped with a multiple exhaust system that does not originate from a common point, both sides shall be tested simultaneously with an approved adapter.
- 8.0 When inspecting a vehicle under windy conditions, the tailpipe shall be shielded from the wind with a suitable cover.
- 9.0 With the tachometer properly attached, the vehicle shall be tested by following the screen prompts, answering questions, and entering required data. Vehicles failing because of excessive exhaust dilution shall repair the dilution problem prior to continuing the emission test. The dilution standard shall be contained in the Analyzer specifications provided by the approved vendor.

- 10.0 The Certified Emissions Inspector shall verify the presence of a gas cap and enter the information into the Analyzer.
- 11.0 Certain vehicles cannot be tested in the high speed (2200-2800 RPM) mode. The test equipment software will maintain a list of these vehicles and perform a modified test.
- 12.0 A Certificate of Compliance shall be issued if the vehicle emissions levels are the same as or less than the applicable cutpoint standards as referenced in Appendix B, and the vehicle has a gas cap present.

Diesel Powered Vehicles Test Procedures

- 1.0 All diesel powered vehicles 2007 and newer shall be tested in accordance with the following procedure:
 - 1.1 Locate the Diagnostic Link Connector (DLC) on the vehicle being tested. Connect the vehicle to the test equipment.

1.1.1 If the DLC is missing, has been tampered with, or is otherwise inoperable, the vehicle fails the test and shall be repaired.

1.1.2 If the DLC is unaccessible, the problem must be remedied before the test can continue.

- 1.2 Turn the ignition switch to the off position for at least 30 seconds.
- 1.3 Visually examine the instrument panel to determine if the malfunction indicator light (MIL) illuminates, at least briefly, when the ignition key is turned to the "key on, engine off" (KOEO) position. Enter your visual inspection result into the test equipment.

1.3.1 If the MIL does not illuminate, the vehicle fails the test and must be repaired.

- 1.4 Turn the ignition switch to the off position for at least 30 seconds.
- 1.5 Start the engine so the vehicle is in the "key on, engine running" (KOER) condition and follow the test equipment screen prompts until the test is complete.
- 1.6 If the vehicle has 1 or more monitors "not ready", follow the attached flowchart to determine whether the readiness check will be marked as pass or fail.
- 1.7 If the MIL is commanded on while the engine is running, regardless of Diagnostic Trouble Codes (DTC's), the vehicle will fail the test and will require repairs.
- 1.8 Certain vehicles have been determined by the EPA to be OBDII deficient. The test equipment software will maintain a list of these vehicles and perform a modified OBDII test.
- 1.9 A Certificate of Compliance will be issued if the vehicle meets the requirements established in this section.

- 2.0 All diesel powered vehicles 1998-2006 shall be subject to a visual anti-tampering inspection. The air pollution control devices identified in the emission decal shall be in place and apparently operable on the vehicle. If the decal is missing the vehicle owner/operator shall have the decal replaced before the inspection can continue.
 - 2.1 The devices listed on the decal must be present and apparently operable to pass the emission inspection.
 - 2.2 If the OBD II system is identified in the emission decal, the procedure in Section 1.1 through 1.5 shall be followed.
 - 2.3 If the decal is missing and is no longer available for replacement the vehicle owner/operator shall provide written documentation to the Department stating such. Approved documentation shall come from an authorized dealer or manufacturer of the vehicle in question.

2.3.1 If the emissions decal is missing and the vehicle meets the requirements of Section 2.3, the following emissions control devices should be present and apparently operable if factory equipped:

Catalyst; Exhaust Gas Recirculation System (EGR); Diesel Particulate System (DPF); Air Injection Reaction System (AIR); Urea System (SCR); and OBD II System.

2.4 A Certificate of Compliance shall be issued if the emissions control devices are in place and apparently operable. An inspection of the OBD II system as referenced in Section 2.2 shall be for informational purposes only and will not determine whether a vehicle passes or fails the emission inspection.

Diesel Readiness Check Flowchart



Appendix E - Technical Specifications and Calibration Gas

1.0 GENERAL

This appendix contains specifications for Emission Inspection System Contractors (hereafter, Contractors) to design Testing Equipment to be used in the Cache County Vehicle Emissions Inspection and Maintenance Program (hereafter, I/M Program). Testing Equipment to be used in the I/M Program must be capable of performing consistent Two-Speed Idle (TSI), and On-Board Diagnostics (OBD) emissions inspections.

1.1 Design Goals

Testing Equipment must be designed and constructed to provide reliable and accurate service in the automotive service environment and have a useful life of at least five years. The software must be designed for maximum operational simplicity and be capable of providing emissions readings or codes that can be used for vehicle diagnostics. A manual, non-test mode should be available to perform vehicle diagnostics. The software must prevent users from clearing Diagnostic Trouble Codes, changing readiness status, or performing other actions that could change the results of an official emissions test. In addition, the Testing Equipment must include security measures that will prevent unauthorized modifications to the software or inspection data, record unauthorized entry, also known as tampering, and prevent subsequent inspections when tampering is detected.

These technical specifications contain the minimum requirements for Testing Equipment used to perform emissions inspections in the I/M Program. Contractors may include additional items with approval from the Bear River Health Department (hereafter, the Department).

1.1.1 Identification Data

A nameplate including the following information must be permanently affixed to the housing of the Testing Equipment:

- Name and address of manufacturer;
- Model description;
- Serial number; and
- Date of assembly.

In addition, the Contractor shall affix a label to the housing of the Testing Equipment that contains a toll-free telephone number for customer service. This telephone number must also be displayed on error messages that recommend the need for service by the manufacturer.

The Testing Equipment must also electronically display:

- Nameplate data;
- Testing Equipment number; and
- Propane Equivalency Factor (PEF).

1.2 Manuals

All Testing Equipment sold or leased by the Contractor must be provided with a current copy of a manual that contains, at a minimum, operating instructions, maintenance instructions, and initial startup instructions. The manual may be provided in an electronic format and should be accessible from the Testing Equipment.

1.3 Certification Requirements

The Contractor shall submit a letter to the Department stating that the Testing Equipment model sold or leased by the Contractor or its authorized representatives satisfies all design and performance criteria described in these specifications. Unless otherwise specified, a copy of the software documentation listed below must be submitted to the Department as part of the certification application. The documentation must include, at a minimum, the following:

- Complete program listing(s);
- Functional specifications;
- Functional flowcharts of the software;
- Example inputs and outputs from all processes;
- Detailed interface information on system components including the identification of protocol and output specifications; and
- File layouts.

To ensure proper maintenance of all Testing Equipment, a full description of the Contractor's service procedures and policies, sample contracts, warranties, and extended service agreements must be provided as part of the certification application. The Contractor shall provide a training plan to the Department that will be used to conduct certification training of potential inspectors on the use of the Testing Equipment. The Contractor shall supply to the Department and maintain at least one piece of Testing Equipment.

1.3.1 Escrow of Software

The Contractor must submit a letter of corporate authorization agreeing to place software source codes and other pertinent technical information in an escrow placement approved by the Department. The Contractor shall contract with the approved escrow company and provide the Department with a copy of the contract including the Department as a beneficiary. Certification of the Testing Equipment will not be valid until this condition has been met.

The Contractor must place in escrow the most recent version of the Testing Equipment software, including but not limited to, the actual software code and related materials
used to meet this specification. The software will be turned over to the Department only if the Contractor defaults or cannot ensure continued performance of the contract.

In the event that the software is transferred, the Department shall protect it from public dissemination and commercial usage to the extent required by law. The software may be used, maintained, and updated by the Department, or its assignee, to support the I/M Program. At a minimum, the Department shall:

- Limit source code access to parties necessary to maintain and update the analyzers;
- Require all parties to sign a non-disclosure agreement before obtaining access to the code; and
- Grant no license permitting an entity to use any part of the codes for any commercial purpose other than to update and operate the analyzers.

The Department is not interested in the disclosure of proprietary information or the detailed inner workings of the software. However, it is essential that the software, schematics, and drawings be available in case the Contractor defaults.

As a prerequisite to certification, the Contractor shall furnish a performance bond to the Department. This bond must be in a form approved by the Department, executed as a surety by a bonding company authorized to do business in the State of Utah, and signed by a licensed resident agent. The performance bond must be for \$250,000 and must remain valid for the entire time period that the Contractor participates in the I/M Program. The performance bond must cover all Testing Equipment that is certified to conduct emissions inspections in the I/M Program.

The performance bond may be used by the Department at any time if the Contractor is in default of the requirements of these specifications, including but not limited, to the following "Events of Default":

- A. The Contractor fails to remedy a breach of covenant, representation, or warranty required by these specifications within thirty (30) days after written notice of such breach has been given to the Contractor by the Department;
- B. The Contractor makes a general assignment for the benefit of creditors, admits in writing its inability to pay debts as they mature, institutes proceedings to be adjudicated upon voluntary bankruptcy, consents to the filing of a bankruptcy proceeding against it, files a petition or answer or consent seeking reorganization, readjustment, arrangement, composition, or similar relief under federal bankruptcy or any other similar applicable law(s), consents to the filing of any such petition, consents to the appointment of a receiver, liquidator, trustee, or assignee in bankruptcy or insolvency of the manufacturer or a substantial part of its property, or takes action to further any of these purposes; or
- C. A court of competent jurisdiction enters a decree or order adjudging the Contractor as bankrupt or insolvent, or approving a properly filed petition seeking

reorganization, readjustment, arrangement, composition, or similar relief for the Contractor under the federal bankruptcy or any other similar applicable law(s), and such decree or order is not discharged or stayed continuously for a period of sixty (60) days; or a decree or order of a court of competent jurisdiction for the appointment of a receiver, liquidator, trustee or assignee in bankruptcy or insolvency of the manufacturer or of a substantial part of its property, or for the liquidation of its affairs, is entered, and such decree or order is not discharged or stayed continuously for a period of sixty (60) days; or any substantial part of the property of the Contractor is sequestered or attached and is not returned to the Contractor or released from such attachment within sixty (60) days thereafter.

To require performance by the surety under the performance bond, the Department shall give written notice of the event of default to the Contractor, specifying the date upon which the surety performance must begin.

The Director or his designee shall release the performance bond once it is determined that the Contractor has satisfactorily completed its obligations in accordance with the terms of these specifications, or at an earlier date, if it is determined by the Director to be in the best interest of the Department.

1.4 Warranty Coverage and Extended Service Agreements

A written warranty coverage agreement, signed by an authorized representative of the Contractor and the I/M Program Station, which provides a complete description of coverage for all systems and components and all Contractor provided services listed below in Contractor Provided Services, must accompany the sale or lease of each unit of Testing Equipment.

The original manufacturer's warranty must be a minimum of one year from the date of purchase. An extended warranty service agreement must be available to the Testing Equipment owner upon the expiration of the manufacturer's original warranty period. Cost disclosures of consumable inventory items and extended warranty service agreements with detailed descriptions of coverage must be available to all Testing Equipment owners.

The cost of extended warranty service agreements must be identified in the Contractor's response to the RFP

1.5 Contractor Provided Services

A Contractor-authorized repair technician is a Testing Equipment service technician that is authorized by the Contractor to perform service on their fleet of Testing Equipment. Only Contractor-authorized repair technicians may access the secure areas on the Testing Equipment.

The Contractor-authorized repair technician shall perform a gas calibration prior to returning an Analyzer to service whenever a component of the emissions measurement system is repaired or replaced. Similarly, the Contractor-authorized repair technician shall perform a leak check each time the Analyzer's sample line is broken and repaired. Contractor-authorized repair technicians shall have software driven menu options or other acceptable method that records the transfer of inspection station, inspector information, and other data from one unit of Testing Equipment to another without manual inputs or the transfer of previous data.

The Department may require the Contractor to conduct on-site or laboratory testing of the Testing Equipment in order to document continued compliance. The Contractor shall supply the I/M Program Station a temporary replacement unit of Testing Equipment that meets the I/M Program requirements if a unit of Testing Equipment is removed from the I/M Program Station for repair or testing. The Contractor shall be responsible for any costs incurred under this requirement.

The Contractor shall correct software features that do not meet these specifications to the satisfaction of the Department. The enhancement of operational software must be specified by the Department and be designed to update through the internet. Unless authorized by the Department, software enhancements must be available for beta testing within 120 days of commencement of a software update contract and receipt of an updated Testing Equipment specification. The Contractor shall not modify any existing Testing Equipment software without obtaining approval from the Department.

The Contractor shall be responsible for training Department officials responsible for oversight of the I/M Program, including but not be limited to, the instruction on all operational, maintenance, and quality control features of the Testing Equipment sampling system, full access to and use of inspection, audit, and calibration menus, and optional programs offered to inspectors. This training must be conducted at the Contractor's expense as a condition of certification, and upon written request by the Department.

The Contractor shall provide the following services to the I/M Program Station as part of any sale, lease, or loan of Testing Equipment:

- Delivery, installation, calibration, and verification of the proper operating condition of the Testing Equipment;
- Two extra sample filters with each TSI Analyzer, and an additional printer cartridge or a certificate redeemable for a printer cartridge for all Testing Equipment;
- A minimum of two hours operation and maintenance training to the owners and operators for each unit of Testing Equipment purchased or leased.

The Contractor shall provide the following services to the I/M Program Station as part of the manufacturer's original warranty and thereafter as a portion of the extended warranty service agreement.

 Full systems support and repair, including temporary provision of units of equal quality and specification;

- Quarterly examination, calibration, and routine maintenance of Analyzer and sampling systems on the TSI Analyzers. Annual examination must be required on the OBD portion of the Testing Equipment.
- On-site service response by a Contractor-authorized repair technician within one business day (Saturday shall be considered a business day), excluding Sundays, national/state holidays (New Year's Day, Martin Luther King, Jr. Day, President's Day, Memorial Day, Independence Day, Pioneer Day, Labor Day, Veteran's Day, Thanksgiving, and Christmas), and other days the purchaser's business might be closed, of a request from the I/M Program Station. The names, toll free telephone number(s), and service facility addresses of the Contractor's representatives responsible for Equipment service must be provided to the I/M Program Station. All system repairs, component replacements, and/or Testing Equipment adjustments, including reset of quality control lockout systems, must be accomplished onsite within a minimum average response time of 8 business hours after a service request has been initiated. If the completion of this work is not possible within this time period, Testing Equipment of equal quality and specifications must be provided until the malfunctioning unit is properly repaired and returned to service.

1.6 Electronic Transmission Security

The Testing Equipment shall utilize a standard protocol encryption method for communications with the host incorporating error detection and not incorporating error correction. The Testing Equipment shall utilize bitsum checking for all messages.

1.7 Tamper Resistance

The controlled access design must be the responsibility of the Contractor, but all security measures must be submitted to the Department for approval. The Testing Equipment operators, Department personnel, and field representatives authorized by the Contractor shall be prevented from creating or changing any inspection results, programs, or data contained on the Testing Equipment. The Contractor shall use appropriate software and hardware provisions to protect I/M files and programs. The file and program protection may consist of mechanical systems in combination with electronic and software systems. The protection features must prevent access to the secured portions of the hard disk containing I/M programs and inspection data. The control key or its functional equivalent, which gives access to the operating system (OS), must not be activated except through the use of a password on the audit menu. The password must be chosen by the Department at the time of certification testing. Other security or protection alternatives may be proposed by the Contractor for approval by the Department.

The Contractor shall, at a minimum, develop tamper resistant features to prevent unauthorized access through the Testing Equipment cabinet. Micro switches, keyed and software controlled locks, and software algorithms requiring the use of an access code must all be used where appropriate. Any unauthorized access to the secured areas of the Testing Equipment must be detected, even when the power is off. A software lockout algorithm must be activated should

tampering occur, which would abort any existing inspection sequence and prevent further inspections until the lockout is cleared by a field representative authorized by the Department. The Contractor shall develop a system to allow Contractor-authorized repair technicians to clear tamper lockouts only during authorized service calls. The lockout system must be designed so that it can be activated from the audit menu by Department personnel. The Contractor may use keyed locks on the cabinet doors to secure the disk drives as long as the locks are built-in and can be changed by authorized personnel should a security problem be identified. A software controlled solenoid lock may also be used on the secured drive door of the Testing Equipment. The solenoid lock may be used instead of or in addition to any key or combination lock that may be provided. The Testing Equipment software must control the solenoid lock and unlatch the doors in response to authorized requests from the audit menu while maintaining the appropriate levels of security.

A tamper file must be created that includes the date, time, type, and location of the tamper lockout, date and time the lockout was cleared, and who cleared the lockout. The tamper lockout type and location must be accessible only through the lockout function of the Testing Equipment's audit menu.

Access to the compact disc drive (CD), if applicable, must be available to I/M Program Station personnel at all times. However, access to the BIOS, I/M related programs, and data must be secured separate from the CD and additional drives. The Contractor shall provide a security method approved by the Department for the CD drive(s) to prevent unauthorized reads, writes, and executable. However, the Contractor may offer Testing Equipment with additional disk drives that can run optional software application programs.

The Testing Equipment must prevent Contractor-authorized repair technicians from performing the following, except in a manner approved by the Department:

- Clearing a state lockout;
- Clearing a lockout for a failed three-day gas calibration or leak check;
- Adding, deleting, or modifying test data;
- Adding, deleting, or modifying I/M Program Station information or an Certified Emissions Inspector's license number; and
- Altering the calibration gas bottle values.

1.8 Automated Inspection Process Software and Displays

The inspection process, data collection, and quality control features of the Testing Equipment must be automated as much as possible. The software must automatically select the emission standards for the vehicle from an internal reference table. Vehicle identification information must be derived from a database accessed over a real time data system to the Testing Equipment. Access to the Vehicle Identification Database (VID) shall be accomplished by entry of the vehicle identification number (VIN) in its entirety. Provisions must be made for manual entry of data for vehicles not in the reference files of the Testing Equipment. The Contractor in consultation with the Department shall customize how the emission testing results are displayed on the Testing Equipment and on the approved paperwork provided to the owner of the vehicle.

2.0 HARDWARE REQUIREMENTS

2.1 Overview

The hardware requirements for the Analyzer must meet or exceed specifications as published by the California Bureau of Automotive Repair (BAR) and contained in the "BAR-97 EMISSIONS INSPECTION SYSTEM SPECIFICATIONS" (BAR-97), dated May 1996, except where reference is made to ASM testing and NOx gas measurement requirements. The Analyzer may include all amendments made to the BAR-97 hardware specifications to present date. Each Analyzer shall be equipped with Bar Code Scanner, Engine Revolutions per Minute Detection System and Real-Time Inspection Testing Monitoring System.

2.2 Accessing the OBD System

The Testing Equipment must include hardware and software necessary to access the on-board computer systems on all model-year 1996 and newer gasoline and natural gas powered vehicles. The Testing Equipment must also be able to access the on-board computer system on all model years 2007 and newer diesel powered vehicles. The equipment design and operation of the Testing Equipment must meet the federal requirements contained in Title 40 of the Code of Federal Regulations (CFR), Chapters 85.2207-2231 and the recommended practices regarding OBD inspections contained in the J1962, J1978 and J1979 published by the Society of Automotive Engineers (SAE). The Testing Equipment must be able to connect to the vehicle's OBD connector and access, at a minimum, the following OBD data:

Service modes: \$01, \$03, \$07, \$09, \$0A

At a minimum, the Testing Equipment must also be capable of communicating with all OBD vehicles that use the following communications protocols:

- International Organization for Standardization (ISO) 9141;
- Variable pulse width (VPW) as defined in the SAE's J1850;
- Pulse width modulation (PWM) as defined in the SAE's J1850;
- Keyword protocol 2000 (KWP); and
- Controller area network (CAN) as defined in the ISO 15765-4.3:2001.

The OBD interrogation process must be fully integrated into the Testing Equipment, automated, and require no inspector intervention to collect and record the OBD data retrieved via the OBD connector link. No separate interface may be used.

2.3 OBD Inspection Equipment

The OBD inspection Equipment apply only to the OBD communication components, which must meet all federal requirements contained in 40 CFR §§85.2207 - 85.2231 and recommended practices contained in the J1962, J1978, and J1979 published by the SAE. The Equipment must meet criteria contained in the EPA's guidance document, "Performing Onboard Diagnostic System Checks as Part of a Vehicle Inspection and Maintenance Program" (EPA, 2001) or EPA's applicable update to this document.

2.4 Bar Code Scanner

The bar code scanner must be able to read a one-dimensional (1-D) and a two-dimensional (2-D) bar code through a windshield and use visible laser diode technology or an equivalent approved by the Department. The bar code scanner must not be able to read Universal Product Code (UPC) 1-D bar codes. The bar code scanner must be able to withstand multiple drops to concrete covering a distance of at least 4 feet and be environmentally sealed to withstand the normal operating conditions of an automotive service environment.

2.5 Engine Revolutions per Minute Detection

Testing Equipment must be equipped with a tachometer, or equivalent software and hardware necessary to detect engine RPM from the original equipment manufacturer (OEM) ignition technologies in use at the time of certification. Possible updates may be required to enable future ignition systems to be monitored for engine RPM. A software "HELP" screen must be available to help the Certified Emissions Inspector locate an RPM signal. The cable-type connection must be at least 25 feet long (measured from the front of the Testing Equipment).

Based on the vehicle identification information available to the Certified Emissions Inspector, the Testing Equipment must display messages indicating when the vehicle under inspection requires a specific type or method of the tachometer pick-up connection. A digital display tachometer must be displayed to measure engine speed. For TSI Analyzers, RPM readings must be recorded on a second-by-second basis for the 10 second or 5 second period that is used to determine the pass or fail status of the TSI emissions inspection, respectively. The tachometer operation must use one of the following means:

- Radio frequency-type transmitter/receiver that requires no direct vehicle connection and can detect engine RPM on vehicles using distributorless ignition systems (DIS);
- Cable-type connection capable of detecting engine RPM of current OEM ignition technology;
- Battery/accessory power connection; or
- Cable-type connection capable of detecting engine RPM via the OBD port.

During the official inspection process the Testing Equipment must prompt the Certified Emissions Inspector to shut the engine off while connecting the cable-type RPM connection. The RPM bypass function must be made available when the live engine RPM is displayed for the first time. If the RPM cannot be obtained, the Certified Emissions Inspector shall be allowed to bypass the RPM. The Certified Emissions Inspector must simultaneously strike at least two keys to activate the RPM bypass. The bypass function must no longer be available once the emission inspection has begun.

The Certified Emissions Inspector may use the previously listed methods for 1996 and newer model-year vehicles if the OBD port is unable to detect engine RPM. Tachometer performance must be no less than a 0.5 second RPM response time with an accuracy of +/-3 percent of actual RPM.

2.6 Real-Time Inspection Testing Monitoring System

All approved Testing Equipment conducting official emission testing shall be equipped with video capturing equipment. An I/M Program Station will be in violation if the video capturing equipment is not properly maintained or installed and capturing images of each inspection. If video equipment is not fully operational, the I/M Program Station must contact the Contractor immediately for repair or replacement.

2.7 Inspection Restrictions Based on Current Calibrations

The Analyzer must:

- prevent TSI emissions inspections if the leak check has not passed in the last 24 hours;
- prevent TSI emissions inspections if the gas calibration has not passed in the last 72 hours;

The Testing Equipment must display appropriate error messages that indicate when a leak check or other calibration is needed to allow TSI inspections to be performed.

2.8 Running Changes and Other Hardware Modifications

Changes to design characteristics, component specifications, or any other modifications to the Testing Equipment hardware must be approved by the Department. The Contractor is responsible for confirming that such changes will have no detrimental effect on performance of the Testing Equipment. The Department may require testing at approved beta test sites prior to the release of the modifications.

All proposed hardware modifications must be thoroughly tested by a third-party before being submitted to the Department, and be accompanied by a cover letter containing the following information:

- Description of all of the proposed modifications to be performed, a parts list, and the installation instructions for the Contractor-authorized repair technician;
- Test data and an engineering evaluation regarding the effects of the proposed modification(s) on the performance and reliability of the Testing Equipment for any modifications to the bench or sample system;

- Timeline showing timeframe in which the modifications are expected to occur and the number of existing units of Testing Equipment that will be updated;
- Description of any special procedures that are needed to perform the hardware modifications; and
- Documentation for any software update that would be required for the proposed hardware modifications.

2.9 Exhaust Gas Analysis Equipment Specifications

This section defines the requirements for the components needed to determine the concentrations of the exhaust gases during the TSI inspections.

2.9.1 Measured Gases

The Analyzer must measure hydrocarbons (HC) as hexane in parts per million (ppm), carbon monoxide (CO), carbon dioxide (CO₂), and oxygen (O₂) in percent. The Analyzer must have a digital display for vehicle engine speed and exhaust concentrations of HC, CO, CO₂, and O₂ and must be capable of measuring exhaust concentrations of HC, CO, CO₂, and O₂ at a minimum sample rate of twice per second.

2.9.2 Warm-up Conditions

The Analyzer must reach stability within 30 minutes from startup at 35 degrees Fahrenheit (°F). The Analyzer must be considered warmed-up when the internal verifications are complete and the zero and span readings for HC, CO, CO₂, and O₂ have stabilized within the allowable accuracy values for five minutes without adjustment. If stabilization has not been reached within an allotted time frame, then the Analyzer must prevent TSI inspection sequences and display a message instructing the Certified Emissions Inspector to call for service. Functional operation of the gas sampling system must remain disabled through an internal lockout until the instrument meets stability and warm-up requirements.

2.9.3 Sampling System Components

A) General:

The sampling system must extract exhaust gas from a subject vehicle, remove particulate matter and aerosols from the sampled gas, drain the condensed water from the sample if necessary, and deliver the resultant gas sample to the Analyzer's sensors for analysis. The sampling system must, at a minimum, consist of a tailpipe probe, flexible sample line, continuously draining water removal system, particulate trap, sample pump, and flow control components. Provisions must be made for the introduction of zero air and calibration gases. Materials that are in contact with the gases sampled must not contaminate or change the composition of the gases to be analyzed, including gases from vehicles not fueled by gasoline. The system must be designed to be corrosion-resistant and to withstand vehicle exhaust.

B) Sample Probe and Hose Criteria:

Sample hose must be 25 feet in length with a tolerance of +/- 0.5 feet when measured from the front of the Analyzer cabinet. The hose must be composed of non-kinking material that will not be affected by or react to the exhaust gases.

Sample hose and probe provided with each Analyzer must withstand exhaust gas temperatures at the probe tip of up to 1,100°F for 10 minutes. Use of dissimilar metals with thermal expansion factors of more than 5 percent must not be used in either the construction of probes or connectors.

A positive means of retention must be incorporated to prevent the probe from slipping out of the tailpipe when in use.

A thermally insulated securely attached hand grip must be provided on the probe to ensure easy probe insertion using one hand.

The probe must be designed so that the tip extends 16 inches into the tailpipe and at least 10 inches into the vehicle's exhaust.

The probe tip must be shielded to avoid inadvertent debris collection and sealed to prevent any sample dilution when it is inserted into the tailpipe. Use of a tailpipe extension is permitted as long as the extension does not change the exhaust back pressure by more than +/- 1 inch of water pressure.

A straight probe tip must be provided that is bent less than 15 degrees, made of stainless steel solid-wall tubing with a 3/16 inch outside diameter, and designed so the connector between the removable probe tip and the rigid portion of tubing is up inside the tailpipe at least three inches to reduce the effects of any leak that might occur.

A probe tip cap suitable for performing a leak check must be provided if the vacuum decay method for performing a leak check is used. Otherwise, all hoses and connectors that are necessary to perform a leak check must be provided.

The sample system must include equipment necessary to inspect vehicles equipped with dual exhaust pipes. The flow in each leg of the dual exhaust probe sample system must be equal.

C) Particulate Filter and Water Trap:

- The particulate filter must be capable of trapping 97 percent of all particulates and aerosols five microns or larger;
- The filter must not absorb or adsorb HC;

- The filter housing must be transparent to allow the operator to observe the filter's condition without removing the housing. The filter must be removable and reliably seal after replacement;
- The water trap must be sized to remove exhaust sample water from vehicles fueled with, or a combination of gasoline, propane, compressed natural gas (CNG), oxygenated fuels, and alternative fuels. The filter bowl, filter, and housing must not react to these fuels or the vehicle's exhaust gases. The condensed water must be continuously and sufficiently drained from the water trap's bowl to prevent condensation in the sample system or in the optical bench's sample cell; and
- Incorporate a back-purge system.

D) Low Flow Indicator:

The Analyzer must lockout official TSI inspections when the sample flow is below the acceptable level. The Analyzer's sample system must be equipped with a flow meter or equivalent device that detects sample flow degradation. The Analyzer must display a low flow condition message when flow rate causes the measurement error for any gas to exceed 3 percent of the gas value used for calibration or audit or causes the analyzer response time to exceed 13 seconds to 90 percent of a step change in input, whichever is less. The sample vacuum may be continuously monitored to detect a low flow condition as an alternative.

E) Analyzer lockout:

The Analyzer must lockout official TSI inspections when the sample flow is below the acceptable level. The Analyzer's sample system must be equipped with a flow meter or equivalent device that must indicate when sample flow degradation for any gas other than NO causes:

- The measurement error to exceed 3 percent of the gas value used for checking; or
- The Analyzer response time to exceed 13 seconds for a 90 percent step change in input.

The sample vacuum may be continuously monitored to detect a low flow condition as an alternative.

3.0 Analyzer Requirements

3.1 Gas Calibration

A) General:

The Analyzer must automatically require and successfully pass a leak check and a gas calibration for HC, CO, CO_2 , and O_2 by a method that is approved by the Department. The Analyzer must not allow an error of more than 2 percent of the readings using the high and low range span gases

for TSI inspections. The Analyzer must automatically prohibit the performance of the tailpipe portion of the vehicle emissions inspection when readings exceed the 2 percent error tolerance. The Analyzer channels must be adjusted to the center of the allowable tolerance range as a result of the gas calibration procedure.

The standard gases to be used to calibrate and audit the Analyzer must meet the requirements in the Federal Clean Air Act, §207(b) and described in Subpart W of Part 85 of Chapter I, Title 40 of the CFR. All standard gases purchased by the I/M Program Station for use in the Analyzer must conform to the requirements established by the BAR for emissions inspection analyzer calibration gases and the National Institute of Standards and Technology (NIST).

B) Gas Calibration Procedure:

- The Analyzer must maintain accuracy between gas calibrations taking into account all errors, including noise, repeatability, drift, linearity, temperature, and barometric pressure;
- The Analyzer must automatically require a zero gas calibration and a high and low range gas calibration for HC, CO, CO₂, and O₂, where applicable. The Analyzer must record the gas reading data prior to the adjustment and other data pertinent to control charting Analyzer performance;
- The gas calibration must be accomplished by the following method: Calibration gases that meet the requirements of Section 3.1: Calibration Gases for TSI Analyzers must be introduced into the calibration port of the Analyzer. The pressure in the sample cell must be the same with the calibration gas flowing as with the sample flowing during an inspection. Once the pressure is the same, the Analyzer must perform a zero gas calibration and a leak check. The leak check must ensure that the entire sample system does not leak.
- 3.2 Calibration Gases for TSI Analyzers

The following gases must be used for the two-point calibration and audit.

A) Low Range Calibration Gas

HC = 200 ppm propane

- CO = 0.5 percent
- $CO_2 = 6.0$ percent
- O_2 = Shop Air
- N₂ = Balance 99.99 percent pure

B) High Range Calibration Gas

HC = 3200 ppm propane

CO = 8.0 percent

 $CO_2 = 12.0$ percent

 O_2 = Shop Air

N₂ = Balance 99.99 percent pure

3.3 Dilution

The flow rate of the Analyzer must not cause more than 10 percent dilution during sampling of vehicle exhaust gases from a 1.6 liter engine at normal idle. Ten percent dilution is defined as a sample of 90 percent exhaust and 10 percent ambient air.

3.4 Calibration Prompts and Gas Usage

The Analyzer must display prompts to guide the inspector through the gas calibration procedure in a manner that minimizes the amount of gas used. The Analyzer must be designed to keep the loss of calibration gas to less than 0.5 liter in 24 hours when the valve on the calibration gas bottle is left open.

3.5 Propane Equivalency Factor

The value of the PEF must range from 0.490 to 0.540 and be displayed in a manner acceptable to the Department for each gas audit and gas calibration point. If an optical bench must be replaced in the field, then the Contractor-authorized repair technician must change any external labels to correspond to the PEF of the new bench. The Analyzer must incorporate an algorithm relating PEF to HC concentration. Corrections to the PEF must be made automatically and the corrected PEF value must range from 0.470 to 0.560.